

#### **Draft Cancer Human Health Hazard Assessment for** Di(2-ethylhexyl) Phthalate (DEHP), Dibutyl Phthalate (DBP), Butyl Benzyl Phthalate (BBP), Diisobutyl Phthalate (DIBP), and **Dicyclohexyl Phthalate (DCHP) Technical Support Document for the Draft Risk Evaluations** CASRNs: 117-81-7 (DEHP), 84-74-2 (DBP), 85-68-7 (BBP), 84-69-5 (DIBP), 84-61-7 (DCHP) May 2025

| 40     | TABLE OF CONTENTS  |    |
|--------|--|----|
| 41     | TABLE OF CONTENTS  | 2  |
| 42     | LIST OF TABLES   | 4  |
| 43     | LIST OF APPENDIX TABLES  | 6  |
| 44     | KEY ABBREVIATIONS AND ACRONYMS   | 7  |
| <br>15 |  | /  |
| 45     |  | 9  |
| 46     | NOTE TO REVIEWERS  | 10 |
| 47     | SUMMARY  | 10 |
| 48     | 1 INTRODUCTION AND SCOPE   | 12 |
| 49     | 2 APPROACH TO IDENTIFYING EPIDEMIOLOGY AND LABORATORY ANIMAL                           |    |
| 50     | DATA   | 14 |
| 51     | <b>3 GENOTOXICITY HAZARD IDENTIFICATION</b>  | 18 |
| 52     | 3.1 Di(2-ethylbexyl) Phthalate (DEHP)  | 18 |
| 53     | 3.2 Butyl Benzyl Phthalate (BBP)   | 19 |
| 54     | 3.3 Dibutyl Phthalate (DBP)  |    |
| 55     | 3.4 Dijsobutyl Phthalate (DIBP)  | 28 |
| 56     | 3.5 Dicyclohexyl Phthalate (DCHP)  | 29 |
| 57     | 3.6 Diisononyl Phthalate (DINP)  | 30 |
| 58     | 3.7 Diisodecvl Phthalate (DIDP)  | 30 |
| 59     | 3.8 Conclusions on Genotoxicity  | 30 |
| 60     | 4 CANCER HAZARD IDENTIFICATION, CHARACTERIZATION, AND MODE OF                          |    |
| 61     | ACTION   | 32 |
| 62     | 4.1 Summary of Available Epidemiological Studies for DEHP, BBP, DBP, DIBP, DCHP, DIN   | Ρ  |
| 63     | and DIDP   | 32 |
| 64     | 4.1.1 Previous Epidemiologic Assessments of Phthalates                                 | 32 |
| 65     | 4.1.1.1 Health Canada (2018a)  | 33 |
| 66     | 4.1.1.2 ATSDR (2022)   | 33 |
| 67     | 4.1.1.3 IARC (2013)  | 33 |
| 68     | 4.1.2 Epidemiologic Studies of Phthalates and Cancer Outcomes (2018–2019) Evaluated by |    |
| 69     | EPA  | 35 |
| 70     | 4.1.2.1 Di(2-ethylhexyl) Phthalate (DEHP)  | 35 |
| 71     | 4.1.2.2 Butyl Benzyl Phthalate (BBP)   | 36 |
| 72     | 4.1.2.3 Dibutyl Phthalate (DBP)  | 36 |
| 73     | 4.1.2.4 Diisobutyl Phthalate (DIBP)  | 36 |
| 74     | 4.1.2.5 Dicyclohexyl Phthalate (DCHP)  | 37 |
| 75     | 4.1.2.6 Diisononyl Phthalate (DINP)  | 37 |
| 76     | 4.1.2.7 Diisodecyl Phthalate (DIDP)  | 37 |
| 77     | 4.1.3 Conclusion   | 37 |
| 78     | 4.2 Overview of Laboratory Animals Studies   | 38 |
| 79     | 4.3 Cancer Hazard Characterization, Mode of Action and Conclusions for DEHP, BBP, DBP, |    |
| 80     | DINP, and DIDP   | 42 |
| 81     | 4.3.1 Di(2-ethylhexyl) Phthalate (DEHP)  | 42 |

| Mari | 2025 |
|------|------|
| May  | 2025 |

|     | Whay 2025  |      |
|-----|--|------|
| 82  | 4.3.1.1 Liver, Pancreatic, and Testicular Tumors (Tumor Triad)                             | . 50 |
| 83  | 4.3.1.1.1 Mode of Action for Liver Tumors in Rats and Mice                                 | . 50 |
| 84  | 4.3.1.1.2 Mode of Action for Pancreatic Acinar Cell Tumors (PACTs)                         | . 56 |
| 85  | 4.3.1.1.3 Mode of Action for Leydig Cell Tumors  | . 57 |
| 86  | 4.3.1.1.4 Inferences from Hypolipidemic Drugs and Other Prototypical PPARα Activators      | . 58 |
| 87  | 4.3.1.1.5 Uncertainties, Limitations, and Human Relevance                                  | . 62 |
| 88  | 4.3.1.1.6 Conclusions Regarding Tumor Triad  | . 63 |
| 89  | 4.3.1.2 Uterine Tumors   | . 63 |
| 90  | 4.3.1.2.1 Conclusions for Uterine Tumors   | . 66 |
| 91  | 4.3.1.3 Mononuclear Cell Leukemia (MNCL)   | . 66 |
| 92  | 4.3.1.3.1 Conclusions for MNCL   | . 67 |
| 93  | 4.3.1.4 Preliminary Cancer Classification for DEHP   | . 69 |
| 94  | 4.3.2 Butyl Benzyl Phthalate (BBP)   | . 71 |
| 95  | 4.3.2.1 Mononuclear Cell Leukemia (MNCL)   | . 74 |
| 96  | 4.3.2.1.1 Conclusions for MNCL   | . 74 |
| 97  | 4.3.2.2 Pancreatic Acinar Cell Tumors (PACTs)  | . 75 |
| 98  | 4.3.2.2.1 Conclusions for Pancreatic Acinar Cell Tumors                                    | . 78 |
| 99  | 4.3.2.3 Urinary Bladder Papillomas and/or Carcinomas                                       | . 79 |
| 100 | 4.3.2.3.1 Conclusions for Urinary Bladder Tumors   | . 82 |
| 101 | 4.3.2.4 Preliminary Cancer Classification for BBP  | . 82 |
| 102 | 4.3.3 Dibutyl Phthalate (DBP)  | . 83 |
| 103 | 4.3.3.1 Pancreatic Acinar Cell Adenomas  | . 85 |
| 104 | 4.3.3.1.1 Conclusions on Pancreatic Acinar Cell Tumors                                     | . 87 |
| 105 | 4.3.3.2 Leydig Cell Adenomas   | . 88 |
| 106 | 4.3.3.2.1 Conclusions on Leydig Cell Tumors  | . 91 |
| 107 | 4.3.3.3 Preliminary Cancer Classification for DBP  | . 91 |
| 108 | 4.3.4 Diisononyl Phthalate (DINP)  | . 92 |
| 109 | 4.3.5 Diisodecyl Phthalate (DIDP)  | . 94 |
| 110 | 5 EVALUATING THE CARCINOGENICITY OF DIBP AND DCHP USING READ-                              |      |
| 111 | ACROSS: WEIGHT OF SCIENTIFIC EVIDENCE ANALYSIS   | . 96 |
| 112 | 5.1 Physical and Chemical Properties   | 97   |
| 112 | 5.2 Absorption Distribution Metabolism and Excretion                                       | 101  |
| 11/ | 5.2 A cute Toxicity  | 101  |
| 115 | 5.4 Evidence of Hormone Perturbation and Developmental and Reproductive Toxicity           | 101  |
| 115 | 5.5 Subchronic Toxicity  | 102  |
| 117 | 5.6 Evidence of Immune System Perturbation   | 105  |
| 118 | 5.7 Genotoxicity   | 105  |
| 110 | 5.8 Mechanistic Studies to Support a Proposed Mode of Action                               | 105  |
| 120 | 5.9 Evidence of Chronic Toxicity and Carcinogenicity From Read-Across to Related Chemicals | 100  |
| 120 | 5.10 Weight of Scientific Evidence Conclusions Regarding Carcinogenicity of DIBP and DCHP  | 100  |
| 121 | Based on Read-across   | 111  |
| 122 | 6 CONCLUSIONS AND NEXT STEPS   | 113  |
| 123 | REFERENCES.  | 115  |
| 125 | APPENDICES   | 141  |
| 126 | Appendix A SUMMARY OF DEHP GENOTOXICITY STUDIES  | 141  |
| 127 | Appendix B RODENT CARCINOGENICITY STUDY SUMMARIES  | 149  |
|     |  |      |

| 128              | B.1 Di(2-e         | thylhexyl) Phthalate (DEHP)  | . 149      |
|------------------|--------------------|--|------------|
| 129              | B.1.1 M            | ice - Oral Exposure Studies  | . 149      |
| 130              | B.1.1.1            | Two-year Dietary Study of B6C3F1 Mice (NTP, 1982a)   | . 149      |
| 131              | B.1.1.2            | Two-year Dietary Study of B6C3F1 Mice (David et al., 2000a; David et al., 1999)                      | . 150      |
| 132              | B.1.2 Ra           | tts - Oral Exposure Studies  | . 150      |
| 133              | B.1.2.1            | Two-year Dietary Study of F344 Rats (NTP, 1982a)   | . 150      |
| 134              | B.1.2.2            | Two-year Dietary Study of F344 Rats (David et al., 2000b; David et al., 1999)                        | . 151      |
| 135              | B.1.2.3            | Ninety-five Week Dietary Study of Male F344 Rats (Rao et al., 1987)                                  | . 153      |
| 136              | B.1.2.4            | Two-year Dietary Study of Male F344 Rats (Rao et al., 1990)  | . 153      |
| 137              | B.1.2.5            | Lifetime Dietary Study of Male Sprague-Dawley Rats (Voss et al., 2005)                               | . 153      |
| 138              | B.1.2.6            | Two-year Dietary Study of Sprague-Dawley Rats (Perinatal and Postweaning                             |            |
| 139              |                    | Exposure Study) (NTP, 2021b)   | . 154      |
| 140              | B.1.2.7            | Two-year Dietary Study of Sprague-Dawley Rats (Postweaning Exposure Study)                           |            |
| 141              |                    | (NTP, 2021b)   | . 159      |
| 142              | B.1.3 Ha           | amsters – Inhalation and Intraperitoneal Studies   | . 164      |
| 143              | B.1.3.1            | Inhalation Study (Schmezer et al., 1988)   | . 164      |
| 144              | B.1.3.2            | Intraperitoneal Injection Study (Schmezer et al., 1988)  | . 164      |
| 145              | B.1.4 Tr           | ansgenic Mice – Oral Exposure Studies  | . 164      |
| 146              | B.1.4.1            | Twenty-six Week Dietary Study of Wild-type and Transgenic RasH2 Mice                                 |            |
| 147              |                    | (Toyosawa et al., 2001)  | . 164      |
| 148              | B.1.4.2            | Twenty-six Week Dietary and 28-week Topical Studies of Tg.AC Mice (Eastin et al.                     | ••         |
| 149              |                    | 2001)  | . 165      |
| 150              | B.1.4.3            | Thirty-nine Week Dietary Study of $Xpa^{-/-}$ Mice, C57BL/6 Mice, and $Xpa^{-/-}/P53^{+/-}$          |            |
| 151              |                    | Mice (Mortensen et al., 2002)  | . 165      |
| 152              | B.1.4.4            | Twenty-two Month Dietary Study of Wild-type and PPAR $\alpha$ -null Sv/129 Mice (Ito et              |            |
| 153              |                    | al., 2007a)  | . 166      |
| 154              | B.2 Butyl          | Benzyl Phthalate (BBP)   | . 166      |
| 155              | B.2.1 St           |  | . 166      |
| 156              | B.2.1.1            | Two-year Dietary Study of B6C3F1 Mice (N1P, 1982b)   | 160        |
| 15/              | B.2.2 St           | udles of Kats  | . 16/      |
| 158              | B.2.2.1            | Two-year Dietary Study of F344/N Rats (NTP, 1982b)   | . 16/      |
| 159              | B.2.2.2<br>D.2.2.2 | Two-year Dietary Study of F344/N Rats (NTP, 1997)  | , 107<br>J |
| 100              | B.2.2.3            | Two-year Dietary Study of F544/N Rais – Study T (Ad Libitum and weight-Matched                       | u<br>160   |
| 101              | D 2 2 4            | Controls Protocol) (NTP, 1997a)  | . 109      |
| 162              | D.2.2.4            | (NTD 1007c)  | 170        |
| 105              | D 2 2 5            | (NIF, 1997a)   | . 170      |
| 104              | D.2.2.3            | I wo-year Dictary Study of 1'344/19 Kats – Study 5 (Lifetiffie Kestficted Feed Drotocol) (NTD 1007a) | 171        |
| 105              | Annondiv C         | SCIENTIEIC UNCEDTAINTIES DELATED TO MONONUCI EAD CELI  | . 1 / 1    |
| 167              | Appendix C         | LEUKEMIA (MNCL) AND LEYDIG CELL TUMORS IN F344 RATS  | . 173      |
| 168              | Appendix D         | SUMMARY OF STUDIES OF DEHP EVALUATING PPARα ACTIVATION   | . 174      |
| 169              |                    |  |            |
| 170              |                    |  |            |
| 171              | LIST OF '          | TABLES   |            |
| - / <del>-</del> |                    |  |            |

#### 

|            | May 2025   |              |
|------------|--|--------------|
| 174        | Table 3-3. Summary of Genotoxicity Studies of DIBP   | . 28         |
| 175        | Table 3-4. Summary of Genotoxicity Studies of DCHP   | . 29         |
| 176        | Table 4-1. Summary of Existing Epidemiologic Assessments of Phthalates Investigating Cancer          |              |
| 177        | Outcomes   | . 32         |
| 178        | Table 4-2. Summary of Database of Available Rodent Carcinogenicity Studies Considered                | . 39         |
| 179        | Table 4-3. Summary of Tumor Types Observed Following Chronic Oral Exposure to Phthalates in          |              |
| 180        | Experimental Rodent Models <sup>a</sup>  | . 41         |
| 181        | Table 4-4. Summary of Cancer Classifications and Listings for DEHP                                   | . 42         |
| 182        | Table 4-5. Summary of Available Carcinogenicity Studies of DEHP in Rodents                           | . 44         |
| 183        | Table 4-6. Summary of Observed Tumors and Effect Levels (LOAEL, mg/kg-day) Across                    |              |
| 184        | Carcinogenicity Studies of DEHP <sup>a</sup>   | . 48         |
| 185        | Table 4-7. Occurrence of Key Events in PPARα MOA in Rats and Mice <sup><i>a</i></sup>                | . 52         |
| 186        | Table 4-8. Summary of Two-year Tumor Findings in Rats Administered Hypolipidemic Drugs               | . 60         |
| 187        | Table 4-9. Incidence of Uterine Tumors in SD Rats Chronically Exposed to DEHP (Perinatal and         |              |
| 188        | Postweaning Exposure Study) (NTP, 2021b) <sup><i>a</i></sup>   | . 64         |
| 189        | Table 4-10. Incidence of Uterine Tumors in SD Rats Chronically Exposed to DEHP in the Diet for Tw    | WO-          |
| 190        | years (NTP, 2021b) <sup><i>a</i></sup>   | . 65         |
| 191        | Table 4-11. Incidence of MNCL in F344 Rats Administered DEHP Through the Diet for Two-Years          |              |
| 192        | (David et al., 2000b; David et al., 1999) <sup><i>a</i></sup>  | . 67         |
| 193        | Table 4-12. Summary of Transcriptional BMD and BMDL Values for Genes Regulated by PPARα in           | 1            |
| 194        | the Liver of Male SD Rats Gavaged with DEHP for Five Days (Gwinn et al., 2020) <sup>a</sup>          | . 71         |
| 195        | Table 4-13. Summary of Cancer Classifications and Listings for BBP                                   | . 72         |
| 196        | Table 4-14. Summary of Available Carcinogenicity Studies of BBP in Rodents                           | . 73         |
| 197        | Table 4-15. Incidence of Non-neoplastic and Neoplastic Findings in the Pancreas of F344/N Rats Fed   | Ĺ            |
| 198        | Diets Containing BBP for Two-years (NTP, 1997b) <sup>a</sup>   | . 76         |
| 199        | Table 4-16. Incidence of Neoplasms and Non-neoplastic Lesions in the Pancreas in F344/N Rats (Ad     |              |
| 200        | Libitum and Weight-Matched Controls Protocols) (NTP, 1997a) <sup>a</sup>                             | . 77         |
| 201        | Table 4-17. Incidence of Non-neoplastic and Neoplastic Findings in the Urinary Bladder in F344/N R   | lats         |
| 202        | Fed Diets Containing BBP for Two-years ( <i>Ad Libitum</i> and Weight-Matched Controls               |              |
| 203        | Protocol) (NTP, 1997b)"  | . 79         |
| 204        | Table 4-18. Incidence of Non-neoplastic and Neoplastic Findings in the Urinary Bladder in F344/N R   | lats         |
| 205        | Fed Diets Containing BBP for Two-years (NTP, 1997a)"   | . 80         |
| 206        | Table 4-19. Incidence of Non-neoplastic and Neoplastic Findings in the Urinary Bladder in F344/N R   | lats         |
| 207        | Treated with BBP (2-year Restricted Feed and Lifetime Restricted Feed Protocols) (N                  | $\Gamma P$ , |
| 208        | $(199/a)^{\circ}$  | . 81         |
| 209        | Table 4-20. Summary of Available Rodent Carcinogenicity Studies of DBP                               | . 84         |
| 210        | Table 4-21. Mean Received Doses (mg/kg-day) for Male and Female SD Rats Exposed to DBP Infol         | lgn          |
| 211        | Table 4.22 Insidence of Neoplestic and Neoplestic Lesions of the Deperson in Male Data in the        | . 85         |
| 212        | Parinetal and Two year Food Study of DPD (NTD 2021a) <sup>4</sup>                                    | 86           |
| 213<br>214 | Table 4.23 Incidence of Interstitial Cell Hyperplasia and Adenomas of the Testis in Male Pats in the | . 80         |
| 214        | Perinatal and Two year Feed Study of DBP (NTP 2021a) <sup><math>a</math></sup>                       | 88           |
| 215        | Table 4-24 Incidence of Interstitial Cell Hyperplasia and Adenomas in Rats Exposed Gestationally to  | . 00         |
| 210        | DBP (Mylchreest et al. 1999) <sup><i>a</i></sup>   | ,<br>89      |
| 218        | Table 4-25. Incidence of Interstitial Cell Hyperplasia and Adenomas in Rats Exposed Gestationally to | )            |
| 219        | DBP (Mylchreest et al. 2000) <sup><i>a</i></sup>   | .90          |
| 220        | Table 4-26. Incidence of Interstitial Cell Hyperplasia and Adenomas in Rats Exposed Gestationally to | )            |
| 221        | DBP (Barlow et al., $2004)^a$  | . 90         |

May 2025

|     | Nuj 2023   |
|-----|--|
| 222 | Table 5-1. Summary of Physical and Chemical Properties of DCHP, DBP, DIBP, BBP, DEHP, DIDP,                      |
| 223 | and DINP   |
| 224 | Table 5-2. Summary of Acute Toxicity Data for DEHP, BBP, DBP, DIBP, DCHP, DINP, and DIDP <sup><i>a</i></sup> 102 |
| 225 | Table 5-3. Summary of Phthalate Potency for Reducing Fetal Testicular Testosterone       103                     |
| 226 | Table 5-4. Summary of Phthalate Syndrome-Related Effects Observed in Studies of Rat <sup>a</sup> 103             |
| 227 | Table 5-5. Summary of EPA Conclusions Regarding Genotoxicity and Mutagenicity of Phthalates 106                  |
| 228 | Table 5-6. Comparative Analysis of PPARα Activation by DIDP, DINP, DEHP, BBP, and DBP 107                        |
| 229 | Table 5-7. Summary of Non-cancer PODs Selected for Use in Human Health Risk Characterization for                 |
| 230 | DCHP, DIBP, DEHP, DBP, BBP, DINP, and DIDP109  |
| 231 | Table 5-8. Summary of Cancer Classifications for DEHP, BBP, DBP, DINP, and DIDP110                               |
| 232 |  |
| 233 |  |

### 234 LIST OF APPENDIX TABLES

| 235        | Table_Apx A-1. Genotoxicity of DEHP In Vitro (Studies Considered by ATSDR (2022)) <sup>a</sup>     |
|------------|--|
| 236        | Table_Apx A-2. Genotoxicity of MEHP In Vitro (Studies Considered by ATSDR (2022)) <sup>a</sup> 144 |
| 237        | Table_Apx A-3. Genotoxicity of DEHP In Vivo (Studies Considered by ATSDR (2022)) <sup>a</sup> 146  |
| 238        | Table_Apx A-4. Genotoxicity of MEHP In Vivo (Studies Considered by ATSDR (2022)) <sup>a</sup> 147  |
| 239        | Table_Apx A-5. Summary of NTP Genotoxicity Testing of DEHP (As Reported in NTP (2021b) 148         |
| 240        | Table_Apx B-1. Incidence of Liver Tumors in Male and Female B6C3F1 Mice Fed Diets Containing       |
| 241        | DEHP for Two Years (NTP, 1982a) <sup><i>a</i></sup>  |
| 242        | Table_Apx B-2. Incidence of Liver Tumors in Male and Female B6C3F1 Mice Fed Diets Containing       |
| 243        | DEHP for Two-years (David et al., 2000a; David et al., 1999) <sup>a</sup>                          |
| 244        | Table_Apx B-3. Incidence of Tumors in Male and Female F344 Rats Fed Diets Containing DEHP for      |
| 245        | Two Years (NTP, 1982a) <sup><i>a</i></sup>   |
| 246        | Table_Apx B-4. Incidence of Tumors in Male and Female F344 Rats Fed Diets Containing DEHP for      |
| 247        | Two-years (David et al., 2000b; David et al., 1999) <sup><i>a</i></sup>                            |
| 248        | Table_Apx B-5. Quantification of Liver Tumors by Size in Male F344 Rats Exposed to DEHP in the     |
| 249        | Diet for 108-weeks (Rao et al., $1990$ ) <sup><i>a</i></sup>                                       |
| 250        | Table_Apx B-6. Incidence of Liver Tumors in Male Sprague-Dawley Rats Chronically Fed Diets         |
| 251        | Containing DEHP (Voss et al., $2005)^a$  |
| 252        | Table_Apx B-7. Incidence of Testicular Tumors in Male Sprague-Dawley Rats Chronically Fed Diets    |
| 253        | Containing DEHP (Voss et al., 2005) <sup><i>a</i></sup>  |
| 254        | Table_Apx B-8. DEHP Intake (mg/kg-day) during the Gestational, Perinatal, and Two-year Phases of   |
| 255        | Chronic Dietary Study of DEHP with Sprague-Dawley Rats (NTP, 2021b) <sup><i>a</i></sup>            |
| 256        | Table_Apx B-9. Incidence of Liver Tumors in SD Rats Chronically Exposed to DEHP (Perinatal and     |
| 257        | Postweaning Exposure Study) (NTP, 2021b) <sup>4</sup>  |
| 258        | Table_Apx B-10. Incidence of Pancreatic Tumors in SD Rats Chronically Exposed to DEHP (Perinatal   |
| 259        | and Postweaning Exposure Study) (NTP, 2021b)"  |
| 260        | Table_Apx B-11. Incidence of Uterine Tumors in SD Rats Chronically Exposed to DEHP (Perinatal and  |
| 261        | Postweaning Exposure Study) (NTP, 2021b)"  |
| 262        | Table_Apx B-12. Incidence of Liver Tumors in SD Rats Exposed to DEHP in the Diet for Two-years     |
| 263        | $(NTP, 2021b)^{*}$   |
| 264<br>265 | Table_Apx B-13. Incidence of Pancreatic Tumors in SD Rats Exposed to DEHP in the Diet for Two-     |
| 265        | $years (N1P, 2021b)^{\circ} \dots 161$   |
| 266        | Table_Apx B-14. Incidence of Testicular Tumors in SD Rats Exposed to DEHP in the Diet for Two-     |
| 207        | years (N1P, 20210) <sup>°</sup> 162  |

|     |             | May 2025   |
|-----|-------------|--|
| 268 | Table_Apx H | 3-15. Incidence of Uterine Tumors in SD Rats Chronically Exposed to DEHP in the Diet for                   |
| 269 |             | Two-years (NTP, $2021b$ ) <sup><i>a</i></sup>  |
| 270 | Table_Apx I | 3-16. Summary of Neoplastic Lesions of the Liver Observed in RasH2 and Wild-type Mice                      |
| 271 |             | Fed Diets Containing DEHP for 26-weeks (Toyosawa et al., 2001) <sup><i>a</i></sup>                         |
| 272 | Table_Apx H | 3-17. Summary of Liver Tumors in Wild-type and <i>Ppara</i> -null Mice Fed Diets Containing                |
| 273 |             | DEHP for 22 Months (Ito et al., 2007a) <sup><i>a</i></sup>   |
| 274 | Table_Apx I | 3-18. Incidence of MNCL in Female F344 Rats Fed Diets Containing BBP for Two-years                         |
| 275 |             | (NTP, 1982b) <sup><i>a</i></sup>   |
| 276 | Table_Apx I | 3-19. Summary of Neoplastic Findings in the Pancreas and Urinary Bladder in F344/N Rats                    |
| 277 |             | Fed Diets Containing BBP for Two-years (NTP, 1997b)"   |
| 278 | Table_Apx I | 3-20. Incidence of Neoplasms and Non-neoplastic Lesions of the Pancreas, Urinary Bladder,                  |
| 279 |             | and MNCL in F344/N Rats (Ad Libitum and Weight-Matched Controls Protocols) (NTP,                           |
| 280 |             | 1997a)"  |
| 281 | Table_Apx I | 3-21. Incidence of Non-neoplastic and Neoplastic Findings in F344/N Rats Treated with                      |
| 282 |             | BBP (2-year Restricted Feed and Lifetime Restricted Feed Protocols) (NTP, 1997a) <sup><i>a</i></sup> . 172 |
| 283 | Table_Apx I | D-1. Summary of NOAEL and LOAEL Values for PPAR $\alpha$ Activation from In Vivo Animal                    |
| 284 |             | Toxicology Studies of DEHP "   |
| 285 |             |  |
| 286 |             |  |
| 287 | KEY AB      | BREVIATIONS AND ACRONYMS   |
| 288 | 2-EH        | 2-ethylhexanol   |
| 289 | ADME        | Adsorption, distribution, metabolism, and excretion  |
| 290 | AhR         | Aryl hydrocarbon receptor  |
| 291 | ANKCL       | Aggressive natural killer cell leukemia  |
| 292 | ATSDR       | Agency for Toxic Substances and Disease Registry   |
| 293 | BBP         | Butyl benzyl phthalate   |
| 294 | CAR         | Constitutive and receptor  |
| 295 | CASRN       | Chemical abstracts service registry number   |
| 296 | СНО         | Chinese hamster ovary  |
| 297 | CI          | Confidence interval  |
| 298 | CPSC        | Consumer Product Safety Commission (U.S.)  |
| 299 | DBP         | Dibutyl phthalate  |
| 300 | DEHP        | Di(2-ethylhexyl) phthalate   |
| 301 | DIBP        | Diisobutyl phthalate   |
| 302 | DIDP        | Diisodecyl phthalate   |
| 303 | DINP        | Diisononyl phthalate   |
| 304 | DNA         | Deoxyribonucleic acid  |
| 305 | ECB         | European Chemicals Bureau  |
| 306 | ECHA        | European Chemicals Agency  |
| 307 | EFSA        | European Food Safety Authority   |

- 308 EPA Environmental Protection Agency (U.S.)
- 309
   F344
   Fischer 344 rat
- 310 GD Gestation day
- 311HRHazard ratio212H. LuIntroduction 1
- 312 IL-1 $\alpha$  Interleukin 1-alpha
- 313 IL-1 $\beta$  Interleukin 1-beta
- 314IRISIntegrated Risk Information System

| 315 | KE     | Key event  |
|-----|--------|--|
| 316 | LGL    | Large granular lymphocyte  |
| 317 | LOAEL  | Lowest-observable-adverse-effect level   |
| 318 | MBP    | Monobutyl phthalate  |
| 319 | MBzP   | Monobenzyl phthalate   |
| 320 | MECPP  | Mono(2-ethyl-5-carboxypentyl) phthalate  |
| 321 | MEHP   | Mono(2-ethylhexyl) phthalate   |
| 322 | MEHHP  | Mono(2-ethyl-5-hydroxyhexyl) phthalate   |
| 323 | MEOHP  | Mono(2-ethyl-5-oxohexyl) phthalate   |
| 324 | MIBP   | Monoisobutyl phthalate   |
| 325 | MNCL   | Mononuclear cell leukemia  |
| 326 | MOA    | Mode of action   |
| 327 | MTD    | Maximum tolerable dose   |
| 328 | NF-ĸB  | Nuclear factor kappa B   |
| 329 | NICNAS | National Industrial Chemicals Notification and Assessment Scheme                     |
| 330 | NOAEL  | No-observed-adverse-effect level   |
| 331 | NTP    | National Toxicology Program (U.S.)   |
| 332 | OCSPP  | Office of Chemical Safety and Pollution Prevention                                   |
| 333 | OECD   | Organisation for Economic Co-operation and Development                               |
| 334 | OPPT   | Office of Pollution Prevention and Toxics  |
| 335 | OR     | Odds ratio   |
| 336 | PACT   | Pancreatic acinar cell tumor   |
| 337 | PBOX   | Peroxisomal β-oxidation  |
| 338 | PECO   | Population, exposure, comparator, and outcome  |
| 339 | PESS   | Potentially exposed or susceptible subpopulation(s)                                  |
| 340 | PND    | Postnatal day  |
| 341 | POD    | Point of departure   |
| 342 | PPARα  | Peroxisome proliferator activated receptor alpha                                     |
| 343 | PPRTV  | Provisional Peer-Reviewed Toxicity Value   |
| 344 | PVC    | Polyvinyl chloride   |
| 345 | PXR    | Pregnane X receptor  |
| 346 | ReCAAP | Rethinking Chronic Toxicity and Carcinogenicity Assessment for Agrochemicals Project |
| 347 | SACC   | Science Advisory Committee on Chemicals  |
| 348 | SCE    | Sister chromatid exchange  |
| 349 | SD     | Sprague Dawley   |
| 350 | SIR    | Standard incidence ratio   |
| 351 | TNFα   | Tumor necrosis factor alpha  |
| 352 | TSCA   | Toxic Substances Control Act   |
| 353 | U.S.   | United States  |
| 354 | WY     | WY 14,643 (also known as prinixic acid)  |
|     |        |  |

### 355 ACKNOWLEDGEMENTS

The Assessment Team gratefully acknowledges the participation, input, and review comments from the 356 357 U.S. Environmental Protection Agency (EPA or the Agency) Office of Pollution Prevention and Toxics (OPPT) and Office of Chemical Safety and Pollution Prevention (OCSPP) senior managers and science 358 359 advisors, as well as intra-agency reviewers. Special acknowledgement is given for the contributions of 360 technical experts from ORD, including Chris Corton. The Agency is also grateful for assistance from EPA contractors ICF (Contract No. 68HERC23D0007). 361 362 363 Docket 364 Supporting information can be found in the public dockets Docket IDs (EPA-HQ-OPPT-2018-0504, EPA-HQ-OPPT-2018-0434, EPA-HQ-OPPT-2018-0503, EPA-HQ-OPPT-2018-0433, and EPA-HQ-365 366 OPPT-2018-0501). 367 Disclaimer 368 369 Reference herein to any specific commercial products, process or service by trade name, trademark, 370 manufacturer, or otherwise does not constitute or imply its endorsement, recommendation, or favoring 371 by the United States Government. 372 373 Authors: Anthony Luz, John Allran, Christelene Horton, Ashley Peppriell, Collin Beachum (Branch 374 Supervisor) 375 376 **Contributors**: Devin Alewel, Keith Jacobs, Susanna Wegner 377 378 Technical Support: Hillary Hollinger and Mark Gibson

### 379 NOTE TO REVIEWERS

380 The non-cancer and cancer human health hazard assessments for diisononyl phthalate (DINP) and 381 diisodecyl phthalate (DIDP) were peer-reviewed by the Science Advisory Committee on Chemicals 382 (SACC) during the July 2024 peer review meeting. EPA's conclusions pertaining to the genotoxicity 383 and carcinogenicity of DIDP and DINP received favorable peer-reviews by the SACC (U.S. EPA, 384 2024q). Further, SACC recommended that, given the limitations and uncertainties regarding 385 mononuclear cell leukemia (MNCL) in Fischer 344 (F344) rats (Appendix C), MNCL should not be 386 considered as a factor in the determination of the cancer classifications for phthalates. Consistent with 387 this recommendation, EPA is not further considering MNCL as a factor in the determination of cancer 388 classifications for phthalates evaluated in this document. Further, SACC supported EPA's decision to 389 evaluate liver tumors in rats and mice caused by a peroxisome proliferator activated receptor alpha 390 (PPAR $\alpha$ ) mode of action (MOA) using a nonlinear, threshold approach.

391

392 <u>EPA is not at this time requesting additional SACC peer-review pertaining to the human health hazards</u>
 393 <u>of DIDP or DINP</u>. DIDP and DINP are included in this document for completeness with respect to the
 394 phthalates undergoing risk evaluation under TSCA. Moreover, the DIDP and DINP cancer evaluations
 395 are important to the overall weight of scientific evidence for DEHP, DBP, BBP, DIBP, and DCHP.
 396

397

398

### SUMMARY

399 This technical support document is in support of the TSCA Draft Risk Evaluations for di(2-ethylhexyl) 400 phthalate (DEHP) (U.S. EPA, 2025e), butyl benzyl phthalate (BBP) (U.S. EPA, 2025c), dibutyl 401 phthalate (DBP) (U.S. EPA, 2025d), diisobutyl phthalate (DIBP) (U.S. EPA, 2025f), and dicyclohexyl 402 phthalate (DCHP) (U.S. EPA, 20241). This document summarizes the genotoxicity and cancer hazards 403 associated with exposure to DEHP, BBP, DBP, DIBP, and DCHP. The genotoxicity and cancer hazards 404 of diisononyl phthalate (DINP) and diisodecyl phthalate (DIDP) have been evaluated by EPA previously 405 (U.S. EPA, 2025a, 2024n), but are briefly summarized in this document to support genotoxicity and 406 cancer hazard comparisons and read-across for the seven phthalate diesters currently being evaluated 407 under TSCA. 408

409 Available studies indicate that DEHP, BBP, DBP, DIBP, DCHP, DINP, and DIDP are not direct acting

- 410 genotoxicants or mutagens (Section 3). Rodent cancer bioassays are available for DEHP, BBP, DBP,
- 411 DINP and DIDP. EPA has previously concluded that DIDP is *Not likely to be carcinogenic to humans*
- 412 (U.S. EPA, 2024n). For DINP (Section 4.3.4), dose-related increases in hepatocellular adenomas and/or
- 413 carcinomas have been consistently observed in rats and mice of both sexes. EPA has previously
- 414 concluded that DINP causes liver tumors in rodents through a peroxisome proliferator activated receptor
- 415 alpha (PPARα) mode of action (MOA) (<u>U.S. EPA, 2025a</u>). Notably, this conclusion was supported by
- the Science Advisory Committee on Chemicals (SACC) during its July 2024 peer review meeting ( $\underline{U.S.}$
- 417 <u>EPA, 2024q</u>). Further, EPA has previously concluded that DINP is *Not Likely to be Carcinogenic to* 418 *Humans* at doses below levels that do not result in PPARα activation (U.S. EPA, 2025a).
- 419
- 420 For both BBP and DBP, EPA has preliminarily concluded that there is *Suggestive Evidence of*
- 421 *Carcinogenic Potential* of BBP and DBP in rodents based on evidence of pancreatic acinar cell
- 422 adenomas in rats (Sections 4.3.2.4 and 4.3.3.3). According to the *Guidelines for Carcinogen Risk*
- 423 Assessment (U.S. EPA, 2005), when there is Suggestive Evidence, "the Agency generally would not
- 424 attempt a dose-response assessment, as the nature of the data generally would not support one."

425 Consistently, EPA did not conduct a dose-response assessment for BBP or DBP and did not 426 quantitatively evaluate either phthalate for carcinogenic risk to human health.

427

428 For DEHP (Section 4.3), dose-related increases in hepatocellular adenomas and/or carcinomas have 429 been observed in rats and mice of both sexes, while dose-related increases in pancreatic acinar cell 430 tumors (PACTs) and Leydig cell tumors have been observed in male rats. As discussed in Section 431 4.3.1.1, EPA has preliminarily concluded that these tumor types, sometimes referred to as the 'tumor 432 triad', are related to PPARa activation. This conclusion is in part informed by inferences from 433 hypolipidemic drugs that lower lipid-levels in humans by activating PPARa, and also induce the tumor triad in rats, but not humans (Section 4.3.1.1.4). For DEHP, EPA has preliminarily concluded that 434 435 DEHP is Not Likely to be Carcinogenic to Humans at doses below levels that do not result in PPARa 436 activation. For both DINP and DEHP, the non-cancer points of departure (PODs) based on effects on the 437 developing male reproductive system consistent with phthalate syndrome and a disruption of androgen 438 action for DEHP or non-cancer liver toxicity (DINP) are lower than the hazard values for PPARa 439 activation identified by EPA. Therefore, EPA has concluded that the non-cancer PODs for DEHP and 440 DINP are expected to adequately account for all chronic toxicity, including carcinogenicity, and cancer 441 risk was not further quantified.

442

443 No chronic toxicity or cancer bioassays are reasonably available for DIBP or DCHP. Therefore, EPA 444 used elements of the Rethinking Chronic Toxicity and Carcinogenicity Assessment for Agrochemicals 445 Project (ReCAAP) weight of evidence framework (Hilton et al., 2022) as an organizational tool to 446 evaluate the extent to which the lack of carcinogenicity studies imparts significant uncertainty on the 447 human health risk assessments for DIBP and DCHP (Section 5). Human health hazards and 448 toxicokinetic properties of DIBP and DCHP were evaluated and compared to DEHP, BBP, DBP, DINP, 449 and DIDP (also referred to as "read-across phthalates" in this document). Overall, based on the weight 450 of scientific evidence, EPA preliminarily concludes that the lack of chronic toxicity data and 451 carcinogenicity bioassays for DIBP and DCHP do not suggest that there are significant remaining 452 scientific uncertainties in the qualitative and quantitative risk characterization for either of these 453 phthalates. Further, EPA has preliminarily concluded that the proposed non-cancer PODs for DIBP and 454 DCHP are health-protective, including for potentially exposed or susceptible subpopulation(s) (PESS). These proposed PODs for DIBP and DCHP are based on effects on the developing male reproductive 455 456 system consistent with a disruption of androgen action and phthalate syndrome that were selected for 457 characterizing risk from acute, intermediate and chronic exposure to DIBP and DCHP. These preliminary conclusions are based on several key weight of scientific evidence considerations (discussed 458 459 in Section 5). First, for the five read-across phthalates, effects on the developing male reproductive 460 system consistent with a disruption of androgen action and phthalate syndrome is a more sensitive and robust endpoint for deriving PODs for use in characterizing risk for acute, intermediate, and chronic 461 462 exposure scenarios than PPAR $\alpha$  mediated effects on the liver. The one exception to this was for DINP, 463 in which chronic non-cancer liver effects were identified as a more sensitive outcome than effects on the 464 developing male reproductive system for deriving a chronic POD. Second, EPA has determined that 465 quantitative cancer risk assessment is not needed for the read-across phthalates.

466

467 EPA is soliciting comments from the SACC and the public on its preliminary cancer classifications for
468 DEHP, BBP, and DBP; and its conclusion that lack of chronic toxicity and carcinogenicity studies are
469 not a significant source of scientific uncertainty for DIBP or DCHP.

### 470 1 INTRODUCTION AND SCOPE

471 In December 2019, the U.S. Environmental Protection Agency (EPA or the Agency) designated di(2-472 ethylhexyl) phthalate (DEHP, Chemical Abstracts Service Registry Number [CASRN] 117-81-7), butyl 473 benzyl phthalate (BBP, CASRN 85-68-7), dibutyl phthalate (DBP, CASRN 84-74-2), diisobutyl 474 phthalate (DIBP, CASRN 85-69-5), and dicyclohexyl phthalate (DCHP, CASRN 84-61-7) as high-475 priority substances for risk evaluation under the Toxic Substances Control Act (TSCA) (U.S. EPA, 476 2019a, b, c, d, e). Additionally, on May 24, 2019, EPA received requests from industry, pursuant to 40 477 CFR 702.37, to conduct risk evaluations for diisononyl phthalate (DINP, CASRNs 28553-12-0 and 478 68515-48-0) (ACC HPP, 2019b) and diisodecyl phthalate (DIDP, CASRNs 26761-40-0 and 68515-49-479 1) (ACC HPP, 2019a). The Agency determined that the requests met the applicable regulatory criteria 480 and requirements, as prescribed under 40 CFR 702.37, and granted the manufacturer-requested risk 481 evaluations for DIDP and DINP on December 2, 2019. As one of the first steps in the risk evaluation 482 process, EPA published the final scope documents for DEHP (U.S. EPA, 2020b), BBP (U.S. EPA, 483 2020a), DBP (U.S. EPA, 2020d), DIBP (U.S. EPA, 2020c), and DCHP (U.S. EPA, 2020e) in August 484 2020, fulfilling TSCA requirements under TSCA section 6(b)(4)(D) and as described in 40 CFR 485 702.41(c)(8). In August 2021, EPA published the final scope documents for DINP (U.S. EPA, 2021b) 486 and DIDP (U.S. EPA, 2021a). 487

488 Following publication of the final scope documents, one of the next steps in the TSCA risk evaluation 489 process is to identify and characterize the human health hazards and conduct dose-response assessments. 490 Non-cancer hazards associated with exposure DEHP, BBP, DBP, DIBP, DCHP, DINP, and DIDP are 491 summarized elsewhere in non-cancer human health hazard technical support documents (U.S. EPA, 492 2025a, k, 2024c, d, e, f, g, n). This technical support document summarizes the genotoxicity and cancer 493 hazards associated with DEHP, BBP, DBP, DIBP, DCHP, DINP, and DIDP. As will be discussed 494 further in Section 3 through Section 5 of this document, varying amounts of genotoxicity, human 495 epidemiologic, and animal cancer bioassays are available for DEHP, BBP, DBP, DIBP, DCHP, DINP, 496 and DIDP. DEHP, BBP, DBP, DINP, and DIDP have the most robust databases that include multiple 497 genotoxicity studies and animal cancer bioassays, while DIBP and DCHP have been evaluated for 498 genotoxicity in a limited number of studies and have not been evaluated for carcinogenicity in any two-499 year cancer bioassays. Therefore, data for DEHP, BBP, DBP, DIBP, DCHP, DINP, and DIDP is 500 summarized in this single document to support read-across and weight of scientific evidence conclusions 501 across the phthalates being evaluated.

- Genotoxicity and cancer hazards associated with exposure to DINP and DIDP have been summarized
  previously by EPA as part of the final human health hazard assessments and final risk evaluations of
  DIDP (U.S. EPA, 2024n, p) and DINP (U.S. EPA, 2025a, k, m). Conclusions from these assessments of
  DIDP and DINP are also briefly summarized and discussed in this technical support document to
  support read-across and weight of scientific evidence conclusions for DEHP, BBP, DBP, DIBP, and
  DCHP. However, EPA is not requesting additional SACC peer-review pertaining to the genotoxicity or
  carcinogenicity of DIDP or DINP at this time.
- 510
- 511 The remainder of this technical support document is organized as follows:
- Section 2 describes EPA's approach for identifying the genotoxicity, epidemiologic, and animal cancer studies discussed throughout this technical support document.
- Section 3 summarizes available genotoxicity data for DEHP, BBP, DBP, DIBP, DCHP, DINP,
   and DIDP.

- Section 4 summarizes available human and animal evidence for the carcinogenicity of DEHP,
   BBP, DBP, DINP, and DIDP. This section includes information pertaining to mode of action (MOA) analysis and EPA's preliminary weight of scientific evidence conclusions and cancer classifications for each phthalate.
- Section 5 describes application of a read-across framework known as the Rethinking Chronic Toxicity and Carcinogenicity for Agrochemicals Project, or the ReCAAP Framework (<u>OECD</u>, 2024; <u>Hilton et al.</u>, 2022), for DIBP and DCHP.
- Appendix A provides additional details on the extensive data on genotoxicity for DEHP.
- Appendix B provides additional details on rodent carcinogenicity studies for DEHP and BBP.
- Appendix C provides discussion of scientific uncertainties related to incidence of mononuclear
   cell leukemia (MNCL) and Leydig cell tumors in Fischer (F344) rats.
- Appendix D provides additional details on studies of DEHP investigating peroxisome
   proliferator activated receptor alpha (PPARα) activation in *in vivo* experimental animal models.

May 2025

# 529 2 APPROACH TO IDENTIFYING EPIDEMIOLOGY AND 530 LABORATORY ANIMAL DATA

531 EPA utilized a similar approach to identifying and integrating human epidemiologic, genotoxicity, 532 experimental animal cancer bioassays, and mechanistic information for DEHP, BBP, DBP, DIBP, 533 DCHP, DINP, and DIDP as previously described in EPA's draft non-cancer human health hazard 534 assessments for DEHP, BBP, DBP, DIBP, and DCHP (U.S. EPA, 2024c, d, e, f, g) and final human health hazard assessments for DINP and DIDP (U.S. EPA, 2025a, k, 2024n). EPA first reviewed 535 536 existing assessments of DEHP, BBP, DBP, DIBP, DCHP, DINP, and DIDP conducted by various 537 regulatory and authoritative agencies. Existing assessments reviewed by EPA are listed below. The 538 purpose of this review was to identify information relevant to assessing carcinogenicity, as well as 539 conclusions pertaining to the genotoxicity and carcinogenicity of these phthalates by various 540 authoritative and regulatory agencies. In addition to the information identified through review of 541 existing phthalate assessments, EPA also considered population, exposure, comparator, and outcome 542 (PECO)-relevant literature identified through the 2019 TSCA literature searches described in the 543 systematic review protocols for DEHP (U.S. EPA, 2025i), BBP (U.S. EPA, 2025g), DBP (U.S. EPA, 2025h), DIBP (U.S. EPA, 2025j), DCHP (U.S. EPA, 2024m), DINP (U.S. EPA, 2025n), and DIDP 544 545 (U.S. EPA, 2024r) in assessing the carcinogenicity of these phthalates.

| 546 | • | Integrated Risk Information System (IRIS), Chemical Assessment Summary, Dibutyl Phthalate; |
|-----|---|--|
| 547 |   | CASRN 84-74-2 (U.S. EPA, 1987);  |

- Integrated Risk Information System (IRIS), Chemical Assessment Summary, Butyl Benzyl
   Phthalate; CASRN 85-68-7 (U.S. EPA, 1988a);
- Integrated Risk Information System (IRIS), Chemical Assessment Summary, Di(2ethylhexyl)phthalate (DEHP); CASRN 117-81-7 (U.S. EPA, 1988b);
- *Provisional Peer Reviewed Toxicity Values for Butyl Benzyl Phthalate* (U.S. EPA, 2002);
- *Toxicological Profile for Di-b-phthalate* (<u>ATSDR, 2001</u>);
- Toxicological Profile for Di(2-Ethylhexyl)Phthalate (DEHP) (ATSDR, 2022);
- Toxicity Review of Di-n-butyl Phthalate (DBP) (U.S. CPSC, 2010b);
- Toxicity Review for Benzyl-n-butyl Phthalate (BBP) (U.S. CPSC, 2010a);
- Toxicity Review of Dicyclohexyl Phthalate (DCHP) (U.S. CPSC, 2010e);
- Toxicity Review of Di(isodecyl) Phthalate (DIDP) (U.S. CPSC, 2010d);
- Toxicity Review of Di(2-ethylhexyl) Phthalate (DEHP) (U.S. CPSC, 2010c);
- Toxicity Review of Diisononyl Phthalate (DINP) (U.S. CPSC, 2010f);
- Toxicity Review of Diisobutyl Phthalate (DiBP, CASRN 84-69-5) (U.S. CPSC, 2011);
- Chronic Hazard Advisory Panel on Phthalates and Phthalate Alternatives (U.S. CPSC, 2014);
- *NTP-CERHR Monograph on the Potential Human Reproductive and Developmental Effects of Di-isodecyl Phthalate (DIDP)* (NTP-CERHR, 2003b);
- *NTP-CERHR Monograph on the Potential Human Reproductive and Developmental Effects of Di-n-Butyl Phthalate (DBP)* (<u>NTP-CERHR, 2003d</u>);

- 567 NTP-CERHR Monograph on the Potential Human Reproductive and Developmental Effects of 568 Butyl Benzyl Phthalate (BBP) (NTP-CERHR, 2003a);
- *NTP-CERHR Monograph on the Potential Human Reproductive and Developmental Effects of Di-isononyl Phthalate (DINP)* (NTP-CERHR, 2003c);
- NTP-CERHR Monograph on the Potential Human Reproductive and Developmental Effects of Di(2-ethylhexyl) Phthalate (DEHP) (NTP-CERHR, 2006);
- Safe Drinking Water and Toxic Enforcement Act of 1986 Proposition 65. Initial Statement of Reasons. Title 27, California Code of Regulations. Proposed amendment to Section 25805(b), Specific Regulatory Levels: Chemicals Causing Reproductive Toxicity. Butyl Benzyl Phthalate (Oral Exposure) (OEHHA, 1986);
- Proposition 65 Maximum Allowable Dose Level (MADL) for Reproductive Toxicity for Di(nbutyl)phthalate (DBP) (OEHHA, 2007);
- Evidence on the Carcinogenicity of Butyl Benzyl Phthalate (<u>OEHHA, 2013b</u>);
- Chemical Listed Effective December 20, 2013 as Known to the State of California to Cause
   Cancer: Diisononyl Phthalate (DINP) (OEHHA, 2013a);
- Application of Systematic Review Methods in an Overall Strategy for Evaluating Low-dose
   Toxicity from Endocrine Active Chemicals (NASEM, 2017);
- Bis(2-ethylhexyl) Phthalate (Environment Canada, 1994);

585

586

- *Canadian Environmental Protection Act Priority Substances List Assessment Report: Dibutyl Phthalate* (EC/HC, 1994);
- 587 Canadian Environmental Protection Act Priority Substances List Assessment Report:
   588 Butylbenzylphthalate (Environment Canada, 2000);
- Supporting Documentation: Carcinogenicity of Phthalates Mode of Action and Human Relevance (Health Canada, 2015);
- 591 State of the Science Report: Phthalate Substance Grouping: Medium-chain Phthalate Esters:
   592 Chemical Abstracts Service Registry Numbers: 84-61-7; 84-64-0; 84-69-5; 523-31-9; 5334-09 593 8;16883-83-3; 27215-22-1; 27987-25-3; 68515-40-2; 71888-89-6 (EC/HC, 2015b);
- State of the Science Report: Phthalates Substance Grouping: Long-chain Phthalate Esters. 1,2-Benzenedicarboxylic Acid, Diisodecyl Ester (Diisodecyl Phthalate; DIDP) and 1,2-Benzenedicarboxylic Acid, Diundecyl Ester (Diundecyl Phthalate; DUP). Chemical Abstracts Service Registry Numbers: 26761-40-0, 68515-49-1; 3648-20-2 (EC/HC, 2015c);
- State of the Science Report: Phthalate Substance Grouping 1,2-Benzenedicarboxylic Acid, Diisononyl Ester; 1,2-Benzenedicarboxylic Acid, di-C8-10-branched Alkyl Esters, C9-rich (Diisononyl Phthalate; DINP). Chemical Abstracts Service Registry Numbers: 28553-12-0 and 601 68515-48-0 (EC/HC, 2015a);
- Supporting Documentation: Evaluation of Epidemiologic Studies on Phthalate Compounds and their Metabolites for Hormonal Effects, Growth and Development and Reproductive Parameters (Health Canada, 2018b);
- Supporting Documentation: Evaluation of Epidemiologic Studies on Phthalate Compounds and their Metabolites for Effects on Behaviour and Neurodevelopment, Allergies, Cardiovascular

May 2025

| 607<br>608        | <i>Function, Oxidative Stress, Breast Cancer, Obesity, and Metabolic Disorders</i> ( <u>Health Canada,</u> <u>2018a</u> );  |
|-------------------|---|
| 609               | • Screening Assessment - Phthalate Substance Grouping (ECCC/HC, 2020);  |
| 610<br>611        | • European Union Risk Assessment Report, vol 36: 1,2-Benzenedicarboxylic Acid, Di-C9-11-<br>Branched Alkyl Esters, C10-Rich and Di-"isodecyl"phthalate (DIDP) (ECB, 2003a);   |
| 612<br>613        | • European Union Risk Assessment Report: 1,2-Benzenedicarboxylic Acid, di-C8-10-Branched Alkyl Esters, C9-rich - and Di-"isononyl" Phthalate (DINP) (ECB, 2003b);   |
| 614<br>615        | • European Union Risk Assessment Report: Dibutyl Phthalate with Addendum to the Environmental Section (ECB, 2004);  |
| 616               | • European Union Risk Assessment Report: Benzyl Butyl Phthalate (BBP) (ECB, 2007);  |
| 617               | • European Union Risk Assessment Report: Bis(2-ethylhexyl)phthalate (DEHP) (ECB, 2008);   |
| 618<br>619<br>620 | • Substance Name: Benzyl Butyl Phthalate, EC Number: 201-622-7, CAS Number: 85-68-7:<br>Member State Committee Support Documentation for Identification of Benzyl Butyl Phthalate<br>(BBP) as a Substance of Very High Concern (ECHA, 2008);                  |
| 621<br>622<br>623 | • Evaluation of New Scientific Evidence Concerning the Restrictions Contained in Annex XVII to Regulation (EC) No 1907/2006 (REACH): Review of New Available Information for Dibutyl Phthalate (DBP) CAS No 84-74-2 Einecs No 201-557-4 (ECHA, 2010a);        |
| 624<br>625<br>626 | • Evaluation of New Scientific Evidence Concerning the Restriction Contained in Annex XVII to Regulation (EC) No. 1907/2006 (REACH): Review of New Available Information for Benzyl Butyl Phthalate (BBP) CAS No. 85-68-7 Einecs no. 201-622-7 (ECHA, 2010b); |
| 627<br>628<br>629 | • Annex XV Restriction Report: Proposal for a Restriction, Version 2. Substance Name: Bis(2-<br>ehtylhexyl)phthlate (DEHP), Benzyl Butyl Phthalate (BBP), Dibutyl Phthalate (DBP), Diisobutyl<br>Phthalate (DIBP) (ECHA, 2011);                               |
| 630<br>631        | • Committee for Risk Assessment (RAC) Opinion on an Annex XV Dossier Proposing Restrictions on Four Phthalates (ECHA, 2012b);   |
| 632<br>633<br>634 | • Committee for Risk Assessment (RAC) Committee for Socio-economic Analysis (SEAC):<br>Background Document to the Opinion on the Annex XV Dossier Proposing Restrictions on Four<br>Phthalates (ECHA, 2012a);   |
| 635<br>636        | • Evaluation of New Scientific Evidence Concerning DINP and DIDP in Relation to Entry 52 of Annex XVII to REACH Regulation (EC) No 1907/2006 (ECHA, 2013);  |
| 637<br>638<br>639 | • Committee for Risk Assessment RAC Opinion Proposing Harmonised Classification and Labelling at EU Level of Dicyclohexyl Phthalate, EC Number: 201-545-9, CAS Number: 84-61-7 (ECHA, 2014);  |
| 640<br>641        | • Opinion on an Annex XV Dossier Proposing Restrictions on Four Phthalates (DEHP, BBP, DBP, DIBP) (ECHA, 2017b);  |
| 642<br>643        | <ul> <li>Annex to the Background Document to the Opinion on the Annex XV Dossier Proposing<br/>Restrictions on Four Phthalates (DEHP, BBP, DBP, DIBP) (ECHA, 2017a);</li> </ul>   |
| 644<br>645<br>646 | • Opinion of the Scientific Panel on Food Additives, Flavourings, Processing Aids and Materials in Contact with Food (AFC) Related to Di-isodecylphthalate (DIDP) for Use in Food Contact Materials (EFSA, 2005e);  |
|                   |   |

- Opinion of the Scientific Panel on Food Additives, Flavourings, Processing Aids and Materials
   in Contact with Food (AFC) on a Request from the Commission Related to Di-isononylphthalate
   (DINP) for Use in Food Contact Materials (EFSA, 2005a);
- Opinion of the Scientific Panel on Food Additives, Flavourings, Processing Aids and Materials
   in Contact with Food (AFC) Related to Butylbenzylphthalate (BBP) for Use in Food Contact
   Materials (EFSA, 2005c);
- Opinion of the Scientific Panel on Food Additives, Flavourings, Processing Aids and Materials
   In Contact With Food (AFC) Related to Bis(2-ethylhexyl)phthalate (DEHP) for Use in Food
   Contact Materials (EFSA, 2005b);
- Opinion of the Scientific Panel on Food Additives, Flavourings, Processing Aids and Materials
   in Contact with Food (AFC) Related to Di-Butylphthalate (DBP) for Use in Food Contact
   Materials (EFSA, 2005d);
- Update of the Risk Assessment of Di-butylphthalate (DBP), Butyl-benzyl-phthalate (BBP), Bis(2ethylhexyl)phthalate (DEHP), Di-isononylphthalate (DINP) and Di-isodecylphthalate (DIDP) for Use in Food Contact Materials (EFSA, 2019);
- Existing Chemical Hazard Assessment Report: Diisobutyl Phthalate (NICNAS, 2008a);
- Phthalates Hazard Compendium: A Summary of Physicochemical and Human Health Hazard
   Data for 24 Ortho-phthalate Chemicals (NICNAS, 2008c);
- Priority Existing Chemical Draft Assessment Report: Diethylhexyl Phthalate (NICNAS, 2010);
- Priority Existing Chemical Assessment Report no. 35: Diisononyl Phthalate (NICNAS, 2012);
- Priority Existing Chemical Assessment Report no. 36: Dibutyl Phthalate (NICNAS, 2013);
- Priority Existing Chemical Assessment Report no. 40: Butyl Benzyl Phthalate (NICNAS, 2015a);
- Priority Existing Chemical Draft Assessment Report: Diisodecyl Phthalate & Di-n-octyl
   Phthalate (NICNAS, 2015b);
- C4-6 Side Chain Transitional Phthalates: Human Health Tier II Assessment (NICNAS, 2016);
- Phthalate Exposure and Male Reproductive Outcomes: A Systematic Review of the Human Epidemiological Evidence (Radke et al., 2018);
- Phthalate Exposure and Female Reproductive and Developmental Outcomes: A Systematic
   Review of the Human Epidemiological Evidence (Radke et al., 2019b);
- Phthalate Exposure and Metabolic Effects: A Systematic Review of the Human Epidemiological
   Evidence (Radke et al., 2019a);
- Phthalate Exposure and Neurodevelopment: A Systematic Review and Meta-analysis of Human
   Epidemiological Evidence (Radke et al., 2020); and
- Hazards of Diisobutyl Phthalate (DIBP) Exposure: A Systematic Review of Animal Toxicology
   Studies (<u>Yost et al., 2019</u>).

#### PUBLIC RELEASE DRAFT May 2025 **HAZARD IDENTIFICATIO**

### 682 3 GENOTOXICITY HAZARD IDENTIFICATION

Understanding the carcinogenic MOA of a chemical substance is an important consideration in
determining the most appropriate approach for cancer dose-response assessment, including use of a
linear vs. nonlinear approach. Consistent with EPA's *Guidelines for Carcinogen Risk Assessment* (U.S.
EPA, 2005), chemical substances with anticipated mutagenic MOAs are assessed with a linear approach.
In this section, EPA reviews available genotoxicity and mutagenicity data for DEHP (Section 3.1), BBP
(Section 3.2), DBP (Section 3.3), DIBP (Section 3.4), DCHP (Section 3.5), DINP (Section 3.6), and
DIDP (Section 3.7).

690

691

### 3.1 Di(2-ethylhexyl) Phthalate (DEHP)

692 The genotoxicity of DEHP and its major metabolites (e.g., mono(2-ethylhexyl) phthalate [MEHP] and 2-693 ethylhexanol [2-EH]) have been evaluated extensively in various *in vitro* and *in vivo* test systems. 694 Available genotoxicity studies have been reviewed by several authoritative and regulatory agencies. The U.S. Consumer Product Safety Commission (U.S. CPSC) (U.S. CPSC, 2010c), European Chemicals 695 696 Agency (ECHA) (ECHA, 2017a, b), European Food Safety Authority (EFSA) (EFSA, 2019), and Australia National Industrial Chemicals Notification and Assessment Scheme (NICNAS) (NICNAS) 697 2010) have concluded that the overall evidence supports the conclusion that DEHP is non-genotoxic and 698 699 non-mutagenic. Similarly, the European Chemicals Bureau (ECB) (ECB, 2008) and Environment 700 Canada (1994) concluded that DEHP and its major metabolites (*i.e.*, MEHP and 2-EH) are not genotoxic 701 or mutagenic.

702

More recently, the database of *in vitro* and *in vivo* genotoxicity studies of DEHP was reviewed by the Agency for Toxic Substances and Disease Registry (ATSDR) (<u>ATSDR</u>, 2022) and National Toxicology Program (NTP) (<u>NTP</u>, 2021b). ATSDR reviewed *in vitro* and *in vivo* genotoxicity studies of DEHP (76 *in vitro* studies and 39 *in vivo* studies) and MEHP (36 *in vitro* studies and 5 *in vivo* studies), which are summarized in Table\_Apx A-1, Table\_Apx A-2, Table\_Apx A-3, and Table\_Apx A-4. Overall, ATSDR concluded:

709

"DEHP has been extensively tested in a variety of genotoxicity assays. Evidence suggests
that DEHP is not mutagenic to bacterial or mammalian cells; however, there is limited
evidence that it may damage DNA and/or result in chromosomal abnormalities (either
directly or indirectly via oxidative stress mechanisms), and it has been shown to induce
morphological transformation. The weight of evidence from these assays indicates that

- morphological transformation. The weight of evidence from these assays indicates that
   DEHP is not a potent genotoxin but may lead to genotoxic effects secondary to oxidative
   stress."
- 716 717

Similarly, NTP (2021b) has tested DEHP in a range of *in vitro* and *in vivo* genotoxicity studies, some of which were not considered as part of the ATSDR assessment and generally found negative results (see Table\_Apx A-5). Overall, NTP concluded "The consensus from published data is that DEHP shows limited evidence of genotoxic potential, and for the sporadic positive results that have been reported, the response is either weak, not reproducible, obtained in a nonstandard test system, or qualified to some degree by the authors."

724

Herein, EPA did not independently re-evaluate the extensive database of *in vitro* and *in vivo*genotoxicity studies of DEHP and its major metabolites. However, a summary of available genotoxicity
studies considered most recently by ATSDR (2022) and conducted by NTP (2021b) are provided in
Appendix A. Overall, EPA agrees with the conclusions of ATSDR, NTP, and other authoritative and

May 2025

regulatory agencies that available evidence indicates that DEHP and its metabolites are not mutagenic,
 but that there is some limited evidence that DEHP may be weakly genotoxic, inducing effects such as
 deoxyribonucleic acid (DNA) damage and/or chromosomal aberrations. As noted by ATSDR, these
 effects may be secondary to oxidative stress.

733

### 734**3.2 Butyl Benzyl Phthalate (BBP)**

BBP has been evaluated for genotoxicity in a number of in vitro and in vivo test systems (see Table 3-1 735 736 for a summary of available assays). BBP did not demonstrate mutagenic activity in four in vitro bacterial 737 reverse mutation assays or in two *in vitro* mouse lymphoma assays with or without metabolic activation. 738 No increases in sister chromatid exchanges (SCE) or chromosomal aberrations were observed in studies 739 of Chinese hamster ovary (CHO) cells treated with BBP with or without metabolic activation (NTP, 740 1997b). BBP did not induce cell transformation in one study of Balb/c-3T3 A31 mouse cells (Monsanto, 741 1985). In a second study of Syrian hamster ovary cells, BBP did not induce a significant increase in 742 transformed foci when cells were incubated for 24 hours, while an increase in transformed foci was 743 observed after 7 days of incubation with BBP, albeit without a clear dose-response relationship (no 744 increase in foci was observed at the highest dose) (Leboeuf et al., 1996).

745

In in vivo studies, BBP did not induce sex-linked recessive lethal mutations in feed or injection studies 746 747 with Drosophila melanogaster (NTP, 1997b) and was negative in dominant lethal assays of B6C3F1 and CD-1 mice (Bishop et al., 1987). BBP did not induce micronuclei formation in one study of female 748 749 Sprague Dawley (SD) rats exposed to BBP via drinking water, albeit at an extremely low dose (*i.e.*, 750 182.6 µg/kg) (Ashby et al., 1997). In contrast, BBP did induce a significant increases in micronuclei formation in male B6C3F1 mice, but only at a very high dose (i.e., increased micronuclei observed at 751 5,000 mg/kg, but not at doses of 1,250-3,750 mg/kg), and only in trials in which cells were harvested 17 752 hours post-exposure, but not in the trial in which cells were harvested 36 hours post-exposure (NTP, 753 754 1997b). Similarly, treatment with high doses of BBP (1,250–5,000 mg/kg) resulted in a weakly positive 755 response in increased SCEs in male B6C3F1 mice in two trials conducted by NTP (1997b). However, in 756 one of the two trials, the positive trend (no statistically significant pairwise comparisons to the control) 757 in increased SCEs was observed only after data from the high-dose group was removed from the 758 analysis because there was no apparent increase in SCE in the high-dose animals.

759

760 Overall, available data support the conclusion that BBP is not likely to be mutagenic. Although BBP

- 761 was weakly positive for increased SCEs and chromosomal aberrations *in vivo*, the effects were only 762 weakly positive and only observed at extremely high doses of BBP (*i.e.*, 5,000 mg/kg). Notably, EP
- weakly positive and only observed at extremely high doses of BBP (*i.e.*, 5,000 mg/kg). Notably, EPA's
   conclusion is consistent with the conclusions of other authoritative and regulatory agencies. The ECB
- $(\underline{\text{ECB}}, 2007)$ , ECHA (2017a, b), and Australia NICNAS (2015a) concluded that BBP is not mutagenic,
- while EFSA (2019) concluded that available data for BBP do not give rise to a concern for genotoxicity.
- 766 Similarly, Environment Canada (2000) concluded "although the weight of evidence of genotoxicity is
- clearly negative, available data are inadequate to conclude unequivocally that BBP is not clastogenic,
- although in available studies it has induced, at most, weak activity." Finally, although U.S. CPSC
  (2010a) did not draw any specific conclusion on the genotoxicity of BBP, U.S. CPSC (2014) did
- 770 conclude that phthalate esters as a class are not genotoxic.

### 771 Table 3-1. Summary of Genotoxicity Studies of BBP

| Test<br>Type                     | Test System (Species/<br>Strain/ Sex)  | Dose/ Duration                                   | Metabolic<br>Activation                         | Result   | Reference   |  |  |  |
|----------------------------------|--|--|---|--|---|--|--|--|
|                                  | In Vitro – Gene Mutation Studies   |  |   |  |   |  |  |  |
| Reverse mutation assay           | <i>S. typhimurium</i> strains TA 98, 100, 1535, 1537   | 100 – 10,000 μg/plate                            | ± Aroclor induced<br>rat or hamster<br>liver S9 | Negative for<br>mutagenicity   | ( <u>NTP, 1997b</u> )   |  |  |  |
| Reverse mutation assay           | <i>S. typhimurium</i> strains TA 98, 100, 1535, 1537   | 333 – 11,550 μg/plate                            | ± Aroclor induced<br>rat or hamster<br>liver S9 | Negative for<br>mutagenicity   | ( <u>NTP, 1997b</u> )   |  |  |  |
| Reverse mutation assay           | <i>S. typhimurium</i> strains<br>TA 98, 100, 1535,<br>1537, 1538; <i>S.</i><br><i>cerevisiae</i> strain D4 | 0.1 – 10 μL/plate                                | ± Aroclor induced<br>rat liver S9               | Negative for<br>mutagenicity   | ( <u>Monsanto, 1976a</u> ) as<br>reported in ( <u>ECB,</u><br><u>2007</u> ) |  |  |  |
| Reverse mutation assay           | <i>S. typhimurium</i> strains<br>TA 98, 100, 1535,<br>1537, 1538   | 0.001 – 10 μL/plate                              | ± Aroclor induced<br>rat liver S9               | Negative for<br>mutagenicity   | ( <u>Monsanto, 1976b</u> )  |  |  |  |
| Mouse lymphoma<br>mutation assay | L5178Y+/- mouse<br>lymphoma cells  | 0, 5, 10, 20, 30, 40, 60 nL/mL                   | ± Aroclor-induced<br>rat liver S9               | Negative for<br>mutagenicity   | ( <u>NTP, 1997b</u> )   |  |  |  |
| Mouse lymphoma<br>mutation assay | L5178Y+/- mouse<br>lymphoma cells  | 0, 0.06, 0.16, 0.32, 0.65, 1.25, 2.5, 5<br>μL/mL | ± Aroclor-induced<br>mouse liver S9             | Negative for<br>mutagenicity   | ( <u>Monsanto, 1976c</u> ) as<br>reported in ( <u>ECB,</u><br><u>2007</u> ) |  |  |  |
|                                  |  | In Vitro – Cytogenetic Stud                      | lies  |  |   |  |  |  |
| SCE                              | CHO cells  | Trial 1: 0, 0.4, 1.25, 4.0 µg/mL                 | Without S9                                      | Trial 1: Equivocal   | ( <u>NTP, 1997b</u> )   |  |  |  |
|                                  |  | Trial 2: 0, 0.4, 1.25, 4.0, 12.5 µg/mL           | Without S9                                      | (trend in increased<br>SCE) (Trial 1);<br>Trial 2: Negative for<br>SCE<br><u>Overall: Negative for</u><br><u>SCE</u> |   |  |  |  |
|                                  |  | Trial 3: 0, 125, 400, 1250 μg/mL                 | With induced<br>liver S9                        | Negative for SCE   |   |  |  |  |
| Chromosomal aberrations          | CHO cells  | 0, 125, 400, 1250 μg/mL                          | ± Aroclor induced<br>liver S9                   | Negative for<br>chromosomal<br>aberrations   | ( <u>NTP, 1997b</u> )   |  |  |  |

| Test<br>Type   | Test System (Species/<br>Strain/ Sex)  | Dose/ Duration  | Metabolic<br>Activation | Result   | Reference                      |
|--|--|---|-------------------------|--|--------------------------------|
|  |  | In Vitro - Other Genotoxicity A   | Issays                  |  | -                              |
| <i>In vitro</i> cell transformation                        | Syrian hamster embryo cells            | Cells treated with 0, 25, 50, 100, 150, 250 $\mu$ g/mL BBP for 24 hours   | Not specified           | No significant<br>increase in<br>transformed foci                                    | (Leboeuf et al., 1996)         |
|  |  | Cells treated with 0, 1, 2, 5, 10, 20 µg/mL<br>BBP for 7 days   | Not specified           | Increased in<br>transformed foci at 2,<br>5, and 10, but not 20<br>µg/mL dose groups |                                |
| <i>In vitro</i> cell transformation                        | Balb/c-3T3 A31 mouse cells             | 0.49 – 8000 nL/mL   | No                      | No significant<br>increase in<br>transformed foci                                    | ( <u>Monsanto, 1985</u> )      |
|  |  | In Vivo Studies   |                         |  |                                |
| Sex-linked recessive                                       | Drosophila<br>melanogaster             | 0, 10,000 ppm in feed   | NA                      | No induction of sex-   | ( <u>NTP, 1997b</u> )          |
| lethal mutations   |  | 0, 10,000 ppm in feed   | NA                      | linked recessive lethal mutations  |                                |
|  |  | 0, 500 ppm (injection)  | NA                      |  |                                |
| Mouse dominant lethal assay                                | B6C3F1 mice                            | Male mice given subcutaneous injections<br>of 400 to 4560 mg/kg BBP on days 1, 5,<br>and 10 and then mated with untreated<br>females. Fetuses examined 17 days after<br>start of mating period. | Negative                | Negative   | ( <u>Bishop et al., 1987</u> ) |
|  | CD-1 mice                              |   | Negative                | Negative   |                                |
| Chromosomal<br>aberrations in femoral<br>bone marrow cells | Female Alpk:AP <sub>f</sub> SD<br>rats | Dams exposed to 0 or approximately 182.6 µg/kg-day BBP via drinking water during gestation and lactation.   | NA                      | Negative for<br>micronuclei  | ( <u>Ashby et al., 1997</u> )  |
| Chromosomal<br>aberrations in femoral<br>bone marrow cells | Male B6C3F1 mice                       | Trial 1: Mice (10/dose) received<br>intraperitoneal injections of 0 (corn oil),<br>1250, 2500, 5000 mg/kg BBP. Cells<br>harvested 17 hours post-exposure.                                       | NA                      | Positive for<br>micronuclei (highest<br>dose only)                                   | ( <u>NTP, 1997b</u> )          |
|  |  | Trial 2: Mice (10/dose) received<br>intraperitoneal injections of 0 (corn oil),<br>1250, 3750, 5000 mg/kg BBP. Cells<br>harvested 17 hours post-exposure.                                       | NA                      | Positive for<br>micronuclei (highest<br>dose only)                                   |                                |
|  |  | Trial 3: Mice (10/dose) received intraperitoneal injections of 0 (corn oil),  | NA                      | Negative for<br>micronuclei  |                                |

| Test<br>Type               | Test System (Species/<br>Strain/ Sex) | Dose/ Duration  | Metabolic<br>Activation | Result   | Reference             |
|----------------------------|---------------------------------------|---|-------------------------|--|-----------------------|
|                            |                                       | 1250, 2500, 5000 mg/kg BBP. Cells harvested 36 hours post-exposure.   |                         |  |                       |
| SCE in femoral bone marrow | Male B6C3F1 mice                      | Mice (5/dose) received intraperitoneal<br>injections of 0 (corn oil), 1250, 2500,<br>5000 mg/kg BBP. Cells harvested 23<br>hours post-exposure. | NA                      | Weakly positive<br>response (positive<br>trend in increased<br>SCEs when highest<br>dose excluded) | ( <u>NTP, 1997b</u> ) |
|                            |                                       | Mice (5/dose) received intraperitoneal<br>injections of 0 (corn oil), 1250, 2500,<br>5000 mg/kg BBP. Cells harvested 42<br>hours post-exposure. | NA                      | Weakly positive<br>response by trends<br>analysis  |                       |
| Abbreviations: BBP = buty  | l benzyl phthalate; CHO = C           | Chinese hamster ovary; NA = not applicable; pp  | m = parts per million;  | SCE = sister chromatid exc   | change                |

### 773 **3.3 Dibutyl Phthalate (DBP)**

The mutagenic and genotoxic potential of DBP has been evaluated in 21 studies (Table 3-2). Available studies include two *in vivo* micronucleus tests in mice, two *in vitro* chromosomal aberration assays, one *in vitro* SCE assay, three *in vitro* mouse lymphoma assays, six bacterial mutation assays, two gene mutation assays (one in *Escherichia coli* and one in *Saccharomyces cerevisiae*), one *in vitro* cell transformation assay, and two comet assays with primary human cells.

779

780 DBP did not induce clastogenic effects or micronuclei formation in two *in vivo* studies of mice (NTP, 781 1995; BASF, 1990) or induce unscheduled DNA repair in E. coli or Bacillus subtilis (Omori, 1976; 782 Kurata, 1975). DBP induced DNA strand breaks in comet assays of primary human lymphocytes, 783 oropharyngeal cells, and mucosal cells (Kleinsasser et al., 2000b; Kleinsasser et al., 2000a). Exposure to 784 DBP did not cause an increase in cell transformation in one *in vitro* study of Balb/c-3T3 A31 mouse 785 cells (Litton Bionetics, 1985). DBP showed no mutagenic activity in gene mutation assays with E. coli 786 and S. cerevisiae (Shahin and Von Borstel, 1977; Omori, 1976; Kurata, 1975). DBP was negative for 787 mutagenic activity both with and without metabolic activation in four out of five reverse mutation assays 788 with several strains of S. typhimurium (NTP, 1995; Zeiger et al., 1985; Kozumbo et al., 1982; Florin et al., 1980; Omori, 1976; Kurata, 1975). Equivocal results were obtained in one bacterial reverse mutation 789 790 assay of S. typhimurium strains TA 100 and TA 1535 that included doses of 100 to 2,000 µg DBP per 791 plate (Agarwal et al., 1985). In TA 1535 a mild increase (<2x) in the number of revertant colonies was 792 observed at the two highest doses in the absence of S9. In TA 100, an increase in the number of 793 reversions was observed in the absence of S9, with a maximum response (<3x) occurring in the low-794 dose group. However, the response was not dose-dependent, was less than a factor of 2 at 200 µg DBP 795 per plate, and the effect plateaued at higher doses. No mutagenic activity was observed with metabolic 796 activation in TA 100 or TA 1535, and no mutagenic activity was observed in other strains with or 797 without metabolic activation. A marginally positive response was also observed in an 8-azaguanine 798 resistance assay with S. typhimurium strain TA 100 in the absence of metabolic activation (Seed, 1982). 799 A marginal increase (<2x) in mutagenic activity was observed at doses of 0.09 and 0.18 mM DBP, 800 which were also cytotoxic (all doses tested in the study resulted in approximately 50% cytotoxicity in 801 the absence of S9). No mutagenic activity was apparent with metabolic activation. 802

Mixed results have been obtained across three *in vitro* mouse lymphoma mutation assays (Barber et al.,
2000; NTP, 1995; Hazleton, 1986). In one of the three studies, a significant increase in mutagenic
activity was observed in the absence of metabolic activation (NTP, 1995). Only two of the three *in vitro*mouse lymphoma mutation assays tested DBP with S9 (Barber et al., 2000; Hazleton, 1986). In both
studies, DBP showed mutagenic activity in the presence of S9.

808

809 DBP did not induce chromosomal aberrations in one in vitro assay with CHO cells (Abe and Sasaki, 810 1977), while an equivocal result was obtained in a second poorly-reported study with Chinese hamster 811 lung fibroblasts (Ishidate and Odashima, 1977). Ishidate and Odashima report a six percent increase in chromosomal aberrations, which study authors characterized as a 'suspicious result.' However, no 812 813 statistical analysis was performed, and it is unclear if the small increase in chromosomal aberrations would be concentration-dependent by trend test, statistically significantly different than the concurrent 814 control, or outside the distribution of historical control data, which are criteria for considering if an in 815 816 *vitro* mammalian chromosomal aberration test is positive under current OECD 473 guidelines (OECD, 817 2016). Finally, treatment with DBP induced a slight (<2x), but statistically significant, increase in SCE 818 in one study of CHO cells, however, the increase in SCEs was not concentration-dependent. 819

820 Available genotoxicity data for DBP has been evaluated by numerous authoritative and regulatory 821 agencies. Based on the weight of evidence, Health Canada (EC/HC, 1994), the ECB (2004), ECHA 822 (2017a, b), Australia NICNAS (2013), and EFSA (2019) concluded that DBP is not genotoxic or 823 mutagenic. U.S. CPSC (2010b) did not draw any specific conclusion on the genotoxicity of DBP, 824 however, U.S. CPSC (2014) did conclude that phthalate esters as a class are not genotoxic. In contrast, 825 ATSDR (2001) concluded that results from available studies "suggest that di-n-butyl phthalate may be 826 weakly mutagenic *in vitro*. The significance of these findings to the intact mammalian organism is not 827 known because in vivo genotoxicity studies have not been conducted." However, in drawing this 828 conclusion, ATSDR did not take into consideration the two in vivo studies of mice that were both 829 negative for micronuclei formation. 830

831 Overall, available studies provide somewhat mixed results. Given the results of the mouse lymphoma 832 assays, it is difficult to conclude unequivocally that DBP is not genotoxic. However, as will be discussed

further in Section 4.3.3, DBP shows equivocal evidence of carcinogenic activity in male rats (based on a

slight increase in pancreatic acinar cell tumors [PACTs]), but no evidence of carcinogenic activity in

- female rats or mice of either sex. <u>Given the results of the *in vivo* and *in vitro* genotoxicity assays of DBP</u>
- 836 and *in vivo* carcinogenicity studies of DBP in rats and mice, EPA does not consider DBP to be a potent
- 837 genotoxicant. However, there is some limited evidence that DBP may be weakly genotoxic in some in
- 838 <u>vitro assays</u>.

### 839 Table 3-2. Summary of Genotoxicity Studies of DBP

| Test<br>Type                        | Test System (Species/<br>Strain/ Sex)                                  | Dose/ Duration   | Metabolic<br>Activation                                  | Result  | Reference   |
|-------------------------------------|--|--|--|---|---|
|                                     | •  | In Vivo Stud   | lies   | -   |   |
| Micronucleus test                   | Male and Female<br>B6C3F1/N mice                                       | 1,250–20,000 ppm DBP in<br>the diet for 3 months<br>(equivalent to 163–4,278<br>mg/kg-day) | NA   | Negative for micronuclei<br>formation in peripheral<br>blood erythrocytes     | ( <u>NTP, 1995</u> )                                |
| Micronucleus test                   | Male and Female<br>NMRI mice   | Mice gavaged once with 333, 1,000, or 3,000 mg/kg DBP in olive oil                         | NA   | Negative for micronuclei<br>formation in femoral<br>erythrocytes              | ( <u>BASF, 1990</u> )                               |
|                                     |  | In Vitro Gene Muta   | tion Studies   |   |   |
| Bacterial reverse<br>mutation assay | <i>S. typhimurium</i> strains TA 98, 100, 1535, 1537                   | 100–10,000 μg/plate  | ± Aroclor induced<br>rat or hamster<br>liver S9          | Negative for mutagenicity   | ( <u>NTP, 1995; Zeiger et</u><br><u>al., 1985</u> ) |
| Bacterial reverse<br>mutation assay | <i>S. typhimurium</i> strains TA 98, 100, 1535, 1537                   | 3 μmol/plate   | ± Aroclor induced<br>rat liver S9                        | Negative for mutagenicity<br>(precipitation of DBP<br>occurred)               | (Florin et al., 1980)                               |
| Bacterial reverse<br>mutation assay | <i>S. typhimurium</i> strains TA 98, TA 100                            | Up to 1,000 μg/plate   | ± Aroclor induced<br>rat liver S9                        | Negative for mutagenicity   | ( <u>Kozumbo et al., 1982</u> )                     |
| Bacterial reverse<br>mutation assay | <i>S. typhimurium</i> strain TA 100                                    | 10,000 μg/plate  | + Aroclor-induced<br>rat liver S9                        | Negative for mutagenicity   | ( <u>Omori, 1976; Kurata,</u><br><u>1975</u> )      |
| Bacterial reverse<br>mutation assay | <i>S. typhimurium</i> strains<br>TA 98, 100, 1535,<br>1537, 1538, 2637 | 100–2,000 μg/plate   | No   | Equivocal in TA 100 and TA 1535, but not in other strains <sup><i>a</i></sup> | ( <u>Agarwal et al., 1985</u> ) <sup>a</sup>        |
|                                     |  |  | + Aroclor induced<br>liver S9 (species<br>not specified) | Negative for mutagenicity   |   |
| Bacterial forward<br>mutation assay | <i>S. typhimurium</i> strain TA 100                                    | 0.045, 0.09, or 0.18 mM  | No   | Marginally positive (weak increases [<2x] at cytotoxic doses)                 | ( <u>Seed, 1982</u> )                               |
|                                     |  |  | + Aroclor-induced<br>rat liver S9                        | Negative for mutagenicity   |   |

### May 2025

| Test<br>Type                         | Test System (Species/<br>Strain/ Sex)       | Dose/ Duration  | Metabolic<br>Activation             | Result   | Reference   |
|--------------------------------------|---|---|-------------------------------------|--|---|
| Gene mutation assay                  | Escherichia coli<br>(uvrA-)                 | 10,000 μg/plate   | No                                  | Negative for mutagenicity  | ( <u>Omori, 1976; Kurata,</u><br><u>1975</u> )    |
| Gene mutation assay                  | S. cerevisiae (Xv 185-<br>14C)              | 10, 20, 100 µL/mL   | ± Aroclor induced<br>mouse liver S9 | Negative for mutagenicity  | ( <u>Shahin and Von</u><br><u>Borstel, 1977</u> ) |
| Mouse lymphoma<br>mutation assay     | L5178Y+/- mouse<br>lymphoma cells           | Trial 1: 15, 30, 40, 50, 60<br>nL/mL<br>Trial 2: 25, 50 nL/mL         | No                                  | Negative for mutagenicity<br>(both trials)   | ( <u>Hazleton, 1986</u> )                         |
|                                      |   | Trial 3: 12.5, 50, 75, 100,<br>150 nL/mL<br>Trial 4: 25, 50, 75 nL/mL | + Aroclor-induced<br>rat liver S9   | Marginally positive (trial<br>3) and positive for<br>mutagenicity (trial 4)                    |   |
| Mouse lymphoma<br>mutation assay     | L5178Y+/- mouse<br>lymphoma cells           | Trial 1: 12, 24, 36, 48, 60<br>μg/mL                                  | No                                  | Positive (increased mutant fraction at 48 µg/mL)   | ( <u>NTP, 1995</u> )                              |
|                                      |   | Trial 2: 0, 30, 38, 46, 54, 62, 70 µg/mL                              | No                                  | Positive (increased mutant fraction at 46 µg/mL)   |   |
| Mouse lymphoma                       | L5178Y+/- mouse                             | 0.015–0.06 μL/mL (-S9)  | No                                  | Negative for mutagenicity  | ( <u>Barber et al., 2000</u> )                    |
| mutation assay                       | lymphoma cells                              | 0.0125–0.15 µL/mL (+S9)   | + Aroclor-induced<br>rat liver S9   | Positive for mutagenicity  |   |
|                                      |   | In Vitro Cytogene   | tics Assays                         |  |   |
| Chromosomal aberrations              | Chinese hamster lung fibroblast cells       | 0.03–1.1 mg/mL for 24 hours   | No                                  | Marginally positive for chromosomal aberrations  | ( <u>Ishidate and</u><br><u>Odashima, 1977</u> )  |
| Chromosomal aberrations              | CHO cells                                   | 0.0001–0.001 M  | No                                  | Negative for chromosomal aberrations   | (Abe and Sasaki, 1977)                            |
| SCE                                  | CHO cells                                   | 0.0001–0.001 M  | No                                  | Marginally positive for<br>SCE (<2x increase, no<br>concentration-dependent -<br>relationship) | (Abe and Sasaki, 1977)                            |
|                                      |   | Other Genotoxic   | ity Assays                          |  |   |
| Bacterial test (indirect DNA-repair) | <i>Escherichia coli</i><br>(pol A-, rec A-) | 10,000 μg/plate   | No                                  | Negative   | ( <u>Omori, 1976; Kurata,</u><br><u>1975</u> )    |
| Bacterial test (indirect DNA-repair) | Bacillus subtilis (Rec<br>A-)               | 10,000 μg/plate   | No                                  | Negative   | ( <u>Omori, 1976; Kurata,</u><br><u>1975</u> )    |

### May 2025

| Test<br>Type  | Test System (Species/<br>Strain/ Sex)  | Dose/ Duration  | Metabolic<br>Activation | Result                                 | Reference                               |
|---|--|---|-------------------------|--|---|
| Cell transformation assay   | Balb/c-3T3 A31 mouse cells   | 0, 3.4, 13.7, 27.5, 55, 82.3<br>nL/mL                     | No                      | Negative                               | (Litton Bionetics, 1985)                |
| Comet assay   | Human: oropharyngeal<br>and nasal mucosa cells<br>from 40 and 30<br>patients, respectively | Cells incubated with 354<br>µmol/mL DBP for 60<br>minutes | No                      | ↑ DNA strand breaks in both cell types | ( <u>Kleinsasser et al.,</u><br>2000a)  |
| Comet assay   | Human: mucosal cells<br>and lymphocytes from<br>60 patients                                | Cells incubated with 354<br>µmol/mL DBP for 60<br>minutes | No                      | ↑ DNA strand breaks in both cell types | ( <u>Kleinsasser et al.</u> ,<br>2000b) |
| Abbreviations: DBP = dibutyl phthalate; CHO = Chinese hamster ovary; NA = not applicable; ppm = parts per million; SCE = sister chromatid exchange<br><sup>a</sup> For TA 100, treatment with DBP increased the number of revertant colonies per plate at all concentrations; however, the response was not concentration-dependent<br>(estimated mean # of revertants/plate at 0, 100, 200, 500, 750, 1,000, 1,500, 2,000 µg/plate: 125, 275, 200, 175, 200, 160, 175, 200, respectively). For TA 1535, a mild<br>(<2x), but statistically significant, increase in mean number of revertant colonies per plate was observed in the two highest dose concentrations. |  |   |                         |  |   |

840

### **3.4 Diisobutyl Phthalate (DIBP)**

Limited genotoxicity testing of DIBP has been conducted (Table 3-3). DIBP was negative for
mutagenicity in four bacterial reverse mutation assays conducted with several strains of *S. typhimurium*both with and without metabolic activation (Sato et al., 1994; Zeiger et al., 1985; Seed, 1982; Simmon et
al., 1977). In contrast, DIBP induced DNA strand breaks in several *in vitro* comet assays with human
mucosal cells and lymphocytes (Kleinsasser et al., 2001; Kleinsasser et al., 2000b; Kleinsasser et al., 2000a).

848

849 Due to limited data, most previous assessments of DIBP have determined that there is insufficient 850 information to determine the genotoxic potential of DIBP (Yost et al., 2019; EC/HC, 2015b; U.S. CPSC, 851 2011; NICNAS, 2008a). In contrast, ECHA (2017a, b) considered genotoxicity data of four phthalates 852 (*i.e.*, DEHP, BBP, DBP, DIBP), while Australia NICNAS (2016) considered data for eight phthalates 853 (*i.e.*, DIBP, DCHP, DBP, BBP, dihexyl phthlate, di(methoxyethyl) phthlate, dialkyl(C7-11-branched 854 and linear) phthalate, diisoheptyl phthlate). Based on the weight of evidence for all phthalates under 855 consideration, ECHA (2017a, b) concluded that DIBP is not mutagenic in *in vitro* tests, while NICNAS 856 (2016) concluded that DIBP is not expected to have mutagenic or genotoxic potential in humans.

As discussed further in Section 3.8, although limited genotoxicity testing of DIBP has been conducted,
<u>EPA does not consider DIBP likely to be genotoxic or mutagenic to humans based on read-across from</u>
<u>DEHP, BBP, DBP, DINP and DIDP</u>.

861 862

| Test<br>Type                         | Test System (Species/<br>Strain/Sex)   | Dose/Duration  | Metabolic<br>Activation                             | Result                                       | Reference   |
|--------------------------------------|--|--|---|--|---|
|                                      |  | In Vitro Gene Mutation As                                  | says  |  |   |
| Reverse<br>mutation                  | <i>S. typhimurium</i> TA 98,<br>TA 100, TA 1535, TA<br>1537                                | 0, 100, 333, 1,000, 3,333,<br>10,000 µg/plate              | ± Aroclor-<br>induced rat or<br>hamster liver<br>S9 | Negative for<br>mutagenicity                 | ( <u>Zeiger et</u><br><u>al., 1985;</u><br><u>Zeiger et</u><br><u>al., 1982</u> ) |
| Reverse<br>mutation                  | S. typhimurium TA 98   | 0.25–500 µmol/plate  | ± Aroclor-<br>induced rat<br>liver S9               | Negative for mutagenicity                    | ( <u>Sato et al.,</u><br><u>1994</u> )  |
| Reverse mutation <sup><i>a</i></sup> | S. typhimurium TA 100  | Not reported <sup><i>a</i></sup>                           | ± S9 <sup><i>a</i></sup>                            | Negative for mutagenicity                    | ( <u>Seed,</u><br><u>1982</u> )   |
| Reverse mutation <sup>b</sup>        | <i>S. typhimurium</i> TA 98,<br>TA 100, TA 1538, TA<br>1537, TA 1535                       | Not reported <sup>b</sup>                                  | ± Aroclor-<br>induced rat<br>liver S9               | Negative for mutagenicity                    | ( <u>Simmon et</u><br><u>al., 1977</u> )  |
|                                      |  | Other Genotoxicity Assa                                    | iys   |  |   |
| In vitro comet<br>assay              | Human: oropharyngeal<br>and nasal mucosa cells<br>from 40 and 30<br>patients, respectively | Cells incubated with 354<br>µmol/mL DIBP for 60<br>minutes | No  | ↑ DNA strand<br>breaks in both<br>cell types | ( <u>Kleinsasser</u><br>et al.,<br>2000a)   |
| In vitro comet<br>assay              | Human: mucosal cells<br>and lymphocytes from<br>60 patients                                | Cells incubated with 354<br>µmol/mL DIBP for 60<br>minutes | No  | ↑ DNA strand<br>breaks in both<br>cell types | ( <u>Kleinsasser</u><br>et al.,<br>2000b)   |

### 863 Table 3-3. Summary of Genotoxicity Studies of DIBP

| Way 2023       |                      |                          |    |                |                       |  |  |
|----------------|----------------------|--------------------------|----|----------------|-----------------------|--|--|
| In vitro comet | Human: oropharyngeal | Cells incubated with 354 | No | ↑ DNA strand   | (Kleinsasser          |  |  |
| assay          | mucosa cells and     | µmol/mL DIBP for 60      |    | breaks in both | <u>et al., 2001</u> ) |  |  |
|                | lymphocytes from 132 | minutes                  |    | cell types     |                       |  |  |
|                | and 49 patients,     |                          |    |                |                       |  |  |
|                | respectively         |                          |    |                |                       |  |  |

<sup>*a*</sup> Seed (<u>1982</u>) tested bacteria for mutations to azaguanine resistance and reversion to histidine prototrophy. Tested concentrations of DIBP were not reported. The maximal concentration tested was determined by either the solubility limit or cytotoxicity exceeding 90% of control values. Study authors report that experiments were conducted with S9 mix, however, assay results for DIBP are reported as negative, and it is unclear if this negative result was for studies with or without S9 mix.

<sup>b</sup> Simmon et al. (1977) report that a "wide range of doses was tested up to 5 mg/plate or a dose which gave a toxic response, whichever was lower."

864

865

870

### 3.5 Dicyclohexyl Phthalate (DCHP)

Limited genotoxicity testing of DCHP has been conducted (Table 3-4). Reasonably available information includes one bacterial reverse mutation study. DCHP was negative for mutagenicity in the one available bacterial reverse mutation assay that was conducted with several strains of *S. typhimurium* both with and without metabolic activation (Zeiger et al., 1985).

871 EPA also identified several additional genotoxicity studies of DCHP reported in the ECHA Dossier

Publication for DCHP (<u>https://chem.echa.europa.eu/100.001.405/dossier-view/4742a866-2c9d-40f3-</u>
 b353-f6d166324c0d/e15fb14e-d558-465c-ba63-11a1b792aa74 e15fb14e-d558-465c-ba63-

11a1b792aa74). In the dossier, registrants report that DCHP was negative for mutagenicity in one study
 that adhered to OECD Guidaling No. 471 (Reptarial Reverse Mutation Test), was negative for induction

that adhered to OECD Guideline No. 471 (Bacterial Reverse Mutation Test), was negative for induction
 of chromosomal aberrations in one study that adhered to OECD Guideline No. 473 (*In Vitro* Mammalian

877 Chromosomal Aberration Test), and was negative for mutagenicity in one study that adhered to OECD

Guideline No. 476 (*In Vitro* Mammalian Cell Gene Mutation Test). However, original study reports
 were not reasonably available to EPA for independent review, so the results of these studies are not
 considered further.

881

612 Given the limited genotoxicity testing that has been conducted for DCHP, Health Canada and U.S.

CPSC refrained from drawing any conclusions regarding the genotoxicity of DCHP (EC/HC, 2015b;
 <u>U.S. CPSC, 2010e</u>). However, U.S. CPSC (2014) has more generally concluded that phthalate esters as a
 class are not genotoxic. As discussed further in Section 3.8, although limited genotoxicity testing of
 DCHP has been conducted, EPA does not consider DCHP likely to be genotoxic or mutagenic to
 humana based on mod across from DEUP, BBB, DBB, DINB and DIDB

887 <u>humans based on read-across from DEHP, BBP, DBP, DINP and DIDP</u>.

888 889

### 890 Table 3-4. Summary of Genotoxicity Studies of DCHP

| Test<br>Type | Test System<br>(Species/<br>Strain/Sex) | Dose/Duration       | Metabolic<br>Activation | Result       | Reference       |
|--------------|---|---------------------|-------------------------|--------------|-----------------|
| Reverse      | S. typhimurium                          | 0, 100, 333, 1,000, | ± Aroclor-              | Negative for | (Zeiger et al., |
| mutation     | TA 98, TA 100,                          | 3,333, 10,000       | induced                 | mutagenicity | <u>1985</u> )   |
|              | TA 1535, TA                             | µg/plate            | rat or                  |              |                 |
|              | 1537                                    |                     | hamster                 |              |                 |
|              |   |                     | liver S9                |              |                 |

### **3.6 Diisononyl Phthalate (DINP)**

EPA has previously evaluated the mutagenic and genotoxic potential of DINP and concluded that the
weight of scientific evidence supports the conclusion that DINP is not likely to be genotoxic or
mutagenic (U.S. EPA, 2025a). This conclusion is based on results from 20 studies, including two *in vivo*micronucleus tests in rodents, one *in vitro* chromosomal aberration assay, two *in vitro* mouse lymphoma
assays, five bacterial reverse mutation assays, one *in vitro* unscheduled DNA synthesis assay, and nine *in vitro* cell transformation assays. Across available studies, DINP was negative for genotoxicity and
mutagenicity.

900

906

Notably, the SACC supported EPA's conclusions regarding the genotoxicity and mutagenicity of DINP
during the July 2024 peer review meeting of DIDP and DINP (<u>U.S. EPA, 2024q</u>). Consistently, Health
Canada, ECHA, Australia NICNAS, U.S. CPSC, and EFSA have also concluded that DINP is not
genotoxic nor is it likely to be genotoxic (<u>ECCC/HC, 2020; EC/HC, 2015a; ECHA, 2013; NICNAS,</u>
2012; U.S. CPSC, 2010f; EFSA, 2005a; ECB, 2003c; U.S. CPSC, 2001).

Readers are directed to EPA's *Cancer Human Health Hazard Assessment for Diisononyl Phthalate* (*DINP*) (U.S. EPA, 2025a) for further discussion of available genotoxicity data for DINP.

### 910 **3.7 Diisodecyl Phthalate (DIDP)**

911 EPA has previously evaluated the mutagenic and genotoxic potential of DIDP and concluded that the 912 weight of scientific evidence supports the conclusion that DIDP is not likely to be genotoxic or 913 mutagenic (U.S. EPA, 2024n). This conclusion is based on results from five studies, including two 914 bacterial reverse mutation assays, two *in vitro* mouse lymphoma assays, and one *in vivo* mouse 915 micronucleus test. Across available studies, DIDP was negative for genotoxicity and mutagenicity. Consistently, existing assessments of DIDP by ECB (2003a), ECHA (2013), Australia NICNAS (2015b, 916 917 2008b, c), Health Canada (EC/HC, 2015c), and U.S. CPSC (2014, 2010d) have also concluded that 918 DIDP is not genotoxic or is not likely to be genotoxic. 919

Readers are directed to EPA's *Human Health Hazard Assessment for Diisodecyl Phthalate (DIDP)* (U.S. EPA, 2024n) for further discussion of available genotoxicity data for DIDP.

922

### 923 **3.8 Conclusions on Genotoxicity**

Overall, available data support the conclusion that BBP (Section 3.2), DINP (Section 3.6), and DIDP
(Section 3.7) are not likely to be genotoxic or mutagenic. As discussed earlier in this section, U.S. CPSC
(2014, 2010a, d, f, 2001), Canada (ECCC/HC, 2020; EC/HC, 2015a, c; Environment Canada, 2000),
Australia NICNAS(2015a, b, 2012, 2008b, c), ECHA (2017a, b, 2013), EFSA (2019, 2005a), and the
European Chemical's Bureau (2007, 2003a, c) have all reached similar conclusions regarding the
genotoxicity of BBP, DINP, and DIDP.

- For DEHP, EPA did not independently evaluate the extensive database of *in vitro* and *in vivo*
- genotoxicity studies of DEHP and its major metabolites (Section 3.1). However, EPA agrees with the
- conclusions of ATSDR (2022), NTP (2021b), U.S. CPSC (2010c), Health Canada (1994), Australia
- 934 NICNAS (2010), ECHA (2017a, b), EFSA (2019), and the European Chemical's Bureau (ECB, 2008),
- and EPA did not identify any new data that would impact the conclusions of these existing assessments.
- 936 Overall, available data indicate that DEHP and its metabolites are not mutagenic, but that there is some
- 937 limited evidence that DEHP may be weakly genotoxic inducing effects such as DNA damage and/or

May 2025

chromosomal aberrations. As noted by ATSDR (2022), these effects may be secondary to oxidative stress.

940

941 For DBP, available studies provide somewhat mixed results (Section 3.3). DBP: did not induce 942 micronuclei formation in two in vivo studies of mice; did not cause in vitro cell transformation in one 943 study of Balb/c-3T2 mouse cells; did not cause gene mutations in studies of E. coli or S. cerevisiae; was 944 not mutagenic in 4 bacterial reverse mutation assays with or without S9; and did not induce 945 chromosomal aberrations in one in vitro assay with CHO cells. Equivocal results were obtained in one 946 bacterial reverse mutation study and in one *in vitro* assay with Chinese hamster lung cells for 947 chromosomal aberrations, while marginally positive results were obtained in one 8-azaguanine 948 resistance assay and in one *in vitro* SCE assay with CHO cells. In *in vitro* mouse lymphoma assays, 949 DBP was mutagenic in 1 of 3 assays without metabolic activation, and in two assays with metabolic 950 activation. Given the results of the mouse lymphoma assays, it is difficult to conclude unequivocally that 951 DBP is not genotoxic. However, as will be discussed further in Section 4.3.3, DBP shows equivocal 952 evidence of carcinogenic activity in male rats (based on a slight increase in PACTs), but no evidence of 953 carcinogenic activity in female rats or mice of either sex. Given the results of the in vivo and in vitro 954 genotoxicity assays of DBP and in vivo carcinogenicity studies of DBP in rats and mice, EPA does not 955 consider DBP to be a potent genotoxicant. However, there is some limited evidence that DBP may be 956 weakly genotoxic in some in vitro assays. Consistently, Health Canada (EC/HC, 1994), the ECB (2004), 957 ECHA (2017a, b), Australia NICNAS (2013), and EFSA (2019) concluded that DBP is not genotoxic or 958 mutagenic, while ATSDR (2001) concluded that DBP may be weakly mutagenic in vitro. 959

Limited genotoxicity testing has been conducted for DIBP (Section 3.4) and DCHP (Section 3.5). DIBP showed no mutagenic activity in four bacterial reverse mutation assays with or without metabolic

activation, while DCHP showed no mutagenic activity in one bacterial reverse mutation assay with or

without metabolic activation. However, for the phthalates evaluated herein, data supports the conclusionthat phthalates are either not genotoxic or mutagenic (as is the case for BBP, DINP, and DIDP) or at

- 965 most weakly genotoxic based on some limited data (as is the case for DEHP and DBP). Overall, based
- 966 on read-across from BBP, DINP, DIDP, DEHP, and DBP, EPA does not consider DIBP or DCHP likely
- to be genotoxic or mutagenic to humans. This conclusion is consistent with that of other assessments,
  which have also generally concluded phthalate esters as a class are not likely to be genotoxic or
- 969 mutagenic (ECHA, 2017a, b; NICNAS, 2016; U.S. CPSC, 2014). Overall, EPA agrees with the
- 970 conclusions of other phthalate assessments, that phthalate esters (*i.e.*, DEHP, BBP, DBP, DIBP, DCHP,
- 971 DINP, DIDP) are not likely to be mutagenic.

May 2025

# 972 4 CANCER HAZARD IDENTIFICATION, CHARACTERIZATION, 973 AND MODE OF ACTION

Section 4.1 summarizes available human epidemiologic data, while Section 4.2 summarizes available
cancer bioassays of experimental animal models. Section 4.3 summarizes EPA's cancer hazard
characterization, including MOA information and EPA's preliminary cancer classifications. No cancer
bioassays are available for DIBP or DCHP. Lack of this data for DIBP and DCHP is addressed in
Section 5 using read-across and elements from the ReCAAP weight of evidence framework (<u>Hilton et al., 2022</u>) as an organizational tool to evaluate the extent to which the lack of carcinogenicity studies
imparts significant uncertainty on the human health risk assessments for DIBP and DCHP.

981

### 982 983 4.1 Summary of Available Epidemiological Studies for DEHP, BBP, DBP, 983 983 DIBP, DCHP, DINP and DIDP

984 This section summarizes available human epidemiologic studies of DEHP, BBP, DBP, DIBP, DCHP, 985 DINP, and DIDP that investigate the association between phthalate exposure and cancer outcomes. 986 Section 4.1.1 provides a summary of conclusions from existing cancer hazard assessments of phthalates 987 by Health Canada (2018a), ATSDR (2022), and IARC (2013), while Section 4.1.2 provides a summary 988 of new epidemiologic studies published between 2018 and 2019 evaluating the association between 989 phthalates (DEHP, BBP, DBP, DIBP, DCHP, DINP, and DIDP) and cancer outcomes in humans. 990 Finally, Section 4.1.3 summarizes EPA's draft conclusions regarding the association between phthalate 991 exposure and cancer outcomes in humans based on available epidemiologic evidence.

992

993

### 4.1.1 Previous Epidemiologic Assessments of Phthalates

EPA reviewed and summarized conclusions from previous assessments that investigated the association
between exposure to DEHP, BBP, DBP, DIBP, DCHP, DINP, and DIDP and cancer outcomes in
humans, including those by Health Canada (2018a), ATSDR (2022), and IARC (2013). The outcomes
evaluated by each assessment are shown in Table 4-1.

998 999

## Table 4-1. Summary of Existing Epidemiologic Assessments of Phthalates Investigating Cancer Outcomes

| Previous Assessment   | Phthalates in Assessment | Outcomes Evaluated  |
|-----------------------|--------------------------|---|
| Health Canada (2018a) | DIBP and its metabolites | Breast cancer   |
| ATSDR ( <u>2022</u> ) | DEHP and its metabolites | <ul><li>Breast cancer</li><li>Prostate cancer</li><li>Thyroid cancer</li></ul>  |
| IARC ( <u>2013</u> )  | DEHP and its metabolites | <ul> <li>Breast cancer</li> <li>Cancer mortality</li> <li>Respiratory cancer mortality</li> <li>Testicular cancer</li> <li>Pancreatic cancer</li> <li>Multiple Myeloma</li> </ul> |

### 4.1.1.1 Health Canada (2018a)

1003 1004 Health Canada evaluated two case-control studies (Martinez-Nava et al., 2013; Lopez-Carrillo et al., 1005 2010) that looked at the relationship between urinary monoisobutyl phthalate (MIBP), a metabolite of 1006 DIBP, and breast cancer outcomes in populations of reproductive-aged women. Lopez-Carrillo et al. 1007 (2010) found no significant association between urinary MIBP and breast cancer. Martinez-Nava et al. 1008 (2013), who evaluated the association between urinary MIBP and breast cancer by PPARGC1B 1009 Ala203Pro alleles, reported a significant negative association between urinary MIBP and breast cancer risk in carriers of the PPARGC1B Ala203Pro G allele, but not the PPAR<sub>y</sub> Pro12Ala C allele. 1010 1011

1012 Overall, Health Canada found inconsistent results for MIBP and breast cancer. Health Canada did not 1013 observe any positive associations, exposure-response relationships, and temporality was not established. Therefore, Health Canada concluded that there was inadequate evidence<sup>1</sup> for the association between 1014 urinary MIBP and risk of breast cancer. Health Canada did not evaluate studies of the association 1015

- 1016 between cancer outcomes and other phthalates (e.g., DINP, DIDP, BBP, DBP, DEHP).
- 1017

#### 1018 4.1.1.2 ATSDR (2022)

1019 ATSDR evaluated the epidemiological evidence for an association between exposure to DEHP (based 1020 on urinary levels of DEHP metabolites) and cancer outcomes. The epidemiological studies evaluated by 1021 ATSDR included one population-based study (Morgan et al., 2016), and nine case-control studies. Six 1022 studies evaluated breast cancer outcomes (Reeves et al., 2019; Mérida-Ortega et al., 2016; Morgan et al., 1023 2016; Holmes et al., 2014; Martinez-Nava et al., 2013; Lopez-Carrillo et al., 2010); one evaluated 1024 prostate cancer (Chuang et al., 2020); and three evaluated thyroid cancer (Liu et al., 2020; Miao et al., 1025 2020; Marotta et al., 2019). The population based study by Morgan et al. did not find an association 1026 between urinary DEHP metabolite levels and breast cancer in the general U.S. population using 1027 NHANES data from 2003 through 2010 (Morgan et al., 2016). The remaining nine case-control studies 1028 evaluated exposure to DEHP after the outcome, cancer, was observed.

1029

1030 Overall, ATSDR (2022) concluded that "There is no information (qualitative or quantitative) on 1031 exposures prior to incidence/diagnosis that could have been involved in tumor induction. Furthermore, 1032 cancer treatments could increase exposure to, and excretion of, phthalates from medical equipment. 1033 Thus, these studies are not useful for evaluating the carcinogenicity of DEHP."

1034

#### 1035 4.1.1.3 IARC (2013)

1036 The IARC workgroup identified one occupational study by Thiess et al. (1978), one case control study 1037 by Lopez-Carrillo et al. (2010), one cohort study by Hagmar et al. (1995; 1990) that looked at the 1038 association between exposure to DEHP (and other phthalates being evaluated under TSCA) and cancer 1039 outcomes in humans.

1040

A case-control study was carried out in northern Mexico by Lopez-Carrillo et al. (2010) to assess the 1041

1042 association between breast cancer and urine levels of nine phthalate metabolites, including four

- 1043 metabolites of DEHP (*i.e.*, MEHP, mono(2-ethyl-5-hydroxyhexyl) phthalate [MEHHP], mono(2-ethyl-1044 5-oxohexyl) phthalate [MEOHP], mono(2-ethyl-5-carboxypentyl) phthalate [MECPP]), one metabolite
- 1045 of DIBP (i.e., MIBP), one metabolite of DBP (monobutyl phthalate [MBP]), and one metabolite of BBP
- 1046 (*i.e.*, monobenzyl phthalate [MBzP). Since there was no information on individual habits with respect to

<sup>&</sup>lt;sup>1</sup> Health Canada defines **inadequate evidence** as "the available studies are of insufficient quality, consistency or statistical power to permit a conclusion regarding the presence or absence of an association."

1047 phthalate exposure, exposure evaluation was dependent on the measurement of urinary metabolite 1048 levels. No significant associations between urine levels of MBP or MIBP and breast cancer were 1049 observed after adjusting for current age, age of menarche, parity, menopause status and other phthalate 1050 metabolites. A significant negative association between urine levels of MBzP and breast cancer was 1051 observed after adjusting for current age, age of menarche, parity, menopause status and other phthalate 1052 metabolites. For DEHP, neither the sum of urinary DEHP metabolites or individual metabolites showed 1053 a significant association with breast cancer, except for MECPP. Urinary levels of MECPP were 1054 significantly associated with increased breast cancer after adjusting for current age, age of menarche, parity, menopause status and other phthalate metabolites (P = 0.047). Although the IARC workgroup 1055 concluded that the study design was appropriate, there were issues with the timing of the exposure 1056 1057 assessment. Biological samples were taken to measure DEHP metabolites in the urine after cancer cases 1058 were diagnosed, but before any treatment was administered. It is unknown if the disease status had an 1059 impact on the levels of these metabolites. No measures of urinary phthalate exposure were measured 1060 prior to diagnosis. This study was limited by the lack of a dose-response for all urinary metabolites, timing of exposure assessment that precludes conclusions related to temporality, and inconsistent 1061 1062 associations of the four DEHP metabolites that were evaluated.

1063

1064 The mortality of 2,031 Swedish employees at a polyvinyl chloride (PVC) processing plant that made floor tiles, thick and thin film floor sheeting, and pipes from PVC was documented in a cohort study by 1065 Hagmar et al. (1995; 1990). The products were made from PVC containing phthalic acid esters, with 1066 1067 DEHP, BBP, and DIDP being the main plasticizer used at the plant. Cumulative exposures to 1068 plasticizers were estimated as the time-weighted average breathing zone levels of total phthalic acid 1069 esters among various types of worker class and were therefore not specific to any individual phthalate, 1070 including DEHP. The PVC-processing workers had a significant excess of respiratory cancer morbidity 1071 (standard incidence ratio (SIR), 2.13; 95% confidence interval (CI): 1.27-3.47; 17 cases) and total 1072 cancer morbidity (SIR, 1.28; 95% CI: 1.01–1.61; 75 cases), but there was no statistically significant 1073 association between cumulative exposure to plasticizers and respiratory cancer morbidity.

1074

1075 The workgroup also evaluated seven case control studies of workers potentially exposed to DEHP, 1076 unspecified combinations of phthalates, or PVC plastics and cancer outcomes in workers (Westberg et al., 2005; Hardell et al., 2004; Ohlson and Hardell, 2000; Hansen, 1999; Hardell et al., 1997; Selenskas 1077 1078 et al., 1995; Heineman et al., 1992). Three population-based case-control studies examined the 1079 relationship between testicular cancer and occupational exposure to PVC plastics or products (exposure 1080 assessment did not evaluate exposure to any specific phthalate) (Westberg et al., 2005; Hansen, 1999; 1081 Hardell et al., 1997). Two of these studies were conducted in Sweden (Westberg et al., 2005; Hardell et 1082 al., 2004), and one was conducted in Denmark (Hansen, 1999). Men who had ever been exposed to 1083 mostly PVC (odds ratio (OR), 0.7; 95% CI: 0.5–1.2) or plastics in general (OR, 1.0; 95% CI: 0.8–1.2) did not have an increased risk of testicular cancer, according to a larger Danish study; however, 1084 1085 exposure to DEHP or any other phthalate was not directly evaluated (Hansen, 1999). The exposure 1086 assessment of these studies were centered on PVC in general rather than exposure to any specific 1087 chemical, which reduces the likelihood of identifying a phthalate-related effect.

1088
1089 A nested case-control study of pancreatic cancer was carried out by Selenskas et al. (1995) on a group of
1090 employees working at a plastic production and research and development facility in New Jersey, USA,
1091 where occupational exposure was assessed by employment history and department of work (Dell and
1092 Teta, 1995). The manufacturing of flexible plastics may have potentially exposed workers to DEHP,
1093 which was identified as being used at this plant. However, only workers who processed vinyl and
1094 polyethylene showed a significant increased risk for pancreatic cancer (relative risk, 7.15; 95% CI:
1095 1.28–40.1). The exposure assessment did not quantitatively evaluate exposure to any specific phthalate.

1096

1097 In a population-based case-control study of Danish men, the association between exposure to

1098 unspecified combinations of phthalates (and other occupational agents) and multiple myeloma was

- 1099 assessed (<u>Heineman et al., 1992</u>). Larger but non-significant ORs for multiple myeloma were linked to
- phthalate exposure: the risk estimate for probable exposure was larger (OR, 2.0; 95% CI: 0.9–4.4; 11
  cases and 21 controls) than the risk estimates for possible exposure (OR, 1.3; 95% CI: 0.9–2.0; 34 cases
- and 94 controls).
- 1103

Overall, while IARC did find some association between exposure to DEHP and cancers such as breast cancer, cancer mortality, respiratory cancer mortality, testicular cancer, and multiple myeloma, the results were generally not statistically significant. The limitations of the studies and/or possible explanations for non-significant results include: the small number of workers exposed to site-specific cancer fatalities or cases; possible confounding by tobacco use or other risk factors; and imprecise exposure estimates.

1110

## 11114.1.2Epidemiologic Studies of Phthalates and Cancer Outcomes (2018–2019) Evaluated<br/>by EPA

EPA also evaluated new epidemiologic studies published between 2018 and 2019 evaluating the association between phthalates (DEHP, BBP, DBP, DIBP, DCHP, DINP, and DIDP) and cancer outcomes in humans. EPA identified five epidemiology studies that evaluated the association between phthalates such as DINP, DIDP, BBP, DBP, DEHP, and DIBP and cancer outcomes, including breast cancer, colorectal cancer, and breast cancer mortality (<u>Trasande et al., 2021</u>; <u>Ahern et al., 2019</u>; <u>Ennis et</u> al., 2019; <u>Reeves et al., 2019</u>; <u>Parada et al., 2018</u>). Results of these studies are discussed further below.

11120

### 4.1.2.1 Di(2-ethylhexyl) Phthalate (DEHP)

Five studies evaluated the association between DEHP and its metabolites and breast cancer and 1121 colorectal cancer outcomes. These included one high confidence study (Ahern et al., 2019) and three 1122 1123 medium confidence studies (Trasande et al., 2021; Reeves et al., 2019; Parada et al., 2018) that 1124 evaluated breast cancer outcomes and one low confidence study (Ennis et al., 2019) that evaluated 1125 colorectal adenocarcinoma. There were no statistically significant findings from the high or low confidence studies that evaluated exposure to DEHP and breast cancer risk. There were significant 1126 1127 results from two of the medium confidence studies (Reeves et al., 2019; Parada et al., 2018). One 1128 medium confidence study (Parada et al., 2018) reported a significant inverse association in multivariable 1129 adjusted hazard ratios (HRs) between urinary MEHP in the 4th and 5th quintiles of breast cancer 1130 specific mortality HR=0.47 (95% CI: 0.25-0.89) and HR=0.54 (95% CI: 0.28-1.04), respectively, compared to the lowest quintile (quintile 1) HR=1 among participants in the Long Island Breast Cancer 1131 Study Project who were diagnosed with breast cancer in 1996 through 1997 and followed for 18+ years. 1132 1133 Additionally, there was an inverse relationship between breast cancer specific mortality and continuous 1134 In-transformed concentrations of MEHP (HR<sub>Ln(MEHP)</sub>=0:79, 95% CI: 0.64–0.98). Statistical significance 1135 was not maintained for other quintiles, and no statistically significant results were reported for breast 1136 cancer incidence. This study also reported the odds of new breast cancer cases among participants in the 1137 Long Island Breast Cancer Study Project for the 3rd vs. 1st quintile of MECPP. Statistical significance 1138 was not maintained for other quintiles or when analyzed continuously. The other medium confidence study (Reeves et al., 2019) reported significantly decreased odds of breast cancer in a Women's Health 1139 1140 Initiative study among participants with positive endocrine receptor and progesterone receptor status for 1141 the 3rd vs. 1st quartile of MEHHP. In non-stratified analyses, no statistically significant results were 1142 reported for MEHHP. This study also reported significant inverse association between MEOHP and

odds of breast cancer with positive estrogen receptor and progesterone receptor status for the 3rd vs. 1st quartile of MEOHP. In non-stratified analyses, no statistically significant results were reported for MEOHP. In the third study by Trasande et al. that looked at the association between DEHP and

- 1146 mortality from all causes, cardiovascular disease and cancer (<u>Trasande et al., 2021</u>), no significant
- association between exposure to DEHP and cancer mortality was found.
- 1148 1149

### 4.1.2.2 Butyl Benzyl Phthalate (BBP)

Five studies evaluated the association between BBP and breast cancer and cancer mortality outcomes. 1150 1151 One high confidence study (Ahern et al., 2019), three medium confidence studies (Trasande et al., 2021; Reeves et al., 2019; Parada et al., 2018) evaluated breast cancer outcomes, and one low confidence study 1152 1153 (Ennis et al., 2019) evaluated colorectal adenocarcinoma and BBP exposure. There were no significant 1154 results from the high or low confidence studies. The three medium confidence studies (Trasande et al., 1155 2021; Reeves et al., 2019; Parada et al., 2018) had some significant results. One medium confidence 1156 study (Parada et al., 2018) of adult women on Long Island reported a significant inverse association between urinary MBzP measured shortly after diagnosis and odds of breast cancer (OR [95% CI] in the 1157 1158 2nd quintile compared to the 1st quintile of MBzP exposure = 0.64 [0.45, 0.91], and in the 4th quintile 1159 compared to the 1st quintile of MBzP exposure = 0.59 [0.41, 0.84]). No significant findings were reported for other quintiles of MBzP or for continuous measurements of MBzP. The other medium 1160 confidence study (Reeves et al., 2019) of postmenopausal women in the U.S. reported a significant 1161 inverse association between urinary MBzP and odds of breast cancer (OR [95% CI] for Q3 vs. Q1 of 1162 MBzP exposure = 0.57 [0.39, 0.84], p-value for trend across quartiles = 0.03; and in women without 1163 1164 estrogen and progesterone hormone receptors for Q3 vs. Q1 of MBzP exposure = 0.23 [0.05, 0.97]). The final medium confidence study (Trasande et al., 2021) reported a significant positive association 1165 between urinary MBzP and cancer mortality in U.S. adults (HR (95% CI) per ln-µmol/L increase in 1166 1167 MBzP = 1.19 [1.04, 1.36]). No significant findings were reported for tertiles of MBzP.

1168

1169

### 4.1.2.3 Dibutyl Phthalate (DBP)

1170 The same five studies evaluated the association between DBP and breast cancer and colorectal cancer outcomes (Trasande et al., 2021; Ahern et al., 2019; Ennis et al., 2019; Reeves et al., 2019; Parada et al., 1171 2018). There were no significant results from the low confidence study, but there were significant results 1172 from the high confidence study (Ahern et al., 2019), and one of the medium confidence studies (Parada 1173 et al., 2018). The high confidence study of Danish women (Ahern et al., 2019) reported a significant 1174 1175 positive association between DBP from phthalate-containing oral medications and risk of invasive breast 1176 cancer in Swedish women with estrogen-receptor positive cancers [HR (95% CI) for medication-related 1177 DBP >10,000 mg vs. unexposed; all breast cancer = 2.0(1.1, 3.6); estrogen receptor-positive breast 1178 cancer = 1.9(1.1, 3.5)]. The medium confidence study (Parada et al., 2018) reported significant inverse 1179 associations between urinary MnBP obtained shortly after diagnosis and breast cancer (OR [95% CI] of 1180 breast cancer for Q4 vs. Q1 of urinary MnBP = 0.65 [0.45, 0.93]).

1181

1182

### 4.1.2.4 Diisobutyl Phthalate (DIBP)

The same five studies evaluated the association between DIBP and breast cancer and colorectal cancer outcomes (<u>Trasande et al., 2021</u>; <u>Ahern et al., 2019</u>; <u>Ennis et al., 2019</u>; <u>Reeves et al., 2019</u>; <u>Parada et al., 2018</u>). There were no significant results from the high or low confidence studies that evaluated breast cancer outcomes. However, there was some significant results from one of the medium confidence

1187 studies (<u>Parada et al., 2018</u>). The medium confidence study (<u>Parada et al., 2018</u>) of adult women on

1188 Long Island reported a significant inverse association between urinary MIBP obtained shortly after
- 1189 diagnosis and odds of breast cancer (OR [95% CI] in the 4th quintile compared to 1st quintile of MIBP 1190 exposure = 0.69 [0.48, 0.99]). No significant findings were reported for other quintiles of MIBP or for
- 1191
- 1192

1193

# 4.1.2.5 Dicyclohexyl Phthalate (DCHP)

continuous measurements of MIBP.

- 1194 EPA did not identify any studies evaluating the association between DCHP (or its metabolites) exposure 1195 and any cancer outcomes.
- 1196

# 4.1.2.6 Diisononyl Phthalate (DINP)

1197 Three medium confidence studies (Trasande et al., 2021; Reeves et al., 2019; Parada et al., 2018) 1198 1199 evaluated the associations between DINP and breast cancer and breast cancer mortality outcomes. Of 1200 these, only one study (Parada et al., 2018) reported significant results. The highest vs. lowest quintiles of 1201 MCOP were associated with breast cancer ORs ranging from 0.71 to 0.73. The highest (vs. lowest) 1202 quintiles of MCOP were associated with breast cancer-specific mortality HR of 0.55 (95% CI: 0.23, 1203 1.35). MCOP concentrations differed by stage (in situ vs. invasive) based on statistically significant 1204 mean differences derived from generalized linear models regressing each of the ln-transformed 1205 creatinine-corrected phthalate metabolite concentrations on age and the covariate. Continuous In-1206 transformed MCOP were associated with HRs of breast cancer-specific mortality of 0.54 (95% CI: 0.33, 0.89), though estimates were imprecise. In follow-up analyses, MCOP had one of the largest inverse 1207 1208 associations for which the highest quintiles were associated with HRs of breast cancer-specific mortality 1209 of 0.55 (95% CI: 0.23, 1.35) relative to the lowest quintiles. The estimate for MCOP was imprecise due 1210 to availability of data for the 320 women with breast cancer.

1211

# 1212

# 4.1.2.7 Diisodecyl Phthalate (DIDP)

1213 Three medium confidence studies evaluated the association between DIDP and breast cancer and breast 1214 cancer mortality outcomes (Trasande et al., 2021; Reeves et al., 2019; Parada et al., 2018). Of those 1215 studies only one study (Parada et al., 2018) had some significant results. Parada et al. (2018) reported a 1216 significant inverse association between urinary MCNP and odds of breast cancer (OR [95% CI]), in the 1217 highest vs. lowest quintile of MCNP; (OR = 0.51 [0.28, 0.92] of adult women in the Long Island Breast 1218 Cancer Study Project (LIBCSP) who were diagnosed with first primary in situ or invasive breast cancer 1219 during the years 1996 to 1997. Breast cancer-specific mortality HRs with multivariable adjustment were 1220 not statistically significant.

1221

1222

#### 4.1.3 Conclusion

In conclusion, Health Canada and ATSDR, determined that the evidence was inadequate to support an 1223 1224 association between phthalate exposure and cancer outcomes, while IARC found no statistically 1225 significant associations between DEHP exposure and cancer outcomes.

1226

1227 Overall, there are a number of sources of uncertainty associated with the available human epidemiologic studies of phthalates and cancer outcomes, including uncertainty associated with exposure 1228

1229 characterization of individual phthalates, source of phthalate exposure, timing of phthalate exposure

1230 (exposure is typically measured after the outcome is reported, meaning temporality can't be established),

- 1231 as well as co-exposure to multiple phthalates, which can confound results. Another uncertainty is that
- 1232 many of the available epidemiologic studies evaluated phthalate exposure after cancer diagnosis and
- cancer treatment had been initiated, which can confound study results because cancer treatment can 1233

increase phthalate exposure from plastic medical equipment. Overall, EPA agrees with the conclusions
 of Health Canada and ATSDR. Given the limitations and uncertainties, EPA concludes that there is
 indeterminant evidence of an association between phthalate exposure and subsequent cancer outcomes.

# 1238 **4.2 Overview of Laboratory Animals Studies**

1239 Of the seven phthalate diesters being evaluated under TSCA, DEHP, BBP, DBP, DINP and DIDP have 1240 been evaluated for carcinogenicity in experimental animal models (see Table 4-2 for a summary of 1241 available cancer bioassays). No studies of experimental animal models evaluating carcinogenicity are 1242 available for DIBP or DCHP, however, the potential carcinogenicity of DIBP and DCHP is further 1243 considered in Section 5 based on read-across from DEHP, BBP, DBP, DINP and DIDP. As can be seen 1244 from Table 4-3, statistically significant increases in several tumor types have been observed in 1245 experimental animal models following chronic oral exposure to DEHP, BBP, DBP, DINP and DIDP. 1246 Observed tumor types include:

- Hepatocellular adenomas and/or carcinomas following exposure to DEHP, DINP, and DIDP;
- Pancreatic acinar cell tumors (PACTs) following exposure to DEHP, BBP, and DBP;
- Testicular Leydig cell adenomas following exposure to DEHP;
- MNCL in F344 rats following exposure to DEHP, BBP, DINP and DIDP;
- Renal tubular cell carcinomas following exposure to DINP; and
- Uterine adenoma, adenocarcinoma, squamous cell carcinoma, or squamous cell papilloma following exposure to DEHP.

1254 Evidence for each of these tumor types for DEHP, BBP, and DBP, including EPA's weight of scientific

evidence conclusions, preliminary cancer classifications, and, when applicable, MOA analyses, are

summarized in Sections 4.3. EPA's weight of scientific evidence conclusions and cancer classifications

for DIDP and DINP have been summarized previously in EPA's *Cancer Human Health Hazard Assessment for Diisononyl Phthalate (DINP)* (U.S. EPA, 2025a) and *Human Health Hazard Assessment*

1259 for Diisodecyl Phthalate (DIDP) (U.S. EPA, 2024n). However, a brief summary of carcinogenic

1260 findings, weight of scientific evidence conclusions, and cancer classifications for DINP and DIDP are

1261 provided in Sections 4.3.4 and 4.3.5, respectively, to facilitate comparisons across phthalates, including

1262 EPA's read-across assessment for DIBP and DCHP in Section 5.

### May 2025

## 1263 Table 4-2. Summary of Database of Available Rodent Carcinogenicity Studies Considered

| Phthalate | Experimental Model  | Exposure<br>Route<br>(Method) | Exposure<br>Duration | # of<br>Studies | Notes                                | Reference(s)   |  |
|-----------|---|-------------------------------|----------------------|-----------------|--------------------------------------|--|--|
| DEHP      | F344/N rat (both sexes)   | Oral (diet)                   | 2 years              | 2               |                                      | ( <u>David et al., 2000b;</u> <u>David et al.,</u><br><u>1999;</u> <u>NTP, 1982a</u> ) |  |
|           | F344 rat (male only)  | Oral (diet)                   | 95–108 weeks         | 2               |                                      | ( <u>Rao et al., 1990;</u> <u>Rao et al., 1987</u> )                                   |  |
|           | SD rat (male only)  | Oral (diet)                   | ≤159 weeks           | 1               | Lifetime exposure study              | ( <u>Voss et al., 2005</u> )   |  |
|           | SD rat (both sexes)   | Oral (diet)                   | 2 years              | 1               | Perinatal & post-weaning exposure    | ( <u>NTP, 2021b</u> )  |  |
|           | SD rat (both sexes)   | Oral (diet)                   | 2 years              | 1               | Post-weaning exposure only           | ( <u>NTP, 2021b</u> )  |  |
|           | B6C3F1/n mice (both sexes)  | Oral (diet)                   | 2 years              | 2               |                                      | ( <u>David et al., 2000a;</u> <u>David et al.,</u><br><u>1999; NTP, 1982a</u> )        |  |
|           | Syrian golden hamster (both sexes)                                    | Inhalation                    | 17-23 months         | 1               | Lifetime exposure study              | (Schmezer et al., 1988)  |  |
|           | Syrian golden hamster (both sexes)                                    | IP injection                  | 17–23 months         | 1               | Lifetime exposure study              | (Schmezer et al., 1988)  |  |
|           | Wild-type & RasH2 mice (both sexes)                                   | Oral (diet)                   | 26 weeks             | 1               |                                      | ( <u>Toyosawa et al., 2001</u> )   |  |
|           | Tg.AC mice (both sexes)   | Oral (diet)                   | 26 weeks             | 1               |                                      | ( <u>Eastin et al., 2001</u> )   |  |
|           | $Xpa^{-/-}$ , wild-type, & $Xpa^{-/-}/p53^{+/-}$<br>mice (both sexes) | Oral (diet)                   | 39 weeks             | 1               |                                      | (Mortensen et al., 2002)   |  |
|           | Wild-type & <i>Ppara</i> -null mice (males only)                      | Oral (diet)                   | 22 months            | 1               |                                      | ( <u>Ito et al., 2007a</u> )   |  |
|           | Tg.AC mice (both sexes)   | Dermal                        | 28 weeks             | 1               |                                      | ( <u>Eastin et al., 2001</u> )   |  |
| BBP       | F344/N rat (both sexes)   | Oral (diet)                   | 2 years              | 2               |                                      | ( <u>NTP, 1997b, 1982b</u> )   |  |
|           | F344/N rat (both sexes)   | Oral (diet)                   | 24–32 months         | 3               | Ad libitum & diet restricted studies | ( <u>NTP, 1997a</u> )  |  |
|           | B6C3F1 mice (both sexes)  | Oral (diet)                   | 2 years              | 1               |                                      | ( <u>NTP, 1982b</u> )  |  |
| DBP       | SD rat (both sexes)   | Oral (diet)                   | 2 years              | 1               | Perinatal & post-weaning exposure    | ( <u>NTP, 2021a</u> )  |  |
|           | B6C3F1 mice (both sexes)  | Oral (diet)                   | 2 years              | 1               |                                      | ( <u>NTP, 2021a</u> )  |  |
| DIBP      |   |                               | No carcin            | ogenicity s     | study available                      |  |  |
| DCHP      | No carcinogenicity study available                                    |                               |                      |                 |                                      |  |  |

May 2025

| Phthalate | Experimental Model                  | Exposure<br>Route<br>(Method) | Exposure<br>Duration | # of<br>Studies | Notes | <b>Reference</b> (s)                        |
|-----------|-------------------------------------|-------------------------------|----------------------|-----------------|-------|---|
| DINP      | F344 rat (both sexes)               | Oral (diet)                   | 2 years              | 2               |       | (Covance Labs, 1998c; Lington et al., 1997) |
|           | SD rat (both sexes)                 | Oral (diet)                   | 2 years              | 1               |       | (Bio/dynamics, 1987)                        |
|           | B6C3F1 mice (both sexes)            | Oral (diet)                   | 2 years              | 1               |       | (Covance Labs, 1998a)                       |
| DIDP      | F344 rat (both sexes)               | Oral (diet)                   | 2 years              | 1               |       | ( <u>Cho et al., 2008</u> )                 |
|           | Wild-type & RasH2 mice (both sexes) | Oral (diet)                   | 26 weeks             | 1               |       | ( <u>Cho et al., 2011</u> )                 |

#### May 2025

#### 1265 Table 4-3. Summary of Tumor Types Observed Following Chronic Oral Exposure to Phthalates in Experimental Rodent Models<sup>a</sup>

| Phthalate    | Hepato<br>Adenom<br>Carci | cellular<br>a and/or<br>noma | Pancreatic<br>Acinar Cell<br>Tumors<br>(PACTs) |           | Pancreatic<br>Acinar Cell<br>Tumors<br>(PACTs)<br>Leydig Cell<br>Tumors |             | Renal Tubular<br>Carcinoma |              | Uterine Adenoma,<br>Adenocarcinoma,<br>Squamous Cell<br>Carcinoma, or<br>Squamous Cell<br>Papilloma |               | MNCL             |       |
|--------------|---------------------------|------------------------------|--|-----------|---|-------------|----------------------------|--------------|---|---------------|------------------|-------|
|              | Rat                       | Mouse                        | Rat  | Mouse     | Rat   | Mouse       | Rat                        | Mouse        | Rat   | Mouse         | Rat              | Mouse |
| DEHP         | Yes                       | Yes                          | Yes  | No        | Yes <sup>c</sup>  | No          | No                         | No           | Yes   | No            | Yes <sup>e</sup> | No    |
| BBP          | No                        | No                           | Yes  | No        | No  | No          | No                         | No           | No  | No            | Yes <sup>e</sup> | No    |
| DBP          | No                        | No                           | Yes  | No        | No  | No          | No                         | No           | No  | No            | No               | No    |
| DIBP         |                           |                              |  |           | Ν   | No carcino  | genicity s                 | study avail  | lable   |               |                  |       |
| DCHP         |                           |                              |  |           | Ν   | No carcino  | genicity s                 | study avail  | lable   |               |                  |       |
| DINP         | Yes                       | Yes                          | No   | No        | No  | No          | Yes <sup>d</sup>           | No           | No  | No            | Yes <sup>e</sup> | No    |
| DIDP         | No                        | Yes <sup>b</sup>             | No   | No        | No  | No          | No                         | No           | No  | No            | Yes <sup>e</sup> | No    |
| "'Vos' india | ates that a s             | tatistically                 | ionifican                                      | tinoransa | in the tur  | oor type he | s been obs                 | arriad in at | least one of  | the available | studios while 'N | ,     |

<sup>*a*</sup> 'Yes' indicates that a statistically significant increase in the tumor type has been observed in at least one of the available studies, while 'No' indicates that no statistically significant increase in the tumor type has been observed in any of the available studies.

<sup>b</sup> Hepatocellular adenomas observed following chronic dietary exposure to DIDP in male rasH2 mice only (discussed further in Section 4.3.5).

<sup>c</sup> Statistically significant increases in Leydig cell tumors have been observed only in male SD rats. As discussed in Appendix C, this tumor type occurs at a high spontaneous background rate in F344 rats, which decreases the utility of this strain to detect treatment-related increases in this tumor. <sup>d</sup> Renal tubular cell carcinomas observed only in male F344 rats following chronic dietary exposure to DINP (discussed further in Section 4.3.4). <sup>e</sup> MNCL has been observed only in F344 rats, which have a high background rate of MNCL in control rats. As discussed further in Appendix C, there are a number of scientific uncertainties associated with MNCL in F344 rats. Consistent with the recommendations of the SACC (<u>U.S. EPA, 2024q</u>), EPA is not further considering MNCL as a factor in the determination of the cancer classifications for phthalates.

# 4.3 Cancer Hazard Characterization, Mode of Action and Conclusions for DEHP, BBP, DBP, DINP, and DIDP

1269 This section characterizes the cancer hazards of DEHP (Section 4.3.1), BBP (Section 4.3.2), and DBP 1270 (Section 4.3.3), including MOA information and EPA's preliminary cancer classifications. Cancer 1271 hazards of DINP and DIDP have been evaluated by EPA previously (U.S. EPA, 2025a, 2024n), but are 1272 briefly summarized in Section 4.3.4 and 4.3.5, respectively, to support cancer hazard comparisons and 1273 read-across. No cancer bioassays are available for DIBP or DCHP. Lack of this data for DIBP and 1274 DCHP is addressed in Section 5 using read-across and elements from the ReCAAP weight of evidence 1275 framework (Hilton et al., 2022) as an organizational tool to evaluate the extent to which the lack of carcinogenicity studies imparts significant uncertainty on the human health risk assessments for DIBP 1276 1277 and DCHP.

1278

## 1279 4.3.1 Di(2-ethylhexyl) Phthalate (DEHP)

DEHP has been evaluated for carcinogenicity by a number of authoritative and regulatory agencies. As
summarized in Table 4-4, DEHP has been classified by IARC as Group 2B (possibly carcinogenic to
humans) (IARC, 2013), by U.S. EPA as Group B2 (probable human carcinogen) (U.S. EPA, 1988b), by
NTP as *reasonably anticipated to be a human carcinogen* (NTP, 2016), and is listed by OEHHA under
California's Proposition 65 as causing cancer (OEHHA, 2022). Despite these cancer listings, DEHP has
not been evaluated quantitatively for cancer risk in assessments by ECB (2008), ECHA (2017a, b),
Australia NICNAS (2010), Health Canada (ECCC/HC, 2020), or U.S. CPSC (2014).

1288

## 1289 **Table 4-4. Summary of Cancer Classifications and Listings for DEHP**

| Agency   | <b>Cancer Classification/ Listing</b>           |  |  |  |  |
|--|---|--|--|--|--|
| NTP ( <u>2016</u> )  | Reasonably anticipated to be a human carcinogen |  |  |  |  |
| IARC ( <u>2013</u> )   | Group 2B (possibly carcinogenic to humans)      |  |  |  |  |
| California OEHHA (2022)  | Listed as carcinogen under Proposition 65       |  |  |  |  |
| U.S. EPA (IRIS) ( <u>1988b</u> )   | Group B2 (probable human carcinogen)            |  |  |  |  |
| IARC = International Agency for Research on Cancer; IRIS = Integrated Risk Information<br>System; NTP = National Toxicology Program; OEHHA = Office of Environmental Health<br>Hazard Assessment |   |  |  |  |  |

1290

1291

In 1988, EPA concluded that DEHP is a *Probable human carcinogen – based on sufficient evidence of carcinogenicity in animals*. Consistent with the guidelines available at the time of the assessment (*i.e., Guidelines for Carcinogen Risk Assessment* (U.S. EPA, 1986)), DEHP was assessed under an
assumption of low-dose linearity. However, since the 1988 Integrated Risk Information System (IRIS)
assessment of DEHP, the science has evolved, and EPA's current *Guidelines for Carcinogen Risk Assessment* (U.S. EPA, 2005) emphasize a data-first approach, rather than use of default options, stating:

- 1298"Rather than viewing default options as the starting point from which departures may be1299justified by new scientific information, these cancer guidelines view a critical analysis of all
- 1300 of the available information that is relevant to assessing the carcinogenic risk as the starting

May 2025

- 1301point from which a default option may be invoked if needed to address uncertainty or the1302absence of critical information."
- Moreover, TSCA requires EPA to use the "best available science," thus the cancer classification and risk
  assessment approach for DEHP has been re-evaluated.
- 1305

1306 DEHP has been evaluated extensively for carcinogenicity in experimental rodent models, including 7 1307 chronic dietary studies of rats, 2 chronic dietary studies of mice, 5 chronic dietary studies of transgenic 1308 mice, 1 chronic inhalation study of hamsters, and 1 chronic intraperitoneal injection study of hamsters. 1309 Available studies and neoplastic findings from each study are summarized in Table 4-5, while study 1310 summaries are provided in Appendix B.1. Across available studies, significant dose-related increases in hepatocellular adenomas and carcinomas have been consistently observed in 7 chronic studies of male 1311 1312 rats, 4 chronic studies of female rats, and both chronic studies of male and female B6C3F1 mice (Table 4-5 and Table 4-6). PACTs have been observed in 3 studies of male SD or F344 rats, while equivocal 1313 1314 evidence for PACTs was observed in 2 studies of female SD rats (but not in 2 studies of female F344 1315 rats), and no evidence of PACTs was reported in 2 studies of male or female B6C3F1 mice (Table 4-5 1316 and Table 4-6). Significant testicular Leydig cell tumors have been observed in one lifetime dietary 1317 exposure study of SD rats (Voss et al., 2005), while equivocal evidence of Leydig cell tumors was 1318 observed in another two-year study of SD rats by NTP (2021b). Leydig cell tumors were not observed in 4 studies of male F344 rats or 2 studies of male B6C3F1 mice; however, as noted in Appendix C, there 1319 1320 is a high spontaneous background rate of this tumor type in F344 rats, making this difficult to detect 1321 treatment-related changes in Leydig cell tumors in this F344 rats. Finally, there is some limited evidence 1322 for uterine tumors in female SD rats in 2 recent studies by NTP (2021b); however, uterine tumors were 1323 not observed in 2 studies of female F344 rats or 2 studies of female B6C3F1 mice. MNCL has been observed in one study of male F344 rats (David et al., 2000b; David et al., 1999), but has not been 1324 1325 observed in any studies of SD rats or B6C3F1 mice. In contrast to studies of rats and mice, no 1326 significant increase in tumors were observed in inhalation and intraperitoneal injection studies of 1327 hamsters (Schmezer et al., 1988).

1328

The remainder of this section includes a summary of evidence for each of these tumor types for DEHP,
including EPA's weight of scientific evidence conclusions and information on MOA, as well as EPA's
preliminary cancer classification. The remainder of the section is organized as follows:

- 1332 Section 4.3.1.1 summaries evidence of liver, pancreatic, and testicular tumors (sometimes 1333 referred to as the 'tumor triad') following chronic oral exposure to DEHP in experimental rodent 1334 models. Information pertaining to MOA for induction of each of these tumor types is provided in 1335 Sections 4.3.1.1.1 through 4.3.1.1.3. Section 4.3.1.1.4 provides information pertaining to 1336 hypolipidemic drugs that are known peroxisome proliferator-activated receptor alpha (PPAR $\alpha$ ) 1337 activators and also cause the tumor triad in rats, but not humans. Evidence from these 1338 hypolipidemic drugs support inferences for DEHP induced liver, pancreatic, and testicular 1339 tumors. Finally, Sections 4.3.1.1.5 and 4.3.1.1.6 summarize information for remaining areas of 1340 uncertainty and EPA's conclusions regarding the tumor triad.
- Section 4.3.1.2 summaries evidence of uterine tumors following chronic oral exposure to DEHP
   in experimental rodent models.
- Section 4.3.1.3 summaries evidence of MNCL following chronic oral exposure to DEHP in experimental rodent models.
- Section 4.3.1.4 summarizes EPA's preliminary cancer classification for DEHP.

## May 2025

## 1346 Table 4-5. Summary of Available Carcinogenicity Studies of DEHP in Rodents

| Brief Study Description   | Tumor Type(s) Observed   |
|---|--|
| Studies o   | f Rats   |
| Male and female F344 rats (50/sex/dose) fed diets containing 0,<br>6000, 12,000 ppm DEHP for 103 weeks (equivalent to<br>approximately 322, 674 mg/kg-day [males]; 394, 774 mg/kg-day<br>[females]) ( <u>NTP, 1982a</u> ) (see Appendix B.1.2.1 for study details).   | - Hepatocellular carcinomas and neoplastic nodules (both sexes)  |
| Male and female F344 rats (55–80/sex/dose) were administered<br>diets containing 0, 100, 500, 2,500, 12,500 ppm DEHP for up to 104<br>weeks (equivalent to 6, 29, 147, 780 mg/kg-day [males]; 7, 36, 182,<br>939 mg/kg-day [females]) (David et al., 2000b; David et al., 1999)<br>(see Appendix B.1.2.2 for study details).  | <ul> <li>Hepatocellular carcinomas and adenomas (both sexes)</li> <li>PACTs (males only)</li> <li>MNCL (males only)</li> </ul>   |
| Male F344 rats (8–10 rats/group) were fed diets containing 0 or 2% DEHP for 95 weeks ( <u>Rao et al., 1987</u> ) (see Appendix B.1.2.3 for study details).  | - Hepatocellular carcinomas and neoplastic nodules   |
| Male F344 rats (10–14 rats/group) were fed diets containing 0 or 2% DEHP for 108 weeks ( <u>Rao et al., 1990</u> ) (see Appendix B.1.2.4 for study details).  | - Hepatocellular carcinomas and neoplastic nodules   |
| Male SD rats were fed diets containing 0 (N=390), 600 (N=180),<br>1,897 (N=100), and 6,000 (N=60) mg DEHP/kg diet. Rats were fed<br>5 g diet/100 g rat/day for 6 days/week and received DEHP-free food<br>on the 7th day only after the rest of their DEHP diet had been<br>consumed (received doses: 0, 30, 95, 300 mg/kg-day over the entire<br>lifetime of rats [up to 159 weeks]) (Voss et al., 2005) (see Appendix<br>B.1.2.5 for study details).                          | <ul> <li>Hepatocellular carcinomas and adenomas (males only)</li> <li>Leydig cell adenomas (males only)</li> </ul>   |
| Time-mated SD rats (45/dose) fed diets containing 0, 300, 1,000,<br>3,000, 10,000 ppm DEHP on GD 6 through PND 21 (weaning).<br>Dams allowed to deliver litters naturally, and at weaning (PND 21),<br>F1 offspring (50/sex/dose) were continued on the same respective<br>diets for 2-years (received dose during 2-year phase of study: 18, 58,<br>189, 678 mg/kg-day [males]; 18, 62, 196, 772 mg/kg-day [females])<br>(NTP, 2021b) (see Appendix B.1.2.6 for study details) | <ul> <li>Hepatocellular carcinomas and adenomas (both sexes)</li> <li>PACTs (Males) (Females: low, statistically non-significant increase in females was considered by NTP to be treatment-related)</li> </ul> |

May 2025

| Brief Study Description  | Tumor Type(s) Observed  |  |  |  |  |
|--|---|--|--|--|--|
|  | - Uterine adenocarcinoma and combined uterus adenoma,<br>adenocarcinoma, squamous cell carcinoma, or squamous cell<br>papilloma (equivocal finding) |  |  |  |  |
| Male and female SD rats (50/sex/dose) were fed diets containing 0,   | - Hepatocellular carcinomas and adenomas (both sexes)   |  |  |  |  |
| 300, 1,000, 3,000, 10,000 ppm DEHP for two-years (equivalent to: 17, 54, 170, 602 mg/kg-day [males]; 17, 60, 177, 646 mg/kg-day [females]) ( <u>NTP, 2021b</u> ) (see Appendix B.1.2.7 for study details)  | - PACTs (Males) (Females: low, statistically non-significant increase in females was considered by NTP to be treatment-related)                     |  |  |  |  |
|  | - Leydig cell adenomas (equivocal finding)  |  |  |  |  |
|  | - Uterine adenocarcinoma and combined uterus adenoma,<br>adenocarcinoma, squamous cell carcinoma, or squamous cell<br>papilloma (Females)           |  |  |  |  |
| Studies of Mice  |   |  |  |  |  |
| Male and female B6C3F1 mice (50/sex/dose) fed diets containing 0, 3,000, 6,000 ppm DEHP for 103 weeks (equivalent to approximately 673, 1,325 mg/kg-day [males]; 799, 1,821 mg/kg-day [females]) (NTP, 1982a) (see Appendix B.1.1.1 for study details)   | - Hepatocellular carcinomas and adenomas (both sexes)   |  |  |  |  |
| Male and female B6C3F1 mice (65–70/sex/dose) fed diets<br>containing 0, 100, 500, 1,500, 6,000 ppm DEHP for 104 weeks<br>(equivalent to: 19, 99, 292, 1,266 mg/kg-day [males]; 24, 117, 354,<br>1,458 mg/kg-day [females]) (David et al., 2000a; David et al., 1999)<br>(see Appendix B.1.1.2 for study details)   | - Hepatocellular carcinomas and adenomas (both sexes)   |  |  |  |  |
| Studies of H   | lamsters  |  |  |  |  |
| Male and female Syrian golden hamsters (80/sex for the control;<br>65/sex for treatment group) were exposed to vapor concentrations of<br>0 or $15 \pm 5 \ \mu g/m^3$ DEHP for 24 hours/day, 5 days/week from 12<br>weeks of age until natural death (around 23 months for males and 17<br>months for females) (Schmezer et al., 1988) (see Appendix B.1.3.1<br>for study details) | - None  |  |  |  |  |

May 2025

| Brief Study Description  | Tumor Type(s) Observed                       |
|--|--|
| Male and female Syrian golden hamsters (25/sex/group) were<br>administered 0 or 3,000 mg DEHP per kilogram body weight via<br>intraperitoneal injection once per week, once every 2 weeks, or once<br>every 4 weeks for life ( <u>Schmezer et al., 1988</u> ) (see Appendix B.1.3.2<br>for study details)  | - None                                       |
| Studies of Tran  | sgenic Mice                                  |
| Male and female transgenic CB6F1-rasH2 mice (15/sex/dose) were<br>fed diets containing 0, 1,500, 3,000, 6,000 ppm DEHP for 26 weeks,<br>while wild-type mice (15/sex/dose) were fed diets containing 0 or<br>6,000 ppm DEHP for 26 weeks ( <u>Toyosawa et al., 2001</u> ) (see<br>Appendix B.1.4.1 for study details)  | - Hepatocellular adenomas (rasH2 males only) |
| Male and female transgenic Tg.AC mice (15/sex/dose) were fed<br>diets containing 0, 1,500, 3,000, 6,000 ppm DEHP for 26 weeks<br>(equivalent to 252, 480, 1,000 mg/kg-day [males]; 273, 545, 1,143<br>mg/kg-day [females]) (Eastin et al., 2001) (see Appendix B.1.4.2 for<br>study details)   | - None                                       |
| Male and female transgenic Tg.AC mice (15/sex/dose) were<br>topically administered doses of 0, 100, 200, 400 mg/kg DEHP to a<br>clipped area of dorsal skin 5 days per week for 28 weeks (Eastin et<br>al., 2001) (see Appendix B.1.4.2 for study details)   | - None                                       |
| Male and female $Xpa^{-/-}$ mice (15/sex/dose) fed diets containing 0,<br>1,500, 3,000, 6,000 ppm DEHP (equivalent to: 204, 408, 862<br>mg/kg-day [males]; 200, 401, 827 mg/kg-day [females]) for 39<br>weeks. Male and female wild-type and $Xpa^{-/-}p53^{+/-}$ mice<br>(15/sex/dose) fed diets containing 0 and 6,000 ppm DEHP for 39<br>weeks (equivalent to 879 (male) and 872 (female) mg/kg-day for<br>wild-type mice; 896 (male) and 796 (female) mg/kg-day for $Xpa^{-/-}/p53^{+/-}$ mice) (Mortensen et al., 2002) (see Appendix B.1.4.3 for<br>study details) | - None                                       |

### May 2025

| Brief Study Description  | Tumor Type(s) Observed   |
|--|--|
| Male wild-type and <i>Ppara</i> -null mice fed diets containing 0, 0.01, 0.05% DEHP for 22 months. ( <u>Ito et al., 2007a</u> ) (see Appendix B.1.4.4 for study details) | - Hepatocellular adenoma, carcinoma, and cholangiocellular carcinoma (combined) ( <i>Ppara</i> -null mice) |

May 2025

## 1348 Table 4-6. Summary of Observed Tumors and Effect Levels (LOAEL, mg/kg-day) Across Carcinogenicity Studies of DEHP<sup>a</sup>

| Study Details<br>(Strain; Sexes Evaluated; N; Duration;<br>Doses (mg/kg-day); Table with Tumor<br>incidence data; Reference)  | Study Details<br>train; Sexes Evaluated; N; Duration;<br>oses (mg/kg-day); Table with Tumor<br>incidence data; Reference) Hepatocellular<br>Adenomas and/or<br>Carcinomas |                  | Pancreati       | c Acinar Cell Tumors<br>(PACTs) | Testicular<br>Leydig<br>Cell<br>Adenomas <sup>b</sup> | Uterine<br>Adenoma,<br>Adenocarcinoma,<br>Squamous Cell<br>Carcinoma,<br>Squamous Cell<br>Papilloma | MN              | CL <sup>c</sup>  |
|---|---|------------------|-----------------|---------------------------------|---|---|-----------------|------------------|
|   | Male  | Female           | Male            | Female                          | Male  | Female  | Male            | Female           |
|   | -   |                  | Studie          | es of Rats                      | -   |   |                 | -                |
| F344; M/F; 50/sex/dose; 2-year; 0, 322,<br>674 (M); 0, 394, 774 (F); Table_Apx B-3;<br>( <u>NTP, 1982a</u> )  | ↑<br>(674)  | ↑<br>(394)       | Not<br>observed | Not observed                    | Not<br>observed                                       | Not observed  | Not<br>observed | Not<br>observed  |
| F344; M/F; 55–80/sex/dose; 2-year; 0, 6, 29, 147, 780 (M); 0, 7, 36, 182, 939 (F); Table_Apx B-4; (David et al., 2000b; David et al., 1999)                                 | ↑<br>(147)  | ↑<br>(939)       | ↑<br>(780)      | Not observed                    | Not<br>observed                                       | Not observed  | ↑<br>(780)      | Not<br>observed  |
| SD; M only; 60–390/dose; lifetime (up to<br>159 weeks); 0, 30, 95, 300; Table_Apx<br>B-6 and Table_Apx B-7; ( <u>Voss et al.</u> ,<br><u>2005</u> )                         | ↑<br>(300)  | Not<br>evaluated | Not<br>observed | Not evaluated                   | ↑<br>(300)  | Not evaluated   | Not<br>observed | Not<br>evaluated |
| SD; M/F; 45/sex/dose; 2-year (perinatal<br>and postweaning); 0, 18, 58, 189, 678 (M);<br>0, 18, 62, 196, 772 (F); Table_Apx B-9 to<br>Table_Apx B-11; ( <u>NTP, 2021b</u> ) | ↑<br>(678)  | ↑<br>(196)       | ↑<br>(189)      | Equivocal <sup>d</sup>          | Not<br>observed                                       | Equivocal   | Not<br>observed | Not<br>observed  |
| SD; M/F; 50/sex/dose; 2-years; 0, 17, 54, 170, 602 (M); 0, 17, 60, 177, 646 (F); Table_Apx B-12 to Table_Apx B-15; ( <u>NTP, 2021b</u> )                                    | ↑<br>(602)  | ↑<br>(646)       | ↑<br>(170)      | Equivocal <sup>d</sup>          | Equivocal   | ↑<br>(646)  | Not<br>observed | Not<br>observed  |
| Studies of Mice   |   |                  |                 |                                 |   |   |                 |                  |
| B6C3F1; M/F; 50/sex/dose; 2-year; 0,<br>673, 1,325 (M); 0, 799, 1,821 (F);<br>Table_Apx B-1; ( <u>NTP, 1982a</u> )  | ↑<br>(673)  | ↑<br>(799)       | Not<br>observed | Not observed                    | Not<br>observed                                       | Not observed  | Not<br>observed | Not<br>observed  |
| B6C3F1; M/F; 65-70/sex/dose; 2-year; 0, 19, 99, 292, 1,266 (M); 0, 24, 117, 354,  | ↑<br>(99)   | ↑<br>(354)       | Not<br>observed | Not observed                    | Not<br>observed                                       | Not observed  | Not<br>observed | Not<br>observed  |

May 2025

| Study Details<br>(Strain; Sexes Evaluated; N; Duration;<br>Doses (mg/kg-day); Table with Tumor<br>incidence data; Reference) | Hepato<br>Adenom<br>Carci | Hepatocellular<br>Adenomas and/or<br>Carcinomas |      | c Acinar Cell Tumors<br>(PACTs) | Testicular<br>Leydig<br>Cell<br>Adenomas <sup>b</sup> | Uterine<br>Adenoma,<br>Adenocarcinoma,<br>Squamous Cell<br>Carcinoma,<br>Squamous Cell<br>Papilloma | MNCL <sup>c</sup> |        |
|--|---------------------------|---|------|---------------------------------|---|---|-------------------|--------|
|  | Male                      | Female  | Male | Female                          | Male  | Female  | Male              | Female |
| 1,458 (F); Table_Apx B-2; ( <u>David et al.,</u><br>2000a; <u>David et al., 1999</u> )                                       |                           |   |      |                                 |   |   |                   |        |
| Abbreviations: E – female: M – male: SD – Sprague Dawley   |                           |   |      |                                 |   |   |                   |        |

Abbreviations: F = female; M = male; SD = Sprague Dawley

<sup>*a*</sup> Cells highlighted in blue indicate studies in which a statistically significant increase in incidence of the tumor was observed, while cells with yellow indicate an equivocal tumor response.

<sup>b</sup> As discussed further in Appendix C, F344/N rats have a high spontaneous background rate of testicular Leydig cell tumors (ranging from 86–87%), which reduces the ability of this strain of rat to detect treatment-related increases in this tumor type.

<sup>*c*</sup> MNCL has been observed only in F344 rats, which have a high background rate of MNCL in control rats. As discussed further in Appendix C, there are a number of scientific uncertainties associated with MNCL in F344 rats. Consistent with the recommendations of the SACC (U.S. EPA, 2024q), EPA is not further considering MNCL as a factor in the determination of the cancer classifications for phthalates.

<sup>d</sup>NTP reported a slight, statistically non-significant increase in pancreatic acinar adenomas and/or carcinomas in female rats. NTP considered this lesion to be treatment related, however, given the low, statistically non-significant effect, EPA considered the finding equivocal.

## 4.3.1.1 Liver, Pancreatic, and Testicular Tumors (Tumor Triad)

Many peroxisome proliferator-activated receptor alpha (PPARα) activators are known to induce
hepatocellular adenomas and/or carcinomas in rats and mice, as well as PACTs and testicular Leydig
cell tumors in rats, but not mice (Klaunig et al., 2003). The induction of liver tumors, PACTs, and
testicular Leydig cell tumors in rats by PPARα activators is often referred to as the 'tumor triad'.

1355 1356 DEHP is an established PPARα activator, and across available chronic dietary studies of rats and mice, there is evidence of the 'tumor triad' in rats, while only liver tumors have been observed in mice. As 1357 1358 shown in Table 4-5 and Table 4-6, chronic dietary exposure to DEHP has been shown to consistently 1359 induce hepatocellular adenomas and/or carcinomas in seven studies of male and/or female rats (NTP, 1360 2021b; Voss et al., 2005; David et al., 2000b; David et al., 1999; Rao et al., 1990; Rao et al., 1987; NTP, 1361 1982a), two studies of male and female B6C3F1 mice (David et al., 2000a; David et al., 1999; NTP, 1362 1982a), and in male transgenic RasH2 mice (Toyosawa et al., 2001). Across studies (Table 4-6), 1363 statistically significant increases in hepatocellular adenomas and/or carcinomas have been observed at doses as low as 147 mg/kg-day (lowest-observable-adverse-effect level [LOAEL]) in male F344 rats 1364 1365 (David et al., 2000b; David et al., 1999), 196 mg/kg-day (LOAEL) in female SD rats (NTP, 2021b), and 1366 99 mg/kg-day (LOAEL) in male B6C3F1 mice (David et al., 2000a; David et al., 1999). Additionally, chronic dietary exposure to DEHP has been shown to induce PACTs in three studies of male rats (NTP, 1367 2021b; David et al., 2000b; David et al., 1999) at doses as low as 170 to 189 mg/kg-day DEHP (NTP, 1368 2021b), while statistically significant increases in Leydig cell adenomas have been observed in one 1369 lifetime dietary exposure study of SD rats at doses as low as 300 mg/kg-day (Voss et al., 2005). 1370 1371

Establishing MOA is an important consideration for determining the most appropriate method to use for cancer risk assessment (application of linear low-dose extrapolation vs. a threshold approach) (U.S. EPA, 2005). EPA further considers the MOA for liver tumors in Section 4.3.1.1.1, while the MOA(s) for PACTs and Leydig cell tumors are discussed further in Section 4.3.1.1.2 and 4.3.1.1.3, respectively. Inferences from hypolipidemic drugs known to activate PPAR $\alpha$  and induce the tumor triad in rats, but not humans, are provided in Section 4.3.1.1.4. Finally, remaining uncertainties and limitations and conclusions regarding the tumor triad are provided in Sections 4.3.1.1.5 and 4.3.1.1.6, respectively.

1380

1350

## 4.3.1.1.1 Mode of Action for Liver Tumors in Rats and Mice

1381 Studies have demonstrated that DEHP can activate PPARa in hepatocytes and cause hepatocellular 1382 adenomas and carcinomas in mice and rats. Existing assessments of DEHP by ECB (2008), ECHA 1383 (2017a, b), NICNAS (2010), Health Canada (2015), and U.S. CPSC (2010c) have postulated that DEHP causes liver tumors in rats and mice through a PPARα MOA. In contrast, ATSDR (2022) concluded that 1384 1385 the "exact mechanism(s) by which DEHP induces hepatic cancer in rodents are not precisely known; 1386 however, the available data suggest that multiple molecular targets and pathways are affected in multiple 1387 liver cell types." In addition to a role for PPARa, ATSDR postulated that other molecular targets may 1388 include constitutive and rostane receptor (CAR) activation or activation of nuclear factor kappa B (NF-1389  $\kappa B$ ) leading to chronic inflammation. PPAR $\alpha$  is a nuclear receptor that controls transcription of genes 1390 involved in fatty acid  $\beta$ -oxidation and peroxisome proliferation.

1391

1392 PPARα activation in hepatocytes in rodent models can cause hepatocellular cancer through a non-

1393 genotoxic MOA that involves activation of Kupffer cells. Activated Kupffer cells secrete cytokines such 1394 as tumor necrosis factor alpha (TNF $\alpha$ ), interleukin 1-alpha (IL-1 $\alpha$ ), and interleukin 1-beta (IL-1 $\beta$ ) that

1395 influence hepatocyte growth and fate. As discussed by Corton et al. (2018; 2014), studies have

1396 demonstrated that Kupffer cell activation following PPARα activation plays a crucial role in several

- tumor precursor effects. These effects include increased DNA synthesis and cell proliferation in both
   normal and preneoplastic hepatocytes, as well as suppression of apoptosis. Altered cell growth and
   survival can facilitate clonal expansion of initiated cells leading to the selective clonal expansion of
   preneoplastic foci cells and ultimately tumor formation.
- 1401
- The PPARα MOA for liver tumorigenesis considered by EPA is described further by Corton et al.
  (2018; 2014). The PPARα MOA includes the following sequence of key events (KEs):
- KE1: activation of PPARα in hepatocytes. PPARα activation can be assessed using transactivation assays or by measuring specific events associated with PPARα activation, such as increased expression of genes involved in fatty acid beta oxidation or peroxisome proliferation, increased activity of palmitoyl-CoA oxidase, increased peroxisomal beta oxidation (PBOX), and/or peroxisome proliferation in hepatocytes. Studies have demonstrated that sustained activation of PPARα can lead to alterations in cell growth pathways.
- KE2: alterations in cell growth pathways. For example, PPARα activation can lead to
   activation of Kupffer cells, which produce and secreted cytokines such as TNFα, IL-1α, and IL 1β. Secreted cytokines can alter hepatocyte fate and perturb hepatocyte growth and survival.
- KE3: perturbation of cell growth and survival. Cytokines secreted by Kupffer cells can increase hepatocyte cell proliferation and inhibit apoptosis. Increased cell proliferation may increase the frequency of spontaneous mutations from increased errors in DNA repair or replication. This can enhance the rate of fixation of DNA damage and/or mutations in tumor suppressor genes or activate oncogenes contributing to the formation of preneoplastic foci.
- KE4: selective clonal expansion of preneoplastic foci cells. Fixation of DNA damage and/or mutations in tumor suppressor genes and/or oncogenes can lead to changes in gene expression (*i.e.*, decreased expression of tumor suppressor genes and increased expression of oncogenes) that facilitate clonal expansion of initiated cells, leading to the formation of hepatic foci, and the apical outcome, hepatocellular adenomas and carcinomas.
- Several modulating factors associated with the PPARα MOA have also been proposed, including increases in reactive oxygen species (ROS) and activation of NF- $\kappa$ B (<u>Corton et al., 2018</u>). These modulating factors are not considered necessary to induce liver tumorigenesis but may modulate the dose-response behavior or the probability of inducing one or more KEs (<u>Corton et al., 2014</u>).
- 1427

1428Evidence supporting a PPARα MOA for DEHP-induced liver tumors in rodents has previously been1429evaluated by Corton et al. (2018; 2014) in a manner consistent with the *Guidelines for Carcinogen Risk*1430Assessment (U.S. EPA, 2005) and the IPCS Mode of Action Framework (IPCS, 2007). EPA reviewed1431the PPARα MOA evaluation reported in the publications by Corton et al. (2018; 2014), which are both1432publicly available.<sup>2</sup> Overall, EPA supports the conclusion reached by Corton et al. that the weight of

- 1433 evidence indicates that DEHP-induces liver tumors in rodents through a PPARα MOA.
- 1434

A brief summary of evidence supporting the PPARα MOA for DEHP-induced liver tumors from Corton
et al. (2018; 2014), including a summary of evidence for KEs in the PPARα MOA, dose-response
concordance, temporal relationship, biological plausibility and coherence, and other carcinogenic MOAs
is provided.

<sup>&</sup>lt;sup>2</sup> Corton et al. (2018) available at <u>https://www.ncbi.nlm.nih.gov/pmc/articles/PMC6092738/;</u> Corton et al. (2014) available at <u>https://www.tandfonline.com/doi/full/10.3109/10408444.2013.835784#abstract</u>

## 1441 Summary of Evidence for KEs in PPARα MOA in Rats and Mice

1442 Table 4-7 provides a summary of the occurrence of KEs in the PPAR $\alpha$  MOA in rats and mice. As can be 1443 seen from Table 4-7, DEHP has been shown to activate PPARα in hepatocytes (KE 1) and alter cell 1444 growth pathways (KE 2) in studies of both rats and mice. DEHP has also been shown to alter cell 1445 hepatocyte cell growth and survival in rats and mice (KE 3). In mice, both acute and chronic 1446 hepatocellular proliferative responses have been observed; however, no studies have evaluated apoptosis 1447 in the liver following exposure to DEHP. In rats, DEHP has been shown to cause acute cell proliferation, with chronic cell proliferation being observed in some, but not all studies. However, lack of a consistent 1448 1449 chronic cell proliferative response is not inconsistent with the PPARa MOA. As discussed by Corton et 1450 al. (2018), PPARa activators tend to "produce transient increases in replicative DNA synthesis during the first few days or weeks of exposure followed by a return to baseline levels." Chronic or sustained 1451 1452 proliferative responses for potent PPAR $\alpha$  activators tend to be much lower compared to acute 1453 proliferative responses. Comparatively, DEHP is a relatively weak PPARα activator, and low levels of chronic hepatic cell proliferation may be difficult to detect over variable background levels, which may 1454 1455 explain some of the inconsistencies in chronic cell proliferation. In rats, studies have also demonstrated 1456 that treatment with DEHP can result in a decrease in apoptosis (part of KE 3). For KE 4 (Clonal Expansion of Preneoplastic Foci), no data are available for DEHP in either rats or mice. Finally, as 1457 1458 discussed earlier, a number of bioassays of rats and mice have consistently demonstrated the chronic

1459 oral exposure to DEHP results in hepatocellular adenomas and carcinomas.

1460 1461

1440

## 1462 Table 4-7. Occurrence of Key Events in PPARα MOA in Rats and Mice <sup>*a*</sup>

|         | KE1:   | KE2:<br>Alteration              | KE3: Perturb                | ations of Cell G<br>Survival    | KE4: Clonal<br>Expansion of | Apical<br>Outcome:    |                 |
|---------|--|---------------------------------|-----------------------------|---------------------------------|-----------------------------|-----------------------|-----------------|
| Species | ecies PPARα of Cell<br>Activation Growth<br>Pathways |                                 | Acute Cell<br>Proliferation | Chronic Cell<br>Proliferation   | Apoptosis                   | Preneoplastic<br>Foci | Liver<br>Tumors |
| Rat     | $\uparrow^b$   | $\uparrow^c$ or NC <sup>d</sup> | ↑ <sup>e</sup>              | $\uparrow^f$ or NC <sup>g</sup> | $\downarrow^h$              |                       | $\uparrow^c$    |
| Mouse   | $\uparrow^i$   | $\uparrow^j$                    | $\uparrow^k$                | $\uparrow^l$                    |                             |                       | $\uparrow^m$    |

<sup>a</sup> Table adapted from Figures 1 and 2 in (Corton et al., 2018) and Tables 5 and 6 in (Corton et al., 2014).

<sup>b</sup> (Corton and Lapinskas, 2005)

<sup>c</sup> (Seo et al., 2004; Isenberg et al., 2001; Thottassery et al., 1992; Conway et al., 1989; Cattley et al., 1987; Lake et al., 1987; Rao et al., 1987; Hinton et al., 1986; Kluwe et al., 1985; Kluwe et al., 1982)

<sup>d</sup> (Seo et al., 2004; Tomaszewski et al., 1990; Conway et al., 1989)

<sup>e</sup> (Hasmall and Roberts, 2000; Hasmall et al., 2000; Isenberg et al., 2000; Soames et al., 1999; Marsman et al., 1988; Busser and Lutz, 1987; Smith-Oliver and Butterworth, 1987)

f (Marsman et al., 1988)

<sup>*g*</sup> (<u>Marsman et al., 1988; Cattley et al., 1987</u>)

<sup>h</sup> (Hasmall et al., 2000)

<sup>*i*</sup>(Corton and Lapinskas, 2005; Bility et al., 2004; Isenberg et al., 2001; Issemann and Green, 1990)

<sup>j</sup> (Lee and Lim, 2011; Dwivedi et al., 1989)

k (Isenberg et al., 2000)

<sup>1</sup>(<u>Ward et al., 1988</u>)

<sup>m</sup> (David et al., 1999; <u>Kluwe et al., 1985; Kluwe et al., 1982</u>)

## 1465 **Dose-Response Concordance**

- 1466 Corton et al. (2014) investigated the dose-response relationships of several KEs in the PPAR $\alpha$  MOA in 1467 the livers of male F344 rats in two studies. In the first study by David et al. (2000b; 1999) (summarized 1468 in Appendix B.1.2.2), F344 rats were fed diets containing 0, 100, 500, 2,500, and 12,500 ppm DEHP for
- 1468 up to 104 weeks (equivalent to 6, 29, 147, 780 mg/kg-day for males). In this study, dose-response
- relationships of palmitoyl-CoA oxidase activity (PBOX) (a surrogate measure of PPARα activation),
- 1471 liver-to-body weights (as a surrogate measure for hepatocyte hyperplasia and hypertrophy), and
- 1472 incidence of combined hepatocellular adenomas and carcinomas were evaluated. In the second study by
- 1473 Isenberg et al. (2000), male F344 rats were fed diets containing 0, 1,000, 6,000, 12,000, and 20,000 ppm
- 1474 (equivalent to approximately 100, 600, 1,200, 2,000 mg/kg-day) DEHP in the diet for two weeks, and
- 1475 hepatocyte DNA synthesis was evaluated.
- 1476

In the study by David et al (2000b; 1999), PBOX was induced at 12,500 ppm (only dose evaluated;
equivalent to approximately 780–939 mg/kg-day) at study weeks 1, 2, and 13 weeks, with induction of
PBOX being higher at weeks 2 and 13, compared to week 1. At 104 weeks, PBOX, was significantly
induced at 2,500 ppm (equivalent to 147–182 mg/kg-day) and above. Similarly, relative liver weights
were significantly increased at 500 ppm (equivalent to 29–36 mg/kg-day) and above after 1 week and at
2,500 ppm (equivalent to 147–182 mg/kg-day) and above after 2, 13, and 104 weeks of exposure.

- 1483 Combined hepatocellular adenomas and carcinomas were significantly increased at 2,500 ppm
- 1484 (equivalent to 147–182 mg/kg-day) and above. In the study by Isenberg et al. (2000), increases in
- 1485 periportal and centrilobular hepatic replicative DNA synthesis were observed after two weeks of 1486 exposure to doses of 6,000 ppm DEHP (equivalent to ~600 mg/kg-day) and above. Further dose
- response modeling of these data sets by Corton et al. (2014) indicated that increases in PBOX, relative
- 1488 liver weight (EC50 = 2,994 ppm) and intercellular communication (EC50 = 2,591 ppm) occur at lower
- 1489 doses compared to combined hepatocellular adenomas and carcinomas (EC50 = 15,940 ppm), while
- induction of DNA synthesis occurred at doses coincident with liver tumors (EC50 = 21,140-25,640ppm); see figure 5 of (Corton et al., 2014). Overall, these findings provide evidence of dose-response
- 1491 ppin); see figure 5 of (Corton et al., 2014). Overall, these findings provide evidence of dose-response 1492 concordance, and evidence that the more proximal the KE is to the apical outcome (*i.e.*, hepatocellular
- adenoma and/or carcinoma), the greater the dose needed to induce the KE.
- 1494 1495

# 1496 <u>Temporal Relationship</u>

1497 Corton et al. (2014) also considered the temporal relationship of KEs in the PPAR $\alpha$  MOA leading to liver tumors. Following oral exposure to DEHP, peroxisomal enzyme activity (a surrogate measure for 1498 1499 PPAR $\alpha$  activation [KE 1]) can be detected with days of treatment, and enzyme activity levels quickly 1500 reach a maximum that is maintained for the duration of treatment (Isenberg et al., 2001; Isenberg et al., 1501 2000; David et al., 1999; Ganning et al., 1990; Barber et al., 1987; Mitchell et al., 1985). Temporal 1502 associations of cell proliferation and inhibition of apoptosis (KE 3) are not as well-established for 1503 DEHP. Acute proliferative responses in the liver have been reported as early as one to two weeks 1504 following administration of DEHP (Isenberg et al., 2001; David et al., 1999; James et al., 1998; Conway et al., 1989; Smith-Oliver and Butterworth, 1987; Mitchell et al., 1985). Low levels of chronic 1505 1506 hepatocellular proliferation have been observed in F344 rats for up to one year (Marsman et al., 1988) and up to 40 weeks in B6C3F1 mice (Ward et al., 1988). In contrast, a significant increase in liver 1507 tumors were only observed after two years of exposure to DEHP (David et al., 2000b; David et al., 1508 1509 1999).

- 1510
- 1511 Providing further evidence of a temporal relationship, *in vivo* data on liver tumor incidence indicate that
- 1512 cessation of exposure may alter liver carcinogenesis. For example, in the study by David et al (2000b;
- 1513 <u>1999</u>), there was a lower incidence of liver adenomas, carcinomas and combined adenomas and

- carcinomas in rats fed diets containing 12,500 ppm DEHP for 78 weeks followed by 26 weeks of control
  diet compared to rats maintained on diets containing 12,500 ppm DEHP for 104 weeks (Table\_Apx
  B-4).
- 1516 1517
- 1518 Overall, reasonably available data provide evidence of a temporal relationship between exposure to
- 1519 DEHP and tumorigenesis in the context of KEs in the PPARα MOA in rodents.
- 1520 1521

# 1522 Strength, Consistency, and Specificity

Corton et al. (2014) also considered the strength, consistency, and specificity of the PPARa MOA. As 1523 1524 discussed by Corton et al., activation of PPARa is the only KE that has high specificity for the PPARa 1525 MOA. KE2, KE3, and KE4 have low specificity to the PPARa MOA, and are common to the neoplastic 1526 process in the rodent liver and may overlap in part with other MOAs in the liver, such as the CAR or 1527 aryl hydrocarbon receptor (AhR) MOAs. For DEHP, there is strong and consistent evidence from available in vivo studies of mice and rats that provide evidence that DEHP can activate PPARa (KE1), 1528 1529 alter hepatocellular growth pathways (KE2), cause perturbations of cell growth and survival, including 1530 induce acute and chronic proliferative responses (KE3), and cause hepatocellular tumors (apical 1531 outcome).

1532

## 1533

# 1534 **Biological Plausibility and Coherence**

1535 Biological plausibility for the PPARα MOA is well-established and is discussed by Corton et al. (2018; 2014). Exposure to DEHP has been shown to result in sustained PPAR $\alpha$  activation, increase hepatic 1536 cellular proliferation, decreased apoptosis in the liver, and cause hepatocellular adenomas and 1537 1538 carcinomas in rats and mice. Further, the PPARa MOA is consistent with the biology of carcinogenesis 1539 and tumor formation. Perturbations in cell growth and survival is an inherent characteristic of tumor 1540 formation and carcinogenesis. Alterations in cellular cell growth and survival can enhance the rate of 1541 fixation of DNA damage and/or mutations in tumor suppressor genes or activate oncogenes, leading to 1542 preferential proliferation of cells within preneoplastic foci, such as hepatocellular foci, leading to tumor 1543 formation and carcinogenesis. 1544

1545

# 1546 Other Modes of Carcinogenic Action

*Mutagenicity*. As discussed in Section 3.1, the genotoxicity and mutagenicity of DEHP and its major 1547 1548 metabolites MEHP and 2-EH have been evaluated extensively in various in vitro and in vivo test 1549 systems. Available genotoxicity studies have been reviewed by several authoritative and regulatory agencies. The U.S. CPSC (U.S. CPSC, 2010c), ECHA (ECHA, 2017a, b), EFSA (EFSA, 2019), and 1550 1551 Australia NICNAS (NICNAS, 2010) have concluded that the overall evidence supports the conclusion 1552 that DEHP is non-genotoxic and non-mutagenic. Similarly, the ECB (ECB, 2008) and Environment 1553 Canada (1994) concluded that DEHP and its major metabolites (*i.e.*, MEHP and 2-EH) are not genotoxic or mutagenic. Similarly, NTP (2021b) has concluded "The consensus from published data is that DEHP 1554 1555 shows limited evidence of genotoxic potential, and for the sporadic positive results that have been 1556 reported, the response is either weak, not reproducible, obtained in a nonstandard test system, or qualified to some degree by the authors." Most recently, ATSDR concluded that "The weight of 1557 1558 evidence from these assays indicates that DEHP is not a potent genotoxin but may lead to genotoxic 1559 effects secondary to oxidative stress." Herein, EPA did not independently re-evaluate the extensive 1560 database of in vitro and in vivo genotoxicity studies of DEHP and its major metabolites. However, EPA 1561 agrees with the conclusions of ATSDR, NTP, and other authoritative and regulatory agencies that 1562 available evidence indicates that DEHP and its metabolites are not mutagenic, but that there is some

limited evidence that DEHP may be weakly genotoxic inducing effects such as DNA damage and/orchromosomal aberrations. As noted by ATSDR, these effects may be secondary to oxidative stress.

1565 1566 Studies of Ppara-Null Mice. Several studies of DEHP have been conducted in Ppara-null mice (Ren et 1567 al., 2010; Eveillard et al., 2009; Ito et al., 2007a). Ito et al. fed wild-type and *Ppara*-null male mice diets 1568 containing 0, 0.01, 0.05 percent DEHP (equivalent to approximately 15 and 75 mg/kg-day) for 22 1569 months (see Appendix B.1.4.4 for study summary). No significant increase in liver tumors was observed 1570 in wild-type mice, while a slight, yet statistically significant increase in combined hepatocellular 1571 adenomas and carcinomas, and cholangiocellular carcinomas was observed in 8 out of 31 high-dose 1572 Ppara-null mice. This result suggests MOAs other than PPARa may be operative in the liver and 1573 contribute to liver tumorigenesis. However, there are a number of limitations associated with the study 1574 by Ito et al. (2007a), which have been discussed extensively elsewhere (Corton et al., 2018; Corton et 1575 al., 2014). First, to achieve statistical significance, Ito et al. combined tumor types originating from 1576 different cell types. It is inappropriate to combine hepatocellular adenomas and carcinomas with hepatoblastomas for purposes of determining statistical significance. However, a statistical re-analysis 1577 1578 by Guyton et al. (2009) found that adenomas and combined adenomas and carcinomas were 1579 significantly increased in high-dose *Ppara*-null mice, addressing this limitation. A second source of 1580 uncertainty stems from the fact that no significant increase in liver tumors was observed in wild-type 1581 mice at either dose tested after 22 months, which complicates the interpretation of the small increase in 1582 liver tumors in *Ppara*-null mice. Further, given the lack of liver tumors in wild-type mice, the small 1583 increase in liver tumors in *Ppara*-null mice may represent a chance finding. This is supported by the fact 1584 that aged *Ppara*-null mice are known to have increased incidence of spontaneous hepatocellular 1585 adenoma and carcinoma in the absence of chemical treatment compared to similarly aged wild-type 1586 mice (Howroyd et al., 2004). Spontaneous occurrence of liver tumors in *Ppara*-null mice appears to be related to increased hepatic lipid accumulation (steatosis) compared to wild-type mice due to decreased 1587 1588 constitutive expression of lipid metabolizing enzymes (Kersten et al., 1999; Leone et al., 1999; Aoyama 1589 et al., 1998). The possibility remains that DEHP is contributing to the mechanism related to the increase 1590 in spontaneously occurring liver tumors. Another possibility is that DEHP is inducing liver tumors 1591 though another nuclear receptor, such as CAR in the absence of PPARa. 1592

1593 Gene expression changes in the liver have also been evaluated by microarrays in wild-type and Ppara-1594 null mice gavaged with 0, 200, or 1,150 mg/kg-day DEHP for 4 days (Ren et al., 2010). A comparison 1595 of gene expression changes in the livers of wild-type and *Ppara*-null mice indicated that PPARa is 1596 required for approximately 94 percent of transcriptional changes. The remaining 6 percent of genes were 1597 predominantly involved in xenobiotic metabolism and are known to be targets of CAR or PXR. 1598 Additionally, CAR-regulated genes were more strongly induced by DEHP in *Ppara*-null mice compared 1599 to wild-type mice, which may indicate that in the absence of PPAR $\alpha$  other nuclear receptors such as 1600 CAR become a dominant pathway for carcinogenesis (Ren et al., 2010). Similar results were obtained in 1601 an gene array study of 320 nuclear receptor target genes in the livers of male wild-type and male *Ppara*-1602 null mice gavaged with 0, 20 or 200 mg/kg-day DEHP for 21 days (Eveillard et al., 2009). In this study, most DEHP-regulated genes in the liver were PPARa-dependent, however, several genes specifically 1603 1604 regulated by CAR were identified.

1605

1606 *Other Nuclear Receptors.* Pregnane X receptor (PXR), CAR, and AhR are known to play a role in liver 1607 homeostasis and disease. Although their precise role, if any, in liver tumorigenesis in response to 1608 chronic exposure to DEHP is unknown. In addition to PPAR $\alpha$ , DEHP has been shown to activate 1609 multiple nuclear receptors that may play a role in liver tumorigenesis. For example, DEHP has been 1610 shown to be a weak inducer of AhR activity *in vitro*. In an AhR-CALUX assay with transfected mouse 1611 hepatoma cells (Hepa1.12cR) exposed to concentrations of  $1 \times 10^{-10}$  to  $1 \times 10^{-4}$  M DEHP, AhR activity

1612 was induced only at the highest concentration of DEHP tested and was only induced 1.75-fold above the 1613 solvent control (Kruger et al., 2008). In another in vitro study, mouse 3T3-L1 fibroblasts were 1614 transfected with mouse or human PPAR $\alpha$ , PPAR gamma (PPAR $\gamma$ ) or PPAR beta (PPAR $\beta$ ) reporters and 1615 exposed to 3 to 200 µM concentrations of MEHP for 24 hours (Bility et al., 2004). MEHP was found to 1616 activate mouse and human PPAR $\alpha$  (lowest activation concentration: 10 µM [mouse] and 30 µM [human]), mouse and human PPAR $\gamma$  (lowest activation concentration: 30  $\mu$ M [mouse] and 10  $\mu$ M 1617 1618 [human]), as well as mouse (but not human) PPAR $\beta$  (lowest activation concentration: 200  $\mu$ M). 1619 DeKeyser et al. (2011) demonstrated that DEHP can activate human PXR as well as certain human CAR splice variants (e.g., CAR2) in various in vitro cell models. Briefly, COS-1 cells were transfected with 1620 1621 the 2B6-XREM-PBREM luciferase reporter and treated with 0 (0.1% DMSO vehicle control), 0.1, 1, or 1622 10  $\mu$ M DEHP for 48 hours. DEHP was found to be strong activator of human CAR2 (EC50 = 0.1  $\mu$ M) 1623 and PXR (EC50 =  $3.8 \mu$ M), but showed little to no activation of CAR1 or CAR3 splice variants (EC50 1624 values could not be determined). Finally, Laurenzana et al. (2016) demonstrated that MEHP can activate 1625 human CAR2 and PXR, as well as human PPARα, PPARβ, and PPARγ in several *in vitro* models. Briefly, COS-1 cells were transfected with the 2B6-XREM-PBREM luciferase reporter (for the CAR2, 1626 1627 CAR3, and PXR assays) or the PPRE luciferase reporter (for the PPAR $\alpha$ , PPAR $\beta$ , and PPAR $\gamma$  assays) 1628 and exposed to 0.1 to 100 µM MEHP for 24 hours. Treatment with MEHP activated the human CAR2 1629 splice variant at 1  $\mu$ M and above, PPAR<sub>y</sub> at 10  $\mu$ M and above, and human PXR, PPAR $\alpha$ , and PPAR $\beta$  at 1630 100 µM, while no human CAR3 activity was detected at any concentration.

1631

As discussed above, gene expression changes in the liver of mice gavaged with DEHP consistent with activation of CAR and PXR have also been noted in several *in vivo* studies (<u>Ren et al., 2010</u>; <u>Eveillard et</u> <u>al., 2009</u>). These *in vivo* studies of mice provide evidence that oral exposure to DEHP can activate CAR and PXR signaling pathways in the liver.

1636 1637

# 1638 Uncertainties and Limitations

1639 There are several limitations and uncertainties associated with the available data set for the PPAR $\alpha$ 1640 MOA. First, no data is available for KE4 for rats or mice. Lack of data for KE4 is a data gap, which 1641 reduces EPA's confidence in the postulated PPARa MOA. Another uncertainty is potential contribution to carcinogenesis by other nuclear receptors. DEHP and its metabolite MEHP have been shown to 1642 1643 activate CAR, PXR, and to a lesser extent AhR in vitro, while transcriptomics studies have also 1644 demonstrated that DEHP can activate CAR and PXR signaling pathways in vivo in mice. However, the majority of transcriptional changes in these studies appear to attributable to PPARa, and to a lesser 1645 1646 extent CAR and PXR (Ren et al., 2010; Eveillard et al., 2009). Despite remaining uncertainties, there is strong evidence to support the PPARa MOA. Available evidence indicates that DEHP is not mutagenic 1647 1648 or a directly genotoxic (Section 3.1). Furthermore, other potential modes of carcinogenic action, such as 1649 activation of CAR, PXR, and AhR, are also non-genotoxic threshold MOAs.

- 1650
- 1651

## 4.3.1.1.2 Mode of Action for Pancreatic Acinar Cell Tumors (PACTs)

Some initial work has been done to establish the MOA for induction of PACTs through PPAR $\alpha$ activation. Klaunig et al. (2003) proposed an initial MOA for induction of PACTs through PPAR $\alpha$ activation in rat. In the proposed MOA, PACTs occur secondary to liver toxicity. However, little work has been done to refine the initially proposed MOA. The MOA for induction of PACTs proposed by Klaunig et al. involves four KEs. The proposed MOA and supporting evidence is discussed in detail in the publication by Klaunig et al. (2003), and is briefly summarized below.

- *KE 1: Activation of PPARa in the liver*. PPARa activation in the liver leads to a decrease in transcription of cholesterol 7α-hydroxylase (CYP7A1), which leads to a disruption of bile acid synthesis. Cholesterol 7α-hydroxylase is the first and rate-limiting enzyme in bile acid synthesis from cholesterol.
- *KE 2a: Decreased bile acid flow*. Treatment with certain PPARα activators such as WY 14,643
   (WY) have been demonstrated to decrease bile acid flow in the liver, which in turn can increase cholecystokinin (CCK).
- *KE2b: Altered bile acid composition*. Treatment with several PPARα activators such as WY, clofibrate, and nafenopin have been shown to alter bile acid composition. Decreased bile acid flow (KE 2a) and/or altered bile acid composition (KE 2b) lead to increases in CCK release from mucosal cells in the intestine into the bloodstream.
- 1670 KE3: Cholestasis. Several PPARa activators such as WY, gemfibrozil, methylclofenopate, and 1671 tibric acid have been shown to produce clinical pathology indicative of cholestasis. Cholestasis is 1672 believed to occur as a consequence of KE 2a and KE 2b. Decreasing bile acid flow (KE 2a) 1673 and/or composition (KE 2b) have been shown to increase CCK levels. Bile acids are believed to 1674 enhance the effectiveness of trypsin, and thus decreased bile acid flow and altered bile acid 1675 composition are believed to reduce the effectiveness of trypsin, which in turn leads to an increase 1676 in monitor peptide binding to M(I) cells in the duodenal mucosa leading to increases in CCK 1677 release.
- *KE4: Increased plasma CCK.* Treatment with the PPARα activator WY has been shown to increase plasma CCK levels, which correlated with cholestasis (KE 3). Increase plasma CCK levels are thought to cause pancreatic acinar cell proliferation, which in turn leads to the apical outcome, PACTs.

1682 Although an MOA has been proposed for PACTs, which involves an increase in CCK that drives 1683 proliferation of pancreatic acinar cells, little work has been done to refine this MOA. Further, data for 1684 the KEs in the proposed MOA are generally not available for DEHP beyond evidence of PPARa 1685 activation in the liver (KE 1) and the apical outcome, PACTs, based on information provided in previous assessments of DEHP. EPA did not further evaluate evidence for DEHP supporting KEs in the MOA 1686 1687 proposed by Klaunig et al. (2003). EPA did not identify any other proposed MOAs for PACTs. 1688 Regardless, the possibility remains that mechanisms other than PPARa may play a role in the observed 1689 PACTs in rats and this is a source of uncertainty.

1690 1691

# 4.3.1.1.3 Mode of Action for Leydig Cell Tumors

Some initial work has been done to establish the MOA for induction of Leydig cell tumors for PPARa 1692 1693 activators. Klaunig et al. (2003) proposed two potential pathways for induction of Leydig cell tumors by 1694 PPAR $\alpha$  activators in the rat, both of which may contribute to Leydig cell tumor formation. As part of the 1695 first pathway, Leydig cell adenomas occur secondary to liver toxicity, and tumorigenesis is driven by 1696 increases in interstitial fluid estradiol and transforming growth factor alpha (TGFa) levels. In the second 1697 pathway, direct inhibition of testis testosterone biosynthesis leads to a disruption of the hypothalamic-1698 pituitary-thyroid axis leading to an increase in Luteinizing hormone and Leydig cell tumors. However, 1699 little work been done to refine the two initially proposed pathways since 2003. The two proposed 1700 pathways for Leydig cell tumorigenesis and supporting evidence is discussed in detail in the publication 1701 by Klaunig et al. (2003), and is briefly summarized below.

- 1702
- 1703 Pathway 1 (Secondary to liver induction)

#### May 2025

- 1704 **KE 1:** Activation of PPARα in the liver.
- *KE 2a: Increased aromatase (CYP19A1).* Aromatase is an enzyme that plays a role in converting androgens to estrogens. Several PPARα activators have been shown to increase hepatic aromatase, as well as estradiol levels, indicating induction of aromatase in Leydig cells.
- *KE 2b: Decreased estradiol metabolism*. Several PPARα activators such as clofibrate,
   gemfibrozil, and WY-14,643 have been shown to reduce estradiol metabolism, which leads to an
   increase in serum estradiol levels.
- *KE 3: Increased serum estradiol levels*. Increased serum estradiol levels may be due to
   increased expression of aromatase (KE 2a) and/or decreased estradiol metabolism (KE 2b).
- *KE 4: Increased interstitial fluid estradiol.* An increase in serum estradiol levels leads to an
   increase in interstitial fluid estradiol levels. Interstitial fluid bathes Leydig cells and seminiferous
   tubules leading to increased estradiol exposure for these cell types.
- *KE 5: Increased transforming growth factor alpha (TGFa) levels in interstitial fluid.* Increases
   in TGFα have been observed in the interstitial fluid for some PPARα activators.
- *KE 6: Increased Leydig Cell Proliferation.* TGFα has been shown to stimulate Leydig-cell proliferation, which can in turn lead to the apical outcome, Leydig cell tumors.

## 1720 Pathway 2 (Direct inhibition of testosterone biosynthesis at the level of the testis)

- **•** *KE* 7: ↓ *Testosterone biosynthesis*.
- *KE 8: Decreased testosterone levels.* Several PPARα activators, including DEHP, have been shown to decrease testosterone levels due to decreases in testosterone biosynthesis.
- *KE 9: Increased Luteinizing hormone levels.* Inhibition of testosterone biosynthesis leads to a disruption of the hypothalamic-pituitary-thyroid axis, leading to increased Luteinizing hormone levels.
- *KE 10: Leydig cell tumorigenesis*. Increases in Luteinizing hormone is established to induce
   Leydig cell tumors.

Although an MOA has been proposed for Leydig cell tumors, little work has been done to refine this
MOA, and EPA did not further evaluate evidence for DEHP supporting KEs in the MOA proposed by
Klaunig et al. (2003). EPA did not identify any other proposed MOAs for Leydig cell tumors.
Regardless, the possibility remains that mechanisms other than PPARα may play a role in the observed
Leydig cell tumors in rats and this is a source of uncertainty.

- 1734
- 1735

## 4.3.1.1.4 Inferences from Hypolipidemic Drugs and Other Prototypical PPARα Activators

1736 1737 Although there is uncertainty pertaining to the precise mechanisms underlying DEHP-induced PACTs 1738 and Leydig cell tumors, there is evidence to suggest that the tumor triad is a fingerprint of chronic 1739 PPARa activation in rats (Klaunig et al., 2003). For example, similar to DEHP, prototypical PPARa 1740 activators such as WY 14,643 (WY, also known as prinixic acid) and hypolipidemic drugs (e.g., 1741 clofibrate, fenofibrate, gemfibrozil) that are commonly prescribed to humans to lower serum cholesterol 1742 and triglyceride levels have also been shown to induce the tumor triad in rats (Table 4-8), but not 1743 humans (discussed further below). Mechanistically, WY and these lipid-lowering agents operate through 1744 activation of PPARa. Notably, these drugs are commonly prescribed at doses several orders of

May 2025

magnitude higher than levels of exposure to DEHP for the general U.S. population based on NHANESurinary biomonitoring data (discussed further below).

1747

1748 Clofibrate (trade name Atromid-S), which was first approved for use as lipid-lowering agent in 1963, 1749 was discontinued in 2002 due to adverse effects unrelated to cancer (*i.e.*, gallstone formation). 1750 Methylclofenapate is a derivative of clofibrate that underwent clinical studies for use as a hypolipidemic 1751 agent but was never approved for use by the FDA. Fenofibrate (trade names Tricor, Antara, Lipofen, 1752 etc.) has been used as a lipid-lowering agent since 1975 and is one of the most commonly prescribed 1753 medications in the U.S. In 2022, fenofibrate was prescribed over 7.8 million times and was the 88th 1754 most prescribed drug in the U.S. (ClinCalc, 2024a). Maximum prescribed doses of fenofibrate are 200 1755 mg/day, equivalent to a dose of 2.5 mg/kg-day for an 80 kg individual. Gemfibrozil (trade name Lopid) 1756 was approved for use as a lipid-lowering agent in 1982 and was the 231st most prescribed drug in the 1757 U.S. in 2022 with over 1.5 million prescriptions (ClinCalc, 2024b). Maximum prescribed doses of 1758 gemfibrozil are 1,200 mg/day, which equates to a dose of 15 mg/kg-day for an 80 kg individual. Notably, slightly higher doses of 30 mg/kg-day gemfibrozil have been shown to induce the tumor triad 1759 1760 in rats (Table 4-8) but have no effect on cancer outcomes in humans (discussed further below). 1761 Comparatively, administered doses of fenofibrate and gemfibrozil are approximately three orders of 1762 magnitude higher than the 95th percentile DEHP daily intake estimate of 4.5 µg/kg-day for all 1763 NHANES participants surveyed in the most recent NHANES cycle between 2017 to 2018 (see EPA's 1764 Draft Environmental Media and General Population and Environmental Exposure for Diethylhexyl Phthalate (DEHP) for further details (U.S. EPA, 2025b)). As can be seen from Table 4-8, clofibrate, 1765 1766 methylclofenapate, fenofibrate, gemfibrozil and WY have all been demonstrated to induce the tumor triad in rats. 1767

1768

Several large retrospective epidemiological studies examined the relationships between chronic 1769 1770 treatment with the hypolipidemic agents gemfibrozil and clofibrate, and liver cancer (reviewed in (Peters 1771 et al., 2005; Klaunig et al., 2003)). In two large studies, there was no reported elevated risk of mortality 1772 from liver cancer associated with over a decade of chronic use of these pharmaceuticals (Tenkanen et 1773 al., 2006; Huttunen et al., 1994; Frick et al., 1987). One possible exception is a cohort in which excess mortality due to a higher incidence of the malignant neoplasms of the "liver, gallbladder and intestines" 1774 was reported in clofibrate-treated subjects. However, death rates among the clofibrate-treated group for 1775 1776 cancer were similar to the official mortality statistics for individuals from the same area; the number of 1777 observed cases of gastrointestinal cancers was very small; and importantly, there was no difference 1778 among groups in a follow-up analysis of the mortality trends in this cohort (WHO, 1978). A meta-1779 analysis of 17 randomized placebo-controlled trials was carried out by Bonovas et al. (2012). The 1780 analysis included 44,929 participants with an average follow-up of 5.2 years from 4 trials for 1781 bezafibrate, 6 trials for clofibrate, 3 trials for fenofibrate, and 4 trials for gemfibrozil. Overall, the 1782 authors found that fibrates have no effect on cancer outcomes in humans. In summary, fibrate drugs 1783 have been on the market since 1977 without an apparent increase in cancer in people taking them 1784 chronically, even at doses approximately three orders of magnitude higher than phthalate exposure 1785 levels for the general U.S. civilian population based on NHANES biomonitoring data.

1786

1787 Collectively, studies of WY and hypolipidemic drugs, which are prototypical PPARα activators, provide
1788 evidence indicating that the tumor triad is a signature of PPARα activation and given that these
1789 hypolipidemic drugs have not been linked to cancer outcomes in humans, raise questions pertaining to

1790 the human relevancy of the tumor triad observed in rats following chronic exposure to DEHP.

May 2025

## 1791 Table 4-8. Summary of Two-year Tumor Findings in Rats Administered Hypolipidemic Drugs

| Drug        | Exposure Route (Method); Duration;<br>Species (Strain); Sexes Tested; Dose<br>Levels (Reference)   | Tumor Incidence (Number of animals with tumors/number<br>examined by dose group)   |
|-------------|--|--|
| Clofibrate  | Oral (not specified); 2 years; Rat<br>(Wistar); Males; 0, 200, 400 mg/kg<br>[(PDR, 1995) as reported in Table 35 of<br>(Klaunig et al., 2003)]                         | Liver (Male): Positive liver tumor finding reported (incidence data not<br>provided)<br>Leydig cell tumor (Male): Positive liver tumor finding reported<br>(incidence data not provided)   |
|             | Oral (dietary); 24–28 months; Rat<br>(F344); Males; 0, 0.5% (v/w) ( <u>Reddy</u><br>and <u>Qureshi, 1979</u> )   | Liver (Male): 0/14, 10/11 (carcinoma)<br>PACT (Male): 0/14, 2/11 (carcinoma)   |
|             | Oral (dietary); 72–97 weeks; Rat (F344);<br>Males; 0, 0.5% (v/w) ( <u>Svoboda and</u><br><u>Azarnoff, 1979</u> )   | Liver (Male): 0/25, 4/25 (carcinoma)<br>PACT (Male): 0/25, 4/11 (combined adenoma and carcinoma)   |
| Fenofibrate | Oral (not specified); 2 years; Rat (not<br>specified); Male and Female; 0, 10, 45,<br>200 mg/kg-day [(PDR, 2002) as reported<br>in Table 35 of (Klaunig et al., 2003)] | Liver (Male): Positive tumor finding in high-dose group (incidence data<br>not provided)<br>Leydig cell tumor (Male): Positive tumor finding in high-dose group<br>(incidence data not provided)<br>PACT (Male): Positive tumor finding in high-dose group (incidence<br>data not provided)<br>Liver (Female): Positive tumor finding in high-dose group (incidence<br>data not provided)<br>PACT (Female): No tumors observed |
| Gemfibrozil | Oral (dietary); 2 years; Rat (SD); Males<br>and Females; 0, 30, 300 mg/kg<br>( <u>Fitzgerald et al., 1981</u> )  | Liver (Male): 1/50, 6/60, 23/50 (combined adenoma and carcinoma)<br>Leydig cell tumor (Male): 1/50, 8/50, 17/50<br>PACT (Male): 0/50, 6/50, 2/50   |
|             |  | PACT (Female): 9/50, 5/50, 0/50<br>PACT (Female): 0/50, 0/50, 0/50   |

### May 2025

| Drug              | Exposure Route (Method); Duration;<br>Species (Strain); Sexes Tested; Dose<br>Levels (Reference)   | Tumor Incidence (Number of animals with tumors/number examined by dose group)  |
|-------------------|--|--|
| Methylclofenapate | Oral (dietary); Rat (Wistar); 2 years;<br>Males and Females; 0, 10, 50, 250 ppm<br>[(Tucker and Orton, 1995) as reported in<br>Table 35 of (Klaunig et al., 2003)] | Liver (Male): 0/24, 0/24, 9/25, 22/23 (carcinoma)<br>Leydig cell tumor (Male): 1/24, 3/24, 10/25, 9/23<br>PACT (Male): 2/24, 5/24, 6/25, 9/23<br>Liver (Female): 0/24, 1/24, 4/25, 20/24 (carcinoma)<br>PACT (Female):0/24, 0/24, 1/25, 2/20 |
| WY-14,643         | Oral (dietary); 2 years; Rat (CD); Males<br>only; 0, 50 ppm (reduced to 25 ppm on<br>study day 301 due to increased mortality)<br>( <u>Biegel et al., 2001</u> )   | Liver (Male): 2/80, 17/67 (combined adenoma and carcinoma)<br>Leydig cell tumor (Male): 0/80, 16/67 (adenoma)<br>PACT (Male): 0/80, 25/67 (adenoma)  |

## 1793

## 4.3.1.1.5 Uncertainties, Limitations, and Human Relevance

1794 There are several limitations and uncertainties associated with the available data set for the occurrence 1795 of liver tumors in mice and rats, and PACTs and Leydig cell tumors in rats. First, there is uncertainty 1796 related to the precise mechanisms underlying PACTs and Levdig cell tumors in rats. Although initial 1797 MOAs that involve PPAR $\alpha$  activation have been proposed for both tumor types (see Sections 4.3.1.1.2) 1798 and 4.3.1.1.3), little work has been done to refine the initially proposed MOAs. This uncertainty reduces 1799 EPA's confidence that DEHP causes PACTs and Leydig cell tumors through PPARα activation. 1800 However, inferences from hypolipidemic drugs help to address this uncertainty. For example, WY, a 1801 selective PPARa activator, and other hypolipemic drugs that reduce serum lipids by activating PPARa, also cause PACTs and Leydig cell tumors in rats, but, as discussed further below, not humans (see 1802 1803 Section 4.3.1.1.4). Regardless, the possibility remains that mechanisms other than PPARa may play a 1804 role in the observed PACTs and Leydig cell tumors in rats, such as activation of other nuclear receptors 1805 or cytotoxicity and regenerative proliferation.

1806

1807 Another source of uncertainty stems from the fact that not all phthalates induce the tumor triad in rats. 1808 As discussed further in subsequent sections of this document, chronic oral exposure to DINP induces 1809 liver tumors in mice and rats, but has not been shown to cause PACTs in F344 rats, SD rats, or B6C3F1 1810 mice (see Section 4.3.4 and (U.S. EPA, 2025a)). Although, as discussed in (U.S. EPA, 2025a), one study of SD rats does provide some limited evidence of a carcinogenic response in the testis following chronic 1811 1812 dietary exposure to DINP (Bio/dynamics, 1987), as demonstrated by a statistically significant increase in Leydig cell hyperplasia (incidence: 4/69 [5.8%] in control vs. 22/70 [31%] in high-dose (553 mg/kg-1813 1814 day) group); however, the incidence of Leydig cell tumors in this study was statistically non-significant (2/69 [2.9%] in controls vs. 7/70 [10%] in high-dose group). Chronic oral exposure to DIDP induces 1815 liver tumors in transgenic rasH2 male mice, but does not induce liver tumors, PACTs, or Levdig cell 1816 1817 tumors in F344 rats (see Section 4.3.5 and (U.S. EPA, 2024n)). As will be discussed further in Section 1818 4.3.2, chronic oral exposure to BBP induces PACTs in F344 rats but does not induce liver tumors or 1819 Leydig cell tumors in F344 rats. Finally, and as will be discussed further in Section 4.3.3, chronic 1820 dietary exposure to DBP induces PACTs in male SD rats, and there is some limited evidence of Leydig 1821 cell hyperplasia in male SD rats, however, statistically significant increases in Levdig cell tumors have 1822 not been observed, nor have liver tumors been observed following chronic exposure to DBP. 1823

1824 Some of the observed inconsistencies in induction of the tumor triad by phthalates may be explained by

1825 the strain of rat tested, doses tested, or differences in phthalate potencies to induce PPAR $\alpha$  activation.

1826 For example, BBP and DIDP have only been evaluated for carcinogenicity in F344 rats (Section 4.3.3

and Section 4.3.5), which is a strain of rats that has a high (ranging from 86–87%) spontaneous
background rate of Leydig cell tumors (Cook et al., 1999), making it difficult to detect treatment-related

1828 background rate of Leydig cell tumors (<u>Cook et al., 1999</u>), making it difficult to detect treatment-related 1829 increases in this tumor type in this strain of rat (discussed further in Appendix C). In the one available

1829 increases in this tunior type in this strain of rat (discussed runner in Appendix C). In the one available 1830 study of DIDP with F344 rats (<u>Cho et al., 2010</u>; <u>Cho et al., 2008</u>), biomarkers of PPAR $\alpha$  activation in

the liver were increased after 12, but not 32 weeks of exposure, indicating that exposure to DIDP did not

sustain PPARα activation, which may explain the lack of observed liver tumors and PACTs in this study
 (Section 4.3.5 and (U.S. EPA, 2024n)). Finally, compared to WY and other hypolipidemic drugs,

1834 phthalates are generally considered weak PPAR $\alpha$  activators (Klaunig et al., 2003; Barber et al., 1987),

1835 although DEHP, DIDP, and DINP do appear to be more potent activators of PPARα *in vivo* in rats

1836 compared to BBP and DBP (<u>Barber et al., 1987</u>). Differences in potency for activating hepatic PPARα

may account for differences in observed liver tumors, PACTs, and Leydig cell tumors across DEHP,
DINP, DIDP, BBP, and DBP.

1840 Another source of uncertainty is human relevance of tumors in the triad. Several panels have been 1841 convened to address the human relevancy of liver tumors in rodents occurring through a PPARa MOA 1842 (Felter et al., 2018; Corton et al., 2014). These panels have generally concluded that the PPARa MOA is 1843 not relevant to humans or unlikely to be relevant to humans based on qualitative and quantitative 1844 differences between species. Consistent with the recommendations of previous panels, most SACC 1845 committee members during the July 2024 peer review meeting of DIDP and DINP supported the 1846 conclusion that liver tumors seen in rodents caused by a PPAR $\alpha$  MOA are not likely to be or are not relevant to humans because "the preponderance of the evidence that PPARa activation in the human 1847 1848 does not trigger, at any dose, the obligatory KEs that would lead to the liver tumors observed in rodents" (U.S. EPA, 2024q). Nevertheless, uncertainty and differing scientific opinions on the human relevance 1849 1850 of the PPARa MOA for liver tumorigenesis remain, despite the related efforts of previous panels and 1851 workshops. Additionally, and as discussed above in Section 4.3.1.1.4, fibrate drugs have been on the 1852 market since 1977 without an apparent increase in cancer in people taking them chronically, even at 1853 doses approximately three orders of magnitude higher than phthalate exposure levels for the general U.S. civilian population based on NHANES biomonitoring data. These findings for fibrate drugs raise 1854 1855 questions pertaining to the human relevance of observed liver tumors, PACTs, and Leydig cell tumors 1856 observed in rats chronically treated with DEHP.

1857

### 1858

### 4.3.1.1.6 Conclusions Regarding Tumor Triad

1859 Despite some remaining uncertainties, the weight of scientific evidence indicates that the tumor triad is 1860 related to PPAR $\alpha$  activation in rats following chronic exposure to DEHP and hypolipidemic drugs. 1861 Given that DEHP is not a direct acting mutagen or genotoxicant (Section 3.1), a non-linear threshold 1862 approach is supported for cancer risk assessment of the tumor triad for DEHP.

1863

1864

1867

## 4.3.1.2 Uterine Tumors

There is some evidence for uterine tumors in female SD rats following chronic oral exposure to DEHP
based on two studies by NTP (2021b).

1868 In the first study, time-mated female SD rats were fed diets containing 0, 300, 1,000, 3,000 or 10,000 1869 ppm DEHP throughout gestation and lactation starting on gestation day (GD) 6. At weaning on postnatal day (PND) 21, Groups of 50 male and female F1 offspring were fed diets containing the same respective 1870 DEHP concentrations for two-years. Received doses for female F1 offspring were 18, 62, 196 and 772 1871 1872 mg/kg-day during the two-year phase of the study. At study termination, there was a significant trend in 1873 increased incidence of uterus endometrium adenocarcinoma and combined incidence of uterus adenoma, 1874 adenocarcinoma, squamous cell carcinoma, or squamous cell papilloma (Table 4-9). However, pair-wise 1875 comparisons to the control were not statistically significant, and NTP characterized the uterine tumors as an equivocal finding. Although DEHP did not significantly affect female survival in any treatment 1876 1877 group, and no DEHP-related clinical findings were observed, body weight gain was significantly lower 1878 in females of the 10,000 ppm group throughout the study, and terminal mean body weight for high-dose 1879 females was 32 percent lower than that of the concurrent control group, indicating exceedance of the 1880 maximum tolerable dose (MTD).

- 1881
- 1882

#### May 2025 Table 4-9. Incidence of Uterine Tumors in SD Rats Chronically Exposed to DEHP (Perinatal and

#### 1883 1884

| Tissue: Tumor Type   | 0 ppm      | 300 ppm   | 1000 ppm  | 3000 ppm   | 10,000 ppm |
|--|------------|-----------|-----------|------------|------------|
| Adenoma <sup>bf</sup>  | 0/50       | 1/50      | 0/50      | 0/50       | 0/48       |
| Adenocarcinoma (overall rate) <sup>bg</sup>  | 3/50 (6%)  | 0/50      | 1/50 (2%) | 3/50 (6%)  | 6/48 (13%) |
| Adenocarcinoma (rate per litter) <sup>c</sup>  | 3/25 (12%) | 0/25      | 1/25 (4%) | 3/25 (12%) | 6/25 (24%) |
| Adenocarcinoma (adjusted rate) <sup>d</sup>  | 7%         | 0%        | 2.4%      | 7%         | 16.4%      |
| Rao-Scott-adjusted Poly-3 test <sup>e</sup>  | p = 0.008  | p = 0.147 | p = 0.325 | p = 0.653  | p = 0.184  |
| Squamous cell carcinoma (includes multiple) $^{h}$   | 0/50       | 1/50      | 0/50      | 0/50       | 1/48       |
| Squamous cell papilloma (includes multiple) <sup><i>i</i></sup>  | 0/50       | 0/50      | 0/50      | 1/50       | 0/48       |
| Adenoma, adenocarcinoma, squamous cell<br>carcinoma, squamous cell papilloma<br>(combined) (overall rate) $^{j}$ | 3/50 (6%)  | 1/50 (2%) | 1/50 (2%) | 3/50 (6%)  | 7/48 (15%) |
| Adenoma, adenocarcinoma, squamous cell<br>carcinoma, squamous cell papilloma<br>(combined) (rate per litter)     | 3/25 (12%) | 1/25 (4%) | 1/25 (4%) | 3/25 (12%) | 7/25 (28%) |
| Adenoma, adenocarcinoma, squamous cell<br>carcinoma, squamous cell papilloma<br>(combined) (adjusted rate)       | 7%         | 2.4%      | 2.4%      | 7%         | 19%        |
| Rao-Scott-adjusted Poly-3 test <sup>e</sup>  | p = 0.005  | p = 0.325 | p = 0.317 | p = 0.651  | p = 0.113  |

<sup>*a*</sup> Adapted from Table 17 in (NTP, 2021b).

Postweaning Exposure Study) (NTP, 2021b)<sup>*a*</sup>

<sup>b</sup> Number of animals with neoplasm or lesion per number of animals necropsied.

<sup>c</sup> Number of litters with neoplasm-bearing animals per number of litters examined at site.

<sup>d</sup> Poly-3 estimated neoplasm incidence after adjustment for intercurrent mortality.

<sup>*e*</sup> Beneath the control incidence is the p value associated with the trend test. Beneath the exposed group incidence are the p values corresponding to pairwise comparisons between the control group and that exposed group. The Rao-Scott test adjusts the Poly-3 test (which accounts for differential mortality in animals that do not reach study termination) for within-litter correlation.

<sup>*f*</sup> Historical control incidence for all routes of 2-year studies (mean  $\pm$  standard deviation): 1/350 (0.29%  $\pm$  0.76%); range: 0–2%.

<sup>*g*</sup> Historical control incidence: 20/350 (5.71%  $\pm$  3.35%); range: 2–10%.

<sup>h</sup> Historical control incidence:  $2/350 (0.57\% \pm 1.51\%)$ ; range: 0-4%.

<sup>*i*</sup> Historical control incidence: 0/350

<sup>*j*</sup> Historical control incidence:  $23/350 (6.57\% \pm 3.41\%)$ ; range: 2-10%.

1885 1886

In the second study, male and female SD rats were fed diets containing 0, 300, 1,000, 3,000, or 10,000 1887 ppm DEHP for two-years (mean received doses: 17, 54, 170, 602 mg/kg-day for males and 17, 60, 177, 1888 646 mg/kg-day for females) (see Appendix B.1.2.7 for full study summary). Survival of male and 1889 female rats to study termination in all treatment groups was commensurate with or greater than that of 1890 1891 control rats, and no exposure-related clinical findings were observed in any treatment groups. Feed 1892 consumption by male and female rats was comparable to across treatment groups, with the exception of 21 percent lower feed consumption for high-dose males during study week one. At study termination, 1893 high-dose male and female rat body weight was approximately 16 and 22 percent lower than respective 1894 1895 controls, providing some indication of exceedance of the MTD for high dose animals. As can be seen 1896 from Table 4-10, treatment with DEHP caused a significant increase in incidence of uterine endometrial 1897 adenocarcinomas and combined uterine adenoma, adenocarcinoma, squamous cell carcinoma, and

adenocarcinomas and combined adenoma, adenocarcinoma, squamous cell carcinoma, and squamous cell papilloma in high-dose females was outside the range of NTP historical controls (see footnotes  $e^{-i}$ in Table 4-10). A significant positive trend in incidence of uterine squamous cell papilloma was also observed, however, pairwise comparisons to the control were not significant. Additionally, chronic uterine inflammation was observed in the 300, 1,000, and 10,000 ppm groups compared to controls, however, the effect was not dose related

- 1904 however, the effect was not dose-related.
- 1905
- 1906

# 1907Table 4-10. Incidence of Uterine Tumors in SD Rats Chronically Exposed to DEHP in the Diet for1908Two-years $(NTP, 2021b)^a$

| Tissue: Tumor Type   | 0 ppm     | 300 ppm   | 1000 ppm   | 3000 ppm   | 10,000 ppm  |
|--|-----------|-----------|------------|------------|-------------|
| Inflammation, Chronic <sup>b</sup>   | 2/50      | 9/50*     | 6/50*      | 8/50       | 8/49*       |
| Adenoma <sup>be</sup>  | 0/50      | 1/50      | 0/50       | 0/50       | 0/49        |
| Adenocarcinoma (overall rate) <sup>b</sup>   | 2/50 (4%) | 2/50 (4%) | 1/50 (2%)  | 4/50 (8%)  | 10/50 (20%) |
| Adenocarcinoma (adjusted rate) <sup>cf</sup>   | 4.7%      | 4.9%      | 2.4%       | 9%         | 23.8%       |
| Poly-3 test <sup>d</sup>   | p < 0.001 | p = 0.678 | p = 0.508N | p = 0.352  | p = 0.011   |
| Squamous cell carcinoma (includes multiple) <sup>g</sup>   | 0/50      | 1/50      | 0/50       | 2/50       | 1/49        |
| Squamous cell papilloma (includes multiple) <sup><i>h</i></sup>  | 0/50      | 0/50      | 0/50       | 0/50       | 2/49        |
| Adenoma, adenocarcinoma, squamous cell<br>carcinoma, squamous cell papilloma<br>(combined) (overall rate) $^{i}$ | 2/50 (4%) | 4/50 (8%) | 1/50 (2%)  | 6/50 (12%) | 13/50 (26%) |
| Adenoma, adenocarcinoma, squamous cell<br>carcinoma, squamous cell papilloma<br>(combined) (adjusted rate)       | 4.7%      | 9.7%      | 2.4%       | 13.4%      | 30.7%       |
| Poly-3 test <sup><math>d</math></sup>  | p < 0.001 | p = 0.315 | p = 0.508N | p = 0.145  | p < 0.001   |

\*Statistically significant at  $p \le 0.05$  by the Poly-3 test.

<sup>a</sup> Adapted from Table 28 in (NTP, 2021b).

<sup>b</sup> Number of animals with neoplasm or lesion per number of animals necropsied.

<sup>c</sup> Poly-3 estimated neoplasm incidence after adjustment for intercurrent mortality.

<sup>d</sup> Beneath the control incidence is the p value associated with the trend test. Beneath the exposed group incidence are the p values corresponding to pairwise comparisons between the control group and that exposed group. The Poly-3 test accounts for differential mortality in animals that do not reach study termination. A negative trend or a lower incidence in

accounts for differential mortality in animals that do not reach study termination. A negative trend or a lower incidence in an exposure group is indicated by N.

<sup>*e*</sup> Historical control incidence for all routes of 2-year studies (mean  $\pm$  standard deviation): 1/350 (0.29%  $\pm$  0.76%); range: 0–2%.

<sup>*f*</sup>Historical control incidence:  $20/350 (5.71\% \pm 3.35\%)$ ; range: 2-10%.

<sup>g</sup> Historical control incidence:  $2/350 (0.57\% \pm 1.51\%)$ ; range: 0-4%.

<sup>*h*</sup> Historical control incidence: 0/350

<sup>*i*</sup> Historical control incidence:  $23/350 (6.57\% \pm 3.41\%)$ ; range: 2-10%.

- 1911 In contrast to the findings of studies of SD rats, no significant increases in uterine tumors were observed
- 1912 in two chronic (two-year) dietary studies of female F344 rats at doses of up to 774 to 939 mg/kg-day
- 1913 (David et al., 2000b; David et al., 1999); two chronic (two-year) dietary studies of female B6C3F1 mice
- 1914 at doses of up to 1,325 to 1,458 mg/kg-day (<u>David et al., 2000a;</u> <u>David et al., 1999;</u> <u>NTP, 1982a</u>); 1
- 1915 inhalation study and 1 intraperitoneal injection study of female Syrian golden hamsters (Schmezer et al.,
- 1916 <u>1988</u>); or in 4 studies of various strains of female transgenic mice (<u>Mortensen et al., 2002</u>; <u>Eastin et al.</u>,
- 1917 <u>2001</u>; <u>Toyosawa et al., 2001</u>) (see Table 4-5 and Table 4-6 for additional study details).

1918

1919

## 4.3.1.2.1 Conclusions for Uterine Tumors

EPA did not identify any human epidemiologic studies that evaluated the association between exposure
to DEHP and uterine cancer (Section 4.1).

1923 Across available carcinogenicity studies of DEHP, there is some limited evidence for uterine tumors in 1924 female SD rats. In the chronic perinatal and post-weaning exposure study by NTP (2021b), a significant 1925 trend in increased incidence of uterus endometrium adenocarcinoma and combined uterus adenoma, 1926 adenocarcinoma, squamous cell carcinoma, or squamous cell papilloma was observed, however, pair-1927 wise comparisons to the control were not statistically significant, and NTP characterized the uterine 1928 tumors as an equivocal finding. Further, body weight gain was significantly lower in high-dose (772 1929 mg/kg-day) females throughout the study, and terminal body weight was 32 percent lower than that of 1930 the concurrent control group, indicating exceedance of the MTD for high-dose females. In a second 1931 study by NTP (2021b), treatment with DEHP caused a significant increase in incidence of uterine 1932 endometrial adenocarcinomas and combined uterine adenoma, adenocarcinoma, squamous cell 1933 carcinoma, and squamous cell papilloma in high-dose (646 mg/kg-day) female rats compared to 1934 concurrent controls. Further incidence of these tumor types in high-dose females was outside the range 1935 of NTP historical controls. However, as with the first NTP study, high-dose female body weight gain 1936 and terminal body weight was significantly reduced by 22 percent compared to concurrent controls, 1937 providing some indication of exceedance of the MTD in the high dose group. Additionally, and as 1938 discussed by NTP (2021b), at present the mechanism(s) underlying the observed uterine neoplasms in 1939 female SD rats is unknown, and further work is required to assess the MOA for this tumor type. 1940

1941 In contrast to the findings of studies of SD rats by NTP (2021b), no significant increases in uterine 1942 tumors were observed in two chronic (two-year) dietary studies of female F344 rats at doses of up to 774 1943 to 939 mg/kg-day (David et al., 2000b; David et al., 1999); two chronic (two-year) dietary studies of 1944 female B6C3F1 mice at doses of up to 1,325 to 1,458 mg/kg-day (David et al., 2000a; David et al., 1945 1999; NTP, 1982a); 1 inhalation study and 1 intraperitoneal injection study of female Syrian golden 1946 hamsters (Schmezer et al., 1988); or in 4 studies of various strains of female transgenic mice (Mortensen 1947 et al., 2002; Eastin et al., 2001; Toyosawa et al., 2001) (see Table 4-5 and Table 4-6 for additional study 1948 details). 1949

1950 Overall, EPA considers there to be slight evidence for DEHP-induced uterine tumors. This is based on 1951 the fact that uterine tumors have only been observed in studies of female SD rats, but not in studies of 1952 female F344 rats, female B6C3F1 mice, or various transgenic strains of female mice. Further, the uterine 1953 tumor response was equivocal in one of the two studies of SD rats, and in both studies of SD rats, 1954 uterine tumors were increased only at high-doses (646-772 mg/kg-day), which coincided with a 22 to 32 1955 percent decrease in terminal body weight indicating exceedance of the MTD. Given the observed 1956 inconsistencies across species and strains of rats, unknown MOA, and the fact that uterine tumors only 1957 occurred at high-doses that exceeded the MTD, EPA considers there to be too much scientific 1958 uncertainty to consider using data for uterine tumors to derive quantitative estimates of cancer risk for 1959 DEHP.

1960

1961

# 4.3.1.3 Mononuclear Cell Leukemia (MNCL)

1962There is some limited evidence for MNCL in F344 rats following chronic oral exposure to DEHP. David1963et al. (2000b; 1999) fed male and female F344 rats diets containing 0, 100, 500, 2,500, or 12,500 ppm

DEHP for two-years (equivalent to 6, 29, 147, 780 mg/kg-day for males; 7, 36, 182, 939 mg/kg-day for females). Increased incidence of MNCL was observed in male (but not female) rats in the 2,500 and 12,500 ppm dose groups compared to concurrent controls (Table 4-11). Further, incidence of MNCL in 2,500 and 12,500 ppm males was outside the range of historical control data from the same laboratory conducting the study (historical control incidence: 128/420 [30%] for males and 82/424 [19%] for females over a 5-year period for rats of the same strain, age and from the same supplier).

- 1970
- 1971

# Table 4-11. Incidence of MNCL in F344 Rats Administered DEHP Through the Diet for Two Years (David et al., 2000b; David et al., 1999)<sup>a</sup>

| Sex    | 0 ppm<br>(M/F: 0/0<br>mg/kg-day) | 100 ppm<br>(M/F: 6/7<br>mg/kg-day) | 500 ppm<br>(M/F: 29/36<br>mg/kg-day) | 2,500 ppm<br>(M/F:<br>147/182<br>mg/kg-day) | 12,500 ppm<br>(M/F: 780/939<br>mg/kg-day) |
|--------|----------------------------------|------------------------------------|--------------------------------------|---|---|
| Male   | 15/65 (23%)                      | 13/50 (26%)                        | 16/55 (27%)                          | 32/65* (49%)                                | 27/65* (42%)                              |
| Female | 14/65 (22%)                      | 17/50 (34%)                        | 11/55 (20%)                          | 16/65 (25%)                                 | 17/65 (26%)                               |

<sup>*a*</sup> Asterisk indicates statistically significant pairwise comparison to the control by Fisher exact test ( $P \le 0.05$ ) as determined by original study authors. Data from Table 5 of (David et al., 1999) and Tables 6 and 7 of (David et al., 2000b).

1974

1975
In contrast to the study by David et al. (2000b; 1999), increased incidence of MNCL was not observed in two other chronic (95–108 weeks) dietary studies of male F344 rats (Rao et al., 1990; Rao et al., 1978 1987) or in one other chronic (two-year) dietary study of male and female F344 rats at doses as high at 674 to 774 mg/kg-day DEHP (NTP, 1982a). Although the two dietary studies by Rao et al. are limited by a small sample size of 8 to 14 rate per dose groups, which may have limited the constituity of the

1979 674 to 774 mg/kg-day DEHP (<u>NTP, 1982a</u>). Although the two dietary studies by Rao et al. are limited
1980 by a small sample size of 8 to 14 rats per dose groups, which may have limited the sensitivity of the
1981 studies, the study by NTP (<u>1982a</u>) was well conducted and similar in design to the study by David et al.
1982 (*i.e.*, male and female F344 rats [50/sex/dose group] were fed diets containing 0, 6,000, or 12,000 ppm
1983 DEHP for 103 weeks). Therefore, even across studies of F344 rats, the evidence for increased incidence
1984 of MNCL following chronic dietary exposure to DEHP is inconsistent and limited to a single study of
1985 male (but not female) F344 rats.

1986

1987 In addition to the noted inconsistencies for MNCL across studies of F344 rats, MNCL was not observed 1988 in 3 chronic (95 to 159 weeks) dietary studies of male and female SD rats exposed to up to 678 to 772 1989 mg/kg-day DEHP (NTP, 2021b; Voss et al., 2005); 2 chronic (two-year) dietary studies of male and 1990 female B6C3F1 mice exposed to up to 1,325 to 1,821 mg/kg-day DEHP (David et al., 2000a; David et 1991 al., 1999; NTP, 1982a); 1 inhalation study and 1 intraperitoneal injection study of Syrian golden 1992 hamsters (Schmezer et al., 1988); or in 5 studies of various strains of transgenic mice (Ito et al., 2007a; 1993 Mortensen et al., 2002; Eastin et al., 2001; Toyosawa et al., 2001) (see Table 4-5 and Table 4-6 for 1994 additional study details).

1995

1996

# 4.3.1.3.1 Conclusions for MNCL

1997 There is some limited evidence for MNCL in F344 rats following chronic oral exposure to DEHP. In 1998 one study of male (but not female) F344 rats, the incidence of MNCL was significantly increased at 1999 doses of 147 and 780 mg/kg-day DEHP compared to concurrent controls and was outside the range of

2000 historical control incidence (David et al., 2000b; David et al., 1999). In contrast, MNCL was not 2001 observed in two other chronic (95-108 weeks) dietary studies of male F344 rats that were limited by 2002 small sample sizes (*i.e.*, included 8–14 rats/group) (Rao et al., 1990; Rao et al., 1987) or in one other 2003 well-conducted chronic (two-year) dietary study of male and female F344 rats at doses as high at 674 to 2004 774 mg/kg-day DEHP (NTP, 1982a). Additionally, MNCL was not observed in 3 chronic (104–159 2005 weeks) dietary studies of male and female SD rats exposed to up to 678 to 772 mg/kg-day DEHP (NTP, 2006 2021b; Voss et al., 2005); 2 chronic (two-year) dietary studies of male and female B6C3F1 mice 2007 exposed to up to 1,325 to 1,821 mg/kg-day DEHP (David et al., 2000a; David et al., 1999; NTP, 1982a); 2008 1 inhalation study and 1 intraperitoneal injection study of Syrian golden hamsters (Schmezer et al., 2009 1988); or in 5 studies of various strains of transgenic mice (Ito et al., 2007a; Mortensen et al., 2002; 2010 Eastin et al., 2001; Toyosawa et al., 2001). Further, there are significant scientific uncertainties related 2011 to the human relevance of MNCL in F344 rats (see Appendix C for a discussion of uncertainties). 2012 2013 In addition to the observed inconsistencies in MNCL across studies of DEHP, there is scientific

2014 uncertainty related to MNCL in F344 rats. As discussed further in Appendix C, MNCL is a 2015 spontaneously occurring neoplasm of the hematopoietic system that reduces the lifespan and is one of 2016 the most common tumor types occurring at a high background rate in the F344 strain of rat (also referred to as Fisher rat leukemia because it is so common) (Thomas et al., 2007). Historical control data from 2017 NTP have demonstrated an increase in the spontaneous background incidence of MNCL in untreated 2018 2019 male and female F344 rats from 7.9 and 2.1 percent in males and females, respectively, in 1971 to 52.5 2020 and 24.2 percent in males and females, respectively, from 1995 through 1998 (Thomas et al., 2007). 2021 Spontaneous incidence of MNCL in other strains of rat appear to be rare, and MNCL does not appear to 2022 occur naturally in mice (Thomas et al., 2007). The F344/N strain of rat was used in NTP two-year 2023 chronic and carcinogenicity bioassays for nearly 30 years (King-Herbert et al., 2010; King-Herbert and 2024 Thayer, 2006). However, in the early 2000s, NTP stopped using the F344/N strain of rat, in large part 2025 because of high background incidence of MNCL and testicular Levdig cell tumors that confounded 2026 bioassay interpretation. NTP subsequently replaced the F344 strain of rats with the Harlan SD strain 2027 (King-Herbert et al., 2010; King-Herbert and Thayer, 2006).

2029 Additional sources of uncertainty include lack of MOA information for induction of MNCL in F344 rats 2030 and uncertainty related to the human correlate to MNCL in F344 rats. Some researchers have suggested 2031 that based on the biological and functional features in the F344 rat, MNCL is analogous to large granular 2032 lymphocyte (LGL) in humans (Caldwell et al., 1999; Caldwell, 1999; Reynolds and Foon, 1984). There 2033 are two major human LGL leukemias, including CD3+ LGL leukemia and CD3- LGL leukemia with 2034 natural killer cell activity (reviewed in (Maronpot et al., 2016; Thomas et al., 2007)). Thomas et al. (2007) contend that MNCL in F344 rats shares some characteristics in common with aggressive natural 2035 2036 killer cell leukemia (ANKCL) in humans, and that ANKCL may be a human correlate. However, 2037 Maronpot et al. (2016) point out that ANKCL is extremely rare with less than 98 cases reported 2038 worldwide, and its etiology is related to infection with Epstein-Barr virus, not chemical exposure. This is 2039 in contrast to MNCL in F344 rats, which is a more common form of leukemia and is not associated with 2040 a viral etiology.

2041

2028

2042 Given the limitations and uncertainties regarding MNCL in F344 rats discussed above, during the July

2043 2024 peer review meeting of the DIDP and DINP human health hazard assessments, the SACC

2044 recommended that "the observation of an increased incidence of MNCL in a chronic bioassay

2045 employing the Fisher 344 rat should not be considered a factor in the determination of the cancer

2046 classification..." and "Most Committee members agreed that given the material presented in a

2047 retrospective review, MNCL and Leydig Cell Tumors, among other tumor responses in F344 rat

2048 *carcinogenicity studies lack relevance in predicting human carcinogenicity (Maronpot et al., 2016)*"
 2049 (U.S. EPA, 2024q). Consistent with the recommendations of the SACC, EPA is not further considering
 2050 MNCL as a factor in the determination of the cancer classification for DEHP.

2051

2052

# 4.3.1.4 Preliminary Cancer Classification for DEHP

2053 Under the *Guidelines for Carcinogen Risk Assessment* (U.S. EPA, 2005), EPA reviewed the weight of 2054 scientific evidence and has preliminarily concluded that DEHP is *Not Likely to be Carcinogenic to* 2055 *Humans* at doses below levels that do not result in PPAR $\alpha$  activation. This draft classification was based 2056 on the following weight of scientific evidence considerations:

- Evidence indicates that DEHP is not a direct acting mutagen or genotoxicant (Section 3.1).
- There is indeterminant evidence of any associations between DEHP exposure and subsequent cancer outcomes in human epidemiologic studies (Section 4.1.3).
- DEHP exposure resulted in treatment related liver tumors (adenomas and/or carcinomas combined) in male and female rats at doses greater than or equal to 147 mg/kg-day DEHP
   (David et al., 2000b; David et al., 1999) and male and female mice at doses greater than or equal to 99 mg/kg-day DEHP (David et al., 2000a; David et al., 1999).
- DEHP exposure resulted in treatment related PACTs in male rats at doses greater than or equal to 170 mg/kg-day (<u>NTP, 2021b</u>).
- DEHP exposure resulted in treatment related Leydig cell tumors in male rats at doses greater than or equal to 300 mg/kg-day (Voss et al., 2005).
- Available MOA data for liver tumors in mice and rats support a PPARα MOA (Section 4.3.1.1.1).
- Limited data are available that potentially indicate a role for other non-genotoxic, threshold
   MOAs, in the liver, including activation of other nuclear receptors (*e.g.*, CAR, PXR, AhR).
- Inferences from hypolipidemic drugs and other prototypical PPARα activators (*e.g.*, WY-14,643) provide evidence indicating that the tumor triad (*i.e.*, hepatocellular tumors, PACTs, and Leydig cell tumors) is a fingerprint of chronic PPARα activation in rats (Section 4.3.1.1.4). However, there is some scientific uncertainty, as not all PPARα activators induce the triad, which may be related to differences in potency for activating PAPRα. Regardless, some uncertainty remains that mechanisms other than PPARα activation may be involved in development of PACTs and Leydig cell tumors.
- 2079 As discussed in Section 4.3.1.2.1, there is slight evidence for DEHP-induced uterine tumors in 2080 female SD rats, but not in studies of F344 female rats, B6C3F1 mice, or various transgenic 2081 strains of female mice. Further, the uterine tumor response was equivocal in one of the two 2082 studies of SD rats, and in both studies of SD rats, uterine tumors were increased only at high-2083 doses (646–772 mg/kg-day), which coincided with a 22 to 32 percent decrease in terminal body weight indicating exceedance of the MTD. Given the observed inconsistencies across species 2084 2085 and strains of rats, unknown MOA, and fact that uterine tumors only occurred at high-doses that 2086 exceeded the MTD. EPA considers there to be too much scientific uncertainty to consider using data for uterine tumors to derive quantitative estimates of cancer risk for DEHP. 2087
- As discussed in Section 4.3.1.3.1, given the limitations and uncertainties regarding MNCL in
   F344 rats, <u>EPA is not considering MNCL as a factor in the determination of the cancer</u>

May 2025

- 2090classification for DEHP. This is consistent with the recommendations of the SACC (U.S. EPA,20912024q).
- 2092 Further, the draft non-cancer point of departure (POD) (NOAEL [no-observed-adverse-effect 2093 level]/LOAEL of 4.8/14 mg/kg-day) based on effects on the developing male reproductive system 2094 consistent with a disruption of androgen action and phthalate syndrome (see Draft Non-cancer Human 2095 Health Hazard Assessment for Diethylhexyl Phthalate (DEHP) (U.S. EPA, 2024f)) that was selected to 2096 characterize risk for acute, intermediate, and chronic exposures scenarios is expected to adequately 2097 account for all chronic toxicity, including carcinogenicity (assuming a threshold MOA), which could potentially result from exposure to DEHP. This draft conclusion is because the non-cancer POD 2098 2099 (NOAEL/LOAEL of 4.8/14 mg/kg-day) is less than the lowest identified thresholds (*i.e.*,
- NOAEL/LOAEL or BMDL values) for tumorigenesis in the liver, pancreas and testis, and is less than
   the lowest identified threshold for PPARα activation. Identified thresholds are as follows:
- 2102 PPARa activation in the Liver. EPA identified 27 studies that evaluated various biomarkers of 2103 PPARα activation (KE 1 in PPARα MOA) in the liver, including 18 studies of rats, 3 studies of 2104 mice, 3 studies of monkeys, 2 studies of hamsters, and 1 study of guinea pigs (Table\_Apx D-1). 2105 As can be seen from Table Apx D-1, the lowest identified NOAEL for PPAR $\alpha$  activation in the 2106 liver were 7.5 mg/kg-day for mice (Isenberg et al., 2000) and for 11 mg/kg-day for rats (Barber 2107 et al., 1987; BIBRA, 1985). These NOAELs are greater than the identified draft non-cancer POD 2108 (NOAEL/LOAEL of 4.8/14 mg/kg-day) based on effects on the developing male reproductive 2109 system. 2110
- 2111 EPA also identified a recent gene expression study conducted by NTP that evaluated biomarkers 2112 of PPARα activation in the liver and conducted BMD modeling of gene expression changes. Gwinn et al. (2020) conducted a transcriptomic dose-response study of DEHP in which male SD 2113 2114 rats were gavaged with 0, 8, 16, 31.25, 62.5, 125, 250, 500, or 1,000 mg/kg-day DEHP for five 2115 days. Animals were sacrificed twenty-four hours after the last exposure, and then gene expression changes in the liver and kidney were evaluated using high-throughput transcriptomics 2116 2117 with the rat Biospyder S1500+ platform. BMD modeling of transcriptional changes was 2118 performed using BMD Express 2.2 and a predefined analysis process that was previously peer-2119 reviewed (NTP, 2018). Transcriptional BMDs were determined based on a benchmark response 2120 of 1 control standard deviation (1SD). Table 4-12 summarizes transcriptional BMD<sub>1SD</sub> and BMDL<sub>1SD</sub> values in the liver for genes known to be regulated by PPAR $\alpha$ . The lowest BMDL<sub>1SD</sub> 2121 2122 was 8.6 mg/kg-day for enoyl-CoA hydratase and 3-hydroxyacyl CoA dehydrogenase (Ehhadh), 2123 which is above the non-cancer POD of 4.8 mg/kg-day.
- 2124 Hepatocellular adenoma and carcinoma (combined). The lowest identified NOAELs/LOAELs 2125 were 29/147 mg/kg-day in male F344 rats (David et al., 2000b; David et al., 1999) and 19/99 mg/kg-day in male B6C3F1 mice (David et al., 2000a; David et al., 1999). Notably, in the 2126 studies by David et al. biomarkers of PPARα activation (*i.e.*, palmitoyl CoA oxidase activity) 2127 2128 were significantly increased at doses equivalent to or less than those that resulted in 2129 tumorigenesis. These NOAELs are greater than the identified draft non-cancer POD 2130 (NOAEL/LOAEL of 4.8/14 mg/kg-day) based on effects on the developing male reproductive 2131 system.
- Pancreatic acinar cell adenoma and carcinoma (combined). The lowest NOAEL/LOAEL was 54/170 mg/kg-day in male SD rats exposed to DEHP in the feed for two-years (postweaning only exposure study) (NTP, 2021b). NTP also conducted benchmark dose (BMD) modeling of pancreatic acinar cell adenoma and carcinoma (combined) incidence data from the perinatal and

postweaning and postweaning only carcinogenicity studies of DEHP with male SD rats. The
lowest BMD and BMDL associated with a 10 percent tumor response were 31 mg/kg-day and 20
mg/kg-day, respectively, in male rats in the postweaning only exposure study of DEHP (see
Table 30, Table 31, and Appendix F in (NTP, 2021b)). This NOAEL and BMDL is greater than
the identified draft non-cancer POD (NOAEL/LOAEL of 4.8/14 mg/kg-day) based on effects on
the developing male reproductive system.

Leydig cell tumors. The lowest NOAEL/LOAEL for Leydig cell tumors in male SD rats was 95/300 mg/kg-day (Voss et al., 2005). This NOAEL is greater than the identified draft non-cancer POD (NOAEL/LOAEL of 4.8/14 mg/kg-day) based on effects on the developing male reproductive system.

2146

#### 2147

# Table 4-12. Summary of Transcriptional BMD and BMDL Values for Genes Regulated by PPARα in the Liver of Male SD Rats Gavaged with DEHP for Five Days (Gwinn et al., 2020)<sup>a</sup>

| Gene Name   | Gene<br>Symbol | Entrez<br>Gene ID | BMD <sub>1SD</sub><br>(mg/kg-day) | BMDL <sub>1SD</sub><br>(mg/kg-day) |
|---|----------------|-------------------|-----------------------------------|------------------------------------|
| Enoyl-CoA hydratase and 3-<br>hydroxyacyl CoA dehydrogenase   | Ehhadh         | 171142            | 11                                | 8.6                                |
| Cytochrome P450, family 4, subfamily a, polypeptide 1   | Cyp4a1         | 50549             | 12                                | 9.0                                |
| Acyl-CoA thioesterase 1   | Acot1          | 50559             | 13                                | 9.5                                |
| CD36 molecule   | Cd36           | 29184             | 30                                | 18                                 |
| Acyl-CoA oxidase 1  | Acox1          | 50681             | 44                                | 28                                 |
| Fatty acid binding protein 1  | fabp1          | 24360             | 77                                | 32                                 |
| Apolipoprotein A1   | Apoa1          | 25081             | 120                               | 58                                 |
| Catalase  | Cat            | 24248             | 124                               | 86                                 |
| Fibroblast growth factor 21   | Fgf21          | 170580            | 815                               | 614                                |
| <sup><i>a</i></sup> Rat S1500 <sup>+</sup> gene expression data and BMDs can be found in NTP's Chemical Effects in Biological Systems (CEBs) database (https://doi.org/10.22427/NTP-DATA-002-00058-0002-0000-7) |                |                   |                                   |                                    |

2150

2151

## 4.3.2 Butyl Benzyl Phthalate (BBP)

BBP has been evaluated for carcinogenicity by a number of authoritative and regulatory agencies. As 2152 summarized in Table 4-13, BBP has been classified by the U.S. EPA Integrated Risk Information 2153 2154 System (IRIS) program as Group C (possible human carcinogen) (U.S. EPA, 1988a); as Likely to be carcinogenic to humans by the U.S. EPA PPRTV (Provisional Peer-reviewed Toxicity Value) program 2155 (U.S. EPA, 2002): by IARC as Group 3 (not classifiable as to its carcinogenicity to humans) (IARC, 2156 1999); and was considered, but not listed by OEHHA under California's Proposition 65 for 2157 2158 carcinogenicity because it "has not been clearly shown to cause cancer" (OEHHA, 2013b). Further, BBP 2159 was not evaluated quantitatively for cancer risk in assessments by ECB (2007), ECHA (2017a, b), 2160 Australia NICNAS (2015a), Health Canada (ECCC/HC, 2020), and U.S. CPSC (2014). 2161

Page 71 of 180

- 2162 The PPRTV program evaluated BBP for carcinogenicity under EPA's 1999 draft *Guidelines for*
- 2163 Carcinogen Risk Assessment (U.S. EPA, 1999). Consistent with the guidelines available at the time of
- the assessment (U.S. EPA, 1999), BBP was assessed under an assumption of low-dose linearity.
- However, since the 2002 PPRTV assessment of BBP, the science has evolved, and EPA's current
- 2166 *Guidelines for Carcinogen Risk Assessment* (U.S. EPA, 2005) emphasize a data-first approach, rather
- than use of default options, stating:
- 2168 "Rather than viewing default options as the starting point from which departures may be
   2169 justified by new scientific information, these cancer <u>guidelines view a critical analysis of all</u>
   2170 <u>of the available information that is relevant to assessing the carcinogenic risk as the starting</u>
   2171 point from which a default option may be invoked if needed to address uncertainty or the
- 2172 *absence of critical information.*"
- 2173 Moreover, TSCA requires EPA to use the 'best available science', thus the cancer classification and risk 2174 assessment approach for BBP has been re-evaluated.
- 2175
- 2176

## 2177 Table 4-13. Summary of Cancer Classifications and Listings for BBP

| Agency   | Cancer Classification/ Listing  |  |  |
|--|---|--|--|
| U.S. EPA (IRIS) ( <u>1988a</u> )   | Group C (possible human carcinogen)   |  |  |
| IARC ( <u>1999</u> )   | Group 3 (not classifiable as to its carcinogenicity to humans)                                |  |  |
| U.S. EPA (PPRTV) ( <u>2002</u> )   | Likely to be carcinogenic to humans   |  |  |
| California OEHHA ( <u>2013b</u> )  | ) Not listed as a carcinogen under Proposition 65 (ha not been clearly shown to cause cancer) |  |  |
| IARC = International Agency for Research on Cancer; IRIS = Integrated Risk Information<br>System; OEHHA = Office of Environmental Health Hazard Assessment; PPRTV =<br>Provisional Peer-Reviewed Toxicity Values |   |  |  |

- BBP has been evaluated for carcinogenicity by NTP in six chronic oral exposure studies, including five studies of F344/N rats and one of B6C3F1 mice (NTP, 1997a, b, 1982b). Available studies of BBP are
- studies of 1544/1 lats and one of Boest 1 line (<u>NIT, 1997a</u>, <u>b</u>, <u>1982b</u>). Available studies of BBF are summarized in Table 4-14 and Appendix B.2. Across available studies, statistically significant increases
- in MNCL and PACTs have been observed in F344/N rats. Additionally, slight, but statistically non-
- significant, increases in urinary bladder papilloma and/or carcinoma have been observed in female
- 2185 F344/N rats. No tumors were observed in one study of male and female B6C3F1 mice (<u>NTP, 1997a</u>).
- 2186 Evidence for MNCL, PACTs, and urinary bladder tumors is discussed further in Sections 4.3.2.1,
- 4.3.2.2, and 4.3.2.3, respectively, while EPA's preliminary cancer classification for BBP is provided in
  Section 4.3.2.4.
- 2189
- 2190
- 2191
- 2192
- 2193
May 2025

#### 2194 Table 4-14. Summary of Available Carcinogenicity Studies of BBP in Rodents

| Brief Study Description   | Tumor Type(s) Observed   |
|---|--|
| Studies of Rat  | S  |
| Male and female F344/N rats (50/sex/dose) fed 0, 6,000, 12,000 ppm BBP for 103 weeks (equivalent to approximately 300 and 600 mg/kg-day) ( <u>NTP, 1982b</u> ) (see Appendix B.2.2.1 for further study details).  | - MNCL (females only) <sup><i>a</i></sup>  |
| Male F344/N rats (60/dose) fed 0, 3,000, 6,000, 12,000<br>ppm BBP and female F344/N rats (60/dose) fed 0, 6,000,<br>12,000, 24,000 ppm BBP for two years (equivalent to<br>120, 240, 500 mg/kg-day [males]; 300, 600, 1,200<br>mg/kg-day [females]) (NTP, 1997b) (see Appendix<br>B.2.2.2 for further study details).   | <ul> <li>PACTs (males only)</li> <li>Transitional epithelium papilloma in<br/>urinary bladder (females only; not<br/>statistically significant)</li> </ul> |
| <ul> <li><u>Study 1 (<i>Ad Libitum</i> and Weight-Matched Control Protocol)</u>: Male F344/N rats (60/sex/dose) fed 0 or 12,000 ppm BBP, while female F344/N rats fed 0 or 24,000 ppm BBP in feed that was available <i>ad libitum</i> for 104 weeks. Two control groups were included: rats fed <i>ad libitum</i> and weight-matched controls (diet restricted such that mean body weight matched the dose group) (NTP, 1997a) (see Appendix B.2.2.3 for further study details).</li> <li><u>Study 2 (Two-year Restricted Feed Protocol)</u>: Male and female F344/N rats (60/sex/dose) were diet restricted to limit the mean body weight of the control group to approximately 85% of controls fed <i>ad libitum</i> in study 1. BBP was administered at the same concentrations as in study 1 for 104 weeks (NTP, 1997a) (see Appendix B.2.2.4 for further study details).</li> <li><u>Study 3 (Lifetime Restricted Feed Protocol)</u>: Male and female F344/N rats (60/sex/dose) were diet restricted and administered BBP as described for studies 1 and 2 until survival fell to 20% (<i>i.e.</i>, 30 months for males, 32 months for females) (NTP, 1997a) (see Appendix B.2.2.5 for further study details).</li> </ul> | <ul> <li>PACTs (males only)</li> <li>Urinary bladder carcinomas/papilloma (females only; not statistically significant)</li> </ul>                         |
| Studies of Mic  | e  |
| Male and female B6C3F1 mice (50/sex/dose) fed 0,<br>6,000, 12,000 ppm BBP for 103 weeks (equivalent to<br>900 and 1,800 mg/kg-day) (NTP, 1982b) (see Appendix<br>B.2.1.1 for further study details).  | - None   |
| <sup><i>a</i></sup> As described further in Appendix B.2, male rats from this stud of high mortality rates that led study authors to terminate the stu  | y were not evaluated for carcinogenicity because<br>dy of male rats between study weeks 29 and 30  |

of high mortality rates that led study authors to terminate the study of male rats between study weeks 29 and 30

May 2025

### 4.3.2.1 Mononuclear Cell Leukemia (MNCL)

Statistically significant increases in the incidence of MNCL have been observed in one out of five
studies of F344/N rats chronically exposed to BBP in the diet for two-years. MNCL was not observed in
one study of male or female B6C3F1 mice treated with up to 1,800 mg/kg-day BBP for two-years (NTP,
<u>1982b</u>).

2200

2195

2201 NTP (1982b) report a statistically significant increase in the incidence of MNCL in female F344/N rats 2202 treated with 600 mg/kg-day BBP in the diet for two-years (Table Apx B-18). In this study, MNCL was 2203 observed in 18/50 (36%) high-dose (600 mg/kg-day) female rats, compared to 7/49 (14%) of controls. 2204 Incidence of MNCL in high-dose females was outside the range of historical control data for female 2205 F344/N rats with "all leukemias" from the laboratory conducting the study (observed in 77/399 [19%]; 2206 range 12–24%). As described further in Appendix B.2, male rats from this study were not evaluated for 2207 carcinogenicity because of high mortality rates that led study authors to terminate the study of male rats 2208 between study weeks 29 and 30.

2209

In contrast to the study by NTP (<u>1982b</u>), no increase in incidence of MNCL was observed in male

F344/N rats treated with up to 500 mg/kg-day BBP or female F344/N rats treated with up to 1,200 mg/kg-day BBP for two wears in a subsequent distant study by NTP (1007b) (Table Arr P 10)

mg/kg-day BBP for two-years in a subsequent dietary study by NTP (<u>1997b</u>) (Table\_Apx B-19).
 Notably, this study was similar in design and tested doses of BBP twice as high as those used in the first

2213 Notably, this study was similar in design and tested doses of BBF twice 2214 NTP study (*i.e.*, 1,200 vs. 600 mg/kg-day for female F344/N rats).

2215

2216 Clear treatment-related increases in MNCL were not observed in a series of three dietary-restriction 2217 studies of F344/N rats reported by NTP (1997a). In the first study (Ad Libitum and Weight-Matched 2218 Control Protocol; Appendix B.2.2.3), incidence of MNCL was comparable between ad libitum fed 2219 control rats and BBP treated male (500 mg/kg-day) and female (1200 mg/kg-day) F344/N rats following 2220 two-years of dietary exposure (MNCL reported in 60-62% of control and BBP-treated males and 2221 38-42% for females). In contrast, lower incidence of MNCL was observed in weight-matched controls 2222 of both sexes (15/50 [30%] for males; 13/50 [26%] for females) (Table Apx B-20). Further, incidence 2223 of MNCL in BBP-treated rats of both sexes was reported by NTP to be within the historical control 2224 ranges for leukemia (all types) in untreated F344/N rats. In the second dietary restriction study of BBP 2225 with F344/N rats (two-year restricted feed protocol; Appendix B.2.2.4), no statistically significant 2226 increase in MNCL was observed in male or female rats treated with 500 and 1,200 mg/kg-day BBP, 2227 respectively, compared to controls (incidence: 21/50 [42%] in control vs. 27/50 [54%] in BBP-treated 2228 males; 16/50 [32%] in control vs. 18/50 [36%] in BBP-treated females) (Table\_Apx B-21) (NTP, 2229 1997a). Similarly, in the lifetime restricted feed study of BBP with F344/N rats (Appendix B.2.2.5), no 2230 statistically significant increase in MNCL was observed in male or female rats treated with 500 and 2231 1,200 mg/kg-day BBP, respectively, compared to controls (incidence: 39/50 [78%] controls vs. 36/50 2232 [72%] BBP-treated males; 29/50 [58%] controls vs. 39/50 [78%] BBP-treated females) (Table Apx 2233 B-21) (NTP, 1997a).

2234

2235

# 4.3.2.1.1 Conclusions for MNCL

Increased incidence of MNCL was observed in one dietary study of female F344/N rats treated with 600
mg/kg-day BBP for two-years (incidence in control and 600 mg/kg-day group: 7/49 [14%], 18/50
[36%]) (NTP, 1982b). In this study, incidence of MNCL in females at 600 mg/kg-day was outside of the

range of NTP historical control data (observed in 77/399 female F344/N rats [19%]; range 12–24%). In

- 2240 contrast, treatment-related increases in MNCL were not observed in four other chronic dietary studies in
- which female F344/N rats dosed with up to 1,200 mg/kg-day BBP (a dose twice as high as the study in

which MNCL was observed), four chronic dietary studies of male F344/N rats dosed with up to 500 mg/kg-day BBP, or in male or female B6C3F1 mice treated with up to 1,800 mg/kg-day BBP for two years (NTP, 1997a, b, 1982b).

2245

2261

2246 In addition to the observed inconsistencies in MNCL across studies of BBP, there is scientific 2247 uncertainty related to MNCL in F344 rats. As discussed further in Appendix C, MNCL is a 2248 spontaneously occurring neoplasm of the hematopoietic system that reduces the lifespan and is one of 2249 the most common tumor types occurring at a high background rate in the F344 strain of rat (also referred 2250 to as Fisher rat leukemia because it is so common) (Thomas et al., 2007). Historical control data from NTP have demonstrated an increase in the spontaneous background incidence of MNCL in untreated 2251 2252 male and female F344 rats from 7.9 and 2.1 percent in males and females, respectively, in 1971 to 52.5 2253 and 24.2 percent in males and females, respectively, from 1995 through 1998 (Thomas et al., 2007). 2254 Spontaneous incidence of MNCL in other strains of rat appear to be rare and MNCL does not appear to 2255 occur naturally in mice (Thomas et al., 2007). The F344/N strain of rat was used in NTP two-year 2256 chronic and carcinogenicity bioassays for nearly 30 years (King-Herbert et al., 2010; King-Herbert and 2257 Thayer, 2006). However, in the early 2000s, NTP stopped using the F344/N strain of rat in large part 2258 because of high background incidence of MNCL and testicular Leydig cell tumors that confounded bioassay interpretation. NTP subsequently replaced the F344 strain of rats with the Harlan SD strain 2259 (King-Herbert et al., 2010; King-Herbert and Thayer, 2006). 2260

2262 Additional sources of uncertainty include lack of MOA information for induction of MNCL in F344 rats 2263 and uncertainty related to the human correlate to MNCL in F344 rats. Some researchers have suggested 2264 that based on the biological and functional features in the F344 rat, MNCL is analogous to LGL in 2265 humans (Caldwell et al., 1999; Caldwell, 1999; Reynolds and Foon, 1984). There are two major human LGL leukemias, including CD3+ LGL leukemia and CD3- LGL leukemia with natural killer cell activity 2266 2267 (reviewed in (Maronpot et al., 2016; Thomas et al., 2007)). Thomas et al. (2007) contend that MNCL in 2268 F344 rats shares some characteristics in common with ANKCL in humans, and that ANKCL may be a 2269 human correlate. However, Maronpot et al. (2016) point out that ANKCL is extremely rare with less 2270 than 98 cases reported worldwide, and its etiology is related to infection with Epstein-Barr virus, not 2271 chemical exposure. This is in contrast to MNCL in F344 rats, which is a more common form of 2272 leukemia and is not associated with a viral etiology. 2273

2274 Given the limitations and uncertainties regarding MNCL in F344 rats discussed above, during the July 2275 2024 peer review meeting of the DIDP and DINP human health hazard assessments, the SACC 2276 recommended that "the observation of an increased incidence of MNCL in a chronic bioassay 2277 employing the Fisher 344 rat should not be considered a factor in the determination of the cancer 2278 classification..." and "Most Committee members agreed that given the material presented in a 2279 retrospective review, MNCL and Levdig Cell Tumors, among other tumor responses in F344 rat 2280 carcinogenicity studies lack relevance in predicting human carcinogenicity (Maronpot et al., 2016)" (U.S. EPA, 2024q). Consistent with the recommendations of the SACC, EPA is not further considering 2281 2282 MNCL as a factor in the determination of the cancer classification for BBP.

2283

2284

#### 4.3.2.2 Pancreatic Acinar Cell Tumors (PACTs)

Statistically significant increases in the incidence of pancreatic acinar cell hyperplasia, adenomas, and combined adenomas and carcinomas have been observed in two out of five studies of F344/N rats chronically exposed to BBP in the diet. Adenomas and carcinomas represent a progression from preneoplastic pancreatic acinar cell hyperplasia, and these pre-neoplastic and neoplastic findings are

- 2289 discussed further below. In contrast to studies of F344/N rats, pancreatic acinar cell hyperplasia,
- adenomas, and carcinomas were not observed in the one study of male or female B6C3F1 mice treated with up to 1,800 mg/kg-day BBP for two-years (NTP, 1982b).

2292 2293 NTP (1997b) reports a statistically significant increase in the incidence of pancreatic acinar cell 2294 hyperplasia, adenomas, and combined adenomas and carcinomas in high-dose (500 mg/kg-day) male 2295 F344/N rats (Table 4-15). Notably, the increase in adenomas and carcinomas was outside the range of 2296 laboratory historical control data (see footnotes b-e in Table 4-15) and occurred at a dose that did not 2297 cause overt toxicity. That is, no effect on survival, clinical observations, or food consumption was 2298 observed in male rats treated with 500 mg/kg-day, although body weight was reduced 4 to 10 percent 2299 throughout most of the study. In contrast, treatment-related increases in pancreatic acinar cell 2300 hyperplasia were not observed in high-dose female rats exposed to up to 1,200 mg/kg-day BBP, 2301 although a marginal, statistically non-significant increase in pancreatic acinar cell adenomas was 2302 observed in 2 out of 50 high-dose (1,200 mg/kg-day) females (Table 4-15).

- 2303
- 2304

# Table 4-15. Incidence of Non-neoplastic and Neoplastic Findings in the Pancreas of F344/N Rats Fed Diets Containing BBP for Two-years (NTP, 1997b)<sup>a</sup>

|   | 0 ppm     | 3,000<br>ppm<br>(M/F:<br>120/NA<br>mg/kg-d) | 6,000<br>ppm<br>(M/F:<br>240/300<br>mg/kg-d) | 12,000 ppm<br>(M/F:<br>500/600<br>mg/kg-d) | 24,000<br>ppm<br>(M/F: NA/<br>1,200<br>mg/kg-d) |
|---|-----------|---|--|--|---|
|   | Male      | Rats  | -  | -  |   |
| Number Examined                                     | 50        | 49  | 50   | 50   | NA  |
| Pancreas, Acinus, Focal Hyperplasia                 | 4/50      | 7/49  | 9/50   | 12/50*                                     | NA  |
| Pancreas, Acinus, Adenoma <sup>b</sup>              | 3/50 (6%) | 2/49 (4%)                                   | 3/50 (6%)                                    | 10/50*<br>(20%)                            | NA  |
| Pancreas, Acinus, Carcinoma <sup>c</sup>            | 0/50      | 0/49  | 0/50   | 1/50 (2%)                                  | NA  |
| Pancreas, Acinus, Adenoma or Carcinoma <sup>d</sup> | 3/50 (6%) | 2/49 (4%)                                   | 3/50 (6%)                                    | 11/50*<br>(22%)                            | NA  |
|   | Female    | Rats  |  |  |   |
| Number Examined                                     | 50        | NA  | 50   | 50   | 50  |
| Pancreas, Acinus, Focal Hyperplasia                 | 1/50      | NA  | 4/50   | 2/50                                       | 0/50  |
| Pancreas, Acinus, Adenoma <sup>e</sup>              | 0/50      | NA  | 0/50   | 0/50                                       | 2/50 (4%)                                       |

NA = Not applicable (dose not tested for this sex)

Asterisk (\*) indicates significant difference ( $P \le 0.05$ ) from the control by the logistic regression test, as calculated by NTP.

<sup>a</sup> Incidence data from Tables 9 and 10 in (NTP, 1997b).

<sup>*b*</sup> Historical incidence for 2-year NTP feed studies with untreated controls (acinus, adenoma, males):  $19/1,191 (1.6\% \pm 2.4\%)$ ; range 0-10%.

<sup>c</sup> Historical incidence (acinus, carcinoma, males): 0/1,919 (0.0%)

<sup>*d*</sup> Historical incidence (acinus, adenoma or carcinoma, males): 19/1,191 (1.6%  $\pm 2.4\%$ ); range 0-10%.

 $^e$  Historical incidence (acinus, adenoma, females): 2/1,194 (0.2%  $\pm$  0.8%); range 0–4%

2308 2309 Similar to the results of NTP (1997b), statistically significant increases in incidence of pancreatic acinar 2310 cell hyperplasia, adenomas, and combined adenomas and carcinomas have been observed in one of three 2311 dietary-restriction studies of F344/N rats (NTP, 1997a). In the first study (ad libitum and weight-2312 matched controls protocol) of BBP, statistically significant increases in the incidences of pancreatic 2313 acinar cell hyperplasia, adenomas, and combined adenomas and carcinomas were observed in high-dose 2314 (500 mg/kg-day) male F344/N rats compared to *ad libitum* and weight-matched controls (Table 4-16). 2315 Notably, the increase in pancreatic tumors occurred at a dose that did not cause overt toxicity. Treatment 2316 of male rats with BBP had no effect on survival, clinical observations, or food consumption compared to 2317 the *ad libitum* controls, although body weight was reduced approximately eight percent in BBP-treated 2318 males throughout most of the study. Pancreatic acinar cell hyperplasia was not observed in high-dose 2319 female rats exposed to up to 1,200 mg/kg-day BBP, although a marginal, statistically non-significant 2320 increase in pancreatic acinar cell adenomas was observed in 2 out of 50 high-dose (1,200 mg/kg-day) 2321 females (Table 4-16). In contrast, no significant increase in pancreatic acinar cell hyperplasia, 2322 adenomas, or carcinomas were observed in male or female rats treated with up to 500 and 1.200 mg/kg-2323 day BBP, respectively, in the two-year and lifetime restricted feed studies of BBP with F344/N rats 2324 (Table Apx B-21).

Finally, no pancreatic acinar cell hyperplasia, adenomas, and carcinomas were observed in another twoyear dietary study of female F344/N rats dosed with up to 600 mg/kg-day BBP (Table\_Apx B-18) (NTP, 1982b). However, the carcinogenicity of BBP was not assessed in male rats in this study due to high rates of mortality, which resulted in all male rats being sacrificed between study weeks 29 and 30.

2330

2325

| Lesion/ Tumor Type                     | Ad<br>Libitum-<br>Fed<br>Control | Weight-<br>Matched<br>Control | 12,000 ppm<br>(males) or 24,000<br>ppm (females) |  |  |
|--|----------------------------------|-------------------------------|--|--|--|
| Male Rats                              |                                  |                               |  |  |  |
| Number Examined                        | 50                               | 50                            | 50   |  |  |
| Pancreas, Acinus, Focal<br>Hyperplasia | 4/50                             | 2/50                          | 12/50  |  |  |
| Pancreas, Acinus, Adenoma              | 3/50 (6%)                        | 0/50                          | 10/50* (20%)                                     |  |  |
| Pancreas, Acinus, Carcinoma            | 0/50                             | 1/50 (2%)                     | 1/50 (2%)  |  |  |
| Pancreas, Adenoma or Carcinoma         | 3/50 (6%)                        | 1/50 (2%)                     | 11/50* (22%)                                     |  |  |
|  | Female Rats                      |                               |  |  |  |
| Number Examined                        | 50                               | 49                            | 50   |  |  |
| Pancreas, Acinus, Focal<br>Hyperplasia | 1/50 (2%)                        | 0/49                          | 0/50   |  |  |
| Pancreas, Acinus, Adenoma              | 0/50                             | 0/49                          | 2/50 (4%)  |  |  |

# Table 4-16. Incidence of Neoplasms and Non-neoplastic Lesions in the Pancreas in F344/N Rats (Ad Libitum and Weight-Matched Controls Protocols) (NTP, 1997a)<sup>a</sup>

| Lesion/ Tumor Type  | Ad<br>Libitum-<br>Fed<br>Control | Weight-<br>Matched<br>Control | 12,000 ppm<br>(males) or 24,000<br>ppm (females) |
|---|----------------------------------|-------------------------------|--|
| Asterisk (*) indicates significant differen as calculated by NTP. | ce (P≤0.05) from                 | the control by the            | logistic regression test,                        |

<sup>a</sup> Incidence data from Tables 3, 4, B1a, and B3a of (NTP, 1997a).

<sup>b</sup> Incidence of MNCL significantly increased compared to weight-matched, but not *ad libitum* fed controls.

2334

2335

### 4.3.2.2.1 Conclusions for Pancreatic Acinar Cell Tumors

2336 Pancreatic adenomas and carcinomas (PACTs) represent a progression from pre-neoplastic pancreatic 2337 acinar cell hyperplasia. EPA did not identify any human epidemiologic studies that evaluated the 2338 association between exposure to BBP and pancreatic cancer (Section 4.1). As discussed in Section 2339 4.3.2.1.1, clear treatment-related increases in pancreatic acinar cell hyperplasia and PACTs have been 2340 observed in two out of four studies of male F344/N rats treated with 500 mg/kg-day BBP (NTP, 1997a, 2341 b). Marginal (statistically non-significant) increases in PACTs were also observed in high-dose (*i.e.*, 1,200 mg/kg-day BBP) female F344/N rats in two studies (NTP, 1997a, b). Studies in which significant 2342 2343 increases in hyperplasia and PACTs were observed utilized ad libitum feeding protocols and reported no 2344 evidence of overt toxicity in male F344/N rats. In contrast, no statistically significant treatment-related 2345 increases in acinar cell hyperplasia or PACTs were noted in male or female F344/N rats treated with 500 2346 and 1,200 mg/kg-day BBP, respectively, in two-year and lifetime restricted feed studies (NTP, 1997a). 2347 However, as discussed by NTP (1997a), feed and/or caloric restriction is known to suppress 2348 tumorigenesis in the pancreas (Roebuck et al., 1993; Roebuck et al., 1981) and thus dietary restriction 2349 may have prevented BBP-induced PACTs in the two-year and lifetime dietary restriction studies.

2350

2351 As discussed previously in Section 4.3.1.1.2, a MOA for induction of PACTs has been proposed, which involves activation of PPARα in the liver (KE 1), leading to deceased bile acid flow (KE2a) and/or bile 2352 2353 acid composition (KE 2b) in the liver leading to increased release of CCK into the bloodstream, which 2354 can lead to cholestasis (KE 3), and increased plasma CCK levels (KE 4), which in turn are believed to 2355 cause increased pancreatic acinar cell proliferation and PACT formation (apical outcome). Evidence 2356 supporting this MOA for BBP is limited, although BBP has been shown to activate PPAR $\alpha$  in the liver. 2357 For example, Barber et al. (1987) demonstrate that BBP and other phthalates (*i.e.*, DEHP, DINP, DIDP, 2358 DBP) can all activate PPARa in the livers of male F344 rats exposed to each phthalate in the diet for 21 days based on induction of hepatic palmitoyl CoA oxidase activity. Although BBP (and DBP) was found 2359 to be a much weaker PPARα activator than DEHP, DINP, and DIDP. Similarly, Bility et al. (2004) 2360 2361 demonstrated that monoester metabolites of BBP and other phthalates (*i.e.*, DEHP, DINP, DIDP, DBP) 2362 can activate both mouse and human PPAR $\alpha$  in vitro; however, for all five phthalates, human PPAR $\alpha$ 2363 was less sensitive to activation compared to mouse PPAR $\alpha$ . Notably, similar trends in potency for 2364 PPARα activation were observed *in vitro* with mouse PPARα as were observed *in vivo* with studies of 2365 rats, with BBP (and DBP) being a considerably weaker PPARα activator than DIDP, DINP and DEHP. 2366 As discussed previously in Section 4.3.1.1, PPAR $\alpha$  activators have been shown to cause the tumor triad 2367 in rats (i.e., liver tumors, PACTs, and Leydig cell tumors), however, no evidence of liver tumors or 2368 Leydig cell tumors were observed following chronic exposure to BBP in any study. The lack of liver 2369 tumors following chronic exposure to BBP may be related to the fact that BBP is a relatively weak 2370 PPAR $\alpha$  activator compared to other phthalates such as DEHP (Section 4.3.1.1), DINP (Section 4.3.4),

and DIDP (Section 4.3.5) that have been shown to cause liver tumors. Additionally, BBP has only been
evaluated for carcinogenicity in F344/N rats, which have a high spontaneous background rate of
testicular Leydig cell tumors (ranging from 86–87%), which reduces the ability of this strain of rat to
detect treatment-related increases in this tumor type (see Appendix C for further discussion).

- Overall, EPA considers there to be evidence to support the conclusion that chronic oral exposure to BBP induces PACTs in F344/N rats.
- 2378

2379

2375

## 4.3.2.3 Urinary Bladder Papillomas and/or Carcinomas

Statistically significant increases in the incidence of transitional epithelium hyperplasia and statistically non-significant increases in papilloma and/or carcinoma in the urinary bladder have been observed in four out of five studies of female F344/N rats chronically exposed to BBP in the diet. Papillomas and carcinomas represent a progression from pre-neoplastic transitional epithelium hyperplasia, and these pre-neoplastic and neoplastic findings are discussed further below. In contrast to studies of F344/N rats, transitional epithelium hyperplasia, papilloma and carcinoma were not observed in the one study of male or female B6C3F1 mice treated with up to 1,800 mg/kg-day BBP for two-years (NTP, 1982b).

2387 2388 NTP (1997b) report a statistically significant increase in the incidence of transitional epithelium 2389 hyperplasia in high-dose (1,200 mg/kg-day) female (but not male) F344/N rats exposed to BBP for two-2390 years (Table 4-17). Transitional epithelium papillomas were observed in two high-dose females and one 2391 control female. Although the increase in papilloma was not statistically significant, the incidence in 2392 high-dose females was outside the range of NTP historical control data (historical incidence of 2393 transitional epithelium papilloma: 4/1,182 [0.3%  $\pm$  0.8%]; range 0–2%). No transitional epithelium papillomas were observed in male F344/N rats at any dose, nor were any transitional epithelium 2394 2395 carcinomas observed at any dose for either sex. Although there was no evidence of overt toxicity or 2396 exceedance of the MTD for male rats at any dose, there was evidence of exceedance of the MTD for 2397 high-dose (1,200 mg/kg-day) female rats, as demonstrated by a 7 to 27 percent reduction in body weight 2398 throughout the duration of the study and a 27 percent reduction in body weight compared to controls at 2399 study termination.

2400 2401

# Table 4-17. Incidence of Non-neoplastic and Neoplastic Findings in the Urinary Bladder in F344/N Rats Fed Diets Containing BBP for Two-years (*Ad Libitum* and Weight-Matched Controls Protocol) (NTP, 1997b)<sup>a</sup>

|   | 0 ppm     | 3,000<br>ppm | 6,000<br>ppm | 12,000 ppm | 24,000<br>ppm |
|---|-----------|--------------|--------------|------------|---------------|
| Male Rats                                       |           |              |              |            |               |
| Number Examined microscopically                 | 50        | 49           | 50           | 50         | NA            |
| Hyperplasia, Transitional Epithelium            | 0/50      | 0/49         | 0/50         | 2/50       | NA            |
| Papilloma, Transitional Epithelium              | 0/50      | 0/49         | 0/50         | 0/50       | NA            |
|   | Female    | e Rats       |              |            |               |
| Number Examined microscopically                 | 50        | NA           | 50           | 50         | 50            |
| Hyperplasia, Transitional Epithelium            | 4/50      | NA           | 0/50         | 1/50       | 10/50*        |
| Papilloma, Transitional Epithelium <sup>b</sup> | 1/50 (2%) | NA           | 0/50         | 0/50       | 2/50 (4%)     |

|  | May   | 2025                                 |                               |                     |                 |
|--|---|--------------------------------------|-------------------------------|---------------------|-----------------|
|  | 0 ppm   | 3,000<br>ppm                         | 6,000<br>ppm                  | 12,000 ppm          | 24,000<br>ppm   |
| NA = Not Applicable (dose not tested for t<br>Asterisk (*) indicates significant difference<br>calculated by NTP.<br><sup>a</sup> Incidence data from Tables 10 and A5 in a<br><sup>b</sup> Historical incidence (transitional epitheliu | his sex)<br>e (P≤0.05) fro<br>(NTP, 1997b)<br>um papilloma` | m the contro<br>).<br>): 4/1,182 (0. | 1 by the logi $3\% \pm 0.8\%$ | stic regression tes | st, as          |
|  |   |                                      |                               | <u> </u>            |                 |
|  |   |                                      |                               |                     |                 |
| Similar to the results of NTP ( $\underline{1997b}$ ), st  | tatistically si   | gnificant in                         | creases in i                  | ncidence of tran    | sitional        |
| epithelium hyperplasia have been observ  | ved in three  | dietary-resti                        | riction stud                  | ies of female (bu   | ut not male)    |
| F344/N rats dosed with 1,200 mg/kg-da  | y for 24- to 3  | 32-months (                          | Table 4-18                    | and Table 4-19      | ) ( <u>NTP,</u> |
| <u>1997a</u> ). Increases in transitional epitheli   | um hyperpla   | asia were ac                         | companied                     | by slight, statist  | tically non-    |
| significant increases in transitional epith  | nelium papill   | loma and/or                          | carcinoma                     | (Table 4-18 and     | d Table 4-19).  |
| In the first study (ad libitum and weight  | -matched co   | ntrols proto                         | col) of BB                    | P, transitional ep  | oithelium       |
| papilloma was observed in two high-dos   | se (1,200 mg  | g/kg-day) fei                        | males and o                   | one control fema    | ıle. No         |
| papilloma was observed in male rats trea   | ated with 50  | 0 mg/kg-da                           | y BBP (Tal                    | ble 4-18). In the   | second study    |
| (two-year restricted feed protocol), trans   | sitional epith  | elium papil                          | loma was o                    | observed in two     | high-dose       |
| (1,200 mg/kg-day) female rats and one l  | high-dose (5  | 00 mg/kg-d                           | ay) male ra                   | t (Table 4-19). I   | Finally, in the |
| third study (lifetime restricted feed proto  | ocol), transit  | ional epithe                         | lium papill                   | oma and carcino     | oma were        |
| each observed in 1 male rat dosed with 5   | 500 mg/kg-d   | lay BBP, wł                          | nile transiti                 | onal epithelium     | papilloma and   |
| carcinoma were observed in 2 and 4 hig   | h-dose (1,20  | 0 mg/kg-da                           | y) female 1                   | ats, respectively   | , with          |
| papilloma noted in 1 of 49 control femal   | les (Table 4-   | 19). Howev                           | ver, across                   | the three dietary   | restriction     |
| studies, the slight increases in incidence   | of transition   | nal epitheliu                        | m papillon                    | ha and/or carcino   | oma was not     |
| statistically significant in any case. Acro  | oss all three s   | studies, ther                        | e was no e                    | vidence of overt    | toxicity to     |
| suggest the MTD was exceeded for male  | es, while ter   | minal body                           | weight for                    | females dosed w     | vith 1,200      |
| mg/kg-day BBP was reduced by 23 to 2   | 9 percent, 1n   | dicating exc                         | ceedance of                   | t the MTD.          |                 |
|  |   |                                      |                               |                     |                 |

Finally, no transitional epithelium hyperplasia or papilloma or carcinoma of the urinary bladder were
observed in a two-year dietary study of female F344/N rats dosed with up to 600 mg/kg-day BBP (NTP,
1982b). However, the highest achieved dose in this study was lower than the dose (*i.e.*, 1,200 mg/kgday) shown to cause transitional epithelium hyperplasia or papilloma and carcinoma in other chronic
dietary studies of female F344/N rats.

2431 2432

# Table 4-18. Incidence of Non-neoplastic and Neoplastic Findings in the Urinary Bladder in F344/N Rats Fed Diets Containing BBP for Two-years (NTP, 1997a)<sup>a</sup>

|                 | Lesion/ Tumor Type                      | Ad<br>Libitum-<br>Fed<br>Control | Weight-<br>Matched<br>Control | 12,000 ppm<br>(males) or<br>24,000 ppm<br>(females) |
|-----------------|---|----------------------------------|-------------------------------|---|
|                 | Male Rats                               |                                  |                               |   |
| Number Examined |   | 50                               | 50                            | 50  |
| Urinary Bladder | Hyperplasia, Transitional<br>Epithelium | 0/50                             | 0/50                          | 2/50  |

|   | Lesion/ Tumor Type                      | Ad<br>Libitum-<br>Fed<br>Control | Weight-<br>Matched<br>Control | 12,000 ppm<br>(males) or<br>24,000 ppm<br>(females) |  |  |
|---|---|----------------------------------|-------------------------------|---|--|--|
|   | Papilloma, Transitional                 | 0/50                             | 0/50                          | 0/50  |  |  |
|   | Epithelium                              |                                  |                               |   |  |  |
|   | Female Rats                             |                                  |                               |   |  |  |
| Urinary Bladder   | Hyperplasia, Transitional<br>Epithelium | 4/50 (8%)                        | 0/50                          | 10/50 (20%)   |  |  |
|   | Papilloma, Transitional<br>Epithelium   | 1/50 (2%)                        | 0/50                          | 2/50 (4%)   |  |  |
| Asterisk (*) indicates significant difference (P $\leq$ 0.05) from the control by the logistic regression test, as calculated by NTP.<br><sup><i>a</i></sup> Incidence data from Tables 3, 4, B1a, and B3a of (NTP, 1997a). |   |                                  |                               |   |  |  |

2435

2436

# Table 4-19. Incidence of Non-neoplastic and Neoplastic Findings in the Urinary Bladder in F344/N Rats Treated with BBP (2-year Restricted Feed and Lifetime Restricted Feed Protocols) (NTP,

2439 **<u>1997a</u>**)<sup>*a*</sup>

|   | 2-Year Restricted Feed<br>Protocol |  | Lifetime Restricted Feed<br>Protocol |  |  |  |
|---|------------------------------------|--|--------------------------------------|--|--|--|
|   | 0 ppm                              | 12,000 ppm<br>(males) or 24,000<br>ppm (females) | 0 ppm                                | 12,000 ppm<br>(males) or 24,000<br>ppm (females) |  |  |
| Male Rats   |                                    |  |                                      |  |  |  |
| Number Examined   | 50                                 | 50   | 50                                   | 50   |  |  |
| Hyperplasia   | 1/50                               | 2/50   | 0/50                                 | 1/50   |  |  |
| Papilloma   | 0/50                               | 1/50 (2%)  | 0/50                                 | 1/50 (2%)  |  |  |
| Carcinomas  | 0/50                               | 0/50   | 0/50                                 | 1/50 (2%)  |  |  |
|   | F                                  | emale Rats                                       |                                      |  |  |  |
| Number Examined   | 50                                 | 50   | 49                                   | 50   |  |  |
| Hyperplasia   | 0/50                               | 14/50*   | 0/49                                 | 16/50*   |  |  |
| Papilloma   | 0/50                               | 2/50 (4%)  | 1/49 (2%)                            | 2/50 (4%)  |  |  |
| Carcinomas  | 0/50                               | 0/50   | 0/49                                 | 4/50 (8%)  |  |  |
| Papilloma or Carcinoma<br>(combined)  | 0/50                               | 2/50 (4%)  | 1/49 (2%)                            | 6/50 (12%)                                       |  |  |
| Asterisk (*) indicates significant difference (P $\leq$ 0.05) from the control by the logistic regression test, as calculated by NTP.<br><sup><i>a</i></sup> Incidence date from Table 7 of (NTP, 1997a). |                                    |  |                                      |  |  |  |

2440

May 2025

#### 4.3.2.3.1 Conclusions for Urinary Bladder Tumors

2442 Transitional epithelium papilloma and carcinoma in the urinary bladder represent a progression of preneoplastic transitional epithelium hyperplasia. As discussed in Section 4.3.2.3, consistent increases in 2443 pre-neoplastic transitional epithelium hyperplasia of the urinary bladder have been observed in four out 2444 2445 of five studies of female F344/N rats chronically exposed to 1,200 mg/kg-day BBP (NTP, 1997a, b). In 2446 a 5th study, no transitional epithelium hyperplasia was observed in female F344/N rats, however, the 2447 highest achieved dose (i.e., 600 mg/kg-day) in this study was lower than in the studies where 2448 hyperplasia was observed (NTP, 1982b). In contrast to studies of female F344/N rats, no significant 2449 increases in transitional epithelium hyperplasia have been observed in male F344/N rats treated with up 2450 to 500 mg/kg-day BBP in four studies (NTP, 1997a, b) or in male or female B6C3F1 mice treated with 2451 up to 1,800 mg/kg-day BBP for two-years (NTP, 1982b).

- 2452 2453 Coinciding with increased incidence of transitional epithelium hyperplasia, marginal, statistically non-2454 significant increases in urinary bladder papilloma and/or carcinoma were also observed in female 2455 F344/N rats treated with high doses of 1,200 mg/kg-day BBP in four studies (NTP, 1997a, b). It is 2456 plausible that the significantly increased incidences of hyperplasia noted in the urinary bladder at 1,200 mg/kg-day are proliferative responses that can lead to the marginal (not significant) increases in urinary 2457 2458 bladder tumors. However, there are several sources of uncertainty associated with this tumor type. First, 2459 the marginal increase in urinary bladder tumors did not reach statistical significance in any study. 2460 Second, the MOA for induction of urinary bladder tumors in F344/N female rats is unknown. Lack of 2461 MOA information makes it difficult to determine human relevancy, and EPA did not identify any human epidemiologic studies that examined the link between BBP (or any other phthalate) exposure and 2462 2463 incidence of bladder cancer. Third, this tumor type has only been observed in one sex of one species 2464 (*i.e.*, female F344/N rats). Significant increases in this tumor type were not observed in male or female 2465 B6C3F1 mice treated with up to 1,800 mg/kg-day BBP or male F344/N rats in four studies. However, 2466 the highest achieved dose in studies of male rats was 500 mg/kg-day, which is considerably lower than 2467 the dose (*i.e.*, 1,200 mg/kg-day) linked with marginal increases in urinary bladder tumors in female F344/N rats, which may explain the sex difference in tumor response. Finally, the marginal (not 2468 2469 significant) increase in urinary bladder tumors in female rats only occurred at a very high-dose (i.e., 2470 1,200 mg/kg-day). In all four studies in which marginal increases in urinary bladder tumors were 2471 observed, there was evidence that the MTD was exceeded, as demonstrated by a 23 to 29 percent 2472 reduction in mean terminal body weight for female rats. Overall, EPA considers there to be too much 2473 scientific uncertainty to consider using data for urinary bladder tumors to derive quantitative estimates 2474 of cancer risk. 2475
- 2476

2441

#### 4.3.2.4 Preliminary Cancer Classification for BBP

2477 Under the Guidelines for Carcinogen Risk Assessment (U.S. EPA, 2005), EPA reviewed the weight of 2478 evidence for the carcinogenicity of BBP and has preliminarily concluded that there is Suggestive 2479 Evidence of Carcinogenic Potential of BBP in rodents. According to the Guidelines for Carcinogen Risk 2480 Assessment (U.S. EPA, 2005), a descriptor of Suggestive Evidence of Carcinogenic Potential is 2481 appropriate "when the weight of evidence is suggestive of carcinogenicity; a concern for potential 2482 carcinogenic effects in humans is raised, but the data are judged not sufficient for a stronger conclusion. 2483 This descriptor covers a spectrum of evidence associated with varying levels of concern for 2484 carcinogenicity, ranging from a positive cancer result in the only study on an agent to a single positive 2485 cancer result in an extensive database that includes negative studies in other species." EPA's 2486 determination is based on evidence of pancreatic acinar cell adenomas in male and female F344 rats. 2487 Further weight of scientific evidence considerations supporting EPA's determination are listed below.

According to the *Guidelines for Carcinogen Risk Assessment* (U.S. EPA, 2005), when there is *Suggestive Evidence* "the Agency generally would not attempt a dose-response assessment, as the nature of the data generally would not support one." Consistently, EPA is not conducting a dose-response assessment for BBP or quantitatively evaluating BBP for carcinogenic risk to humans.

- BBP is not likely to be genotoxic or mutagenic (Section 3.2).
- Significant treatment-related increases in incidence of pancreatic acinar cell hyperplasia, adenomas, and combined adenomas and carcinomas have been observed in two chronic dietary studies of male F344/N rats treated with 500 mg/kg-day BBP for two-years (NTP, 1997a, b). The MTD was not exceeded for high-dose males in either study (*i.e.*, no treatment-related effects on survival, food consumption, or clinical findings; mean body weight was within 10% that of concurrent controls both studies).
- Marginal (statistically non-significant) increases in incidence of pancreatic acinar cell adenomas were observed in two chronic dietary studies of female F344/N rats treated with 1,200 mg/kg-day BBP for two-years (NTP, 1997a, b).
- In two-year and lifetime dietary restriction studies of BBP, no significant increase in acinar cell hyperplasia or pancreatic tumors was observed in male or female F344/N rats exposed to 500 and 1,200 mg/kg-day BBP, respectively (NTP, 1997b). However, as discussed in Section 4.3.2.2, dietary restriction can suppress tumorigenesis in the pancreas (Roebuck et al., 1993; Roebuck et al., 1981) and therefore dietary restriction may have suppressed BBP-induced tumorigenesis in the pancreas in these studies.
- PACTs have also been observed in male rats following chronic oral exposure to toxicologically similar phthalates, including DEHP (Section 4.3.1.1) and DBP (Section 4.3.3.1). Occurrence of PACTs following chronic exposure to these phthalates increases EPA's confidence in the conclusion that chronic oral exposure to BBP causes PACTs in rats.
- No carcinogenic activity of BBP was observed in the one study of male and female B6C3F1
   mice treated with up to 1,800 mg/kg-day BBP for two-years (NTP, 1982b).
- Similar conclusions have been reached by other authoritative and regulatory agencies. California OEHHA concluded BBP has not been clearly shown to cause cancer and did not list BBP as a carcinogen under Proposition 65 (OEHHA, 2013b). IARC (1999) classified BBP as Group 3 (no classifiable as to its carcinogenicity to humans). The U.S. EPA IRIS program previously classified BBP as Group C (possible human carcinogen) (U.S. EPA, 1988a).
- Herein, EPA has preliminarily concluded that there is *Suggestive Evidence of Carcinogenic Potential* of
  BBP in rodents and consistent with the Guidelines for Carcinogen Risk Assessment (U.S. EPA, 2005)
  did not conduct a dose-response assessment or evaluate BBP quantitatively for cancer risk.
- 2522

2523

### 4.3.3 Dibutyl Phthalate (DBP)

DBP has previously been classified as Group D (not classifiable as to human carcinogenicity) by U.S.
EPA (1987). Similarly, assessments of DBP by other regulatory and authoritative bodies have concluded
that there is insufficient information to evaluate DBP for carcinogenicity, primarily due to the lack of
two-year rodent cancer bioassays at the time of the assessments (NICNAS, 2013; U.S. CPSC, 2010b;
ECB, 2004). However, EPA identified two new cancer bioassays of DBP (NTP, 2021a), which have not
been considered in previous assessments of DBP but are considered by EPA herein.

2530

2531 DBP has been evaluated for carcinogenicity in two chronic oral exposure studies (1 in rats, 1 in mice) 2532 published in an NTP Technical Report (NTP, 2021a), and an additional three studies of rats have 2533 evaluated DBP for carcinogenicity in the male reproductive system following gestational only exposure 2534 to DBP (Barlow et al., 2004; Mylchreest et al., 2000; Mylchreest et al., 1999). Available studies of DBP 2535 are summarized in Table 4-20. Across studies, there is some limited evidence for the carcinogenicity of 2536 DBP, which is based on marginal increases in the incidence of pancreatic acinar cell adenomas and 2537 statistically non-significant incidence of Leydig cell adenomas following chronic and/or gestational 2538 exposure to DBP. Evidence for acinar cell adenomas and Leydig cell adenomas following exposure to 2539 DBP is discussed further in Sections 4.3.3.1 and 4.3.3.2, respectively, while EPA's preliminary cancer 2540 classification for DBP is provided in Section 4.3.3.3.

2541 2542

## 2543 **Table 4-20. Summary of Available Rodent Carcinogenicity Studies of DBP**

| Brief Study Description  | Tumor Type(s) Observed   |
|--|--|
| Studies of Rats  | -  |
| Time-mated female SD rats (50/sex/dose) fed 0, 300, 1,000, 3,000, 10,000 ppm DBP during gestation and lactation.<br>Postweaning F1 offspring fed diets with same concentrations of DBP for 2 years (equivalent to 16, 54, 152, 510 mg/kg-day [males]; 17, 57, 169, 600 mg/kg-day [females]) ( <u>NTP, 2021a</u> ). | <ul> <li>Pancreatic acinus adenomas (males<br/>only; equivocal response)</li> <li>Leydig cell adenoma (not<br/>statistically significant)</li> </ul> |
| Timed pregnant SD rats (9–10 per dose) gavaged with 0,<br>100, 250, 500 mg/kg-day DBP from GD 12–21 and allowed<br>to deliver litters naturally. Testes of male F1 offspring<br>examined microscopically on PND 100 or PND 105<br>(Mylchreest et al., 1999).   | - Leydig cell adenoma (not<br>statistically significant)   |
| Timed pregnant SD rats (19–20 per dose, 11 in high-dose<br>group) gavaged with 0, 0.5, 5, 50, 100, 500 mg/kg-day DBP<br>from GD 12–21 and allowed to deliver litters naturally.<br>Testes of male F1 offspring examined microscopically on<br>PND 110 (Mylchreest et al., 2000).                                   | - Leydig cell adenoma (not<br>statistically significant)   |
| Time-mated pregnant CRL:CD(SD)BR rats gavaged with 0, 100, 500 mg/kg-day DBP from GD 12–21 and allowed to deliver litters naturally. Male F1 offspring were necropsied at PND 180, PND 370, or PND 540 (Barlow et al., 2004).  | - Leydig cell adenoma (not<br>statistically significant)   |
| Studies of Mice  |  |
| Adult male and female B6C3F1/N mice (50/sex/dose) fed 0, 1,000, 3,000, 10,000 ppm DBP for 2-years (equivalent to 112, 347, 1,306 mg/kg-day [males]; 105, 329, 1,393 mg/kg-day [females]) ( <u>NTP, 2021a</u> ).  | - None   |

2544

May 2025

### 4.3.3.1 Pancreatic Acinar Cell Adenomas

2546 Pancreatic acinar cell adenomas have been observed in one chronic dietary study of SD rats (NTP, 2021a). Time-mated (F<sub>0</sub>) SD rats were fed diets containing 0, 300, 1000, 3000, or 10,000 ppm DBP 2547 2548 starting on GD 6 (45-47 dams/dose) continuously throughout gestation and lactation. Dams were 2549 allowed to deliver litters naturally, and on PND4, litters were culled to eight pups per litter (4 per sex). 2550 At weaning on PND 21, 25 litters per dose group were selected, and 2 males and 2 females were 2551 selected and fed diets containing the same respective DBP concentrations for two years. Treatment with 2552 DBP had no effect on pregnancy status, maternal survival, gestation length, number of dams that 2553 littered, or maternal body weight and weight gain during gestation. During lactation, mean body weights were reduced less than six percent in dams of the high-dose group. Mean received doses of DBP in units 2554 2555 of mg/kg-day during gestation, lactation, and the main two-year study are shown in Table 4-21. In the 2556 two-year rat study, no exposure-related effects on survival or clinical observations were reported, however, terminal body weight was reduced by 3.5 and 10.6 percent for high-dose males and females, 2557 2558 respectively.

2559 2560

2545

# Table 4-21. Mean Received Doses (mg/kg-day) for Male and Female SD Rats Exposed to DBP Through the Diet (NTP, 2021a)

| Study Phase                           | 0 ppm | 300 ppm | 1000 ppm | 3000 ppm | 10,000 ppm |
|---------------------------------------|-------|---------|----------|----------|------------|
| F <sub>0</sub> Dams on GD 6–21        | 0     | 22      | 72       | 214      | 740        |
| F <sub>0</sub> Dams on PND 1–14       | 0     | 47      | 155      | 466      | 1,514      |
| F1 Males (2-year study)               | 0     | 16      | 54       | 152      | 510        |
| F <sub>1</sub> Females (2-year study) | 0     | 17      | 57       | 169      | 600        |

2563

2564 2565 No treatment-related neoplastic lesions were observed in female rats at any dose. In males, there was a statistically significant dose-related trend in increased pancreatic acinus adenomas (Table 4-22). The 2566 2567 incidence of acinus adenomas was slightly higher in the 10,000 ppm group compared to concurrent controls (overall incidence: 4/49 (8%) in control vs. 10/49 (20%) in 10,000 ppm group), however, the 2568 2569 pairwise comparison to the concurrent control was not statistically significant. Two acinus carcinomas 2570 were observed in control males (2/49) but were not observed in any males treated with DBP. The 2571 incidence of acinus adenomas in the 10,000 ppm group was within NTP historical control range (0-28%) for studies of SD rats on the same diet. Time to first occurrence of acinus adenomas was 2572 2573 unaffected by treatment with DBP (first observed in control and 10,000 ppm males on study days 676 2574 and 684, respectively). The incidence of acinus hyperplasia was unaffected by treatment with DBP (Table 4-22). Under the conditions of the study, NTP concluded there was "equivocal evidence of 2575 carcinogenic activity of di-n-butyl phthalate (DBP) in male Hsd:Sprague Dawley<sup>®</sup> SD<sup>®</sup> rats based on 2576 marginal increases in the incidence of pancreatic acinus adenomas" and "no evidence of carcinogenic 2577 activity of DBP in female Hsd:Sprague Dawley<sup>®</sup> SD<sup>®</sup> rats at exposure concentrations of 300, 1,000, 2578 2579 3,000, or 10,000 ppm." 2580

In contrast to the study of SD rats (<u>NTP, 2021a</u>), exposure to DBP did not induce pancreatic tumors (or any other neoplastic findings) in male and female B6C3F1/N mice administered up to 1,306 to 1,393 mg/kg-day DBP through the diet for two-years (NTP, 2021a). Under the conditions of the study, NTP

#### May 2025

concluded that there was "no evidence of carcinogenic activity of DBP in male or female B6C3F1/N
 mice…"

2586

# Table 4-22. Incidence of Neoplastic and Non-neoplastic Lesions of the Pancreas in Male Rats in the Perinatal and Two-year Feed Study of DBP (<u>NTP, 2021a</u>)<sup>a</sup>

|   | 0 ppm                              | 300 ppm    | 1,000 ppm  | 3,000<br>ppm  | 10,000<br>ppm   |
|---|------------------------------------|------------|------------|---------------|-----------------|
| N (# animals with tissue examined microscopically)          | 49                                 | 50         | 50         | 50            | 49              |
| Acinus, hyperplasia   | 19 <sup>b</sup> (2.3) <sup>c</sup> | 21 (2.1)   | 18 (2.1)   | 23 (2.0)      | 18 (2.1)        |
| Acinus, Adenoma, Multiple                                   | 2                                  | 1          | 0          | 0             | 2               |
| Acinus, Adenoma (Includes Multiple) <sup>d</sup>            |                                    |            |            |               |                 |
| Overall Rate <sup>e</sup>                                   | 4/49 (8%)                          | 4/50 (8%)  | 3/50 (6%)  | 1/50 (2%)     | 10/49<br>(20%)  |
| Rate per litters <sup>f</sup>                               | 4/25 (16%)                         | 4/25 (16%) | 3/25 (12%) | 1/25 (4%)     | 9/25 (36%)      |
| Adjusted rate <sup>g</sup>                                  | 9.7%                               | 8.9%       | 6.8%       | 2.3%          | 24.1%           |
| Terminal rate <sup>h</sup>                                  | 2/27 (7%)                          | 3/38 (8%)  | 3/31 (10%) | 1/34 (3%)     | 8/33 (24%)      |
| First incidence (days)                                      | 676                                | 565        | 729 (T)    | 729 (T)       | 684             |
| Rao-Scott-adjusted Poly-3 test <sup>i</sup>                 | p = 0.010                          | p = 0.595N | p = 0.472N | p =<br>0.192N | p = 0.094       |
| Acinus, Carcinoma <sup><i>i</i></sup>                       | 2                                  | 0          | 0          | 0             | 0               |
| Acinus, Adenoma or Carcinoma (Combined) <sup><i>j</i></sup> |                                    |            |            |               |                 |
| Overall rate  | 6/49 (12%)                         | 4/50 (8%)  | 3/50 (6%)  | 1/50 (2%)     | 10/49 (20%<br>) |
| Rate per litters  | 6/25 (24%)                         | 4/25 (16%) | 3/25 (12%) | 1/25 (4%)     | 9/25 (36%)      |
| Adjusted rate   | 14.3%                              | 8.9%       | 6.8%       | 2.3%          | 24.1%           |
| Terminal rate   | 2/27 (7%)                          | 3/38 (8%)  | 3/31 (10%) | 1/34 (3%)     | 8/33 (24%)      |
| First incidence (days)                                      | 611                                | 565        | 729 (T)    | 729 (T)       | 684             |
| Rao-Scott-adjusted Poly-3 test                              | p = 0.024                          | p = 0.349N | p = 0.243N | p =<br>0.072N | p = 0.217       |

May 2025

|   | 0 ppm            | 300 ppm          | 1,000 ppm       | 3,000<br>ppm        | 10,000<br>ppm |
|---|------------------|------------------|-----------------|---------------------|---------------|
| (T) = terminal euthanasia.  | •                |                  |                 |                     |               |
| <sup>a</sup> Adapted from Table 13 in (NTP, 2021a)                    |                  |                  |                 |                     |               |
| <sup>b</sup> Number of animals with lesion                            |                  |                  |                 |                     |               |
| <sup>c</sup> Average severity grade of lesions in affected animals    | s in parenthese  | s: 1 = minimal   | 2 = mild, 3 =   | moderate, 4 =       | marked.       |
| <sup>d</sup> Historical control incidence for all routes of 2-year s  | studies (mean ±  | standard devi    | ation): 60/488  | $(11.58\% \pm 9.2)$ | 25%); range:  |
| 0%-28%.   |                  |                  | ,               |                     |               |
| <sup>e</sup> Number of animals with neoplasm per number of an         | imals necropsi   | ed.              |                 |                     |               |
| <sup>f</sup> Number of litters with tumor-bearing animals per num     | mber of litters  | examined at ar   | natomical site. |                     |               |
| <sup>g</sup> Poly-3-estimated neoplasm incidence after adjustme       | nt for intercurr | ent mortality.   |                 |                     |               |
| <sup>h</sup> Observed incidence at study termination.                 |                  | •                |                 |                     |               |
| <sup>i</sup> Beneath the control incidence is the p value associate   | ed with the trea | nd test. Beneatl | h the exposed g | group incidend      | ces are the p |
| values corresponding to pairwise comparisons betwee                   | en the control g | roup and that e  | exposed group.  | The Rao-Sco         | ott test      |
| adjusts the Poly-3 test, which accounts for differential              | l mortality in a | nimals that do   | not reach study | v termination,      | for within-   |
| litter correlation. A negative trend or a lower incidence             | e in an exposu   | re group is ind  | icated by N.    |                     |               |
| <i>i</i> Historical control incidence: $1/488 (0.80\% \pm 1.420\%)$ . | range: 0–4%.     |                  |                 |                     |               |
| $^{\circ}$ Thistorical control incluence: 4/488 (0.8% ± 1.42%),       |                  |                  |                 |                     |               |

2591 2592 Pancreatic adenomas and carcinomas (PACTs) represent a progression from pre-neoplastic pancreatic 2593 acinar cell hyperplasia. EPA did not identify any human epidemiologic studies that evaluated the 2594 association between exposure to DBP and pancreatic cancer (Section 4.1). Pancreatic acinar cell 2595 adenomas have been observed in one chronic dietary study of DBP with a male SD rats at doses that did 2596 not result in overt toxicity (NTP, 2021a). Treatment with DBP caused a significant trend in increased 2597 incidence of pancreatic acinar cell adenomas in male SD rats fed diets containing DBP for two-years; 2598 however, pairwise comparisons to concurrent controls were not statistically significant (incidence of 2599 adenomas in control and 10,000 ppm (equivalent to 510 mg/kg-day) groups: 4/49 [8%], 10/49 [20%]). 2600 Incidence of pancreatic acinar cell adenoma in high-dose males was within NTP historical control range (0-28%), and treatment with DBP did not reduce the time to onset of pancreatic tumors in high-dose 2601 2602 male rats (days to first incidence: 676 vs. 684). Further, treatment with DBP did not increase the 2603 incidence of pancreatic acinar cell hyperplasia, which is a preneoplastic lesion that precedes tumorigenesis in the pancreas. Overall, NTP concluded there was "equivocal evidence" of carcinogenic 2604 activity of DBP in male rats based on the observed pancreatic acinar cell tumors. 2605 2606

2590

2607 As discussed previously in Section 4.3.1.1.2, a MOA for induction of PACTs has been proposed, which involves activation of PPAR $\alpha$  in the liver (KE 1), leading to deceased bile acid bile acid flow (KE2a) 2608 2609 and/or bile acid composition (KE 2b) in the liver leading to increased release of CCK into the 2610 bloodstream, which can lead to cholestasis (KE 3) and increased plasma CCK levels (KE 4), which in 2611 turn are believed to cause increased pancreatic acinar cell proliferation and PACT formation (apical 2612 outcome). Evidence supporting this MOA for DBP is limited, although DBP has been shown to activate 2613 PPAR $\alpha$  in the liver. For example, Barber et al. (1987) demonstrate that DBP and other phthalates (*i.e.*, DEHP, DINP, DIDP, BBP) can all activate PPARa in the livers of male F344 rats exposed to each 2614 2615 phthalate in the diet for 21 days based on induction of hepatic palmitoyl CoA oxidase activity. Although 2616 DBP (and BBP) was found to be a much weaker PPAR $\alpha$  activator than DEHP, DINP, and DIDP. Similarly, Bility et al. (2004) demonstrated that monoester metabolites of DBP and other phthalates (*i.e.*, 2617 2618 DEHP, DINP, DIDP, BBP) can activate both mouse and human PPAR $\alpha$  in vitro, however, for all five 2619 phthalates, human PPARa was less sensitive to activation compared to mouse PPARa. Notably, similar trends in potency for PPARa activation were observed in vitro with mouse PPARa as were observed in 2620

- vivo with studies of rats, with DBP (and BBP) being a considerable weaker PPARa activator than DIDP, 2621 2622 DINP and DEHP. As discussed previously in Section 4.3.1.1, PPARa activators have been shown to 2623 cause the tumor triad in rats (*i.e.*, liver tumors, PACTs, and Leydig cell tumors), however, no evidence 2624 of liver tumors were observed following chronic exposure to DBP in mice or rats. The lack of liver 2625 tumors following chronic exposure to DBP may be related to the fact that DBP is a relatively weak 2626 PPARα activator compared to other phthalates such as DEHP (Section 4.3.1.1), DINP (Section 4.3.4), 2627 and DIDP (Section 4.3.5) that have been shown to cause liver tumors. As will be discussed further in 2628 Section 4.3.3.2, there is some limited evidence for a carcinogenic response in the testis.
- 2629
- In contrast to the study of male SD rats, no PACTs (or any other neoplastic findings) were observed in the one study of male and female B6C3F1 mice exposed to up to 1,306 to 1,393 mg/kg-day DBP through the diet for two-years or in female SD rats exposed to up to 600 mg/kg-day DBP through the diet for two-years (NTP, 2021a).
- 2634
- Overall, <u>EPA considers there to be limited evidence to support the conclusion that chronic oral exposure</u>
   to <u>DBP causes pancreatic tumors in rats</u>. However, read-across from other toxicologically similar
   phthalates (*i.e.*, DEHP [Section 4.3.1.1] and BBP [Section 4.3.2.1.1]) that induce pancreatic acinar cell
   tumors in rats provides additional evidence to support the conclusion that phthalates, including DBP, can
   cause pancreatic acinar cell adenomas in rats.
- 2640

2641

# 4.3.3.2 Leydig Cell Adenomas

Levdig cell hyperplasia and/or adenomas have been reported in four studies of SD rats (NTP, 2021a; 2642 Barlow et al., 2004; Mylchreest et al., 2000; Mylchreest et al., 1999), but not in male B6C3F1 mice 2643 2644 dosed with up to 1,306 mg/kg-day DBP for two years (NTP, 2021a). In the first study of SD rats by NTP 2645 (2021a), which was described previously in Section 4.3.3.1, a statistically significant increase in diffuse 2646 and focal interstitial cell hyperplasia was observed in high-dose males (10,000 ppm in the diet, 2647 equivalent to 510 mg/kg-day) compared to concurrent control males (incidence of focal hyperplasia: 2648 11/50 [22%] for high-dose males vs. 1/49 [2%] for controls; Table 4-23). A slight, statistically non-2649 significant, increase in interstitial cell tumors was also observed, but without clear relationship to dose 2650 (Table 4-23).

2651 2652

#### Table 4-23. Incidence of Interstitial Cell Hyperplasia and Adenomas of the Testis in Male Rats in the Perinatal and Two-year Feed Study of DBP (NTP, 2021a)<sup>*a*</sup>

|  | 0 ppm     | 300 ppm    | 1,000 ppm         | 3,000 ppm  | 10,000 ppm |
|--|-----------|------------|-------------------|------------|------------|
| N (# animals with tissue examined microscopically)                                   | 49        | 50         | 50                | 47         | 50         |
| Interstitial cell, hyperplasia, diffuse, bilateral <sup>d</sup>                      | 0**       | 0          | $1^{b} (2.0)^{c}$ | 0          | 9** (2.2)  |
| Interstitial cell, hyperplasia, focal (includes bilateral) <sup><math>d</math></sup> | 1* (3.0)  | 7* (1.6)   | 5 (1.2)           | 3 (1.7)    | 11** (1.5) |
| Testis, Adenoma  |           |            |                   |            |            |
| Overall Rate <sup><i>e</i></sup>   | 2/49 (4%) | 5/50 (10%) | 1/50 (2%)         | 4/47 (9%)  | 5/50 (10%) |
| Rate per litters <sup>f</sup>  | 2/25 (8%) | 5/25 (20%) | 1/25 (4%)         | 4/25 (16%) | 4/25 (16%) |
| Adjusted rate <sup>g</sup>   | 4.9%      | 11.2%      | 2.2%              | 9.8%       | 12%        |
| Terminal rate <sup>h</sup>   | 2/27 (7%) | 4/38 (11%) | 0/31 (0%)         | 4/32 (13%) | 4/33 (12%) |

|   | 1.14) =0=0 |         |           |           |            |
|---|------------|---------|-----------|-----------|------------|
|   | 0 ppm      | 300 ppm | 1,000 ppm | 3,000 ppm | 10,000 ppm |
| First incidence (days)                      | 729 (T)    | 685     | 621       | 729 (T)   | 595        |
| Rao-Scott-adjusted Poly-3 test <sup>i</sup> | P=0.214    | P=0.287 | P=0.492N  | P=0.362   | P=0.255    |

(T) = terminal euthanasia.

<sup>a</sup> Adapted from Table 15 in (NTP, 2021a) and PO8: Statistical Analysis of Primary Tumors

<sup>b</sup> Number of animals with lesion

<sup>c</sup> Average severity grade of lesions in affected animals in parentheses: 1 = minimal, 2 = mild, 3 = moderate, 4 = marked.

<sup>d</sup> Statistical significance for the vehicle control group indicates a significant trend test, while statistical significance for an

exposure group indicates a significant pairwise test compared to the vehicle control group. \* indicates statistical significance (p  $\leq 0.05$ ) from the vehicle control group by the Rao-Scott adjusted Poly-3 test; \*\*p  $\leq 0.01$ .

<sup>e</sup> Number of animals with neoplasm per number of animals necropsied.

<sup>*f*</sup>Number of litters with tumor-bearing animals per number of litters examined at anatomical site.

<sup>g</sup> Poly-3-estimated neoplasm incidence after adjustment for intercurrent mortality.

<sup>*h*</sup> Observed incidence at study termination.

<sup>*i*</sup> Beneath the control incidence is the p value associated with the trend test. Beneath the exposed group incidences are the p values corresponding to pairwise comparisons between the control group and that exposed group. The Rao-Scott test adjusts the Poly-3 test, which accounts for differential mortality in animals that do not reach study termination, for within-litter correlation. A negative trend or a lower incidence in an exposure group is indicated by N.

#### 2655 2656

Three studies, all of similar design and conducted by the same laboratory (*i.e.*, the Chemical Industry Institute of Toxicology, CIIT), have reported slight, statistically non-significant increases in Leydig cell adenomas following gestation only exposure to DBP in SD rats. In the first study, Mylchreest et al. (1999) gavaged timed pregnant SD rats (9–10/dose) from GD 12 to 21 with 0, 100, 250, and 500 mg/kgday DBP and allowed to deliver litters naturally. Testes of F1 males were then examined microscopically at sexual maturity on PND 100 to PND 105. Low, statistically non-significant increases in Leydig cell hyperplasia and adenomas were observed in high-dose F1 males (Table 4-24).

2665

# Table 4-24. Incidence of Interstitial Cell Hyperplasia and Adenomas in Rats Exposed Gestationally to DBP (Mylchreest et al., 1999)<sup>a</sup>

| Lesion  | 0<br>mg/kg-day      | 100<br>mg/kg-day | 250<br>mg/kg-day | 500<br>mg/kg-day |
|---|---------------------|------------------|------------------|------------------|
| No. of animals (litters)                            | 51 (10)             | 51 (9)           | 55 (10)          | 45 (9)           |
| Leydig cell hyperplasia                             | 0 (0)               | 0 (0)            | 1 (1)            | 5 (2)            |
| Leydig cell adenomas                                | 0 (0)               | 0 (0)            | 0 (0)            | 2(1)             |
| <sup><i>a</i></sup> Adapted from Table 3 in (Mylchi | reest et al., 1999) | ).               |                  |                  |

2668

2669

In a second study, Mylchreest et al. (2000) gavaged timed pregnant SD rats (19–20/dose, 11 in the highdose group) from GD 12 through 21 with 0, 0.5, 5, 50, 100, and 500 mg/kg-day DBP and allowed to
deliver litters naturally. Testes of F1 males were then examined microscopically at sexual maturity on
PND 110. Similar to the first study, low, statistically non-significant increases in Leydig cell hyperplasia
and adenomas were observed in F1 males at 500 mg/kg-day (Table 4-25).

2675

2676

| Lesion   | 0<br>mg/kg-<br>day | 0.5<br>mg/kg-<br>day | 5<br>mg/kg-<br>day | 50<br>mg/kg-<br>day | 100<br>mg/kg-<br>day | 500<br>mg/kg-<br>day |  |  |
|--|--------------------|----------------------|--------------------|---------------------|----------------------|----------------------|--|--|
| No. of animals (litters)   | 134 (19)           | 118 (20)             | 103 (19)           | 120 (20)            | 140 (20)             | 58 (11)              |  |  |
| Interstitial cell hyperplasia  | 0 (0)              | 0 (0)                | 0 (0)              | 0 (0)               | 0 (0)                | 14 (5)               |  |  |
| Interstitial cell adenomas   | 0 (0)              | 0 (0)                | 0 (0)              | 0 (0)               | 0 (0)                | 1 (1)                |  |  |
| <sup><i>a</i></sup> Adapted from Table 3 in (Mylchreest et al., 2000). |                    |                      |                    |                     |                      |                      |  |  |

# Table 4-25. Incidence of Interstitial Cell Hyperplasia and Adenomas in Rats Exposed Gestationally to DBP (Mylchreest et al., 2000)<sup>a</sup>

2679

2680 In a third study, Barlow et al. (2004) gavaged time-mated pregnant CRL:CD(SD)BR rats with 0, 100, 2681 2682 and 500 mg/kg-day DBP on GDs 12 through 21 and then allowed dams to deliver litters naturally. Male F1 offspring were weaned on PND 21, and then necropsied at PND 180, PND 370, or PND 540. Low, 2683 statistically non-significant incidence of Leydig cell hyperplasia was observed in F1 males, including 2684 unilateral hyperplasia in three control males on PND 540, one to two low-dose males on PND 370 or 2685 2686 PND 540, and one to three high-dose males on PND 180, PND 370, or PND 540, while bilateral 2687 hyperplasia was observed in three low-dose males on PND 540 (Table 4-26). Similarly, low, statistically 2688 non-significant increases in Leydig cell adenomas (unilateral) were observed, including in one control male on PND 370 and PND 540, and one low-dose F1 male on PND 540. No adenomas were observed 2689 2690 in high-dose F1 males at any timepoint.

2691

2692

Table 4-26. Incidence of Interstitial Cell Hyperplasia and Adenomas in Rats Exposed
 Gestationally to DBP (Barlow et al., 2004)<sup>a</sup>

|  | PND 180    |            | PND 370    |            |           | PND 540    |        |         |           |
|--|------------|------------|------------|------------|-----------|------------|--------|---------|-----------|
| DBP (mg/kg-day)                          | 0          | 100        | 500        | 0          | 100       | 500        | 0      | 100     | 500       |
| No. of animals<br>(litters)              | 60<br>(10) | 65<br>(10) | 45<br>(11) | 61<br>(10) | 61<br>(9) | 74<br>(11) | 45 (9) | 49 (10) | 35<br>(8) |
| LC hyperplasia<br>(unilateral)           | 0 (0)      | 0 (0)      | 1 (1)      | 0 (0)      | 1 (1)     | 3 (3)      | 3 (1)  | 2 (1)   | 1 (1)     |
| LC hyperplasia<br>(bilateral)            | 0 (0)      | 0 (0)      | 0 (0)      | 0 (0)      | 0 (0)     | 0 (0)      | 0 (0)  | 3 (2)   | 0 (0)     |
| LC adenoma<br>(unilateral)               | 0 (0)      | 0 (0)      | 0 (0)      | 1 (1)      | 0 (0)     | 0 (0)      | 1 (1)  | 1 (1)   | 0 (0)     |
| LC adenoma<br>(bilateral)                | 0 (0)      | 0 (0)      | 0 (0)      | 0 (0)      | 0 (0)     | 0 (0)      | 0 (0)  | 1 (1)   | 0 (0)     |
| <sup><i>a</i></sup> Adapted from Table 2 | in (Barl   | ow et al., | 2004).     |            |           |            |        |         |           |

2695

May 2025

#### 4.3.3.2.1 Conclusions on Leydig Cell Tumors

2697 EPA did not identify any human epidemiologic studies that evaluated the association between exposure to DBP and testicular cancer (Section 4.1). As discussed above in Section 4.3.3.2, significant treatment-2698 2699 related increases in Leydig cell hyperplasia has been observed in one study of SD rats dosed with 510 2700 mg/kg-day DBP for two-years (NTP, 2021a), while three studies of SD rats reported slight, but statistically non-significant, increases in Leydig cell hyperplasia (Barlow et al., 2004; Mylchreest et al., 2701 2702 2000; Mylchreest et al., 1999). As discussed by NTP (2021a), Leydig cell hyperplasia is suggestive of 2703 systemic hormonal disturbance, including disturbance of the hypothalamus-pituitary-gonad axis. More 2704 specifically, decreased systemic testosterone levels may cause a decrease in negative feedback of 2705 testosterone on the hypothalamus-pituitary-gonad axis, which in turn can lead to increased luteinizing 2706 hormone that might have resulted in a stimulatory response of the Leydig cells (NTP, 2021a). This 2707 response would be consistent with pathway two of the MOA for Leydig cell tumors previously 2708 discussed in Section 4.3.1.1.3.

2709

2696

2710 Leydig cell adenomas represent a progression from pre-neoplastic Leydig cell hyperplasia. Leydig cell adenomas have been observed in F1 male offspring in three studies of similar design and from the same 2711 2712 laboratory (*i.e.*, CIIT) of SD rats exposed gestationally to up to 500 mg/kg-day DBP on GD 12 through 2713 GD 21 (Barlow et al., 2004; Mylchreest et al., 2000; Mylchreest et al., 1999). However, incidence of 2714 Leydig cell adenomas observed across all three studies was low (limited to 1-2 males per study) and did 2715 not reach statistical significance. Given that all three studies were designed to investigate the effects of 2716 gestation-only exposure to DBP on GD 12 through GD 21, the trend in Levdig cell adenomas is notable. 2717 However, in a subsequent study of SD rats by NTP, which included gestational and chronic (2-year) 2718 postnatal exposure, no significant increase in Leydig cell adenomas were observed in male SD rats 2719 exposed to up to 740 mg/kg-day DBP during gestation (GDs 6-21) and up to 510 mg/kg-day DBP for a 2720 further two-years (NTP, 2021a). Additionally, Leydig cell tumors were not observed in male B6C3F1 2721 mice treated with up to 1,306 mg/kg-day DBP for two-years, however, this study did not include 2722 gestational exposure to DBP (NTP, 2021a). 2723

EPA considers the low, statistically non-significant increase in Leydig cell adenomas reported by Mylchreest et al. (2000; 1999), Barlow et al. (2004), and NTP (2021a), which were not observed in chronic studies of male mice that achieved higher doses of DBP, to be of uncertain toxicological significance. Overall, <u>EPA considers there to be indeterminant scientific evidence to conclude that</u> gestational and/or chronic oral exposure to DBP induce Leydig cell adenomas in rats.

2729

2730

#### 4.3.3.3 Preliminary Cancer Classification for DBP

Under the Guidelines for Carcinogen Risk Assessment (U.S. EPA, 2005), EPA reviewed the weight of 2731 2732 evidence for the carcinogenicity of DBP and has preliminarily concluded that there is *Suggestive* 2733 Evidence of Carcinogenic Potential of DBP in rodents. According to the Guidelines for Carcinogen Risk 2734 Assessment (U.S. EPA, 2005), a descriptor of Suggestive Evidence of Carcinogenic Potential is 2735 appropriate "when the weight of evidence is suggestive of carcinogenicity; a concern for potential 2736 carcinogenic effects in humans is raised, but the data are judged not sufficient for a stronger conclusion. 2737 This descriptor covers a spectrum of evidence associated with varying levels of concern for carcinogenicity, ranging from a positive cancer result in the only study on an agent to a single positive 2738 2739 cancer result in an extensive database that includes negative studies in other species." EPA's 2740 determination is based on evidence of pancreatic acinar cell adenomas in male SD rats. Further weight 2741 of scientific evidence considerations supporting EPA's determination are listed below. According to the 2742 Guidelines for Carcinogen Risk Assessment (U.S. EPA, 2005), when there is Suggestive Evidence "the

Agency generally would not attempt a dose-response assessment, as the nature of the data generally would not support one." Consistently, EPA is not conducting a dose-response assessment for DBP or quantitatively evaluating DBP for carcinogenic risk to humans.

- DBP showed no carcinogenic activity in one study of male and female B6C3F1 mice exposed to up to 1,306 to 1,393 mg/kg-day DBP through the diet for two-years (NTP, 2021a).
- DBP showed no carcinogenic activity in one study of female SD rats exposed to up to 600 mg/kg-day DBP through the diet for two-years (NTP, 2021a).
- Treatment with DBP caused a significant increase in incidence of pancreatic acinar cell
   adenomas in male SD rats fed diets containing DBP for two-years at doses that did not result in
   overt toxicity (NTP, 2021a).
- Read-across from other toxicologically similar phthalates (*i.e.*, DEHP [Section 4.3.1.1] and BBP
   [Section 4.3.2.1.1]), which have also been shown to induce pancreatic acinar cell tumors in rats,
   provides additional evidence to support the conclusion that phthalates, including DBP, may
   cause pancreatic acinar cell adenomas in rats.

Herein, EPA has preliminarily concluded that there is *Suggestive Evidence of Carcinogenic Potential* of
DBP in rodents and consistent with the *Guidelines for Carcinogen Risk Assessment* (U.S. EPA, 2005)

did not conduct a dose-response assessment or evaluate DBP quantitatively for cancer risk.

2760

2761

## 4.3.4 Diisononyl Phthalate (DINP)

EPA has previously evaluated DINP for carcinogenicity in its *Cancer Human Health Hazard Assessment for Diisononyl Phthalate (DINP)* (U.S. EPA, 2025a). EPA's cancer assessment for DINP
was peer-reviewed by the SACC during its July 2024 meeting (U.S. EPA, 2024q). A brief summary of
carcinogenic findings and weight of evidence conclusions for DINP, which reflect recommendations
from the SACC (U.S. EPA, 2024q) and public comments, are provided below.

2767

DINP has been evaluated for carcinogenicity in two studies of male and female F344 (Covance Labs,
<u>1998c</u>; Lington et al., 1997), one study of SD rats (Bio/dynamics, 1987), and one study of male and
female B6C3F1 mice (Covance Labs, 1998b). Across available studies, statistically significant increases
in liver tumors, MNCL, and kidney tumors have been reported. EPA's conclusions regarding each of
these tumor types and EPA's cancer classification for DINP are provided below.

2773 MNCL. Following chronic dietary exposure to DINP, MNCL has been observed in two studies 2774 of male and female F344 rats (Covance Labs, 1998c; Lington et al., 1997), but not in SD rats (Bio/dynamics, 1987) or B6C3F1 mice of either sex (Covance Labs, 1998b). As discussed in the 2775 2776 Cancer Human Health Hazard Assessment for Diisononyl Phthalate (DINP) (U.S. EPA, 2025a) there are several sources of uncertainty associated with MNCL in F344 rats. First, MNCL has a 2777 2778 high background rate of spontaneous occurrence in F344 rats. Historical control data from NTP 2779 (1995–1998) show a background rate of MNCL of 52.5 percent in males and 24.2 percent in 2780 females (Thomas et al., 2007). F344 strain of rat was used in NTP 2-year chronic and carcinogenicity bioassays for nearly 30 years (King-Herbert et al., 2010; King-Herbert and 2781 2782 Thayer, 2006). However, in the early 2000s NTP stopped using the F344 strain of rat, in part 2783 because of high background incidence of MNCL and testicular Leydig cell tumors, and replaced 2784 the F344 strain of rats with the Harlan SD strain (King-Herbert et al., 2010; King-Herbert and 2785 Thayer, 2006). Additional sources of uncertainty include lack of MOA information and 2786 uncertainty related to the human correlate to MNCL in F344 rats. Given these uncertainties,

- 2787 SACC recommended that "the observation of an increased incidence of MNCL in a chronic bioassay employing the Fisher 344 rat should not be considered a factor in the determination of 2788 the cancer classification..." and "Most Committee members agreed that given the material 2789 2790 presented in a retrospective review, MNCL and Leydig Cell Tumors, among other tumor 2791 responses in F344 rat carcinogenicity studies lack relevance in predicting human 2792 carcinogenicity (Maronpot et al., 2016)" (U.S. EPA, 2024q). Consistent with the 2793 recommendations of the SACC, and based on the above discussion, EPA did not consider MNCL 2794 as a factor in its determination of the cancer classification for DINP.
- 2795 *Kidney Tumors*. Following chronic dietary exposure to DINP, renal tubule cell carcinomas have 2796 been reported in two studies of male (but not female) F344 rats (Covance Labs, 1998c; Lington et al., 1997). Kidney tumors were not observed in male or female SD rats or B6C3F1 mice fed 2797 2798 diets containing DINP for two-years (Covance Labs, 1998b; Bio/dynamics, 1987). Overall, EPA concluded that much of the available literature supports an  $\alpha_{2u}$ -globulin MOA to explain the 2799 incidences of renal tubule cell carcinomas observed in male rats exposed to DINP. EPA does not 2800 consider kidney tumors arising through a  $\alpha_{2u}$ -globulin MOA to be human relevant (U.S. EPA, 2801 2802 1991). Therefore, EPA did not consider it appropriate to derive quantitative estimates of cancer 2803 hazard for data on kidney tumors observed in these studies and did not further consider kidney 2804 tumors as a factor in the determination of the cancer classification for DINP. This conclusion 2805 was supported by the SACC. In its final report to EPA, the SACC states "The Agency has provided substantial evidence that the kidney tumors produced by DINP are due to a 2u-globulin 2806 2807 MOA and correctly classified them as not relevant to humans" (U.S. EPA, 2024q). See Section 2808 3.2.3 of (U.S. EPA, 2025a) for further details.
- 2809 *Liver Tumors*. Following chronic dietary exposure to DINP, hepatocellular adenomas (or 2810 neoplastic nodules) and/or carcinomas were consistently observed in male and female F344 rats 2811 (Covance Labs, 1998c; Lington et al., 1997), female SD rats (Bio/dynamics, 1987), and B6C3F1 mice of both sexes (Covance Labs, 1998b). Overall, EPA concluded that there is strong evidence 2812 to support the conclusion that DINP causes liver tumors in rodents through a non-genotoxic. 2813 2814 threshold, peroxisome proliferator-activated receptor alpha (PPAR $\alpha$ ) MOA (see Section 4 of (U.S. EPA, 2025a) for further discussion). This conclusion was supported by the SACC during 2815 2816 their July 2024 peer review meeting (U.S. EPA, 2024q).
- 2817 *Cancer Classification for DINP*. Under the *Guidelines for Carcinogen Risk Assessment* (U.S. • 2818 EPA, 2005), EPA reviewed the weight of evidence and determined that DINP is Not Likely to be Carcinogenic to Humans at doses below levels that do not result in PPARa activation (KE 1 in 2819 2820 the PPARa MOA) (see Section 4.8 of (U.S. EPA, 2025a) for further details). Further, the non-2821 cancer chronic POD (NOAEL/LOAEL of 15/152 mg/kg-day based on non-cancer liver effects 2822 (see Draft Non-cancer Human Health Hazard Assessment for Diisononyl Phthalate (DINP) 2823 (U.S. EPA, 2025k)) will adequately account for all chronic toxicity, including carcinogenicity, 2824 which could potentially result from exposure to DINP. In one study of male mice (Kaufmann et al., 2002), biomarkers of PPARa activation were significantly increased at 117 mg/kg-day, 2825 2826 which is less than the chronic LOAEL of 152 mg/kg-day based on non-cancer liver effects. Although, the study by Kaufman et al. did not test sufficiently low doses to establish a NOAEL 2827 for PPARα activation, other studies of mice have established a NOAEL of 75 mg/kg-day for 2828 2829 PPAR $\alpha$  activation (Smith et al., 2000). Therefore, the non-cancer chronic POD of 15 mg/kg-day 2830 is considered protective of PPARa activation.
- 2831

4.3.5 Diisodecyl Phthalate (DIDP)

EPA has previously evaluated DIDP for carcinogenicity in its *Human Health Hazard Assessment for Diisodecyl Phthalate (DIDP)* (U.S. EPA, 2024n). EPA's cancer assessment for DIDP was peer-reviewed
by the SACC during its July 2024 meeting (U.S. EPA, 2024q). A brief summary of carcinogenic
findings and weight of evidence conclusions for DIDP, which reflect recommendations from the SACC
(U.S. EPA, 2024q) and public comments, are provided below.

DIDP has been evaluated for carcinogenicity in one two-year dietary study of male and female F344 rats (Cho et al., 2010; Cho et al., 2008) and in one 26-week dietary study of male and female wild-type and transgenic CB6F1-RasH2 mice (Cho et al., 2011). Across available studies, statistically significant increases in MNCL were observed in high-dose (479–620 mg/kg-day) male and female F344 rats, while hepatocellular adenomas were observed in high-dose (1,500 mg/kg-day) male transgenic CB6F1-RasH2 mice (Cho et al., 2011).

2845

2849

2832

2838

Under the *Guidelines for Carcinogen Risk Assessment* (U.S. EPA, 2005), EPA reviewed the weight of
 evidence for the carcinogenicity of DIDP and concluded that DIDP is not likely to be carcinogenic to
 humans. This conclusion is based on the following:

2850 "Weight of scientific evidence considerations supporting EPA's determination are listed below.
2851 Consistent with this cancer classification, EPA is not conducting a dose-response assessment for DIDP
2852 or evaluating DIDP for carcinogenic risk to humans.

- 2853 Hepatocellular adenomas were observed only in high-dose male CB6F1-rasH2 transgenic mice 2854 at 1,500 mg/kg-day but not in female transgenic mice or in wild-type male or female mice, which 2855 are more appropriate for use in human health risk assessment (Cho et al., 2011). However, in the 2856 studies of wild-type and transgenic mice, the highest dose tested, 1,500 mg/kg-day, was above the limit dose. This is demonstrated by the fact that terminal body weight was reduced 27 and 12 2857 2858 percent in male and female wild-type mice, respectively, and 31 and 15 percent in male and female transgenic mice, respectively, at 1,500 mg/kg-day. Per EPA's Guidelines for Carcinogen 2859 2860 Risk Assessment (U.S. EPA, 2005) "signs of treatment-related toxicity associated with an excessive 2861 high dose may include (a) significant reduction of body weight gain (e.g., greater than 10%)." 2862 Further, EPA's Guidelines for Carcinogen Risk Assessment state that "overt toxicity or qualitatively altered toxicokinetics due to excessively high doses may result in tumor effects that 2863 2864 are secondary to the toxicity rather than directly attributable to the agent."
- No evidence of carcinogenic activity was observed in male or female CB6F1-rasH2 transgenic mice dosed with 150 or 495 mg/kg-day DIDP (Cho et al., 2011). Evidence of overt treatment-related toxicity associated with exceedance of the limit dose was not apparent at these dose levels.
- EPA acknowledges that increased MNCL was observed in male and female F344 rats treated with DIDP for two years (Cho et al., 2010; Cho et al., 2008). However, MNCL was only observed at in the high-dose group and coincided with high mortality. No other preneoplastic or neoplastic findings were observed in any tissue for either sex at any dose.
- MNCL has a high rate of spontaneous occurrence in F344 rats. Although the historical control data are not available for the laboratory that conducted this study, historical control data from NTP (1995–1998) show 52.5 percent in males and 24.2 percent in females (Thomas et al., 2007). The F344 strain of rat was used in NTP 2-year chronic and carcinogenicity bioassays for nearly 30 years (King-Herbert et al., 2010; King-Herbert and Thayer, 2006). However, in the early

- 2878 2000s, NTP stopped using the F344 strain of rat, in part because of high background incidence
  2879 of MNCL and testicular Leydig cell tumors, and replaced the F344 strain of rats with the Harlan
  2880 Sprague Dawley strain (King-Herbert et al., 2010; King-Herbert and Thayer, 2006). Consistent
  2881 with recommendations of the SACC (U.S. EPA, 2024q), EPA is not further considering MNCL as
  2882 a factor in the determination of the cancer classification for DIDP because this is likely a strain2883 specific effect.
- EPA's weight of scientific evidence conclusion is consistent with Health Canada (EC/HC, 2015c), U.S. CPSC (2014, 2010d), NICNAS (2015b), and ECHA (2013). None of these regulatory agencies have evaluated DIDP for carcinogenic risk to human health."

# May 2025 5 EVALUATING THE CARCINOGENICITY OF DIBP AND DCHP USING READ-ACROSS: WEIGHT OF SCIENTIFIC EVIDENCE ANALYSIS

PUBLIC RELEASE DRAFT

No chronic toxicity or cancer bioassays are available for DIBP or DCHP in the published literature. EPA
therefore evaluated the relevance of read-across approaches to assess potential cancer hazards of DIBP
and DCHP based on cancer bioassays and MOA information available for other phthalates being
evaluated under TSCA (*i.e.*, DEHP, BBP, DBP, DINP, DIDP).

2895 Hilton et al. (2022) published a weight of evidence-based framework for determining the need for rodent 2896 cancer bioassays for agrochemicals lacking chronic and/or carcinogenicity studies - known as the 2897 Rethinking Chronic Toxicity and Carcinogenicity Assessment for Agrochemicals Project, or the 2898 ReCAAP Framework. Although developed specific for agrochemicals, EPA believes many of the same 2899 scientific principles in the ReCAAP Framework apply to TSCA risk evaluations. As such, elements of 2900 the ReCAAP Framework is used as an organizational tool to evaluate the extent to which the lack of 2901 carcinogenicity studies imparts significant uncertainty on the human health risk assessments for DIBP 2902 and DCHP.

2903

2904 The ReCAAP framework takes into consideration multiple lines of evidence including information 2905 pertaining to nomenclature, physical and chemical properties; exposure and use patterns; absorption, 2906 distribution, metabolism, and excretion (ADME) properties; and toxicological data (e.g., genetic 2907 toxicity, acute toxicity, subchronic toxicity, hormone perturbation, immunotoxicity, and MOA). The 2908 framework was developed by a workgroup comprised of scientists from academia, government 2909 (including EPA), non-governmental organizations, and industry stakeholders. Recently, the Organisation 2910 for Economic Co-operation and Development (OECD) has published several Integrated Approach to 2911 Testing and Assessment (IATA) case studies demonstrating applicability of the weight of evidence 2912 ReCAAP framework (OECD, 2024).

2913

2914 Herein, EPA used some, but not all, elements of the ReCAAP framework and OECD case studies. 2915 Elements of the ReCAAP framework considered herein include physical and chemical properties 2916 (Section 5.1), ADME properties (Section 5.2), acute toxicity (Section 5.3), evidence of hormone 2917 perturbation, developmental and reproductive toxicity (Section 5.4), subchronic toxicity (Section 5.5), 2918 immune systemic perturbation (Section 5.6), genotoxicity (Section 5.7), MOA (Section 5.8), and 2919 evidence of chronic toxicity and carcinogenicity from read-across chemicals (Section 5.9). Read-across 2920 to other structurally and toxicologically similar phthalate diesters currently being evaluated under TSCA 2921 (i.e., DEHP, BBP, DBP, DIBP, DCHP, DINP, DIDP) were considered as part of the current weight of 2922 evidence and read-across approach. The weight of evidence narrative provided in this section represents 2923 a brief synthesis of available information for DIBP, DCHP, and the five read-across phthalates (DEHP, 2924 BBP, DBP, DINP, DIDP). Complete human health hazard and physical and chemical property 2925 information for the seven phthalates being evaluated under TSCA is provided in individual phthalate 2926 technical support documents, including:

- Draft Non-cancer Human Health Hazard Assessment for Diethylhexyl Phthalate (DEHP) (U.S. EPA, 2024f);
- Draft Non-cancer Human Health Hazard Assessment for Butyl benzyl phthalate (BBP) (U.S. EPA, 2024c);

#### May 2025

| 2931 • | Draft Non-cancer Human Health Hazard Assessment for Dibutyl Phthalate (DBP) (U.S. EPA, |
|--------|--|
| 2932   | <u>2024d</u> );  |

- Draft Non-cancer Human Health Hazard Assessment for Diisobutyl phthalate (DIBP) (U.S.
   EPA, 2024g);
- Draft Non-Cancer Human Health Hazard Assessment for Dicyclohexyl Phthalate (DCHP) (U.S.
   <u>EPA, 2024e</u>);
- Non-Cancer Human Health Hazard Assessment for Diisononyl Phthalate (DINP) (U.S. EPA, 2025k);
- Cancer Human Health Hazard Assessment for Diisononyl Phthalate (DINP) (U.S. EPA, 2025a);
- Human Health Hazard Assessment for Diisodecyl Phthalate (DIDP) (U.S. EPA, 2024n);
- Draft Physical and Chemical Property Assessment and Fate and Transport Assessment for Diethylhexyl Phthalate (DEHP) (U.S. EPA, 2024h);
- Draft Physical Chemistry and Fate and Transport Assessment for Butyl Benzyl Phthalate (BBP)
   (U.S. EPA, 2024i);
- Draft Physical Chemistry and Fate and Transport Assessment for Dibutyl Phthalate (DBP) (U.S. EPA, 2024a);
- Draft Physical Chemistry and Fate and Transport Assessment for Diisobutyl phthalate (DIBP)
   (U.S. EPA, 2024k);
- Draft Physical Chemistry and Fate and Transport Assessment for Dicyclohexyl Phthalate (DCHP) (U.S. EPA, 2024j);
- Physical Chemistry Assessment for Diisononyl Phthalate (DINP) (U.S. EPA, 20251); and
- Physical Chemistry Assessment for Diisodecyl Phthalate (DIDP) (U.S. EPA, 20240).
- 2953

## 2954 **5.1 Physical and Chemical Properties**

Table 5-1 summarizes the physical and chemical properties of DIBP and DCHP, as well as read-across chemicals DEHP, BBP, DBP, DINP, and DIDP. Based on the physical and chemical properties of DIBP and DCHP, and the read-across chemicals DEHP, BBP, DBP, DINP, and DIDP, the following conclusions can be drawn.

- DEHP, BBP, DBP, DIBP, DINP, and DIDP are liquid, while DCHP is a solid at room temperature.
- DEHP, BBP, DBP, DIBP, DCHP, DINP, and DIDP have very low to slight solubility in water.
   DEHP, DINP and DIDP have very low water solubility (0.003 mg/L for DEHP; 0.00061 mg/L for DINP; 0.00017 mg/L for DIDP), while BBP, DBP, DIBP, and DCHP are slightly soluble in water (2.3 mg/L for BBP; 11.2 mg/L for DBP; 6.2 mg/L for DIBP; 1.48 mg/L for DCHP).
- Sorption to organics present in sediment and suspended and dissolved solids present in water is expected to be a dominant process given the range of identified log K<sub>oc</sub> values (2.09 to 5.78) across DEHP, BBP, DBP, DIBP, DCHP, DINP, and DIDP.

- Given the range of water solubility values and range of log K<sub>oc</sub> values for DEHP, BBP, DBP, DBP, DIBP, DCHP, DINP, and DIDP, these phthalates are unlikely to exhibit mobility in soils.
- Phthalates generally have low volatility. Based on physical and chemical properties (*i.e.*, melting point, boiling point, Henry's Law Coefficient), DCHP is classified as a non-volatile organic compound, while DEHP, BBP, DBP, DIBP, DINP, and DIDP are marginally classified as semi-volatile organic compounds. However, volatilization of DEHP, BBP, DBP, DIBP, DINP, and DIDP from water to air or soil to air is expected to be peoligible.
- 2974 DIDP from water-to-air or soil-to-air is expected to be negligible.

#### May 2025

### 2975 Table 5-1. Summary of Physical and Chemical Properties of DCHP, DBP, DIBP, BBP, DEHP, DIDP, and DINP

| Table 5-1. Builling  | of I hysical and                             |   | per tites of Defin                           | , <b>DDI</b> , <b>DIDI</b> , <b>D</b>        | <b>DI , DEIII , DI</b>                       | JI, and DIM                                  |  |
|--|--|---|--|--|--|--|--|
| Property   | DEHP<br>( <u>U.S. EPA,</u><br><u>2024h</u> ) | BBP<br>( <u>U.S. EPA,</u><br><u>2024i</u> ) | DBP<br>( <u>U.S. EPA,</u><br><u>2024a</u> )  | DIBP<br>( <u>U.S. EPA,</u><br><u>2024k</u> ) | DCHP<br>( <u>U.S. EPA,</u><br><u>2024j</u> ) | DINP<br>( <u>U.S. EPA,</u><br><u>20251</u> ) | DIDP<br>( <u>U.S. EPA,</u><br><u>2024o</u> ) |
| Molecular formula  | $C_{24} H_{38} O_4$                          | $C_{19}H_{20}O_4$                           | $C_{16}H_{22}O_4$                            | $C_{16}H_{22}O_4$                            | $C_{20}H_{26}O_4$                            | $C_{26}H_{42}O_4$                            | $C_{28}H_{46}O_4$                            |
| Molecular Weight<br>(g/mol)                                    | 390.56                                       | 312.37                                      | 278.35                                       | 278.35                                       | 330.43                                       | 418.62                                       | 446.7  |
| Physical state of the chemical                                 | Colorless, oily<br>liquid                    | Clear oil, liquid                           | Colorless to<br>faint yellow,<br>oily liquid | Colorless, clear,<br>viscous liquid          | White, granular solid                        | Clear Liquid                                 | Clear Liquid                                 |
| Melting Point (°C)   | -55  | -35   | -35  | -64  | 66   | -48  | -50  |
| Boiling Point (°C)   | 384  | 370   | 340  | 296.5  | 225  | >400   | >400   |
| Density (g/cm <sup>3</sup> )                                   | 0.981  | 1.119                                       | 1.0459 to<br>1.0465                          | 1.049  | 1.383  | 0.97578                                      | 0.967  |
| Vapor Pressure<br>(mmHg)                                       | 1.42×10-7                                    | 8.25×10 <sup>-6</sup>                       | 2.01×10 <sup>-5</sup>                        | 4.76×10 <sup>-5</sup>                        | 8.69×10 <sup>-7</sup>                        | 5.40×10 <sup>-7</sup>                        | 5.28×10 <sup>-7</sup>                        |
| Water Solubility (ng/L)  | 3,000  | 2,690,000                                   | 11,200,000                                   | 6,200,000                                    | 30,000 -<br>1,480,000                        | 610  | 170  |
| Log K <sub>OW</sub>  | 7.6  | 4.73  | 4.5  | 4.34   | 4.82   | 8.8  | 10.21<br>(estimated)                         |
| Log K <sub>OA</sub> (estimated using EPI Suite <sup>TM</sup> ) | 10.76  | 9.2   | 8.63   | 9.47   | 10.23  | 11.9   | 13.0   |
| Log K <sub>OC</sub>  | 3.75-5.48                                    | 2.09-2.91                                   | 3.16-4.19                                    | 2.5-3.14                                     | 3.46-4.12                                    | 5.5-5.7                                      | 5.04-5.78                                    |
| Henry's Law Constant<br>(atm-m <sup>3</sup> /mol)              | 1.71×10 <sup>-5</sup>                        | 7.61×10 <sup>-7</sup>                       | 1.81×10 <sup>-6</sup>                        | 1.83×10 <sup>-7</sup>                        | 9.446×10 <sup>-8</sup>                       | 9.14×10 <sup>-5</sup>                        | 21.3×10 <sup>-5</sup>                        |
| Flash Point (°C)   | 206  | 199   | 157.22                                       | 185  | 207  | 213  | >200   |
| Autoflammability (°C)  | 390  | -   | 402.778                                      | 432  | No data                                      | 400  | 402  |
| Viscosity (cP)   | 57.94  | 55  | 20.3   | 41   | Not applicable (solid)                       | 77.6   | 87.797                                       |
| Overall Environmental<br>Persistence                           | Low  | Low   | Low  | Low  | Low  | Low  | Low  |

| Property                                 | DEHP<br>( <u>U.S. EPA,</u><br><u>2024h</u> ) | BBP<br>( <u>U.S. EPA,</u><br><u>2024i</u> ) | DBP<br>( <u>U.S. EPA,</u><br><u>2024a</u> ) | DIBP<br>( <u>U.S. EPA,</u><br><u>2024k</u> ) | DCHP<br>( <u>U.S. EPA,</u><br><u>2024j</u> ) | DINP<br>( <u>U.S. EPA,</u><br><u>20251</u> ) | DIDP<br>( <u>U.S. EPA,</u><br><u>2024o</u> ) |
|--|--|---|---|--|--|--|--|
| Bioaccumulation<br>Factor (Log BAF A-G)  | 3.02   | 1.60  | 2.20  | 1.41   | 2.14   | 1.14   | 2.06   |
| Bioconcentration<br>Factor (Log BCF A-G) | 2.09   | 2.88  | 2.20  | 1.41   | 2.13   | 0.39   | 1.04   |

2976

# 2977 **5.2** Absorption, Distribution, Metabolism, and Excretion

The ADME properties of DIBP, DCHP, and the five read-across chemicals (DEHP, BBP, DBP, DINP, and DIDP) following oral exposure are discussed briefly below. Readers are directed to the human health hazard assessments for DEHP (U.S. EPA, 2024f), BBP (U.S. EPA, 2024c), DBP (U.S. EPA, 2024g), DCHP (U.S. EPA, 2024e), DINP (U.S. EPA, 2025k), and DIDP (U.S. 2982 EPA, 2024n) for more detailed summaries of ADME properties.

2984 Limited information is available pertaining to the ADME properties of DIBP and DCHP. No in vivo 2985 studies of experimental animal models or controlled human exposure studies are available that have 2986 evaluated the ADME properties of DCHP. However, *in vitro* studies have demonstrated that DCHP is 2987 rapidly hydrolyzed to its corresponding monoester, monocyclohexyl phthalate (MCHP). Further, human biomonitoring studies have measured MCHP in urine, demonstrating that DCHP can be metabolized to 2988 2989 MCHP and excreted in urine in humans (U.S. EPA, 2024e). Similarly, no in vivo studies of experimental 2990 animal models are available that have evaluated the ADME properties of DIBP. However, in a 2991 controlled human oral exposure study of DIBP, approximately 90 percent of administered DIBP was 2992 recovered in urine within 24 hours. DIBP was excreted primarily as the monoester metabolite, 2993 monoisobutyl phthalate (MIBP, accounted for approximately 70 percent of excreted DIBP), while 2994 several other oxidated derivatives of MIBP (i.e., 2OH-MIBP and 3OH-MIBP) were found to be minor 2995 urinary metabolites accounting for around 20 percent of excreted DIBP. Overall, this study indicates 2996 rapid and near complete oral absorption of DIBP, which is metabolized to MIBP and can then undergo 2997 further oxidative metabolism before being rapidly eliminated in urine (U.S. EPA, 2024g).

2999 For the five read-across phthalates (*i.e.*, DEHP, BBP, DBP, DINP and DIDP), more extensive databases of studies evaluating ADME properties are available, including controlled human oral exposure studies, 3000 3001 studies of rats and mice, as well as in vitro metabolism studies. Available data indicate that following oral exposure, DEHP, BBP, DBP, DINP and DIDP are rapidly absorbed and systemically distributed. 3002 3003 For input into the draft risk evaluations for DEHP, BBP, DBP, DINP and DIDP (as well as for DIBP and 3004 DCHP), EPA assumed 100 percent oral absorption. Further, available studies indicate that DEHP, BBP, 3005 DBP, DINP and DIDP are all rapidly metabolized into monoester metabolites by esterases in the gut or 3006 other tissues following absorption. Monoester metabolites then undergo further oxidative metabolism 3007 and/or can also be conjugated with glucuronic acid before being excreted in urine, or to a lesser extent 3008 feces. Many unique, but also some common metabolites across phthalates have been identified. For 3009 example, phthalic acid is a potential metabolite of DEHP, BBP, DBP, DINP and DIDP (as well as of DIBP and DCHP). Available studies of rats and mice have shown that these five read-across phthalates 3010 3011 are nearly completely excreted within 72 to 96 hours. Given the rapid elimination kinetics, DEHP, BBP, 3012 DBP, DINP and DIDP are not considered bioaccumulative.

3013

2983

2998

# 3014 **5.3 Acute Toxicity**

The acute toxicity of DIBP and DCHP, and read-across chemicals DEHP, BBP, DBP, DINP, and DIDP have been evaluated extensively by various authoritative and regulatory agencies, including U.S. CPSC (2014, 2011, 2010a, b, c, d, e, f), ECB (2008, 2007, 2004, 2003a, c), ECHA (2017a, 2013), Australia NICNAS (2016, 2015a, b, 2013, 2012, 2010, 2008a, b, c), and ATSDR (2022, 2001). Table 5-2 summarizes some of the available acute oral LD<sub>50</sub>, dermal LD<sub>50</sub>, and inhalation LC<sub>50</sub> values, as well as results from skin irritation, eye irritation, and skin sensitization testing for the seven phthalate diesters being evaluated under TSCA. Across existing assessments of phthalates, there is consensus that DEHP,

- BBP, DBP, DIBP, DCHP, DINP, and DIDP are not acutely toxic in terms of lethality via the oral,
- dermal, or inhalation exposure routes. However, as will be discussed further in Sections 5.4 and 5.9,
   DEHP, BBP, DBP, DIBP, DCHP, DINP, and DIDP are all developmental toxicants, and EPA considers
- developmental effects such as reduced offspring survival in the case of DIDP and effects on the
- 3026 developing male reproductive system consistent with phthalate syndrome in the cases of DEHP, BBP,
- 3027 DBP, DIBP, DCHP, and DINP relevant for assessing risk from acute duration exposures.
- 3028
- Further, DEHP, BBP, DBP, DIBP, DCHP, DINP, and DIDP are not considered corrosive and cause no or minimal irritant effects to the eye or skin. Finally, phthalates are considered to have low skin
- sensitizing potential, with the one possible exception being DCHP. As discussed in EPA's *Draft Non*-
- 3032 Cancer Human Health Hazard Assessment for Dicyclohexyl Phthalate (DCHP) (U.S. EPA, 2024e),
- 3033 DCHP tested positive as a dermal sensitizer in one local lymph node assay and is classified
- 3034 (harmonised) as a sensitizer in the European Union (ECHA, 2014). However, only the ECHA robust
- 3035 study summary was available to EPA for review, and the original study report was not available to EPA
- for independent review. Therefore, EPA considers there to be indeterminant evidence to draw aconclusion on the skin sensitizing potential of DCHP.
- 3038
- 3039

3040Table 5-2. Summary of Acute Toxicity Data for DEHP, BBP, DBP, DIBP, DCHP, DINP, and3041DIDP<sup>a</sup>

|                                       |                         |                       |                     |                         |                                | 1                               |                  |
|---------------------------------------|-------------------------|-----------------------|---------------------|-------------------------|--------------------------------|---------------------------------|------------------|
|                                       | DEHP                    | BBP                   | DBP                 | DIBP                    | DCHP                           | DINP                            | DIDP             |
| Oral LD <sub>50</sub><br>(mg/kg)      | 30,600–<br>40,000 (rat) | 2330–<br>20,400 (rat) | 6300- 8000<br>(rat) | 16,000–<br>60,320 (rat) | >3200 (rat)                    | >10,000<br>(rat) <sup>b</sup>   | >29,100          |
| Dermal<br>LD <sub>50</sub><br>(mg/kg) | 24,750<br>(rabbit)      | 6700 (rat)            | >20,000<br>(rabbit) | No study                | >300<br>(rabbit)               | >3,160<br>(rabbit) <sup>b</sup> | >2910<br>(rat)   |
| Inhalation<br>LC <sub>50</sub> (mg/L) | >10.62 (rat)            | No study              | ≥15.68 (rat)        | No study                | >3.2 (rat)                     | >4.4<br>(rat) <sup>b</sup>      | >12.54<br>(rat)  |
| Skin<br>Irritation                    | Minimal<br>effect       | Minimal<br>effect     | Minimal<br>effect   | Minimal effect          | Minimal effect                 | Minimal effect <sup>b</sup>     | Minimal effect   |
| Eye<br>Irritation                     | Minimal<br>effect       | Minimal<br>effect     | Minimal<br>effect   | Not a eye irritant      | Minimal effect                 | Minimal effect <sup>b</sup>     | Minimal effect   |
| Skin<br>Sensitization                 | Not a sensitizer        | Not a sensitizer      | Not a sensitizer    | Not a sensitizer        | Insufficient data <sup>c</sup> | Not a sensitizer <sup>b</sup>   | Not a sensitizer |

<sup>a</sup> Data from Table 4 of (<u>NICNAS, 2008c</u>) unless otherwise noted.

<sup>b</sup> Data from (<u>U.S. EPA, 2025k; ECHA, 2013; NICNAS, 2012; ECB, 2003c</u>).

<sup>c</sup> Only the ECHA robust study summary was available to EPA for review (ECHA, 2014), and the original study report was not available to EPA for independent review. Therefore, EPA considers there to be indeterminant evidence to draw a conclusion on the skin sensitizing potential of DCHP.

3042

# 3043 3043 3044 5.4 Evidence of Hormone Perturbation, and Developmental and Reproductive Toxicity

Hormone perturbation, as well as subsequent developmental and reproductive toxicity, is a hallmark of
exposure to certain phthalate diesters, including DIBP and DCHP, and the read-across chemicals DEHP,
BBP, DBP, and DINP (but not DIDP). As discussed in EPA's *Draft Proposed Approach for Cumulative Risk Assessment of High-Priority Phthalates and a Manufacturer-Requested Phthalate under the Toxic*

3049 Substances Control Act (U.S. EPA, 2023), and in the human health hazard assessments for DEHP (U.S. EPA, 2024f), BBP (U.S. EPA, 2024c), DBP (U.S. EPA, 2024d), DIBP (U.S. EPA, 2024g), DCHP (U.S. 3050 3051 EPA, 2024e), and DINP (U.S. EPA, 2025k), these phthalates are antiandrogenic. Studies in rats have 3052 demonstrated that exposure to DIBP and DCHP, and the read-across chemicals DEHP, BBP, DBP, and 3053 DINP, during the critical window of development can disrupt testosterone biosynthesis in the fetal testis, 3054 leading to decreased male anogenital distance, increased male nipple retention, and seminiferous tubule 3055 atrophy (Table 5-4). Severe reproductive tract malformations such as hypospadias and cryptorchidism, 3056 sperm effects, and decreases in male fertility have also been observed for some of these phthalates 3057 (Table 5-4). Although qualitatively these phthalates are toxicologically similar, important differences in potency are apparent based on reductions in fetal testicular testosterone, with DCHP being the most 3058 3059 potent, followed by DBP, DEHP, DIBP, BBP, and DINP being the least potent (Table 5-3) (see (U.S. EPA, 2024b) for further details). 3060

3061 3062

3063

#### Table 5-3. Summary of Phthalate Potency for Reducing Fetal Testicular Testosterone

| Phthalate   | BMD <sub>40</sub> (mg/kg-day) for Reduced<br>Fetal Testicular Testosterone <sup>a</sup> |
|---|---|
| DCHP  | 90  |
| DBP   | 149   |
| DEHP  | 178   |
| DIBP  | 279   |
| BBP   | 284   |
| DINP  | 699   |
| <sup><i>a</i></sup> BMD <sub>40</sub> = ben<br>reduction in fer | chmark dose (BMD) associated with a 40% tal testicular testosterone.                    |

3064

3065

3066 In contrast to DEHP, BBP, DBP, DIBP, DCHP, and DINP, DIDP is not antiandrogenic and does not disrupt fetal testis testosterone biosynthesis in studies of rats (U.S. EPA, 2024n, 2023). However, as 3067 discussed in EPA's Human Health Hazard Assessment for Diisodecyl Phthalate (DIDP) (U.S. EPA, 3068 2024n), DIDP is a developmental toxicant and has been shown to induce skeletal and visceral variations 3069 in fetal rats in prenatal developmental studies, as well as reduce F1 and F2 offspring survival, body 3070 3071 weight, and body weight gain in several two-generation studies of reproduction. Similar developmental 3072 effects as observed for DIDP have also been observed for DEHP, BBP, DBP, DIBP, DCHP, and DINP, 3073 albeit at higher doses than those that cause antiandrogenic effects on the developing male reproductive 3074 system.

3075 3076

#### 3077 Table 5-4. Summary of Phthalate Syndrome-Related Effects Observed in Studies of Rat<sup>a</sup>

| Phthalate Syndrome-Related Effect                                       | DEHP | BBP | DBP | DIBP | DCHP | DINP | DIDP |
|---|------|-----|-----|------|------|------|------|
| ↓ Steroidogenic gene and <i>Insl3</i><br>expression in the fetal testis | ~    | ~   | ~   | ~    | ~    | ~    | x    |
| ↓ Fetal testis testosterone   | ~    | ~   | ~   | ~    | ~    | ~    | x    |

| Phthalate Syndrome-Related Effect                   | DEHP | BBP          | DBP          | DIBP         | DCHP         | DINP | DIDP |
|---|------|--------------|--------------|--------------|--------------|------|------|
| ↓ Anogenital distance                               | ~    | $\checkmark$ | $\checkmark$ | $\checkmark$ | $\checkmark$ | i    | x    |
| Nipple retention                                    | ~    | $\checkmark$ | $\checkmark$ | ~            | $\checkmark$ | i    | x    |
| Hypospadias   | ~    | $\checkmark$ | √            | ~            | √            | x    | х    |
| Seminiferous tubule atrophy                         | ✓    | $\checkmark$ | $\checkmark$ | ~            | ✓            | i    | x    |
| Multinucleated gonocytes (MNGs)                     | ~    | $\checkmark$ | ~            | ~            | √            | ~    | _    |
| $\downarrow$ Reproductive organ weight <sup>b</sup> | ~    | $\checkmark$ | ~            | ~            | √            | i    | x    |
| Testicular pathology <sup>c</sup>                   | ✓    | $\checkmark$ | ~            | ~            | √            | ~    | x    |
| Epididymal agenesis                                 | ~    | $\checkmark$ | ~            | ~            | -            | ~    | x    |
| Gubernaculum agenesis                               | ~    | _            | ~            | _            | _            | —    | x    |
| Undescended testes                                  | ~    | $\checkmark$ | $\checkmark$ | ~            | x            | x    | x    |
| Sperm effects <sup>d</sup>                          | ~    | $\checkmark$ | $\checkmark$ | _            | $\checkmark$ | ~    | x    |
| $\downarrow$ Male fertility <sup>e</sup>            | ~    | $\checkmark$ | $\checkmark$ | —            | x            | x    | x    |

 $\checkmark$  = Studies available, effects observed.

x = Studies available, no effects observed.

i = Studies available, inconsistent effects observed.

– = No study available.

<sup>a</sup> Adapted from Table 3-22 in EPA's Draft Proposed Approach for Cumulative Risk Assessment of High-Priority Phthalates and a Manufacturer-Requested Phthalate under the Toxic Substances Control Act (U.S. EPA, 2023).

<sup>b</sup> May include decreased absolute testis, epididymis, seminal vesicle, and/or prostate weight.

<sup>c</sup> May include, but is not limited to, Leydig cell aggregation, interstitial cell hyperplasia or adenoma, Sertoli cell only. tubules, and/or epididymal oligospermia or azoospermia.

<sup>d</sup> May include, but is not limited to, decreased sperm motility and/or concentration.

<sup>e</sup> May include, but is not limited to decreased mating, pregnancy, and/or fertility indices.

3078

3079

## 5.5 Subchronic Toxicity

Although hormone perturbation (*i.e.*, disruption of testis testosterone biosynthesis) and effects on the developing male reproductive system have been identified as the most sensitive non-cancer effects for DEHP, BBP, DBP, DIBP, and DCHP, the liver has also been consistently identified as a target organ for DIBP, DCHP, and the five read-across phthalates.

3084

3085 As discussed in Section 3.3 of the Draft Non-Cancer Human Health Hazard Assessment for

3086 *Dicyclohexyl Phthalate (DCHP)* (U.S. EPA, 2024e), intermediate and subchronic duration exposure 3087 studies have consistently demonstrated that oral exposure to DCHP can cause dose-related increases in

3088 relative liver weight in rats, as well as cause increases in hepatocellular hypertrophy and serum

3089 chemistry markers of liver toxicity (*i.e.*, ALT, AST) (<u>Ahbab et al., 2017</u>; <u>Saillenfait et al., 2009</u>;

3090 <u>Yamasaki et al., 2009; Hoshino et al., 2005; Lake et al., 1982</u>). As will be discussed further in Section

3091 5.8, there is some mechanistic evidence that DCHP can activate PPAR $\alpha$  in the liver, and it is possible

that PPARα activation underlies the observed liver effects of DCHP. For DIBP, there is less evidence

3093 for liver toxicity in rodents following oral exposure. As discussed by Yost et al. (2019), there is robust

- 3094 evidence that oral exposure to DIBP can increase relative liver weight in multiple studies of rats and
- 3095 mice (Wang et al., 2017; Oishi and Hiraga, 1980a, b, c, d; University of Rochester, 1954, 1953).

3096 However, available studies have generally not evaluated serum chemistry markers of liver toxicity or 3097 conducted histopathologic evaluations of the liver following oral exposure to DIBP.

3098

3099 For the five read-across phthalates (DEHP, BBP, DBP, DINP and DIDP), there is consistent evidence of 3100 dose-related liver toxicity following subchronic oral exposure. Observed effects include, increases in 3101 relative liver weights, increases in serum markers of liver toxicity (e.g., ALT, AST, ALP, GGT), and 3102 non-cancer histopathologic findings such as hepatocellular hypertrophy, focal necrosis, and spongiosis 3103 hepatis (limited to studies of F344 rats). Further, and as will be discussed further in Section 5.8, there is 3104 evidence that all of these phthalates can activate PPARa, which is mechanistically linked to many of the 3105 observed non-cancer liver effects. One exception to this is the observed increase in spongiosis hepatis in 3106 male F344 rats, which is not believed to be mechanistically linked to PPARa activation. Non-cancer 3107 liver effects are discussed further in the human health hazard assessments for DEHP (U.S. EPA, 2024f), 3108 BBP (U.S. EPA, 2024c), DBP (U.S. EPA, 2024d), DINP (U.S. EPA, 2025k), and DIDP (U.S. EPA, 3109 2024n).

3110

# 3111 **5.6 Evidence of Immune System Perturbation**

As discussed by Hilton et al. (2022), immune system suppression can increase the likelihood of cancer 3112 3113 in humans. DIBP and DCHP, and the read-across chemicals DEHP, BBP, DBP, DINP, and DIDP have 3114 been evaluated extensively by various authoritative and regulatory agencies, including U.S. CPSC 3115 (2014, 2011, 2010a, b, c, d, e, f), NTP Center for the Evaluation of Risks to Human Reproduction (NTP-3116 CERHR) (2006, 2003a, b, c, d), ECB (2008, 2007, 2004, 2003a, c), ECHA (2017a, 2013), Australia 3117 NICNAS (2016, 2015a, b, 2013, 2012, 2010, 2008a, b, c), ATSDR (2022, 2001), EFSA (2019, 2005a, b, 3118 c, d, e), the National Research Council (NRC) (2008), and NASEM (2017). Immune system suppression 3119 has not been identified as a hazard of concern for DIBP or DCHP, or any of the read-across chemicals 3120 by any authoritative or regulatory agencies. However, immune adjuvant effects (i.e., enhanced immune 3121 response) have been identified for several phthalates, including DEHP (U.S. EPA, 2024f), DBP (U.S. 3122 EPA, 2024d), DINP (U.S. EPA, 2025k), and DIDP (U.S. EPA, 2024n). 3123

# 3124 **5.7 Genotoxicity**

Genotoxicity data for DIBP and DCHP, and the read-across chemicals DEHP, BBP, and DBP is
discussed in Sections 3.1 through 3.8 of this document, while genotoxicity data for DINP and DIDP is
summarized in EPA's *Cancer Human Health Hazard Assessment for Diisononyl Phthalate (DINP)* (U.S.
EPA, 2025a) and *Human Health Hazard Assessment for Diisodecyl Phthalate (DIDP)* (U.S. EPA,
2024n). Table 5-5 provides a summary of EPA's conclusions regarding the genotoxicity and
mutagenicity of DIBP and DCHP, and the read-across chemicals DEHP, BBP, DBP, DINP, and DIDP.

- 3131
- As discussed in Sections 3.4 and 3.5 of this document, limited genotoxicity testing of DIBP and DCHP has been conducted. DIBP showed no mutagenic activity in four bacterial reverse mutation assays with
- 3134 or without metabolic activation (Section 3.4), while DCHP showed no mutagenic activity in one
- bacterial reverse mutation assay with or without metabolic activation (Sections 3.5). Other phthalates
- have been evaluated more extensively for genotoxicity in a broader array of *in vitro* and *in vivo* assays.
  Available data for BBP, DINP, and DIDP support the conclusion that these phthalates are not genotoxic
- 3137 Available data for BBP, DINP, and DIDP support the conclusion that these phthalates are not genotoxi 3138 or mutagenic. For DEHP, available data indicate that DEHP and its metabolites are not direct acting
- 3139 mutagens; however, there is some limited evidence that DEHP and its metabolites are not direct acting
- effects such as DNA damage and/or chromosomal aberrations. As noted by ATSDR (2022), these

effects may be secondary to oxidative stress. As discussed in Section 3.3, DBP was positive for

- 3142 mutagenic activity in several *in vitro* mouse lymphoma assays; however, DBP showed no mutagenic
- activity in other *in vitro* bacterial reverse mutation assays or gene mutation assays with *E. coli* and *S.*
- 3144 *cerevisiae*. Further, DBP showed equivocal carcinogenic activity in a two-year bioassay of male SD rats,
- and no evidence of carcinogenic activity in two-year studies of female SD rats or B6C3F1 mice (Section
- 4.3.3). Overall, the weight of evidence indicates that DBP is not a potent genotoxicant but may be
- 3147 weakly genotoxic in some *in vitro* assays.
- 3148

Overall, based on the available genotoxicity data for DIBP and DCHP, and on the genotoxicity data for the read-across chemicals DEHP, BBP, DBP, DINP, and DIDP, EPA does not consider DIBP or DCHP likely to be genotoxic or mutagenic. This conclusion is consistent with other assessments, which have also concluded that phthalate esters as a class are not genotoxic or mutagenic (ECHA, 2017a, b; NICNAS, 2016; U.S. CPSC, 2014).

3154

#### 3155 3156

# Table 5-5. Summary of EPA Conclusions Regarding Genotoxicity and Mutagenicity of Phthalates

| Phthalate | <b>EPA Conclusion (Section or Reference for Additional Information)</b>  |
|-----------|--|
| DEHP      | Evidence indicates that DEHP and its metabolites are not mutagenic. There is some limited evidence that DEHP may be weakly genotoxic inducing effects such as DNA damage and/or chromosomal aberrations. These effects may be secondary to oxidative stress (Section 3.1). |
| BBP       | Not likely to be genotoxic or mutagenic (Section 3.2)  |
| DBP       | Limited evidence that DBP may be weakly genotoxic in some <i>in vitro</i> assays (Section 3.3)   |
| DIBP      | Not likely to be genotoxic or mutagenic (based on read-across) (Section 3.4 and 3.8)   |
| DCHP      | Not likely to be genotoxic or mutagenic (based on read-across) (Section 3.5 and 3.8)   |
| DINP      | Not likely to be genotoxic or mutagenic (U.S. EPA, 2025a) (Section 3.6).   |
| DIDP      | Not likely to be genotoxic or mutagenic (U.S. EPA, $2024n$ ) (Section 3.7).  |

#### 3157

# **5.8 Mechanistic Studies to Support a Proposed Mode of Action**

For the read-across chemicals DEHP and DINP, EPA has concluded that liver tumors observed in
rodents occur through a PPARα MOA (see Section 4.3.1.1.1 for DEHP and Section 4.3.4 and (U.S.
EPA, 2025a) for DINP). Further, for DEHP, EPA has concluded the tumor triad (liver tumors, PACTs,
Leydig cell tumors) in rats is related to PPARα activation following chronic exposure to DEHP and
some hypolipidemic drugs (discussed in Sections 4.3.1.1.4–4.3.1.1.6).

3164

3165 In addition to DEHP and DINP, comparative *in vivo* and *in vitro* studies have also consistently

demonstrated that the read-across chemicals BBP, DBP, and DIDP, can also activate PPARa. For

example, Barber et al. (<u>1987</u>) demonstrate that DEHP, BBP, DBP, DINP, and DIDP, can all activate

3168 PPARα in the livers of male F344 rats exposed to each phthalate in the diet for 21 days. Compared to

3169 hypolipidemic drugs, all five phthalates were found to be relatively weak PPARα activators based on

induction of hepatic palmitoyl CoA oxidase activity, although DEHP, DINP, and DIDP were found to be
 stronger PPARα activators than BBP and DBP (Table 5-6). Similarly, Bility et al. (2004) demonstrated

3172 that monoester metabolites of DEHP, BBP, DBP, DINP, and DIDP, can activate both mouse and human PPARa in vitro, however, for all five monoester metabolites, human PPARa was less sensitive to 3173 3174 activation than mouse PPARa (Table 5-6). Notably, similar trends in potency for PPARa activation 3175 were observed *in vitro* with mouse PPAR $\alpha$  as were observed *in vivo* with studies of rats (*i.e.*, DIDP  $\approx$ 3176 DINP > DEHP >> BBP  $\approx$  DBP) (Table 5-6). Further, the two weakest PPAR $\alpha$  activators (*i.e.*, BBP and 3177 DBP) in vivo and in vitro did not induce liver tumors in chronic studies of rats or mice. 3178 3179 As discussed in Section 3.3 of the Draft Non-Cancer Human Health Hazard Assessment for Dicyclohexyl Phthalate (DCHP) (U.S. EPA, 2024e), only one study of DCHP was identified by EPA 3180 that evaluated PPAR $\alpha$  activation. Briefly, Saillenfait et al. (2009) gavaged pregnant SD rats with 0, 250, 3181 3182 500, and 750 mg/kg-day DCHP on GDs 6 through 20 and sacrificed dams on GD 21. Maternal hepatic palmitoyl CoA oxidase activity (a biomarker for PPARa activation) increased 75 to 108 percent at 250 3183 mg/kg-day and above, indicative of a weak induction of PPAR $\alpha$  activation, while relative liver weight 3184 increased 23 to 35 percent at 500 mg/kg-day and above. Several additional repeat-dose oral exposure 3185 studies of DCHP with rats provide additional indirect evidence consistent with PPAR $\alpha$  activation in the 3186

studies of DCHP with rats provide additional indirect evidence consistent with PPARα activation in the
liver, including increases in relative liver weight and hepatocellular hypertrophy (<u>Ahbab et al., 2017;</u>
<u>Saillenfait et al., 2009; Yamasaki et al., 2009; Hoshino et al., 2005; Lake et al., 1982</u>).

EPA did not identify any *in vivo* or *in vitro* studies that directly evaluated PPARα activation following
exposure to DIBP. However, as discussed by Yost et al. (2019), there is robust evidence that oral
exposure to DIBP can increase relative liver weight in repeat-dose oral exposure studies of rats and
mice. Although not direct evidence, increased relative liver weight is consistent with PPARα activation.

3195 3196

| Table 5-6. Comparativ | e Analysis of PPARα | Activation by DIDP | , DINP | , DEHP, | , BBP, | and DBP |
|-----------------------|---------------------|--------------------|--------|---------|--------|---------|
|                       |                     |                    |        |         |        |         |

| Parent Phthalate<br>(Metabolite)  | <i>In vivo</i> Induction of<br>Hepatic Palmitoyl<br>CoA Oxidase<br>Activity <sup><i>ab</i></sup> (Barber<br><u>et al., 1987</u> ) | Lowest In Vitro Activation<br>Concentration for Mouse<br>PPARα (Maximal fold-<br>induction) <sup>c</sup> (Bility et al.,<br><u>2004</u> ) | Lowest In Vitro Activation<br>Concentration for Human<br>PPARα (Maximal fold-<br>induction) <sup>c</sup> (Bility et al.,<br><u>2004</u> ) |  |  |
|---|---|---|---|--|--|
| DEHP (mono(2-<br>ethylhexyl) phthalate)                                   | 15  | 10 µM (11.1)  | 30 µM (4.8)   |  |  |
| BBP (monobenzyl phthalate)  | 2   | 100 µM (12.3)   | 200 µM (2.5)  |  |  |
| DBP (monobutyl phthalate)   | 3   | 100 µM (3.7)  | 200 µM (2.4)  |  |  |
| DINP (monoisononyl phthalate)   | 11  | 3 µM (27.1)   | 10 µM (5.8)   |  |  |
| DIDP (monoisodecyl phthalate)   | 17  | 3 µM (26.9)   | 30 µM (3.9)   |  |  |
| <sup><i>a</i></sup> Units: [(nmoles/min/mg)/ $\mu$ moles/kg/day)] × 10E–3 |   |   |   |  |  |

<sup>b</sup> Based on dosing with parent phthalate.

<sup>c</sup> Based on exposure to metabolite of parent phthalate.

3197

# 5.9 Evidence of Chronic Toxicity and Carcinogenicity From Read-Across to Related Chemicals

3200 No chronic toxicity or carcinogenicity studies of DIBP or DCHP are available. Chronic toxicity and 3201 carcinogenicity studies are available for the five read-across chemicals DEHP, BBP, DBP, DINP, and 3202 DIDP. For the these read-across chemicals, EPA has consistently identified developmental toxicity as a 3203 more sensitive and robust outcome for characterizing risk to human health from acute, intermediate, and 3204 chronic exposures. This is demonstrated by the PODs selected by EPA to characterize risk to human 3205 health for these durations (Table 5-7). The only exception to this is for DINP, in which non-cancer liver 3206 effects observed in a two-year dietary study of F344 rats were identified as a more sensitive and relevant 3207 effect for setting the chronic POD compared to developmental toxicity (Table 5-7).

3208

3209 Further, although available carcinogenicity data support differing cancer classifications for the read-3210 across chemicals DEHP, BBP, DBP, DINP, and DIDP (summarized in Table 5-8), EPA has determined 3211 that quantitative cancer risk assessment is not needed for the read-across phthalates.. For DIDP, EPA has 3212 concluded that DIDP is Not Likely to be Carcinogenic to Humans and cancer risk was not quantitatively 3213 evaluated (Section 4.3.5 and (U.S. EPA, 2024n)). For both BBP and DBP, EPA has preliminarily 3214 concluded that there is Suggestive Evidence of Carcinogenic Potential in rodents based on increased incidence of PACTs in rats (see Sections 4.3.2.4 and 4.3.3.3). According to the Guidelines for 3215 3216 Carcinogen Risk Assessment (U.S. EPA, 2005), when there is Suggestive Evidence "the Agency 3217 generally would not attempt a dose-response assessment, as the nature of the data generally would not 3218 support one." Consistently, EPA did not conduct a cancer dose-response assessment for BBP or DBP 3219 and did not quantitatively evaluate either phthalate for carcinogenic risk. For DEHP (Section 4.3), 3220 treatment-related increases in hepatocellular adenomas and/or carcinomas have been observed in rats 3221 and mice of both sexes, while treatment-related increases in PACTs and Leydig cell tumors have been 3222 observed in male rats. As discussed in Section 4.3.1.1, EPA has preliminarily concluded that these tumor 3223 types are related to PPARα activation, and EPA has preliminarily concluded that DEHP is Not Likely to 3224 be Carcinogenic to Humans at doses below levels that do not result in PPAR $\alpha$  activation. Finally, for 3225 DINP (Section 4.3.4), treatment-related increases in hepatocellular adenomas and/or carcinomas have 3226 been consistently observed in rats and mice of both sexes. EPA has previously concluded that DINP causes liver tumors in rodents through a PPARa MOA (U.S. EPA, 2025a). Notably, this conclusion was 3227 3228 supported by the SACC during the July 2024 peer review meeting (U.S. EPA, 2024q). EPA further 3229 concluded that DINP is Not Likely to be Carcinogenic to Humans at doses below levels that do not result 3230 in PPAR $\alpha$  activation (U.S. EPA, 2025a). Further, for both DINP and DEHP, the non-cancer PODs based 3231 on developmental toxicity (DEHP) or non-cancer liver toxicity (DINP) are lower than the hazard values 3232 for PPAR $\alpha$  activation identified by EPA. Therefore, EPA has preliminarily concluded that the non-3233 cancer PODs for DEHP and DINP are expected to adequately account for all chronic toxicity, including 3234 carcinogenicity.
Table 5-7. Summary of Non-cancer PODs Selected for Use in Human Health Risk Characterization for DCHP, DIBP, DEHP, DBP,
 BBP, DINP, and DIDP

| Phthalate | Relevant<br>Exposure<br>Scenario(s) | Target Organ System   | POD (HED)<br>(mg/kg-day)        | Benchmark<br>MOE                                       | Effect   | Reference                            |
|-----------|-------------------------------------|---|---------------------------------|--|--|--------------------------------------|
| DEHP      | Acute,<br>intermediate,<br>chronic  | Developing male<br>reproductive system<br>(phthalate syndrome-related<br>effects) | NOAEL = 4.8<br>(1.1)            | $UF_{A} = 3^{a}$ $UF_{H} = 10$ $Total UF = 30$         | ↑ total reproductive tract<br>malformations in F1 and F2<br>offspring  | ( <u>U.S. EPA,</u><br><u>2024f</u> ) |
| BBP       | Acute,<br>intermediate,<br>chronic  | Developing male<br>reproductive system<br>(phthalate syndrome-related<br>effects) | NOAEL = 50<br>(12)              | $UF_{\rm A} = 3^{a}$ $UF_{\rm H} = 10$ $Total UF = 30$ | Phthalate syndrome-related<br>effects ( <i>e.g.</i> , $\downarrow$ AGD; $\downarrow$ fetal<br>testicular testosterone; $\downarrow$<br>reproductive organ weights;<br>Leydig cell effects)     | ( <u>U.S. EPA,</u><br><u>2024c</u> ) |
| DBP       | Acute,<br>intermediate,<br>chronic  | Developing male<br>reproductive system<br>(phthalate syndrome-related<br>effects) | BMDL <sub>5</sub> = 9<br>(2.1)  | $UF_{A} = 3^{a}$ $UF_{H} = 10$ $Total UF = 30$         | ↓ fetal testicular testosterone<br>in rats   | ( <u>U.S. EPA,</u><br><u>2024d</u> ) |
| DIBP      | Acute,<br>intermediate,<br>chronic  | Developing male<br>reproductive system<br>(phthalate syndrome-related<br>effects) | BMDL <sub>5</sub> = 24<br>(5.7) | $UF_{A} = 3^{a}$ $UF_{H} = 10$ $Total UF = 30$         | $\downarrow$ <i>ex vivo</i> fetal testicular<br>testosterone production in rats  | ( <u>U.S. EPA,</u><br><u>2024g</u> ) |
| DCHP      | Acute,<br>intermediate,<br>chronic  | Developing male<br>reproductive system<br>(phthalate syndrome-related<br>effects) | NOAEL = 10<br>(2.4)             | $UF_{A} = 3^{a}$ $UF_{H} = 10$ $Total UF = 30$         | Phthalate syndrome-related<br>effects ( <i>e.g.</i> , ↓ fetal testicular<br>testosterone; ↓AGD; Leydig<br>cell effects; ↓ mRNA and/or<br>protein expression of<br>steroidogenic genes; ↓INSL3) | ( <u>U.S. EPA,</u><br><u>2024e</u> ) |
| DINP      | Acute,<br>intermediate              | Developing male<br>reproductive system<br>(phthalate syndrome-related<br>effects) | BMDL <sub>5</sub> = 49<br>(12)  | $UF_A = 3^a$ $UF_H = 10$ $Total UF = 30$               | ↓ fetal testicular testosterone<br>in rats   | ( <u>U.S. EPA,</u><br><u>2025k</u> ) |

#### May 2025

| Phthalate   | Relevant<br>Exposure<br>Scenario(s) | Target Organ System  | POD (HED)<br>(mg/kg-day) | Benchmark<br>MOE                                       | Effect   | Reference                            |
|---|-------------------------------------|--|--------------------------|--|--|--------------------------------------|
|   | Chronic                             | Liver Toxicity   | NOAEL = 15<br>(3.5)      | $UF_{\rm A} = 3^{a}$ $UF_{\rm H} = 10$ $Total UF = 30$ | <ul> <li>↑ liver weight, ↑ serum</li> <li>chemistry, histopathology</li> <li>(<i>e.g.</i>, focal necrosis,</li> <li>spongiosis hepatis)</li> </ul> |                                      |
| DIDP  | Acute,<br>intermediate,<br>chronic  | Developmental toxicity<br>(decreased F2 offspring<br>survival) | NOAEL = 38<br>(9.0)      | $UF_{\rm A} = 3^a$ $UF_{\rm H} = 10$ $Total UF = 30$   | Reduced F2 offspring survival<br>on PND1 and PND4  | ( <u>U.S. EPA,</u><br><u>2024n</u> ) |
| <sup><i>a</i></sup> EPA used allometric body weight scaling to the three-quarters power to derive human equivalent doses (HEDs). Consistent with EPA Guidance (U.S. EPA, 2011), the UF <sub>A</sub> was reduced from 10 to 3. |                                     |  |                          |  |  |                                      |

3237 3238

### 3239 **Table 5-8. Summary of Cancer Classifications for DEHP, BBP, DBP, DINP, and DIDP**

| Phthalate | EPA Cancer Classification (Section or Reference for Additional Information)   |  |
|-----------|---|--|
| DEHP      | <i>Not Likely to be Carcinogenic to Humans</i> at doses below levels that do not result in PPARα activation (Section 4.3.1.4) (Draft)                 |  |
| BBP       | Suggestive Evidence of Carcinogenic Potential (Section 4.3.2.4) (Draft)   |  |
| DBP       | Suggestive Evidence of Carcinogenic Potential (Section 4.3.3.3) (Draft)   |  |
| DINP      | <i>Not Likely to be Carcinogenic to Humans</i> at doses below levels that do not result in PPARα activation (U.S. EPA, 2025a) (Section 4.3.4) (Final) |  |
| DIDP      | Not Likely to Be Carcinogenic to Humans (U.S. EPA, 2024n) (Section 4.3.5) (Final)   |  |

3240

# 5.10 Weight of Scientific Evidence Conclusions Regarding Carcinogenicity of DIBP and DCHP Based on Read-across

3243 Based on the weight of scientific evidence, EPA preliminarily concludes that the lack of chronic toxicity 3244 and carcinogenicity bioassays for DIBP and DCHP do not suggest that there are significant remaining 3245 scientific uncertainties in the qualitative and quantitative risk characterization for either of these 3246 phthalates. Further, EPA has preliminarily concluded that the non-cancer PODs for DIBP and DCHP 3247 based on effects on the developing male reproductive system consistent with a disruption of androgen 3248 action and phthalate syndrome that were selected for characterizing risk from acute, intermediate and 3249 chronic exposure to DIBP and DCHP are health-protective PODs, including for PESS. These 3250 conclusions are based on the following weight of scientific evidence considerations:

- The toxicological profiles of DCHP, DIBP, and five read-across chemicals (DEHP, BBP, DBP, DINP, and DIDP) were evaluated (Section 5).
- Following oral exposure, phthalates are rapidly absorbed, metabolized, systemically distributed and excreted in urine, and to a lesser extent feces. Studies of rodents and humans have demonstrated near complete excretion within 72 to 96 hours. Based on the rapid elimination kinetics, phthalates are not considered bioaccumulative (Section 5.2).
- DIBP, DCHP and the five read-across chemicals DEHP, BBP, DBP, DINP, and DIDP are not considered to be direct-acting genotoxicants or mutagens (Section 5.7).
- There is no evidence for immune suppression in experimental animal studies of DIBP, DCHP 3260 and the five read-across chemicals DEHP, BBP, DBP, DINP, and DIDP (Section 5.6).
- DIBP and DCHP, and four of the five read-across chemicals (DEHP, BBP, DBP, DINP, but not DIDP), are antiandrogenic and can disrupt fetal testicular testosterone biosynthesis in rats leading to a spectrum of effects on the developing male reproductive system consistent with phthalate syndrome (Section 5.4).
- Intermediate and subchronic duration studies identify the liver as a target organ of phthalate toxicity, including for DIBP, DCHP, and the five read-across phthalates (DEHP, BBP, DBP, DINP, and DIDP). Evidence of PPARα activation in the liver is also apparent (Sections 5.5 and 5.8).
- 3269 Of the five read-across phthalates (DEHP, BBP, DBP, DINP, DIDP) that have chronic toxicity • 3270 studies, in only one case (DINP) did a chronic toxicity study support a more sensitive POD for 3271 use in risk characterization than a POD derived from developmental toxicity studies. For DIDP, 3272 developmental toxicity (decreased F2 offspring survival) was identified as the most sensitive 3273 outcome and was used in characterize risk from acute, intermediate, and chronic duration 3274 exposures. For DEHP, BBP, DBP, DIBP, and DCHP, effects on the developing male 3275 reproductive system consistent with a disruption of androgen action were identified as the most 3276 sensitive and robust outcomes for use in risk characterization for acute, intermediate, and chronic 3277 exposure scenarios. For DINP, antiandrogenic effects were the most sensitive outcome for acute 3278 and intermediate exposure durations, while non-cancer liver effects were identified as the most sensitive effect for chronic exposure durations. 3279
- Although available carcinogenicity data support differing cancer classifications for the readacross chemicals DEHP, BBP, DBP, DINP, and DIDP (see Table 5-8), EPA has determined that quantitative cancer risk assessment is not needed for the read-across phthalates (Section 5.9).

3283 EPA has concluded that DIDP is Not Likely to Be Carcinogenic to Humans, while there is ٠ 3284 Suggestive Evidence of Carcinogenic Potential for BBP and DBP based on increased incidence 3285 of PACTs in rats. For DBP, BBP, and DIDP, EPA did not quantitatively evaluate cancer risk. 3286 For DEHP and DINP, EPA concluded that these phthalates are *Not Likely to be Carcinogenic to* Humans at doses below levels that do not result in PPARa activation. For both DEHP and DINP, 3287 3288 non-cancer PODs based on developmental toxicity (DEHP) or non-cancer liver toxicity (DINP) are lower than the hazard values for PPARa activation, and EPA has concluded that the non-3289 3290 cancer PODs for DEHP and DINP are expected to adequately account for all chronic toxicity, including carcinogenicity (Section 5.9). 3291

### 3292 6 CONCLUSIONS AND NEXT STEPS

3293 Available studies indicate that DEHP, BBP, DBP, DIBP, DCHP, DINP, and DINP are not direct acting 3294 genotoxicants or mutagens (Section 2). Cancer bioassays are available for DEHP, BBP, DBP, DINP and 3295 DIDP. EPA has previously concluded that DIDP is Not Likely to be Carcinogenic to Humans (U.S. 3296 EPA, 2024n). Herein, EPA has preliminarily concluded that there is Suggestive Evidence of 3297 Carcinogenic Potential of BBP and DBP in rodents based on evidence of pancreatic acinar cell 3298 adenomas in rats (Sections 4.3.2.4 and 4.3.3.3). According to the Guidelines for Carcinogen Risk 3299 Assessment (U.S. EPA, 2005), when there is Suggestive Evidence "the Agency generally would not 3300 attempt a dose-response assessment, as the nature of the data generally would not support one." 3301 Consistently, EPA did not conduct a dose-response assessment for BBP or DBP and did not 3302 quantitatively evaluate either phthalate for carcinogenic risk to human health. 3303 3304 For DINP (Section 4.3.4), treatment-related increases in hepatocellular adenomas and/or carcinomas 3305 have been consistently observed in rats and mice of both sexes. EPA has previously concluded that DINP causes liver tumors in rodents through a PPARa MOA (U.S. EPA, 2025a). Notably, this 3306 3307 conclusion was supported by the SACC during their July 2024 peer review meeting (U.S. EPA, 2024q). 3308 Further, EPA has previously concluded that DINP is Not Likely to be Carcinogenic to Humans at doses 3309 below levels that do not result in PPARa activation and that the non-cancer POD based on liver toxicity 3310 will adequately account for all chronic toxicity, including carcinogenicity, which could potentially result 3311 from exposure to DINP (U.S. EPA, 2025a). For DEHP (Section 4.3), treatment-related increases in 3312 hepatocellular adenomas and/or carcinomas have been observed in rats and mice of both sexes, while 3313 treatment-related increases in PACTs and Levdig cell tumors have been observed in male rats. As 3314 discussed in Section 4.3.1.1, EPA has preliminarily concluded that these tumor types, sometimes 3315 referred to as the 'tumor triad,' are related to PPAR $\alpha$  activation. This conclusion is in part informed by 3316 inferences from hypolipidemic drugs that lower lipid-levels in humans by activating PPARa, and also 3317 induce the tumor triad in rats, but not humans (Section 4.3.1.1.4). For DEHP, EPA has preliminarily 3318 concluded that DEHP is Not Likely to be Carcinogenic to Humans at doses below levels that do not 3319 result in PPARa activation. Further, for DEHP, the non-cancer POD based on developmental toxicity is 3320 lower than the hazard values for PPARα activation identified by EPA. Therefore, EPA has concluded that the non-cancer PODs for DEHP is expected to adequately account for all chronic toxicity, including 3321 3322 carcinogenicity, and cancer risk was not further quantified. 3323

3324 No chronic toxicity or cancer bioassays are available for DIBP or DCHP. Herein, EPA used elements of 3325 the ReCAAP weight of evidence framework as an organizational tool to evaluate the extent to which the 3326 lack of carcinogenicity studies imparts significant uncertainty on the human health risk assessments for 3327 DIBP and DCHP (Section 5). Human health hazards and toxicokinetic properties of DIBP and DCHP 3328 were evaluated and compared to DEHP, DBP, BBP, DINP, and DIDP. Overall, based on the weight of 3329 scientific evidence, EPA preliminarily concludes that the lack of chronic toxicity and carcinogenicity 3330 bioassays for DIBP and DCHP do not suggest that there are significant remaining scientific uncertainties 3331 in the qualitative and quantitative risk characterization for either of these phthalates. Further, EPA has 3332 preliminarily concluded that the proposed non-cancer PODs for DIBP and DCHP are health-protective, 3333 including for PESS. These proposed PODs for DIBP and DCHP are based on effects on the developing 3334 male reproductive system consistent with a disruption of androgen action and phthalate syndrome that 3335 were selected for characterizing risk from acute, intermediate and chronic exposure to DIBP and DCHP. 3336 These preliminary conclusions are based on several key weight of scientific evidence considerations 3337 (discussed in Section 5). First, for the five read-across phthalates, effects on the developing male 3338 reproductive system consistent with a disruption of androgen action and phthalate syndrome is a more

3339 sensitive and robust endpoint for deriving PODs for use in characterizing risk for acute, intermediate,

- and chronic exposure scenarios than PPAR $\alpha$  mediated effects on the liver. The one exception to this was for DINP, in which chronic non-cancer liver effects were identified as a more sensitive outcome than
- developmental toxicity for deriving a chronic POD. Second, EPA has determined that quantitative
- 3343 cancer risk assessment is not needed for the read-across phthalates.
- 3344

3345 EPA is soliciting comments from the SACC and the public on its preliminary cancer classifications for

- 3346 DEHP, DBP, and BBP; and its conclusion that lack of chronic toxicity and carcinogenicity studies are
- not a significant source of scientific uncertainty for DCHP or DIBP. EPA's conclusions pertaining to the
- genotoxicity and carcinogenicity of DIDP and DINP received favorable peer-reviews by the SACC
   during the July 2024 peer review meeting (U.S. EPA, 2024q). Therefore, EPA is not requesting
- additional SACC peer-review pertaining to the human health hazards of DIDP or DINP in 2025.

### 3351 **REFERENCES**

| 3352         | <u>Abe, S; Sasaki, M. (1977)</u> . Chromosome aberrations and sister chromatid exchanges in Chinese hamster   |
|--------------|---|
| 3353         | cells exposed to various chemicals. J Natl Cancer Inst 58: 1635-1641.   |
| 3354         | http://dx.doi.org/10.1093/jnci/58.6.1635  |
| 3355         | ACC HPP. (2019a). Manufacturer request for risk evaluation Di-isodecyl Phthalate (DIDP). American   |
| 3356         | Chemistry Council High Phthalates Panel.  |
| 3357         | ACC HPP. (2019b). Manufacturer Request for Risk Evaluation Di-isononyl Phthalate (DINP).  |
| 3358         | (730R19001). American Chemistry Council High Phthalates Panel :: ACC HPP  |
| 3359         | https://nepis.epa.gov/exe/ZyPURL.cgi?Dockey=P100YEGF.txt  |
| 3360         | Agarwal, DK; Lawrence, WH; Nunez, LJ; Autian, J. (1985). Mutagenicity evaluation of phthalic acid   |
| 3361         | esters and metabolites in Salmonella typhimurium cultures. J Toxicol Environ Health 16: 61-69.  |
| 3362         | http://dx.doi.org/10.1080/15287398509530719   |
| 3363         | Ahbab, MA; Güven, C; Koçkaya, EA; Barlas, N. (2017). Comparative developmental toxicity   |
| 3364         | evaluation of di- n-hexyl phthalate and dicyclohexyl phthalate in rats. Toxicol Ind Health 33:  |
| 3365         | 696-716. http://dx.doi.org/10.1177/0748233717711868   |
| 3366         | Ahern, TP; Broe, A; Lash, TL; Cronin-Fenton, DP; Ulrichsen, SP; Christiansen, PM; Cole, BF; Tamimi,   |
| 3367         | RM; Sørensen, HT; Damkier, P. (2019). Phthalate Exposure and Breast Cancer Incidence: A   |
| 3368         | Danish Nationwide Cohort Study. J Clin Oncol 37: 1800-1809.   |
| 3369         | http://dx.doi.org/10.1200/JCO.18.02202  |
| 3370         | Ahmed, RS; Price, SC; Grasso, P; Hinton, RH. (1989). Effects of intermittent feeding of rats with di-2-   |
| 3371         | ethylhexylphthalate [Abstract]. Biochem Soc Trans 17: 1073-1074.  |
| 3372         | http://dx.doi.org/10.1042/bst0171073  |
| 3373         | Al-Saleh, I; Coskun, S; Al-Doush, I; Al-Rajudi, T; Al-Rouqi, R; Abduljabbar, M; Al-Hassan, S. (2019).   |
| 3374         | Exposure to phthalates in couples undergoing in vitro fertilization treatment and its association   |
| 3375         | with oxidative stress and DNA damage. Environ Res 169: 396-408.   |
| 3376         | http://dx.doi.org/10.1016/j.envres.2018.11.018  |
| 3377         | Albro, PW; Corbett, JT; Schroeder, JL; Jordan, S; Matthews, HB. (1982). Pharmacokinetics, interactions  |
| 3378         | with macromolecules and species differences in metabolism of DEHP. Environ Health Perspect  |
| 3379         | 45: 19-25. https://heronet.epa.gov/heronet/index.cfm/reference/download/reference_id/673560   |
| 3380         | Anderson, SP; Cattley, RC; Corton, JC. (1999). Hepatic expression of acute-phase protein genes during   |
| 3381         | carcinogenesis induced by peroxisome proliferators. Mol Carcinog 26: 226-238.   |
| 3382         | http://dx.doi.org/10.1002/(SICI)1098-2744(199912)26:4<226::AID-MC2>3.0.CO;2-Q   |
| 3383         | <u>Aoyama, T; Peters, JM; Iritani, N; Nakajima, T; Furihata, K; Hashimoto, T; Gonzalez, FJ</u> . (1998).  |
| 3384         | Altered constitutive expression of fatty acid-metabolizing enzymes in mice lacking the  |
| 3385         | peroxisome proliferator-activated receptor alpha (PPARalpha). J Biol Chem 273: 5678-5684.   |
| 3386         | http://dx.doi.org/10.1074/jbc.273.10.5678   |
| 3387         | Ashby, J; Tinwell, H; Lefevre, PA; Odum, J; Paton, D; Millward, SW; Tittensor, S; Brooks, AN. (1997).   |
| 3388         | Normal sexual development of rats exposed to butyl benzyl phthalate from conception to  |
| 3389         | weaning. Regul Toxicol Pharmacol 26: 102-118. <u>http://dx.doi.org/10.1006/rtph.1997.1159</u>   |
| 3390         | Astill, B; Barber, E; Lington, A; Moran, E; Mulholland, A; Robinson, E; Scheider, B. (1986). Chemical   |
| 3391         | industry voluntary test program for phthalate esters: Health effects studies. Environ Health  |
| 3392         | Perspect 65: $329-336$ . <u>http://dx.doi.org/10.230//3430200</u><br>ATSDD (2001) Tarriada sized and file for dia a bated abthelate (Us date. Southersher 2001) [ATSDD Tarr |
| 2204         | ATSUK. (2001). Toxicological profile for al-n-butyl phthalate (Update, September 2001) [ATSUK Tox   |
| 3394<br>2205 | Promej. Auania, GA: U.S. Department of Health and Human Services, Public Health Service.  |
| 3393<br>2204 | <u>Intp://www.atsur.cuc.gov/toxprofiles/tp.asp/id=859&amp;tid=16/</u><br>ATSDD (2022) Toxicological profile for di(2 athylboyyl) white late (DEUD) [ATSDD Tox Profile]      |
| 2207         | <u>AISUK. (2022)</u> . Toxicological profile for di (2-einyinexyi)phthalate (DEHP) [AISUK Tox Profile].   |
| 222/         | (US2/412/-A). Attanta, UA. <u>https://www.atsdr.cdc.gov/10xProfiles/tp9.pdf</u>   |

| 3398         | Autian, J. (1982). Antifertility effects and dominant lethal assays for mutagenic effects of DEHP.  |
|--------------|---|
| 3399         | Environ Health Perspect 45: 115-118. http://dx.doi.org/10.1289/ehp.8245115  |
| 3400         | Barakat, R; Lin, PC; Park, CJ; Best-Popescu, C; Bakry, HH; Abosalem, ME; Abdelaleem, NM; Flaws,   |
| 3401         | JA; Ko, C. (2018). Prenatal Exposure to DEHP Induces Neuronal Degeneration and  |
| 3402         | Neurobehavioral Abnormalities in Adult Male Mice. Toxicol Sci 164: 439-452.   |
| 3403         | http://dx.doi.org/10.1093/toxsci/kfy103   |
| 3404         | Barber, ED; Astill, BD; Moran, EJ; Schneider, BF; Gray, TJB; Lake, BG; Evans, JG. (1987).   |
| 3405         | Peroxisome induction studies on seven phthalate esters. Toxicol Ind Health 3: 7-24.   |
| 3406         | http://dx.doi.org/10.1177/074823378700300203  |
| 3407         | Barber, ED; Cifone, M; Rundell, J; Przygoda, R; Astill, BD; Moran, E; Mulholland, A; Robinson, E;   |
| 3408         | Schneider, B. (2000). Results of the L5178Y mouse lymphoma assay and the Balb/3t3 cell in   |
| 3409         | vitro transformation assay for eight phthalate esters. J Appl Toxicol 20: 69-80.  |
| 3410         | http://dx.doi.org/10.1002/(SICI)1099-1263(200001/02)20:1<69::AID-JAT630>3.0.CO;2-2  |
| 3411         | Barlow, NJ; McIntyre, BS; Foster, PM. (2004). Male reproductive tract lesions at 6, 12, and 18 months   |
| 3412         | of age following in utero exposure to di(n-butyl) phthalate. Toxicol Pathol 32: 79-90.  |
| 3413         | http://dx.doi.org/10.1080/01926230490265894   |
| 3414         | BASF. (1990). Cytogenic study in vivo of dibutylphthalat in mice micronucleus test: Single oral   |
| 3415         | administration. (26M0449/894382). Ludwigshafen, Germany.  |
| 3416         | BIBRA. (1985). Rat liver and lipid effects of representative phthalate esters. (EPA/OTS Doc   |
| 3417         | #40+8526207). Surrey, United Kingdom: British Industrial Biological Research Association.   |
| 3418         | BIBRA. (1990). An investigation of the effect of di-(2-ethylhexyl) phthalate on rat hepatic peroxisomes   |
| 3419         | with cover letter [TSCA Submission]. (86-91000007729). Houston, TX: Exxon Chemical  |
| 3420         | Americas. https://ntrl.ntis.gov/NTRL/dashboard/searchResults/titleDetail/OTS0530399.xhtml   |
| 3421         | Biegel, LB; Hurtt, ME; Frame, SR; O'Connor, JC; Cook, JC. (2001). Mechanisms of extrahepatic tumor  |
| 3422         | induction by peroxisome proliferators in male CD rats. Toxicol Sci 60: 44-55.   |
| 3423         | http://dx.doi.org/10.1093/toxsci/60.1.44  |
| 3424         | Bility, MT; Thompson, JT; McKee, RH; David, RM; Butala, JH; Vanden Heuvel, JP; Peters, JM.  |
| 3425         | (2004). Activation of mouse and human peroxisome proliferator-activated receptors (PPARs) by  |
| 3426         | phthalate monoesters. Toxicol Sci 82: 170-182. <u>http://dx.doi.org/10.1093/toxsci/kth253</u>   |
| 3427         | <u>Bio/dynamics. (1987)</u> . A chronic toxicity carcinogenicity feeding study in rats with Santicizer 900 with                                     |
| 3428         | cover letter dated 06/05/8/ [TSCA Submission]. (EPA/OTS Doc #868/0000362). St. Louis,   |
| 3429         | MO: Monsanto Company.   |
| 3430         | https://ntrl.ntis.gov/NTRL/dashboard/searchResults/titleDetail/OTS05131/2.xhtml   |
| 3431         | Bishop, JB; Teat, CM; Bhoosan, B. (1987). Assessment of fetal death rate among in utero progeny of  |
| 3432         | BoC3F1 and CD-1 mice after subcutaneous injections of males with butyl benzyl phthalate   |
| 3433         | (BBP) [Abstract]. Environ Mutagen 9: 15.  |
| 3434         | Boerrigter, ME. (2004). Mutagenicity of the peroxisome proliferators clothorate, wyeth 14,643 and di-2-   |
| 3435         | Consing serversis 2: 7, http://dx.doi.org/10.1186/1477.2162.2.7   |
| 3430         | Carcinogenesis 3: 7. <u>http://dx.doi.org/10.1186/14/7-5165-5-7</u>   |
| 3437         | Bonovas, S; Nikolopoulos, GK; Bagos, PG. (2012). Use of fibrates and cancer fisk: a systematic review   |
| 3438         | and meta-analysis of 1 / long-term randomized placebo-controlled trials [Review]. PLoS ONE /:   |
| 2439         | Prin A.E. Nuclea, A. Hassman, IV, Salla, DM: Jahiran, MD: Walliam, NJ. (2005). Insidenses of salasted   |
| 344U<br>2441 | DIIX, AE, NYSKA, A; HASEIHAH, JK; SEHS, DIVI; JOKIHEH, MP; WAIKEF, NJ. (2003). Incidences of selected   |
| 3441<br>3442 | National Toxicology Program Toxicol Pathol 22: 477 492  |
| 3442<br>3112 | 1000000000000000000000000000000000000   |
| 3443<br>3444 | <u>Iup.//ux.u01.01g/10.1000/01720230370701030</u><br>Busser MT: Lutz WK (1987) Stimulation of DNA synthesis in rat and mouse liver by verious tymer |
| 3444<br>2445 | promotors Corrinogonosis 8: 1422 1427 http://dx.doi.org/10.1002/corrin/9.10.1422  |
| J44J         | promoters. Carcinogenesis 6. 1455-1457. <u>http://ux.doi.org/10.1095/carcin/6.10.1455</u>   |

| 3446 | Butterworth, BE. (1984). The genetic toxicology of di(2-ethylhexyl)phthalate (DEHP). CIIT Activities    |
|------|---|
| 3447 | 4: 1-8.   |
| 3448 | Butterworth, BE; Bermudez, E; Smith-Oliver, T; Earle, L; Cattley, R; Martin, J; Popp, JA; Strom, S;     |
| 3449 | Jirtle, R; Michalopoulos, G. (1984). Lack of genotoxic activity of di(2-ethylhexyl)phthalate            |
| 3450 | (DEHP) in rat and human hepatocytes. Carcinogenesis 5: 1329-1335.                                       |
| 3451 | http://dx.doi.org/10.1093/carcin/5.10.1329  |
| 3452 | Caldwell, DJ. (1999). Review of mononuclear cell leukemia in F-344 rat bioassays and its significance   |
| 3453 | to human cancer risk: A case study using alkyl phthalates [Review]. Regul Toxicol Pharmacol             |
| 3454 | 30: 45-53. http://dx.doi.org/10.1006/rtph.1999.1305   |
| 3455 | Caldwell, DJ; Eldridge, SR; Lington, AW; McKee, RH. (1999). Retrospective evaluation of alpha 2u-       |
| 3456 | globulin accumulation in male rat kidneys following high doses of diisononyl phthalate. Toxicol         |
| 3457 | Sci 51: 153-160. http://dx.doi.org/10.1093/toxsci/51.1.153  |
| 3458 | Cattley, RC; Conway, JG; Popp, JA. (1987). Association of persistent peroxisome proliferation and       |
| 3459 | oxidative injury with hepatocarcinogenicity in female F-344 rats fed di(2-ethylhexyl)phthalate          |
| 3460 | for 2 years. Cancer Lett 38: 15-22. http://dx.doi.org/10.1016/0304-3835(87)90195-9                      |
| 3461 | Cattley, RC; Glover, SE. (1993). Elevated 8-hydroxydeoxyguanosine in hepatic DNA of rats following      |
| 3462 | exposure to peroxisome proliferators: Relationship to carcinogenesis and nuclear localization.          |
| 3463 | Carcinogenesis 14: 2495-2499. http://dx.doi.org/10.1093/carcin/14.12.2495                               |
| 3464 | Cattley, RC; Richardson, KK; Smith-Oliver, T; Popp, JA; Butterworth, BE. (1986). Effect of              |
| 3465 | peroxisome proliferator carcinogens on unscheduled DNA synthesis in rat hepatocytes                     |
| 3466 | determined by autoradiography. Cancer Lett 33: 269-277. http://dx.doi.org/10.1016/0304-                 |
| 3467 | 3835(86)90066-2   |
| 3468 | Cattley, RC; Smith-Oliver, T; Butterworth, BE; Popp, JA. (1988). Failure of the peroxisome proliferator |
| 3469 | WY-14,643 to induce unscheduled DNA synthesis in rat hepatocytes following in vivo                      |
| 3470 | treatment. Carcinogenesis 9: 1179-1183. http://dx.doi.org/10.1093/carcin/9.7.1179                       |
| 3471 | Chang, YJ; Tseng, CY; Lin, PY; Chuang, YC; Chao, MW. (2017). Acute exposure to DEHP metabolite,         |
| 3472 | MEHP cause genotoxicity, mutagenesis and carcinogenicity in mammalian Chinese hamster                   |
| 3473 | ovary cells. Carcinogenesis 38: 336-345. http://dx.doi.org/10.1093/carcin/bgx009                        |
| 3474 | Cho, W; Han, B; Ahn, B; Nam, K; Choi, M; Oh, S; Kim, S; Jeong, J; Jang, D. (2008). Peroxisome           |
| 3475 | proliferator di-isodecyl phthalate has no carcinogenic potential in Fischer 344 rats. Toxicol Lett      |
| 3476 | 178: 110-116. http://dx.doi.org/10.1016/j.toxlet.2008.02.013  |
| 3477 | Cho, W; Han, BS; Ahn, B; Nam, Ki; Choi, M; Oh, SY; Kim, S; Jeong, J; Jang, DD. (2010).                  |
| 3478 | Corrigendum to "Peroxisome proliferator di-isodecyl phthalate has no carcinogenic potential in          |
| 3479 | Fischer 344 rats" [Toxicol. Lett. 178 (2008) 110-116]. Toxicol Lett 197: 156-156.                       |
| 3480 | http://dx.doi.org/10.1016/j.toxlet.2010.05.014  |
| 3481 | Cho, W; Jeong, J; Choi, M; Park, SN; Han, BS; Son, WC. (2011). 26-Week carcinogenicity study of di-     |
| 3482 | isodecyl phthalate by dietary administration to CB6F1-rasH2 transgenic mice. Arch Toxicol 85:           |
| 3483 | 59-66. <u>http://dx.doi.org/10.1007/s00204-010-0536-6</u>   |
| 3484 | Choi, S; Park, S; Jeong, J; Cho, E; Phark, S; Lee, M; Kwak, D; Lim, J; Jung, W; Sul, D. (2010).         |
| 3485 | Identification of toxicological biomarkers of di(2-ethylhexyl) phthalate in proteins secreted by        |
| 3486 | HepG2 cells using proteomic analysis. Proteomics 10: 1831-1846.   |
| 3487 | http://dx.doi.org/10.1002/pmic.200900674  |
| 3488 | Chuang, SC; Chen, HC; Sun, CW; Chen, YA; Wang, YH; Chiang, CJ; Chen, CC; Wang, SL; Chen, CJ;            |
| 3489 | Hsiung, CA. (2020). Phthalate exposure and prostate cancer in a population-based nested case-           |
| 3490 | control study. Environ Res 181: 108902. http://dx.doi.org/10.1016/j.envres.2019.108902                  |
| 3491 | ClinCalc. (2024a). ClinCalc: Fenofibrate, Drug Usage Statistics, United States, 2013 - 2022 [Website].  |
| 3492 | https://clincalc.com/DrugStats/Drugs/Fenofibrate  |

|      | 114 2025   |
|------|--|
| 3493 | ClinCalc. (2024b). ClinCalc: Gemfibrozil, Drug Usage Statistics, United States, 2013 - 2022 [Website]. |
| 3494 | https://clincalc.com/DrugStats/Drugs/Gemfibrozil   |
| 3495 | Conway, JG; Tomaszewski, KE; Olson, MJ; Cattley, RC; Marsman, DS; Popp, JA. (1989). Relationship       |
| 3496 | of oxidative damage to the hepatocarcinogenicity of the peroxisome proliferators di(2-                 |
| 3497 | ethylhexyl)phthalate and Wy-14,643. Carcinogenesis 10: 513-519.  |
| 3498 | http://dx.doi.org/10.1093/carcin/10.3.513  |
| 3499 | Cook, JC; Klinefelter, GR; Hardisty, JF; Sharpe, RM; Foster, PM. (1999). Rodent leydig cell            |
| 3500 | tumorigenesis: A review of the physiology, pathology, mechanisms and relevance to humans               |
| 3501 | [Review]. Crit Rev Toxicol 29: 169-261. http://dx.doi.org/10.1080/10408449991349203                    |
| 3502 | Corton, JC; Cunningham, ML; Hummer, BT; Lau, C; Meek, B; Peters, JM; Popp, JA; Rhomberg, L;            |
| 3503 | Seed, J; Klaunig, JE. (2014). Mode of action framework analysis for receptor-mediated toxicity:        |
| 3504 | The peroxisome proliferator-activated receptor alpha (PPARα) as a case study [Review]. Crit            |
| 3505 | Rev Toxicol 44: 1-49. http://dx.doi.org/10.3109/10408444.2013.835784                                   |
| 3506 | Corton, JC; Lapinskas, PJ. (2005). Peroxisome proliferator-activated receptors: Mediators of phthalate |
| 3507 | ester-induced effects in the male reproductive tract? [Review]. Toxicol Sci 83: 4-17.                  |
| 3508 | http://dx.doi.org/10.1093/toxsci/kfi011  |
| 3509 | Corton, JC; Peters, JM; Klaunig, JE. (2018). The PPARα-dependent rodent liver tumor response is not    |
| 3510 | relevant to humans: addressing misconceptions [Review]. Arch Toxicol 92: 83-119.                       |
| 3511 | http://dx.doi.org/10.1007/s00204-017-2094-7  |
| 3512 | Covance Labs. (1998a). Oncogenicity study in mice with di(isononyl)phthalate including ancillary       |
| 3513 | hepatocellular proliferation & biochemical analyses: Part 1 of 2, volumes 1-3. (OTS0556283-3).         |
| 3514 | Philadelphia, PA: Aristech Chemical Corp.  |
| 3515 | Covance Labs. (1998b). Support: oncogenicity study in mice with di(isononyl)phthalate including        |
| 3516 | ancillary hepatocellular proliferation and biochemical analyses with cover letter dated                |
| 3517 | 11/18/1998 [2598-105] [TSCA Submission]. (2598-105). Philadelphia, PA: Aristech Chem                   |
| 3518 | Corp. https://ntrl.ntis.gov/NTRL/dashboard/searchResults/titleDetail/OTS05562833.xhtml                 |
| 3519 | Covance Labs. (1998c). Support: Oncogenicity study in rats with di(isononyl) phthalate including       |
| 3520 | ancillary hepatocellular proliferation & biochemical analyses with cover [TSCA Submission].            |
| 3521 | (EPA/OTS Doc #89980000308). Philadelphia, PA: Aristech Chemical Corp.                                  |
| 3522 | https://ntrl.ntis.gov/NTRL/dashboard/searchResults/titleDetail/OTS05562832.xhtml                       |
| 3523 | David, RM; Moore, MR; Cifone, MA; Finney, DC; Guest, D. (1999). Chronic peroxisome proliferation       |
| 3524 | and hepatomegaly associated with the hepatocellular tumorigenesis of di(2-ethylhexyl)phthalate         |
| 3525 | and the effects of recovery. Toxicol Sci 50: 195-205. http://dx.doi.org/10.1093/toxsci/50.2.195        |
| 3526 | David, RM: Moore, MR: Finney, DC: Guest, D. (2000a), Chronic toxicity of di(2-ethylhexyl)phthalate     |
| 3527 | in mice. Toxicol Sci 58: 377-385. http://dx.doi.org/10.1093/toxsci/58.2.377                            |
| 3528 | David, RM: Moore, MR: Finney, DC: Guest, D. (2000b), Chronic toxicity of di(2-ethylhexyl)phthalate     |
| 3529 | in rats. Toxicol Sci 55: 433-443. http://dx.doi.org/10.1093/toxsci/55.2.433                            |
| 3530 | Dekeyser, JG: Laurenzana, EM: Peterson, EC: Chen, T: Omiecinski, CJ. (2011). Selective phthalate       |
| 3531 | activation of naturally occurring human constitutive androstane receptor splice variants and the       |
| 3532 | pregnane X receptor. Toxicol Sci 120: 381-391. http://dx.doi.org/10.1093/toxsci/kfq394                 |
| 3533 | Dell, L: Teta, MJ. (1995). Mortality among workers at a plastics manufacturing and research            |
| 3534 | development facility: 1946-1988. Am J Ind Med 28.  |
| 3535 | Dirven, HAA: Theuws, JLG: Jongeneelen, FJ: Bos, RP. (1991). Non-mutagenicity of 4 metabolites of       |
| 3536 | di(2-ethylhexyl)phthalate (DEHP) and 3 structurally related derivatives of di(2-                       |
| 3537 | ethylhexyl)adipate (DEHA) in the Salmonella mutagenicity assay. Mutat Res 260. 121-130                 |
| 3538 | http://dx.doi.org/10.1016/0165-1218(91)90088-4   |

| 3539 | Diwan, BA; Ward, JM; Rice, JM; Colburn, NH; Spangler, EF. (1985). Tumor-promoting effects of di(2-   |
|------|--|
| 3540 | ethylhexyl)phthalate in JB6 mouse epidermal cells and mouse skin. Carcinogenesis 6: 343-347.         |
| 3541 | http://dx.doi.org/10.1093/carcin/6.3.343   |
| 3542 | Douglas, GR; Hugenholtz, AP; Blakey, DH. (1986). Genetic toxicology of phthalate esters: Mutagenic   |
| 3543 | and other genotoxic effects. Environ Health Perspect 65: 255-262.                                    |
| 3544 | http://dx.doi.org/10.1289/ehp.8665255  |
| 3545 | Dwivedi, RS; Alvares, K; Nemali, MR; Subbarao, V; Reddy, MK; Usman, MI; Rademaker, AW;               |
| 3546 | Reddy, JK; Rao, MS. (1989). Comparison of the peroxisome proliferator-induced pleiotropic            |
| 3547 | response in the liver of nine strains of mice. Toxicol Pathol 17: 16-26.                             |
| 3548 | http://dx.doi.org/10.1177/01926233890171P103   |
| 3549 | Eastin, WC; Mennear, JH; Tennant, RW; Stoll, RE; Branstetter, DG; Bucher, JR; McCullough, B;         |
| 3550 | Binder, RL; Spalding, JW; Mahler, JF. (2001). Tg.AC genetically altered mouse: assay working         |
| 3551 | group overview of available data. Toxicol Pathol 29 Suppl: 60-80.                                    |
| 3552 | http://dx.doi.org/10.1080/019262301753178483   |
| 3553 | EC/HC. (1994). Canadian environmental protection act priority substances list assessment report:     |
| 3554 | Dibutyl phthalate. Ottawa, Ontario: Environment Canada, Health Canada.                               |
| 3555 | https://www.canada.ca/content/dam/hc-sc/migration/hc-sc/ewh-semt/alt_formats/hecs-                   |
| 3556 | sesc/pdf/pubs/contaminants/psl1-lsp1/phthalate_dibutyl_phtalate/butyl_phthalate-eng.pdf              |
| 3557 | EC/HC. (2015a). State of the science report: Phthalate substance grouping 1,2-Benzenedicarboxylic    |
| 3558 | acid, diisononyl ester; 1,2-Benzenedicarboxylic acid, di-C8-10-branched alkyl esters, C9-rich        |
| 3559 | (Diisononyl Phthalate; DINP). Chemical Abstracts Service Registry Numbers: 28553-12-0 and            |
| 3560 | 68515-48-0. Gatineau, Quebec. https://www.ec.gc.ca/ese-  |
| 3561 | ees/default.asp?lang=En&n=47F58AA5-1   |
| 3562 | EC/HC. (2015b). State of the science report: Phthalate substance grouping: Medium-chain phthalate    |
| 3563 | esters: Chemical Abstracts Service Registry Numbers: 84-61-7; 84-64-0; 84-69-5; 523-31-9;            |
| 3564 | 5334-09-8;16883-83-3; 27215-22-1; 27987-25-3; 68515-40-2; 71888-89-6. Gatineau, Quebec:              |
| 3565 | Environment Canada, Health Canada. <u>https://www.ec.gc.ca/ese-ees/4D845198-761D-428B-</u>           |
| 3566 | A519-75481B25B3E5/SoS_Phthalates%20%28Medium-chain%29_EN.pdf   |
| 3567 | EC/HC. (2015c). State of the Science Report: Phthalates Substance Grouping: Long-chain Phthalate     |
| 3568 | Esters. 1,2-Benzenedicarboxylic acid, diisodecyl ester (diisodecyl phthalate; DIDP) and 1,2-         |
| 3569 | Benzenedicarboxylic acid, diundecyl ester (diundecyl phthalate; DUP). Chemical Abstracts             |
| 3570 | Service Registry Numbers: 26761-40-0, 68515-49-1; 3648-20-2. Gatineau, Quebec:                       |
| 3571 | Environment Canada, Health Canada. <u>https://www.ec.gc.ca/ese-</u>                                  |
| 3572 | ees/default.asp?lang=En&n=D3FB0F30-1   |
| 3573 | ECB. (2003a). European Union risk assessment report, vol 36: 1,2-Benzenedicarboxylic acid, Di-C9-11- |
| 3574 | Branched alkyl esters, C10-Rich and Di-"isodecyl"phthalate (DIDP). In 2nd Priority List. (EUR        |
| 3575 | 20785 EN). Luxembourg, Belgium: Office for Official Publications of the European                     |
| 3576 | Communities.   |
| 3577 | http://publications.jrc.ec.europa.eu/repository/bitstream/JRC25825/EUR%2020785%20EN.pdf              |
| 3578 | ECB. (2003b). European Union risk assessment report: 1,2-Benzenedicarboxylic acid, di-C8-10-         |
| 3579 | branched alkyl esters, C9-rich - and di-"isononyl" phthalate (DINP). In 2nd Priority List,           |
| 3580 | Volume: 35. (EUR 20784 EN). Luxembourg, Belgium: Office for Official Publications of the             |
| 3581 | European Communities. <u>http://bookshop.europa.eu/en/european-union-risk-assessment-report-</u>     |
| 3582 | <u>pbEUNA20784/</u>  |
| 3583 | ECB. (2003c). European union risk assessment report: DINP. European Commission.                      |
| 3584 | http://publications.jrc.ec.europa.eu/repository/handle/JRC25827                                      |
| 3585 | ECB. (2004). European Union Risk Assessment Report: Dibutyl phthalate with addendum to the           |
| 3586 | environmental section - 2004. (EUR 19840 EN). Luxembourg: European Union, European                   |

| 3587         | Chemicals Bureau, Institute for Health and Consumer Protection.  |
|--------------|--|
| 3588         | https://echa.europa.eu/documents/10162/ba7f7c39-dab6-4dca-bc8e-dfab7ac53e37  |
| 3589         | ECB. (2007). European Union Risk Assessment Report: Benzyl butyl phthalate (CAS No: 85-68-7,   |
| 3590         | EINECS: 201-622-7). (EUR 22773 EN). Luxembourg: European Commission.   |
| 3591         | https://echa.europa.eu/documents/10162/bad5c928-93a5-4592-a4f6-e02c5e89c299  |
| 3592         | ECB. (2008). European Union risk assessment report: Bis(2-ethylhexyl)phthalate (DEHP) [Standard]. In   |
| 3593         | 2nd Priority List. (EUR 23384 EN). Luxembourg: Office for Official Publications of the   |
| 3594         | European Communities. https://op.europa.eu/en/publication-detail/-/publication/80eaeafa-5985-  |
| 3595         | <u>4481-9b83-7b5d39241d52</u>  |
| 3596         | ECCC/HC. (2020). Screening assessment - Phthalate substance grouping. (En14-393/2019E-PDF).  |
| 3597         | Environment and Climate Change Canada, Health Canada.  |
| 3598         | https://www.canada.ca/en/environment-climate-change/services/evaluating-existing-  |
| 3599         | substances/screening-assessment-phthalate-substance-grouping.html  |
| 3600         | ECHA. (2008). Substance name: Benzyl butyl phthalate, EC number: 201-622-7, CAS number: 85-68-7:   |
| 3601         | Member state committee support document for identification of benzyl butyl phthalate (BBP) as  |
| 3602         | a substance of very high concern. Helsinki, Finland.   |
| 3603         | https://echa.europa.eu/documents/10162/b4024445-00ce-4849-9ab0-69aed88c51f2  |
| 3604         | ECHA. (2010a). Evaluation of new scientific evidence concerning the restrictions contained in Annex  |
| 3605         | XVII to Regulation (EC) No 1907/2006 (REACH): Review of new available information for  |
| 3606         | dibutyl phthalate (DBP) CAS No 84-74-2 Einecs No 201-557-4 (pp. 18).   |
| 3607         | ECHA. (2010b). Evaluation of new scientific evidence concerning the restrictions contained in Annex  |
| 3608         | XVII to regulation (EC) No. 1907/2006 (REACH): Review of new available information for   |
| 3609         | benzyl butyl phthalate (BBP) CAS No. 85-68-7 Einecs No. 201-622-7 (pp. 15).  |
| 3610         | ECHA. (2011). Annex XV restriction report: Proposal for a restriction, version 2. Substance name:  |
| 3611         | bis(2-ehtylhexyl)phthlate (DEHP), benzyl butyl phthalate (BBP), dibutyl phthalate (DBP),   |
| 3612         | diisobutyl phthalate (DIBP). Copenhagen, Denmark: Danish Environmental Protection Agency ::  |
| 3613         | Danish EPA. https://echa.europa.eu/documents/10162/c6781e1e-1128-45c2-bf48-8890876fa719  |
| 3614         | ECHA. (2012a). Committee for Risk Assessment (RAC) Committee for Socio-economic Analysis   |
| 3615         | (SEAC): Background document to the Opinion on the Annex XV dossier proposing restrictions  |
| 3616         | on four phthalates. Helsinki, Finland. <u>http://echa.europa.eu/documents/10162/3bc5088a-a231-</u>   |
| 3617         | <u>498e-86e6-8451884c6a4f</u>  |
| 3618         | ECHA. (2012b). Committee for Risk Assessment (RAC) Opinion on an Annex XV dossier proposing  |
| 3619         | restrictions on four phthalates. (ECHA/RAC/RES-O-0000001412-86-07/F). Helsinki, Finland:   |
| 3620         | European Chemicals Agency :: ECHA. <u>https://echa.europa.eu/documents/10162/77ct7d29-ba63-</u>  |
| 3621         | <u>4901-aded-59cf/5536e06</u>  |
| 3622         | ECHA. (2013). Evaluation of new scientific evidence concerning DINP and DIDP in relation to entry 52   |
| 3623         | of Annex XVII to REACH Regulation (EC) No 1907/2006. Helsinki, Finland.  |
| 3624         | http://echa.europa.eu/documents/10162/31b406/e-de40-4044-93e8-9c9ff1960/15   |
| 3625         | ECHA. (2014). Committee for Risk Assessment RAC Opinion proposing harmonised classification and  |
| 3626         | labelling at EU level of Dicyclonexyl phthalate, EC number: 201-545-9, CAS number: 84-61-7.  |
| 3627         | https://heronet.epa.gov/heronet/index.cfm/reference/download/reference_id/10328890   |
| 3628         | ECHA. (2017a). Annex to the Background document to the Opinion on the Annex XV dossier   |
| 3629         | proposing restrictions on four phinalates (DEHP, BBP, DBP, DBP). (ECHA/RAC/RES-O-  |
| 303U         | 0000001412-80-140/F; ECHA/SEAC/KES-0-0000001412-86-154/F).   |
| 3031<br>2622 | nups://neronet.epa.gov/neronet/index.cim/reference/download/reference_id/10328892  |
| 3032<br>2622 | <u><b>DUDA</b></u> . (20170). Opinion on an Annex A v dossier proposing restrictions on four pathalates (DEHP,<br><b>DDD</b> DDD (ECHA/DAC/DES O 0000001412.96 140/E). Heleinel: Einland |
| 3033         | DDF, DDF, DIDF). (ECHA/KAC/KES-U-UUUUUU1412-80-140/F). Helsinki, Finland.  |
| 3034         | https://ecna.europa.eu/documents/10162/e39983ad-1016-1402-7992-8a032b5b82aa  |

- 3635 <u>EFSA. (2005a)</u>. Opinion of the scientific panel on food additives, flavourings, processing aids and
   3636 materials in contact with food (AFC) on a request from the commission related to di 3637 isononylphthalate (DINP) for use in food contact materials. Question N° EFSA-q-2003-194 (pp.
   3638 1-18). <u>http://dx.doi.org/10.2903/j.efsa.2005.244</u>
- 3639 <u>EFSA. (2005b)</u>. Opinion of the Scientific Panel on food additives, flavourings, processing aids and
   3640 materials in contact with food (AFC) related to Bis(2-ethylhexyl)phthalate (DEHP) for use in
   3641 food contact materials. EFSA J 3: 243. <u>http://dx.doi.org/10.2903/j.efsa.2005.243</u>
- 3642 <u>EFSA. (2005c)</u>. Opinion of the Scientific Panel on food additives, flavourings, processing aids and
   3643 materials in contact with food (AFC) related to Butylbenzylphthalate (BBP) for use in food
   3644 contact materials. 3. <u>http://dx.doi.org/10.2903/j.efsa.2005.241</u>
- 3645 <u>EFSA. (2005d)</u>. Opinion of the Scientific Panel on food additives, flavourings, processing aids and
   3646 materials in contact with food (AFC) related to di-Butylphthalate (DBP) for use in food contact
   3647 materials. 3: 242. <u>http://dx.doi.org/10.2903/j.efsa.2005.242</u>
- 3648 <u>EFSA. (2005e)</u>. Opinion of the Scientific Panel on food additives, flavourings, processing aids and
   3649 materials in contact with food (AFC) related to Di-isodecylphthalate (DIDP) for use in food
   3650 contact materials. EFSA J 3: 245. <u>http://dx.doi.org/10.2903/j.efsa.2005.245</u>
- 3651 <u>EFSA. (2019)</u>. Update of the risk assessment of di-butylphthalate (DBP), butyl-benzyl-phthalate (BBP),
   3652 bis(2-ethylhexyl)phthalate (DEHP), di-isononylphthalate (DINP) and di-isodecylphthalate
   3653 (DIDP) for use in food contact materials. EFSA J 17: ee05838.
   3654 http://dx.doi.org/10.2903/j.efsa.2019.5838
- 3655 <u>Elliott, BM; Elcombe, CR. (1987)</u>. Lack of DNA damage or lipid peroxidation measured in vivo in the
   3656 rat liver following treatment with peroxisomal proliferators. Carcinogenesis 8: 1213-1218.
   3657 <u>http://dx.doi.org/10.1093/carcin/8.9.1213</u>
- 3658 <u>Ennis, ZN; Pottegård, A; Ahern, TP; Hallas, J; Damkier, P</u>. (2019). Exposure to phthalate-containing
   3659 prescription drugs and the risk of colorectal adenocarcinoma: A Danish nationwide case-control
   3660 study. Pharmacoepidemiol Drug Saf 28: 528-535. <u>http://dx.doi.org/10.1002/pds.4759</u>
- 3661 <u>Environment Canada. (1994)</u>. Bis(2-ethylhexyl) phthalate. In Priority Substances List Assessment
   3662 Report. (RISKLINE/1995020013). Ottawa, Canada: Minister of Supply and Services Canada.
   3663 <u>http://www.hc-sc.gc.ca/ewh-semt/pubs/contaminants/psl1-lsp1/bis\_2\_ethylhexyl/index-eng.php</u>
- 3664 <u>Environment Canada. (2000)</u>. Canadian environmental protection act priority substances list assessment
   3665 report: Butylbenzylphthalate. Ottawa, Ontario: Environment Canada, Health Canada.
   3666 <u>https://www.canada.ca/content/dam/hc-sc/migration/hc-sc/ewh-semt/alt\_formats/hecs-</u>
   2667
- 3667 <u>sesc/pdf/pubs/contaminants/psl2-lsp2/butylbenzylphthalate/butylbenzylphthalate-eng.pdf</u>
   3668 Erkekoglu, P; Giray, B; Durmaz, E; Ozmert, E; Kizilgun, M; Derman, O; Yurdakok, K. (2010a).
- Erkekoglu, P; Giray, B; Durmaz, E; Ozmert, E; Kizilgun, M; Derman, O; Yurdakok, K. (2010a).
   Evaluation of the correlation between plasma amylase and lipase levels and phthalate exposure
   in pubertal gynecomastia patients. Turk Pediatri Arsivi 45: 366-370.
   http://dx.doi.org/10.4274/tpa.45.366
- 3672 Erkekoglu, P; Rachidi, W; De Rosa, V; Giray, B; Favier, A; Hincal, F. (2010b). Protective effect of
   3673 selenium supplementation on the genotoxicity of di(2-ethylhexyl)phthalate and mono(2 3674 ethylhexyl)phthalate treatment in LNCaP cells. Free Radic Biol Med 49: 559-566.
   3675 http://dx.doi.org/10.1016/j.freeradbiomed.2010.04.038
- 3676 Eveillard, A; Mselli-Lakhal, L; Mogha, A; Lasserre, F; Polizzi, A; Pascussi, JM; Guillou, H; Martin,
   3677 PG; Pineau, T. (2009). Di-(2-ethylhexyl)-phthalate (DEHP) activates the constitutive androstane
   3678 receptor (CAR): a novel signalling pathway sensitive to phthalates. Biochem Pharmacol 77:
   3679 1735-1746. <u>http://dx.doi.org/10.1016/j.bcp.2009.02.023</u>
- 3680 <u>Fahrig, R; Steinkamp-Zucht, A. (1996)</u>. Co-recombinogenic and anti-mutagenic effects of
   3681 diethylhexylphthalate, inactiveness of pentachlorophenol in the spot test with mice. Mutat Res
   3682 354: 59-67. http://dx.doi.org/10.1016/0027-5107(96)00036-X

| 3683 | Felter, SP; Foreman, JE; Boobis, A; Corton, JC; Doi, AM; Flowers, L; Goodman, J; Haber, LT; Jacobs,  |
|------|--|
| 3684 | A; Klaunig, JE; Lynch, AM; Moggs, J; Pandiri, A. (2018). Human relevance of rodent liver   |
| 3685 | tumors: Key insights from a toxicology forum workshop on nongenotoxic modes of action  |
| 3686 | [Review]. Regul Toxicol Pharmacol 92: 1-7. http://dx.doi.org/10.1016/j.yrtph.2017.11.003   |
| 3687 | Fitzgerald, JE; Sanver, JL; Schardein, JL; Lake, RS; McGuire, EJ; de la Iglesia, FA. (1981). Carcinogen  |
| 3688 | bioassay and mutagenicity studies with the hypolipidemic agent gemfibrozil. J Natl Cancer Inst   |
| 3689 | 67: 1105-1116.   |
| 3690 | Florin, I; Rutberg, L; Curvall, M; Enzell, CR. (1980). Screening of tobacco smoke constituents for   |
| 3691 | mutagenicity using the Ames' test. Toxicology 15: 219-232. http://dx.doi.org/10.1016/0300-   |
| 3692 | 483X(80)90055-4  |
| 3693 | Franken, C; Lambrechts, N; Govarts, E; Koppen, G; Den Hond, E; Ooms, D; Voorspoels, S; Bruckers,   |
| 3694 | L; Loots, I; Nelen, V; Sioen, I; Nawrot, TS; Baeyens, W; Van Larebeke, N; Schoeters, G.  |
| 3695 | (2017). Phthalate-induced oxidative stress and association with asthma-related airway  |
| 3696 | inflammation in adolescents. Int J Hyg Environ Health 220: 468-477.  |
| 3697 | http://dx.doi.org/10.1016/j.ijheh.2017.01.006  |
| 3698 | Frick, M; Elo, O; Haapa, K; Heinonen, O; Heinsalmi, P; Helo, P; Huttunen, J; Kaitaniemi, P; Koskinen,  |
| 3699 | P; Manninen, V. (1987). Helsinki Heart Study: Primary-prevention trial with gemfibrozil in   |
| 3700 | middle-aged men with dyslipidemia: Safety of treatment, changes in risk factors, and incidence   |
| 3701 | of coronary heart disease. N Engl J Med 317: 1237-1245.  |
| 3702 | http://dx.doi.org/10.1056/NEJM198711123172001  |
| 3703 | Ganning, AE; Olsson, MJ; Brunk, U; Dallner, G. (1990). Effects of prolonged treatment with phthalate   |
| 3704 | ester on rat liver. Pharmacol Toxicol 67: 392-401. <u>http://dx.doi.org/10.1111/j.1600-</u>  |
| 3705 | <u>0773.1990.tb00851.x</u>   |
| 3706 | Gupta, RC; Goel, SK; Earley, K; Singh, B; Reddy, JK. (1985). 32P-postlabeling analysis of peroxisome   |
| 3707 | proliferator-DNA adduct formation in rat liver in vivo and hepatocytes in vitro. Carcinogenesis  |
| 3708 | 6: 933-936. <u>http://dx.doi.org/10.1093/carcin/6.6.933</u>  |
| 3709 | Guyton, KZ; Chiu, WA; Bateson, TF; Jinot, J; Scott, CS; Brown, RC; Caldwell, JC. (2009). A   |
| 3710 | reexamination of the PPAR-alpha activation mode of action as a basis for assessing human   |
| 3711 | cancer risks of environmental contaminants [Review]. Environ Health Perspect 117: 1664-1672.   |
| 3712 | http://dx.doi.org/10.1289/ehp.0900758  |
| 3713 | Gwinn, WM; Auerbach, SS; Parham, F; Stout, MD; Waidyanatha, S; Mutlu, E; Collins, B; Paules, RS;   |
| 3714 | Merrick, BA; Ferguson, S; Ramaiahgari, S; Bucher, JR; Sparrow, B; Toy, H; Gorospe, J;  |
| 3715 | Machesky, N; Shah, RR; Balik-Meisner, MR; Mav, D; Phadke, DP; Roberts, G; Devito, MJ.  |
| 3/16 | (2020). Evaluation of 5-day in Vivo Rat Liver and Kidney With High-throughput  |
| 3/1/ | Transcriptomics for Estimating Benchmark Doses of Apical Outcomes. Toxicol Sci 1/6: 343-   |
| 3/18 | 354. <u>http://dx.doi.org/10.1093/toxsci/ktaa081</u>   |
| 3/19 | Hagmar, L; Akesson, B; Nielsen, J; Andersson, C; Linden, K; Attewell, R; Moller, 1. (1990). Mortality  |
| 3720 | and cancer morbidity in workers exposed to low levels of vinyl chloride monomer at a polyvinyl   |
| 3721 | chioride processing plant. Am J Ind Med 17: 553-565.   |
| 3122 | Hugmar I : Mikozzy 7: Walindar H (1005) Cancer incidence in Swedish starilant workers avposed to   |
| 3723 | <u>Haginar, L. Mikoczy, Z. Weinider, H. (1995)</u> . Cancel incidence in Swedish sternant workers exposed to athylono oxide. Occup Environ Mod 52: 154–156. http://dx.doi.org/10.1126/oom 52.3.154   |
| 3724 | Honson I (1000) Rick for testioular concer after accurational exposure to plastics [] atter]. Int I Concer   |
| 3725 | <u><b>Hallsell, J. (1999)</b></u> . Kisk for restricting cancer after occupational exposure to plastics [Letter]. Int J Cancer 82: 011 012 http://dx.doi.org/10.1002/(sigi)1007.0215(10000000)82:6<011::gid.iio23>3.0.go;2   |
| 3720 | $62. 711-712. \underline{\text{nup.//ux.uoi.org/10.1002/(Stor)1097-0213(19990909)82.0<911atu-ijc23>3.0.c0;2-0}{0}$   |
| 3727 | $\stackrel{\vee}{=}$<br>Hardell I: Malmovist N: Ohlson CG: Westberg H: Frikeson M (2004) Testicular cancer and   |
| 3720 | occupational exposure to polyvinyl chloride plastics: A case-control study. Int I Cancer 100.  |
| 3730 | 425-429 http://dx.doi.org/10.1002/jic.11709  |
| 5150 | $123 - 127 \cdot \frac{100}{100} \cdot \frac{1000}{2} \cdot \frac{1000}{100} \cdot 1$ |

| Hardell, L; Ohlson, CG; Fredrikson, M. (1997). Occupational exposure to polyvinyl chloride as a risk  |    |
|---|----|
| factor for testicular cancer evaluated in a case-control study. Int J Cancer 73: 828-830.             |    |
| http://dx.doi.org/10.1002/(sici)1097-0215(19971210)73:6<828::aid-ijc10>3.0.co;2-0                     |    |
| Hasmall, SC; James, NH; Macdonald, N; Soames, AR; Roberts, RA. (2000). Species differences in         |    |
| response to diethylhexylphthalate: suppression of apoptosis, induction of DNA synthesis and           |    |
| peroxisome proliferator activated receptor alpha-mediated gene expression. Arch Toxicol 74: 85        | -  |
| 91. http://dx.doi.org/10.1007/s002040050657   |    |
| Hasmall, SC; Roberts, RA. (2000). The nongenotoxic hepatocarcinogens diethylhexylphthalate and        |    |
| methylclofenapate induce DNA synthesis preferentially in octoploid rat hepatocytes. Toxicol           |    |
| Pathol 28: 503-509. http://dx.doi.org/10.1177/019262330002800401                                      |    |
| Hayashi, F; Motoki, Y; Tamura, H; Watanabe, T; Ogura, T; Esumi, H; Suga, T. (1998). Induction of      |    |
| hepatic poly(ADP-ribose) polymerase by peroxisome proliferators, non-genotoxic                        |    |
| hepatocarcinogens. Cancer Lett 127: 1-7. http://dx.doi.org/10.1016/S0304-3835(98)00002-0              |    |
| Hazleton. (1986). Mutagenicity of 1C in a mouse lymphoma mutation assay final report on dimethyl      |    |
| phthalate and dibutyl phthalate with cover letter dated 102786. (EPA/OTS Doc #8EHQ-1086-              |    |
| 0620). Washington, DC: Chemical Manufacturers Association.  |    |
| https://ntrl.ntis.gov/NTRL/dashboard/searchResults/titleDetail/OTS0513454.xhtml                       |    |
| Health Canada. (2015). Supporting documentation: Carcinogenicity of phthalates - mode of action and   |    |
| human relevance. In Supporting documentation for Phthalate Substance Grouping. Ottawa, ON.            |    |
| Health Canada. (2018a). Supporting documentation: Evaluation of epidemiologic studies on phthalate    |    |
| compounds and their metabolites for effects on behaviour and neurodevelopment, allergies,             |    |
| cardiovascular function, oxidative stress, breast cancer, obesity, and metabolic disorders. Ottawa    | ı, |
| ON.   |    |
| Health Canada. (2018b). Supporting documentation: Evaluation of epidemiologic studies on phthalate    |    |
| compounds and their metabolites for hormonal effects, growth and development and                      |    |
| reproductive parameters. Ottawa, ON.  |    |
| Heineman, EF; Olsen, JH; Pottern, LM; Gomez, M; Raffn, E; Blair, A. (1992). Occupational risk factors | S  |
| for multiple myeloma among Danish men. Cancer Causes Control 3: 555-568.                              |    |
| http://dx.doi.org/10.1007/BF00052753  |    |
| Hilton, GM; Adcock, C; Akerman, G; Baldassari, J; Battalora, M; Casey, W; Clippinger, AJ; Cope, R;    |    |
| Goetz, A; Hayes, AW; Papineni, S; Peffer, RC; Ramsingh, D; Williamson Riffle, B; Sanches da           |    |
| Rocha, M; Ryan, N; Scollon, E; Visconti, N; Wolf, DC; Yan, Z; Lowit, A. (2022). Rethinking            |    |
| chronic toxicity and carcinogenicity assessment for agrochemicals project (ReCAAP): A                 |    |
| reporting framework to support a weight of evidence safety assessment without long-term rodent        | t  |
| bioassays. Regul Toxicol Pharmacol 131: 105160. http://dx.doi.org/10.1016/j.yrtph.2022.105160         | )  |
| Hinton, RH; Mitchell, FE; Mann, A; Chescoe, D; Price, SC; Nunn, A; Grasso, P; Bridges, JW. (1986).    |    |
| Effects of phthalic acid esters on the liver and thyroid. Environ Health Perspect 70: 195-210.        |    |
| http://dx.doi.org/10.2307/3430356   |    |
| Hodgson, JR; Myhr, BC; McKeon, M; Brusick, DJ. (1982). Evaluation of di-(2-ethylhexyl)phthalate and   | d  |
| its major metabolites in the primary rat hepatocyte unscheduled DNA synthesis assay [Abstract].       |    |
| Environ Mutagen 4: 388. http://dx.doi.org/10.1002/em.2860040316                                       |    |
| Holmes, AK; Koller, KR; Kieszak, SM; Sjodin, A; Calafat, AM; Sacco, FD; Varner, DW; Lanier, AP;       |    |
| Rubin, CH. (2014). Case-control study of breast cancer and exposure to synthetic environmental        |    |
| chemicals among Alaska Native women. Int J Circumpolar Health 73: 25760.                              |    |
| http://dx.doi.org/10.3402/ijch.v73.25760  |    |
| Hoshino, N; Iwai, M; Okazaki, Y. (2005). A two-generation reproductive toxicity study of dicyclohexyl | l  |
| phthalate in rats. J Toxicol Sci 30: 79-96. http://dx.doi.org/10.2131/jts.30.s79                      |    |
| · · · · · · · · · · · · · · · · · · ·   |    |

| Howr          | oyd, P; Swanson, C; Dunn, C; Cattley, RC; Corton, JC. (2004). Decreased longevity and           |
|---------------|---|
|               | enhancement of age-dependent lesions in mice lacking the nuclear receptor peroxisome            |
|               | proliferator-activated receptor alpha (PPARalpha). Toxicol Pathol 32: 591-599.                  |
|               | http://dx.doi.org/10.1080/01926230490515283   |
| <u>Hsu, I</u> | C; Kuo, YT; Leon Guo, Y; Chen, JR; Tsai, SS; Chao, HR; Teng, YN; Pan, MH. (2016). The           |
|               | adverse effects of low-dose exposure to Di(2-ethylhexyl) phthalate during adolescence on sperm  |
|               | function in adult rats. Environ Toxicol 31: 706-712. http://dx.doi.org/10.1002/tox.22083        |
| Huttu         | nen, J; Heinonen, O; Manninen, V; Koskinen, P; Hakulinen, T; Teppo, L; Mänttäri, M; Frick, M.   |
|               | (1994). The Helsinki Heart Study: An 8.5-year safety and mortality follow-up. J Intern Med 235: |
|               | 31-39. http://dx.doi.org/10.1111/j.1365-2796.1994.tb01029.x                                     |
| IARC          | . (1999). Butyl benzyl phthalate [IARC Monograph]. In IARC Monographs on the Evaluation of      |
|               | Carcinogenic Risks to Humans, vol 73 (pp. 115-129). (RISKLINE/2000010015). Lyon, France.        |
|               | http://monographs.iarc.fr/ENG/Monographs/vol73/index.php  |
| IARC          | (2013), Di(2-ethylhexyl)phthalate [IARC Monograph]. In IARC Monographs on the Evaluation        |
|               | of Carcinogenic Risks to Humans, vol 101 (pp. 149-284). Lyon, France.                           |
| IPCS.         | (2007). Harmonization project document no. 4: Part 1: IPCS framework for analysing the          |
|               | relevance of a cancer mode of action for humans and case-studies: Part 2: IPCS framework for    |
|               | analysing the relevance of a non-cancer mode of action for humans. Geneva, Switzerland: World   |
|               | Health Organization.  |
|               | http://www.who.int/ipcs/methods/harmonization/areas/cancer_mode.pdf?ua=1                        |
| Isenbe        | erg, JS; Kamendulis, LM; Ackley, DC; Smith, JH; Pugh, G, Jr.; Lington, AW; McKee, RH;           |
|               | Klaunig, JE. (2001). Reversibility and persistence of di-2-ethylhexyl phthalate (DEHP)- and     |
|               | phenobarbital-induced hepatocellular changes in rodents. Toxicol Sci 64: 192-199.               |
|               | http://dx.doi.org/10.1093/toxsci/64.2.192   |
| Isenbe        | erg, JS; Kamendulis, LM; Smith, JH; Ackley, DC; Pugh, G, Jr.; Lington, AW; Klaunig, JE. (2000). |
|               | Effects of Di-2-ethylhexyl phthalate (DEHP) on gap-junctional intercellular communication       |
|               | (GJIC), DNA synthesis, and peroxisomal beta oxidation (PBOX) in rat, mouse, and hamster         |
|               | liver. Toxicol Sci 56: 73-85. http://dx.doi.org/10.1093/toxsci/56.1.73                          |
| Ishida        | te, M, Jr.; Odashima, S. (1977). Chromosome tests with 134 compounds on Chinese hamster cells   |
|               | in vitro: A screening for chemical carcinogens. Mutat Res 48: 337-353.                          |
|               | http://dx.doi.org/10.1016/0027-5107(77)90177-4  |
| Issem         | ann, I; Green, S. (1990). Activation of a member of the steroid hormone receptor superfamily by |
|               | peroxisome proliferators. Nature 347: 645-650. http://dx.doi.org/10.1038/347645a0               |
| Ito, Y        | ; Yamanoshita, O; Asaeda, N; Tagawa, Y; Lee, CH; Aoyama, T; Ichihara, G; Furuhashi, K;          |
|               | Kamijima, M; Gonzalez, FJ; Nakajima, T. (2007a). Di(2-ethylhexyl)phthalate induces hepatic      |
|               | tumorigenesis through a peroxisome proliferator-activated receptor alpha-independent pathway.   |
|               | J Occup Health 49: 172-182. http://dx.doi.org/10.1539/joh.49.172                                |
| Ito, Y        | ; Yamanoshita, O; Kurata, Y; Kamijima, M; Aoyama, T; Nakajima, T. (2007b). Induction of         |
|               | peroxisome proliferator-activated receptor alpha (PPAR[alpha])-related enzymes by di(2-         |
|               | ethylhexyl) phthalate (DEHP) treatment in mice and rats, but not marmosets. Arch Toxicol 81:    |
|               | 219-226. http://dx.doi.org/10.1007/s00204-006-0141-x  |
| James         | NH: Soames, AR: Roberts, RA. (1998). Suppression of hepatocyte apoptosis and induction of       |
|               | DNA synthesis by the rat and mouse hepatocarcinogen diethylhexylphlathate (DEHP) and the        |
|               | mouse hepatocarcinogen 1.4-dichlorobenzene (DCB). Arch Toxicol 72: 784-790.                     |
|               | http://dx.doi.org/10.1007/s002040050574   |
| Kanki         | K: Nishikawa, A: Masumura, K: Umemura, T: Imazawa, T: Kitamura, Y: Nohmi, T: Hirose, M          |
|               | (2005). In vivo mutational analysis of liver DNA in gpt delta transgenic rats treated with the  |
|               |   |

| 3825 | hepatocarcinogens N-nitrosopyrrolidine, 2-amino-3-methylimidazo[4,5-f]quinoline, and di(2-              |
|------|---|
| 3826 | ethylhexyl)phthalate. Mol Carcinog 42: 9-17. http://dx.doi.org/10.1002/mc.20061                         |
| 3827 | Kanode, R; Chandra, S; Sharma, S. (2017). Application of bacterial reverse mutation assay for detection |
| 3828 | of non-genotoxic carcinogens. Toxicol Mech Meth 27: 376-381.  |
| 3829 | http://dx.doi.org/10.1080/15376516.2017.1300616   |
| 3830 | Karabulut, G; Barlas, N. (2018). Genotoxic, histologic, immunohistochemical, morphometric and           |
| 3831 | hormonal effects of di-(2-ethylhexyl)-phthalate (DEHP) on reproductive systems in pre-pubertal          |
| 3832 | male rats. Toxicology Research 7: 859-873. http://dx.doi.org/10.1039/c8tx00045j                         |
| 3833 | Kaufmann, W; Deckardt, K; McKee, RH; Butala, JH; Bahnemann, R. (2002). Tumor induction in mouse         |
| 3834 | liver: Di-isononyl phthalate acts via peroxisome proliferation. Regul Toxicol Pharmacol 36: 175-        |
| 3835 | 183. <u>http://dx.doi.org/10.1006/rtph.2002.1575</u>  |
| 3836 | Kersten, S; Seydoux, J; Peters, J; Gonzalez, F; Desvergne, B; Wahli, W. (1999). Peroxisome              |
| 3837 | proliferator-activated receptor alpha mediates the adaptive response to fasting. J Clin Invest 103:     |
| 3838 | 1489-1498. http://dx.doi.org/10.1172/JCI6223  |
| 3839 | Kim, S; Park, GY; Jeong, JS; Ki Taek, N; Kyung-Min, L; Sun-Ha, J; Yoo, YJ; Yun-Sil, L. (2019). Di-2-    |
| 3840 | ethylhexylphthalate promotes thyroid cell proliferation and DNA damage through activating               |
| 3841 | thyrotropin-receptor-mediated pathways in vitro and in vivo. Food Chem Toxicol 124: 265-272.            |
| 3842 | http://dx.doi.org/10.1016/j.fct.2018.12.010   |
| 3843 | King-Herbert, A; Thayer, K. (2006). NTP workshop: Animal models for the NTP rodent cancer               |
| 3844 | bioassay: Stocks and strains - Should we switch? Toxicol Pathol 34: 802-805.                            |
| 3845 | http://dx.doi.org/10.1080/01926230600935938   |
| 3846 | King-Herbert, AP; Sills, RC; Bucher, JR. (2010). Commentary: update on animal models for NTP            |
| 3847 | studies. Toxicol Pathol 38: 180-181. http://dx.doi.org/10.1177/0192623309356450                         |
| 3848 | Kirby, PE; Pizzarello, RF; Lawlor, TE; Haworth, SR; Hodgson, JR. (1983). Evaluation of di-(2-           |
| 3849 | ethylhexyl)phthalate and its major metabolites in the Ames test and L5178Y mouse lymphoma               |
| 3850 | mutagenicity assay. Environ Mutagen 5: 657-663. <u>http://dx.doi.org/10.1002/em.2860050504</u>          |
| 3851 | Klaunig, JE; Babich, MA; Baetcke, KP; Cook, JC; Corton, JC; David, RM; Deluca, JG; Lai, DY;             |
| 3852 | McKee, RH; Peters, JM; Roberts, RA; Fenner-Crisp, PA. (2003). PPARalpha agonist-induced                 |
| 3853 | rodent tumors: modes of action and human relevance [Review]. Crit Rev Toxicol 33: 655-780.              |
| 3854 | http://dx.doi.org/10.1080/713608372   |
| 3855 | Kleinsasser, NH; Harreus, UA; Kastenbauer, ER; Wallner, BC; Sassen, AW; Staudenmaier, R;                |
| 3856 | Rettenmeier, AW. (2004). Mono(2-ethylhexyl)phthalate exhibits genotoxic effects in human                |
| 3857 | lymphocytes and mucosal cells of the upper aerodigestive tract in the comet assay. Toxicol Lett         |
| 3858 | 148: 83-90. http://dx.doi.org/10.1016/j.toxlet.2003.12.013  |
| 3859 | Kleinsasser, NH; Kastenbauer, ER; Weissacher, H; Muenzenrieder, RK; Harreus, UA. (2000a).               |
| 3860 | Phthalates demonstrate genotoxicity on human mucosa of the upper aerodigestive tract. Environ           |
| 3861 | Mol Mutagen 35: 9-12. http://dx.doi.org/10.1002/(SICI)1098-2280(2000)35:1<9::AID-                       |
| 3862 | <u>EM2&gt;3.0.CO;2-1</u>  |
| 3863 | Kleinsasser, NH; Wallner, BC; Kastenbauer, ER; Muenzenrieder, RK; Harreus, UA. (2000b).                 |
| 3864 | Comparing the genotoxic sensitivities of human peripheral blood lymphocytes and mucosa cells            |
| 3865 | of the upper aerodigestive tract using the Comet assay. Mutat Res 467: 21-30.                           |
| 3866 | http://dx.doi.org/10.1016/S1383-5718(00)00022-X   |
| 3867 | Kleinsasser, NH; Wallner, BC; Kastenbauer, ER; Weissacher, H; Harréus, UA. (2001). Genotoxicity of      |
| 3868 | di-butyl-phthalate and di-iso-butyl-phthalate in human lymphocytes and mucosal cells. Teratog           |
| 3869 | Carcinog Mutagen 21: 189-196. http://dx.doi.org/10.1002/tcm.1007  |
| 3870 | Kluwe, WM; Huff, JE; Matthews, HB; Irwin, R; Haseman, JK. (1985). Comparative chronic toxicities        |
| 3871 | and carcinogenic potentials of 2-ethylhexyl-containing compounds in rats and mice.                      |
| 3872 | Carcinogenesis 6: 1577-1583. http://dx.doi.org/10.1093/carcin/6.11.1577                                 |

| 3873         | Kluwe, WM; McConnell, EE; Huff, JE; Haseman, JK; Douglas, JF; Hartwell, WV. (1982).  |
|--------------|--|
| 3874         | Carcinogenicity testing of phthalate esters and related compounds by the National Toxicology   |
| 3875         | Program and the National Cancer Institute. Environ Health Perspect 45: 129-133.  |
| 3876         | http://dx.doi.org/10.2307/3429396  |
| 3877         | Kornbrust, DJ; Barfknecht, TR; Ingram, P; Shelburne, JD. (1984). Effect of di(2-ethylhexyl) phthalate  |
| 3878         | on DNA repair and lipid peroxidation in rat hepatocytes and on metabolic cooperation in Chinese  |
| 3879         | hamster V-79 cells. J Toxicol Environ Health 13: 99-116.   |
| 3880         | http://dx.doi.org/10.1080/15287398409530484  |
| 3881         | Kozumbo, WJ; Kroll, R; Rubin, RJ. (1982). Assessment of the mutagenicity of phthalate esters. Environ  |
| 3882         | Health Perspect 45: 103-109. http://dx.doi.org/10.2307/3429391   |
| 3883         | Kruger, T; Long, M; Bonefeld-Jørgensen, EC. (2008). Plastic components affect the activation of the  |
| 3884         | aryl hydrocarbon and the androgen receptor. Toxicology 246: 112-123.   |
| 3885         | http://dx.doi.org/10.1016/j.tox.2007.12.028  |
| 3886         | Kurata, H. (1975). Studies on the mutagenic effects of phthalates. Report to Ministry of Health and  |
| 3887         | Welfare (Japan), Scientific Research on Food Hygiene Program. (as cited in Omori 1976).  |
| 3888         | Kurata, H.   |
| 3889         | Lake, B; Kozlen, S; Evans, J; Gray, T; Young, P; Gangolli, S. (1987). Effect of prolonged  |
| 3890         | administration of clofibric acid and di-(2-ethylhexyl)phthalate on hepatic enzyme activities and   |
| 3891         | lipid peroxidation in the rat. Toxicology 44: 213-228. <u>http://dx.doi.org/10.1016/0300-</u>  |
| 3892         | <u>483X(87)90151-X</u>   |
| 3893         | Lake, BG; Foster, JR; Collins, MA; Stubberfield, CR; Gangolli, SD; Srivastava, SP. (1982). Studies on  |
| 3894         | the effects of orally administered dicyclohexyl phthalate in the rat. Acta Pharmacol Toxicol 51:   |
| 3895         | 217-226. http://dx.doi.org/10.1111/j.1600-0773.1982.tb01017.x  |
| 3896         | Lake, BG; Gray, TJ; Foster, JR; Stubberfield, CR; Gangolli, SD. (1984). Comparative studies on di-(2-  |
| 3897         | ethylhexyl) phthalate-induced hepatic peroxisome proliferation in the rat and hamster. Toxicol   |
| 3898         | Appl Pharmacol 72: 46-60. <u>http://dx.doi.org/10.1016/0041-008X(84)90248-5</u>  |
| 3899         | Laurenzana, EM; Coslo, DM; Vigilar, MV; Roman, AM; Omiecinski, CJ. (2016). Activation of the   |
| 3900         | Constitutive Androstane Receptor by Monophthalates. Chem Res Toxicol 29: 1651-1661.  |
| 3901         | http://dx.doi.org/10.1021/acs.chemrestox.6b00186   |
| 3902         | Le Hégarat, L; Mourot, A; Huet, S; Vasseur, L; Camus, S; Chesné, C; Fessard, V. (2014). Performance  |
| 3903         | of comet and micronucleus assays in metabolic competent HepaRG cells to predict in vivo  |
| 3904         | genotoxicity. Toxicol Sci 138: 300-309. <u>http://dx.doi.org/10.1093/toxsci/kfu004</u>   |
| 3905         | Leboeuf, RA; Kerckaert, GA; Aardema, MJ; Gibson, DP; Brauninger, R; Isfort, RJ. (1996). The pH 6.7   |
| 3906         | Syrian hamster embryo cell transformation assay for assessing the carcinogenic potential of  |
| 3907         | chemicals [Review]. Mutat Res 356: 85-127. <u>http://dx.doi.org/10.1016/0027-510/(95)00199-9</u>   |
| 3908         | Lee, J; Lim, K1. (2011). Plant-originated glycoprotein (24 kDa) has an inhibitory effect on proliferation  |
| 3909         | of BNL CL.2 cells in response to di(2-ethylhexyl)phthalate. Cell Biochem Funct 29: 496-505.  |
| 3910         | <u>http://dx.doi.org/10.1002/cbi.1///</u>  |
| 3911         | Lee, JW; Lee, SJ; Gye, MC; Moon, EY. (2019). Genotoxicity and glucose tolerance induction by   |
| 3912         | acetyltrietnylcitrate, substitute plasticizer compared to di(2-ethylnexyl)phthalate. Sci Rep 9:  |
| 3913         | $\frac{12237. \text{ http://dx.doi.org/10.1038/s41598-019-48599-y}{12237. \text{ http://dx.doi.0018/s41598-019-48599-y}{12237.  http://dx.doi.0018/s41598-019-48599-019-48599-019-48598-019-48598-019-48598-019-4859-019-48598-019-48588-019-48598-019-48588-019-48588-019-48588-019-48588-019-48588-019-48588-019-48588-019-48588-019-48588-019-48588-019-48588-019-485888-019-485888-019-485888-019-48588-019-4888-019-48888-019-488888$ |
| 3914         | Leone, IC; Weinneimer, CJ; Kelly, DP. (1999). A critical role for the peroxisome proliferator-activated  |
| 3915         | receptor alpha (PPARalpha) in the cellular fasting response: the PPARalpha-null mouse as a   |
| 3910<br>2017 | model of fatty acid oxidation disorders. Proc Nati Acad Sci USA 96: 7473-7478.   |
| 371/<br>2010 | Intp.//ux.doi.org/10.10/5/pilas.90.15./4/5<br>Lington AW: Bird MC: Districk DT: Styphilafield WA: Scale DA (1007) Chaomic toxicity and   |
| 2010         | Lington, Aw, Dird, MO; Flutinck, KT; Studdleneid, WA; Scala, KA. (1997). Unformet loxicity and   |
| 2020         | bttp://dv.doi.org/10.1002/towoi/26.1.70  |
| 3720         | <u>nup.//ux.u01.01g/10.1095/10x801/50.1./9</u>   |

| 3921 | Litton Bionetics. (1985). Evaluation of 1C dibutyl phthalate in the in vitro transformation of BALB/3T3  |
|------|--|
| 3922 | cells assay final report. (EPA/OTS Doc #40-8526194). Washington, DC: Chemical                            |
| 3923 | Manufacturers Association.   |
| 3924 | https://ntrl.ntis.gov/NTRL/dashboard/searchResults/titleDetail/OTS0508504.xhtml                          |
| 3925 | Liu, C; Deng, YL; Zheng, TZ; Yang, P; Jiang, XQ; Liu, EN; Miao, XP; Wang, LQ; Jiang, M; Zeng, Q.         |
| 3926 | (2020). Urinary biomarkers of phthalates exposure and risks of thyroid cancer and benign                 |
| 3927 | nodule. J Hazard Mater 383: 121189. http://dx.doi.org/10.1016/j.jhazmat.2019.121189                      |
| 3928 | Lopez-Carrillo, L; Hernandez-Ramirez, RU; Calafat, AM; Torres-Sanchez, L; Galvan-Portillo, M;            |
| 3929 | Needham, LL; Ruiz-Ramos, R; Cebrian, ME. (2010). Exposure to phthalates and breast cancer                |
| 3930 | risk in northern Mexico. Environ Health Perspect 118: 539-544.   |
| 3931 | http://dx.doi.org/10.1289/ehp.0901091  |
| 3932 | Lu, Z; Zhang, C; Han, C; An, Q; Cheng, Y; Chen, Y; Meng, R; Zhang, Y; Su, J. (2019). Plasticizer         |
| 3933 | Bis(2-ethylhexyl) Phthalate Causes Meiosis Defects and Decreases Fertilization Ability of                |
| 3934 | Mouse Oocytes in Vivo. J Agric Food Chem 67: 3459-3468. [Journal of agricultural and food                |
| 3935 | chemistry], http://dx.doi.org/10.1021/acs.jafc.9b00121   |
| 3936 | Lutz, WK. (1986). Investigation of the potential for binding of di(2-ethylhexyl) phthalate (DEHP) to rat |
| 3937 | liver DNA in vivo. Environ Health Perspect 65: 267-269. http://dx.doi.org/10.2307/3430193                |
| 3938 | Maronpot, RR; Nyska, A; Foreman, JE; Ramot, Y. (2016). The legacy of the F344 rat as a cancer            |
| 3939 | bioassay model (a retrospective summary of three common F344 rat neoplasms) [Review]. Crit               |
| 3940 | Rev Toxicol 46: 641-675. http://dx.doi.org/10.1080/10408444.2016.1174669                                 |
| 3941 | Marotta, V; Russo, G; Gambardella, C; Grasso, M; La Sala, D; Chiofalo, MG; D'Anna, R; Puzziello, A;      |
| 3942 | Docimo, G; Masone, S; Barbato, F; Colao, A; Faggiano, A; Grumetto, L. (2019). Human                      |
| 3943 | exposure to bisphenol AF and diethylhexylphthalate increases susceptibility to develop                   |
| 3944 | differentiated thyroid cancer in patients with thyroid nodules. Chemosphere 218: 885-894.                |
| 3945 | http://dx.doi.org/10.1016/j.chemosphere.2018.11.084  |
| 3946 | Marsman, DS; Cattley, RC; Conway, JG; Popp, JA. (1988). Relationship of hepatic peroxisome               |
| 3947 | proliferation and replicative DNA synthesis to the hepatocarcinogenicity of the peroxisome               |
| 3948 | proliferators di(2-ethylhexyl)phthalate and [4-chloro-6-(2,3-xylidino)-2-pyrimidinylthio]acetic          |
| 3949 | acid (Wy-14,643) in rats. Cancer Res 48: 6739-6744.  |
| 3950 | Martinez-Nava, GA; Burguete-Garcia, A; Lopez-Carrillo, L; Hernandez-Ramirez, RU; Madrid-Marina,          |
| 3951 | V; Cebrian, ME. (2013). PPAR gamma and PPARGC1B polymorphisms modify the association                     |
| 3952 | between phthalate metabolites and breast cancer risk. Biomarkers 18: 493-501.                            |
| 3953 | http://dx.doi.org/10.3109/1354750X.2013.816776   |
| 3954 | Mauthe, RJ; Gibson, DP; Bunch, RT; Custer, L. (2001). The syrian hamster embryo (SHE) cell               |
| 3955 | transformation assay: review of the methods and results. Toxicol Pathol 29 Suppl: 138-146.               |
| 3956 | http://dx.doi.org/10.1080/019262301753178546   |
| 3957 | Mérida-Ortega, Á; Hernández-Alcaraz, C; Hernández-Ramírez, RU; García-Martínez, A; Trejo-                |
| 3958 | Valdivia, B; Salinas-Rodríguez, A; Svensson, K; Cebrián, ME; Franco-Marina, F; López-                    |
| 3959 | Carrillo, L. (2016). Phthalate exposure, flavonoid consumption and breast cancer risk among              |
| 3960 | Mexican women. Environ Int 96: 167-172. http://dx.doi.org/10.1016/j.envint.2016.08.023                   |
| 3961 | Miao, H; Liu, X; Li, J; Zhang, L; Zhao, Y; Liu, S; Ni, S; Wu, Y. (2020). Associations of urinary         |
| 3962 | phthalate metabolites with risk of papillary thyroid cancer. Chemosphere 241: 125093.                    |
| 3963 | http://dx.doi.org/10.1016/j.chemosphere.2019.125093  |
| 3964 | Mikalsen, SO; Holen, I; Sanner, T. (1990). Morphological transformation and catalase activity of Syrian  |
| 3965 | hamster embryo cells treated with hepatic peroxisome proliferators, TPA and nickel sulphate.             |
| 3966 | Cell Biol Toxicol 6: 1-13. http://dx.doi.org/10.1007/BF00135022  |

May 2025

3967 Mitchell, FE; Price, SC; Hinton, RH; Grasso, P; Bridges, JW. (1985). Time and dose-response study of the effects on rats of the plasticizer di(2-ethylhexyl) phthalate. Toxicol Appl Pharmacol 81: 371-3968 3969 392. http://dx.doi.org/10.1016/0041-008x(85)90409-0 3970 Monsanto. (1976a). Microbial plate assay. (Project no. BO-76-017). Monsanto. (1976b). Mutagenicity plate assay: Santicizer 160. (Project no. LF-76-124C). 3971 3972 Monsanto. (1976c). Mutagenicity testing of S-160 and benzyl chloride; Project no. BO-76-243-244. 3973 (Project no. BO-76-243-244). 3974 Monsanto. (1985). Evaluation of Santicizer 160 in the in vitro transformation of Balb/3T3 cell assay. 3975 (Project no. XX-85-069). 3976 Morgan, M; Deoraj, A; Felty, Q; Roy, D. (2016). Environmental estrogen-like endocrine disrupting 3977 chemicals and breast cancer. Mol Cell Endocrinol 457: 89-102. 3978 http://dx.doi.org/10.1016/j.mce.2016.10.003 Mortensen, A; Bertram, M; Aarup, V; Sorensen, IK. (2002). Assessment of carcinogenicity of di(2-3979 3980 ethylhexyl)phthalate in a short-term assay using Xpa-/- and Xpa-/-/p53+/- mice. Toxicol Pathol 3981 30: 188-199. http://dx.doi.org/10.1080/019262302753559524 Mylchreest, E; Sar, M; Cattley, RC; Foster, PMD. (1999). Disruption of androgen-regulated male 3982 3983 reproductive development by di(n-butyl) phthalate during late gestation in rats is different from 3984 flutamide. Toxicol Appl Pharmacol 156: 81-95. http://dx.doi.org/10.1006/taap.1999.8643 Mylchreest, E; Wallace, DG; Cattley, RC; Foster, PM. (2000). Dose-dependent alterations in androgen-3985 3986 regulated male reproductive development in rats exposed to di(n-butyl) phthalate during late 3987 gestation. Toxicol Sci 55: 143-151. http://dx.doi.org/10.1093/toxsci/55.1.143 3988 NASEM. (2017). Application of systematic review methods in an overall strategy for evaluating low-3989 dose toxicity from endocrine active chemicals. In Consensus Study Report. Washington, D.C.: 3990 The National Academies Press. http://dx.doi.org/10.17226/24758 3991 NICNAS. (2008a). Existing chemical hazard assessment report: Diisobutyl phthalate. Sydney, Australia: 3992 National Industrial Chemicals Notification and Assessment Scheme. 3993 https://www.nicnas.gov.au/\_\_data/assets/pdf\_file/0006/4965/DIBP-hazard-assessment.pdf 3994 NICNAS. (2008b). Existing chemical hazard assessment report: Diisodecyl phthalate. Sydney, 3995 Australia: Australian Government Department of Health and Ageing. https://www.industrialchemicals.gov.au/sites/default/files/Diisodecyl%20phthalate%20DIDP.pdf 3996 3997 NICNAS. (2008c). Phthalates hazard compendium: A summary of physicochemical and human health 3998 hazard data for 24 ortho-phthalate chemicals. Sydney, Australia: Australian Department of 3999 Health and Ageing, National Industrial Chemicals Notification and Assessment Scheme. 4000 https://www.regulations.gov/document/EPA-HQ-OPPT-2010-0573-0008 4001 NICNAS. (2010). Priority existing chemical draft assessment report: Diethylhexyl phthalate. (PEC32). 4002 Sydney, Australia: Australian Department of Health and Ageing. 4003 https://www.industrialchemicals.gov.au/sites/default/files/PEC32-Diethylhexyl-phthalate-4004 DEHP.pdf 4005 NICNAS. (2012). Priority existing chemical assessment report no. 35: Diisononyl phthalate. (PEC35). 4006 Sydney, Australia: Australian Government Department of Health and Ageing. 4007 https://www.industrialchemicals.gov.au/sites/default/files/PEC35-Diisononyl-phthalate-4008 DINP.pdf 4009 NICNAS. (2013). Priority existing chemical assessment report no. 36: Dibutyl pthalate. (PEC36). 4010 Sydney, Australia: Australian Department of Health, National Industrial Chemicals Notification 4011 and Assessment Scheme. https://www.industrialchemicals.gov.au/sites/default/files/PEC36-4012 Dibutyl-phthalate-DBP.pdf

| 4013 | NICNAS. (2015a). Priority existing chemical assessment report no. 40: Butyl benzyl phthalate.          |
|------|--|
| 4014 | (PEC40). Sydney, Australia: Australian Government Department of Health and Ageing.                     |
| 4015 | https://heronet.epa.gov/heronet/index.cfm/reference/download/reference_id/3664467                      |
| 4016 | NICNAS. (2015b). Priority existing chemical draft assessment report: Diisodecyl Phthalate & Di-n-octyl |
| 4017 | Phthalate. Sydney, Australia: Australian Department of Health and Ageing, National Industrial          |
| 4018 | Chemicals Notification and Assessment Scheme.  |
| 4019 | https://www.industrialchemicals.gov.au/sites/default/files/PEC39-Diisodecyl-phthalate-DIDP-            |
| 4020 | Di-n-octyl-phthalate-DnOP.pdf  |
| 4021 | NICNAS. (2016). C4-6 side chain transitional phthalates: Human health tier II assessment. Sydney,      |
| 4022 | Australia: Australian Department of Health, National Industrial Chemicals Notification and             |
| 4023 | Assessment Scheme. https://www.industrialchemicals.gov.au/sites/default/files/C4-                      |
| 4024 | 6%20side%20chain%20transitional%20phthalates_Human%20health%20tier%20II%20assessm                      |
| 4025 | <u>ent.pdf</u>   |
| 4026 | NRC. (2008). Phthalates and cumulative risk assessment: The task ahead. Washington, DC: National       |
| 4027 | Academies Press. http://dx.doi.org/10.17226/12528  |
| 4028 | NTP-CERHR. (2003a). NTP-CERHR monograph on the potential human reproductive and                        |
| 4029 | developmental effects of butyl benzyl phthalate (BBP). (NIH Publication No. 03-4487).                  |
| 4030 | https://ntp.niehs.nih.gov/ntp/ohat/phthalates/bb-phthalate/bbp_monograph_final.pdf                     |
| 4031 | NTP-CERHR. (2003b). NTP-CERHR monograph on the potential human reproductive and                        |
| 4032 | developmental effects of di-isodecyl phthalate (DIDP). (NIH 03-4485). Research Triangle Park,          |
| 4033 | NC: National Toxicology Program Center for the Evaluation of Risks to Human Reproduction.              |
| 4034 | http://ntp.niehs.nih.gov/ntp/ohat/phthalates/didp/didp_monograph_final.pdf                             |
| 4035 | NTP-CERHR. (2003c). NTP-CERHR monograph on the potential human reproductive and                        |
| 4036 | developmental effects of di-isononyl phthalate (DINP) (pp. i-III90). (NIH Publication No. 03-          |
| 4037 | 4484). Research Triangle Park, NC: National Toxicology Program Center for the Evaluation of            |
| 4038 | Risks to Human Reproduction.   |
| 4039 | http://ntp.niehs.nih.gov/ntp/ohat/phthalates/dinp/dinp_monograph_final.pdf                             |
| 4040 | NTP-CERHR. (2003d). NTP-CERHR Monograph on the Potential Human Reproductive and                        |
| 4041 | Developmental Effects of Di-n-Butyl Phthalate (DBP) (pp. 169). Research Triangle Park, NC:             |
| 4042 | Center for the Evaluation of Risks to Human Reproduction/National Toxicology Program-                  |
| 4043 | National Institute of Environmental Health Sciences.   |
| 4044 | https://heronet.epa.gov/heronet/index.cfm/reference/download/reference_id/1332562                      |
| 4045 | <u>NTP-CERHR. (2006)</u> . NTP-CERHR monograph on the potential human reproductive and                 |
| 4046 | developmental effects of di(2-ethylhexyl) phthalate (DEHP) [NTP]. (NIH Publication No. 06-             |
| 4047 | 4476). Research Triangle Park, NC. http://cerhr.niehs.nih.gov/evals/phthalates/dehp/DEHP-              |
| 4048 | <u>Monograph.pdf</u>   |
| 4049 | NTP. (1982a). Carcinogenesis bioassay of di(2-ethylhexyl)phthalate (CAS No. 117-81-7) in F344 rats     |
| 4050 | and B6C3F1 mice (feed studies) [NTP]. In National Toxicology Program Technical Report (pp.             |
| 4051 | 1-127). (NTP-80-37). Research Triangle Park, NC.   |
| 4052 | http://ntp.niehs.nih.gov/ntp/htdocs/LT_rpts/tr217.pdf  |
| 4053 | NTP. (1982b). Carcinogensis bioassay of butyl benzyl phthalate (CAS No. 85-68-7) in F344/N rats and    |
| 4054 | B6C3F1 mice (feed study). In NTP Technical Report (pp. 1-98). (ISSN 0888-8051                          |
| 4055 | NTP TR 213). Research Triangle Park, NC. <u>http://www.ncbi.nlm.nih.gov/pubmed/12778222</u>            |
| 4056 | NTP. (1995). NTP technical report on the toxicity studies of dibutyl phthalate (CAS No. 84-74-2)       |
| 4057 | administered in feed to F344/N rats and B6C3F1 mice [NTP] (pp. 1-G5). (ISSN 1521-4621                  |
| 4058 | NIH Publication 95-3353). Research Triangle Park, NC.  |
| 4059 | http://ntp.niehs.nih.gov/ntp/htdocs/ST_rpts/tox030.pdf   |
|      |  |

| 4060 | <u>NTP. (1997a)</u> . Effect of dietary restriction on toxicology and carcinogenesis studies in F344/N rats and |
|------|---|
| 4061 | B6C3F1 mice [NTP]. (TR-460). Research Triangle Park, NC.  |
| 4062 | http://ntp.niehs.nih.gov/?objectid=070A674C-9392-953F-80A7F3BC5DCCB3FB  |
| 4063 | <u>NTP. (1997b)</u> . Toxicology and carcinogenesis studies of butyl benzyl phthalate (CAS No. 85-68-7) in      |
| 4064 | F344/N rats (feed studies). (TR-458). Research Triangle Park, NC.   |
| 4065 | http://ntp.niehs.nih.gov/results/pubs/longterm/reports/longterm/tr400499/abstracts/tr458/index.ht               |
| 4066 |   |
| 4067 | <u>NTP. (2016)</u> . Report on Carcinogens, 14th edition: Di(2-ethylhexyl) phthalate. Research Triangle Park,   |
| 4068 | NC: U.S. Department of Health and Human Services, National Toxicology Program.                                  |
| 4069 | <u>NTP. (2018)</u> . National Toxicology Program approach to genomic dose-response modeling. (NTP               |
| 4070 | Research Report No. 5). <u>https://ntp.niehs.nih.gov/ntp/results/pubs/rr/reports/rr05_508.pdf</u>               |
| 4071 | <u>NTP. (2021a)</u> . NTP technical report on the toxicology and carcinogenesis studies of di-n-butyl phthalate |
| 4072 | (CASRN 84-74-2) administered in feed to Sprague Dawley (HSD: Sprague Dawley® SD®) rats                          |
| 4073 | and B6C3F1/n mice. (Technical Report 600). Research Triangle Park, NC.  |
| 4074 | http://dx.doi.org/10.22427/NTP-TR-600   |
| 4075 | <u>NTP. (2021b)</u> . NTP technical report on the toxicology and carcinogenesis studies of di(2-ethylhexyl)     |
| 4076 | phthalate (CASRN 117-81-7) administered in feed to Sprague Dawley (Hsd:Sprague Dawley®                          |
| 4077 | SD®) rats [NTP]. (TR 601). U.S. Department of Health and Human Services.  |
| 4078 | https://ntp.niehs.nih.gov/ntp/htdocs/lt_rpts/tr601_508.pdf  |
| 4079 | <u>Oberly, TJ; Bewsey, BJ; Probst, GS. (1985)</u> . Tests for the induction of forward mutation at the          |
| 4080 | thymidine kinase locus of L5178Y mouse lymphoma cells in culture. In J Ashby; FJ de Serres;                     |
| 4081 | M Draper; M Ishidate, Jr.; BH Margolin; BE Matter; MD Shelby (Eds.), (pp. 569-582).                             |
| 4082 | Amsterdam, The Netherlands: Elsevier.   |
| 4083 | <u>OECD. (2016)</u> . Test No. 473: In vitro mammalian chromosomal aberration test. Paris, France.              |
| 4084 | http://dx.doi.org/10.1787/9789264264649-en  |
| 4085 | <u>OECD. (2024)</u> . Case study on the use of Integrated Approaches for Testing and Assessment (IATA) for      |
| 4086 | chronic toxicity and carcinogenicity of agrichemicals with exemplar case studies - ninth review                 |
| 4087 | cycle (2023). In Series on Testing and Assessment No 402. Paris, France: OECD Publishing.                       |
| 4088 | http://dx.doi.org/10.178//c3b9ac3/-en   |
| 4089 | <u>OEHHA. (1986)</u> . Safe Drinking Water and Toxic Enforcement Act of 1986 Proposition 65. Initial            |
| 4090 | Statement of Reasons. Title 27, California Code of Regulations. Proposed amendment to Section                   |
| 4091 | 25805(b), Specific Regulatory Levels: Chemicals Causing Reproductive Toxicity. Butyl benzyl                     |
| 4092 | phthalate (oral exposure). California: California Environmental Protection Agency, Office of                    |
| 4093 | Environmental Health Hazard Assessment. <u>https://oehha.ca.gov/media/downloads/proposition-</u>                |
| 4094 | <u>65/chemicals/060112bbpisor.pdf</u>   |
| 4095 | <u>OEHHA. (2007)</u> . Proposition 65 Maximum Allowable Dose Level (MADL) for reproductive toxicity for         |
| 4096 | di(n-butyl)phthalate (DBP). California: California Environmental Protection Agency, Office of                   |
| 4097 | Environmental Health Hazard Assessment, Reproductive and Cancer Hazard Assessment                               |
| 4098 | Section. https://oenna.ca.gov/media/downloads/proposition-65/chemicals/dbpmadi06290/.pdf                        |
| 4099 | <u>OEHHA. (2013a)</u> . Chemical listed effective December 20, 2013 as known to the state of California to      |
| 4100 | cause cancer: Diisononyi phthalate (DINP) [website]. <u>https://oenna.ca.gov/proposition-</u>                   |
| 4101 | 65/crnf/cnemical-listed-effective-december-20-2013-known-state-california-cause-cancer                          |
| 4102 | <u>OEHHA. (2013b)</u> . Evidence on the carcinogenicity of butyl benzyl phthalate. California: California       |
| 4103 | Environmental Protection Agency, Office of Environmental Health Hazard Assessment,                              |
| 4104 | Reproductive and Cancer Hazard Assessment Branch.   |
| 4105 | nups://oenna.ca.gov/media/downioads/proposition-65/chemicals/bbpnid10042013.pdf                                 |
| 4106 | <u>OEHHA. (2022)</u> . Proposition 65: Di(2-ethylnexyl)phthalate (DEHP) [Website].                              |
| 4107 | https://oehha.ca.gov/proposition-65/chemicals/di2-ethylhexylphthalate-dehp                                      |

| 4108 | Ohlson, CG; Hardell, L. (2000). Testicular cancer and occupational exposures with a focus on   |
|------|--|
| 4109 | xenoestrogens in polyvinyl chloride plastics. Chemosphere 40: 1277-1282.   |
| 4110 | http://dx.doi.org/10.1016/S0045-6535(99)00380-X  |
| 4111 | Oishi, S; Hiraga, K. (1980a). Effect of phthalic acid esters on mouse testes. Toxicol Lett 5: 413-416.   |
| 4112 | http://dx.doi.org/10.1016/0378-4274(80)90024-7   |
| 4113 | Oishi, S; Hiraga, K. (1980b). Effects of phthalic acid monoesters on mouse testes. Toxicol Lett 6: 239-  |
| 4114 | 242. http://dx.doi.org/10.1016/0378-4274(80)90126-5  |
| 4115 | Oishi, S; Hiraga, K. (1980c). Testicular atrophy induced by phthalic acid esters: Effect on testosterone   |
| 4116 | and zinc concentrations. Toxicol Appl Pharmacol 53: 35-41. http://dx.doi.org/10.1016/0041-   |
| 4117 | <u>008X(80)90378-6</u>   |
| 4118 | Oishi, S; Hiraga, K. (1980d). Testicular atrophy induced by phthalic acid monoesters: Effects of zinc  |
| 4119 | and testosterone concentrations. Toxicology 15: 197-202. <u>http://dx.doi.org/10.1016/0300-</u>  |
| 4120 | <u>483X(80)90053-0</u>   |
| 4121 | Okai, Y; Higashi-Okai, K. (2000). Enhancing effect of a plastic plasticizer, di-2-ethylhexyl phthalate on  |
| 4122 | umu C gene expression in Salmonella typhimurium (TA 1535/pSK 1002). J UOEH 22: 305-315.  |
| 4123 | http://dx.doi.org/10.7888/juoeh.22.305   |
| 4124 | <u>Omori, Y. (1976)</u> . Recent progress in safety evaluation studies on plasticizers and plastics and their  |
| 4125 | controlled use in Japan. Environ Health Perspect 17: 203-209. <u>http://dx.doi.org/10.230//342862/</u>   |
| 4126 | Pant, K; Sly, J; Bruce, S; Scott, A; Carmichael, P; San, R. (2010). Syrian Hamster Embryo (SHE) cell   |
| 4127 | transformation assay with and without X-ray irradiation of feeder cells using Di(2-  |
| 4128 | ethylnexyl)phthalate (DEHP) and N-nitroso-N-methylnitroguanidine (MINNG). Mutat Res 698:   |
| 4129 | 6-10. <u>http://dx.doi.org/10.1016/j.mrgentox.2010.02.01/</u><br>Darada H: Cammon MD: Chan I: Calafat AM: Nayayut AI: Sontalla DM: Walff MS: Taitalhaum  |
| 4130 | Parada, H; Gammon, MD; Chen, J; Calaral, AW; Neugul, AI; Santella, KW; Wolli, MS; Telleloaum,<br>SL (2018) Urinerry Dethalata Matabalita Concentrations and Presst Concer Incidence and                |
| 4131 | <u>SL</u> . (2016). Officially Finitiatate Metabolite Concentrations and Breast Cancer Incluence and<br>Survival following Project Concer: The Long Island Project Concer Study Project Environ Health |
| 4132 | Perspect 126: 047013 http://dx.doi.org/10.1289/EHP2083   |
| 4133 | Park SY: Choi I (2007) Cytotoxicity genotoxicity and ecotoxicity assay using human cell and  |
| 4135 | environmental species for the screening of the risk from pollutant exposure. Environ Int 33: 817-  |
| 4136 | 822. http://dx doi org/10.1016/i envint 2007.03.014  |
| 4137 | Parry, JM: Arni, P: Brooks, T: Carere, A: Ferguson, L: Heinisch, J: Inge-Vechtomov, S: Loprieno, N:  |
| 4138 | Nestmann, E: von Borstel, R. (1985). Summary report on the performance of the yeast and  |
| 4139 | aspergillus assays. In J Ashby: FJ de Serres: M Draper: M Ishidate. Jr.: BH Margolin: BESMD  |
| 4140 | Matter (Eds.), (pp. 25-46). Amsterdam, The Netherlands: Elsevier.  |
| 4141 | PDR. (1995). Physician's Desk Reference, 11th ed. Oradell, NJ: Medical Economics Data.   |
| 4142 | PDR. (2002). Physician's Desk Reference, 18th edition  |
| 4143 | Physician's Desk Reference: Tricor (fenofibrate). Oradell, NJ: Medical Economics Data.   |
| 4144 | Peters, JM; Cheung, C; Gonzalez, FJ. (2005). Peroxisome proliferator-activated receptor-alpha and liver  |
| 4145 | cancer: where do we stand [Review]. J Mol Med 83: 774-785. http://dx.doi.org/10.1007/s00109-   |
| 4146 | 005-0678-9   |
| 4147 | Phillips, BJ; Anderson, D; Gangolli, SD. (1986). Studies on the genetic effects of phthalic acid esters on   |
| 4148 | cells in culture. Environ Health Perspect 65: 263-266. http://dx.doi.org/10.2307/3430192   |
| 4149 | Phillips, BJ; James, TE; Gangolli, SD. (1982). Genotoxicity studies of di(2-ethylhexyl)phthalate and its   |
| 4150 | metabolites in CHO cells. Mutat Res 102: 297-304. http://dx.doi.org/10.1016/0165-  |
| 4151 | <u>1218(82)90139-2</u>   |
| 4152 | Pogribny, I; Tryndyak, V; Boureiko, A; Melnyk, S; Bagnyukova, T; Montgomery, B; Rusyn, I. (2008).  |
| 4153 | Mechanisms of peroxisome proliferator-induced DNA hypomethylation in rat liver. Mutat Res  |
| 4154 | 644: 17-23. http://dx.doi.org/10.1016/j.mrfmmm.2008.06.009   |

| 4155 | Poon, R; Lecavalier, P; Mueller, R; Valli, VE; Procter, BG; Chu, I. (1997). Subchronic oral toxicity of |
|------|---|
| 4156 | di-n-octyl phthalate and di(2-ethylhexyl) phthalate in the rat. Food Chem Toxicol 35: 225-239.          |
| 4157 | http://dx.doi.org/10.1016/S0278-6915(96)00064-6   |
| 4158 | Priston, RAJ; Dean, BJ. (1985). Tests for the induction of chromosome aberrations, polyploidy and       |
| 4159 | sister-chromatid exchanges in rat liver (RL4) cells. Prog Mutat Res 5: 387-395.                         |
| 4160 | Probst, GS; Hill, LE. (1985). Tests for the induction of DNA-repair synthesis in primary cultures of    |
| 4161 | adult rat hepatocytes. Prog Mutat Res 5: 381-386.   |
| 4162 | Pugh, G; Isenberg, J; Kamendulis, L; Ackley, D; Clare, L; Brown, R; Lington, A; Smith, J; Klaunig, J.   |
| 4163 | (2000). Effects of di-isononyl phthalate, di-2-ethylhexyl phthalate, and clofibrate in cynomolgus       |
| 4164 | monkeys. Toxicol Sci 56: 181-188. http://dx.doi.org/10.1093/toxsci/56.1.181                             |
| 4165 | Putman, DL; Moore, WA; Schechtman, LM; Hodgson, JR. (1983). Cytogenetic evaluation of di-(2-            |
| 4166 | ethylhexyl)phthalate and its major metabolites in Fischer 344 rats. Environ Mol Mutagen 5: 227-         |
| 4167 | 231. http://dx.doi.org/10.1002/em.2860050211  |
| 4168 | Radke, EG: Braun, JM: Meeker, JD: Cooper, GS. (2018). Phthalate exposure and male reproductive          |
| 4169 | outcomes: A systematic review of the human epidemiological evidence [Review]. Environ Int               |
| 4170 | 121: 764-793. http://dx.doi.org/10.1016/i.envint.2018.07.029  |
| 4171 | Radke, EG: Braun, JM: Nachman, RM: Cooper, GS. (2020). Phthalate exposure and neurodevelopment:         |
| 4172 | A systematic review and meta-analysis of human epidemiological evidence [Review]. Environ               |
| 4173 | Int 137: 105408. http://dx.doi.org/10.1016/i.envint.2019.105408   |
| 4174 | Radke, EG; Galizia, A: Thaver, KA: Cooper, GS, (2019a). Phthalate exposure and metabolic effects: A     |
| 4175 | systematic review of the human epidemiological evidence [Review]. Environ Int 132: 104768.              |
| 4176 | http://dx.doi.org/10.1016/j.envint.2019.04.040  |
| 4177 | Radke, EG; Glenn, BS; Braun, JM; Cooper, GS. (2019b). Phthalate exposure and female reproductive        |
| 4178 | and developmental outcomes: A systematic review of the human epidemiological evidence                   |
| 4179 | [Review]. Environ Int 130: 104580. http://dx.doi.org/10.1016/j.envint.2019.02.003                       |
| 4180 | Rao, MS; Usuda, N; Subbarao, V; Reddy, JK. (1987). Absence of gamma-glutamyl transpeptidase             |
| 4181 | activity in neoplastic lesions induced in the liver of male F-344 rats by di-(2-                        |
| 4182 | ethylhexyl)phthalate, a peroxisome proliferator. Carcinogenesis 8: 1347-1350.                           |
| 4183 | http://dx.doi.org/10.1093/carcin/8.9.1347   |
| 4184 | Rao, MS; Yeldandi, AV; Subbarao, V. (1990). Quantitative analysis of hepatocellular lesions induced by  |
| 4185 | di(2-ethylhexyl)phthalate in F-344 rats. J Toxicol Environ Health A 30: 85-89.                          |
| 4186 | http://dx.doi.org/10.1080/15287399009531413   |
| 4187 | Reddy, JK; Moody, DE; Azarnoff, DL; Rao, MS. (1976). Di-(2-ethylhexyl)phthalate: an industrial          |
| 4188 | plasticizer induces hypolipidemia and enhances hepatic catalase and carnitine acetyltransferase         |
| 4189 | activities in rat and mice. Life Sci 18: 941-945. http://dx.doi.org/10.1016/0024-3205(76)90412-4        |
| 4190 | Reddy, JK; Qureshi, SA. (1979). Tumorigenicity of the hypolipidaemic peroxisome proliferator ethyl-     |
| 4191 | alpha-p-chlorophenoxyisobutyrate (clofibrate) in rats. Br J Cancer 40: 476-482.                         |
| 4192 | http://dx.doi.org/10.1038/bjc.1979.203  |
| 4193 | Reddy, JK; Reddy, MK; Usman, MI; Lalwani, ND; Rao, MS. (1986). Comparison of hepatic                    |
| 4194 | peroxisome proliferative effect and its implication for hepatocarcinogenicity of phthalate esters,      |
| 4195 | di(2-ethylhexyl) phthalate, and di(2-ethylhexyl) adipate with a hypolipidemic drug. Environ             |
| 4196 | Health Perspect 65: 317-327. http://dx.doi.org/10.2307/3430199  |
| 4197 | Reeves, KW; Santana, MD; Manson, JE; Hankinson, SE; Zoeller, RT; Bigelow, C; Sturgeon, SR;              |
| 4198 | Spiegelman, D; Tinker, L; Luo, J; Chen, B; Meliker, J; Bonner, MR; Cote, ML; Cheng, TD;                 |
| 4199 | Calafat, AM. (2019). Urinary phthalate biomarker concentrations and postmenopausal breast               |
| 4200 | cancer risk. J Natl Cancer Inst 111: 1059-1067. http://dx.doi.org/10.1093/jnci/djz002                   |
| 4201 | Ren, H; Aleksunes, L; Wood, C; Vallanat, B; George, M; Klaassen, C; Corton, J. (2010).                  |
| 4202 | Characterization of peroxisome proliferator-activated receptor alphaindependent effects of              |
|      |   |

| 3 PPARalpha activators in the rodent liver: di-(2-ethylhexyl) phthalate also activates the             |        |
|--|--------|
| 4 constitutive-activated receptor. Toxicol Sci 113: 45-59. <u>http://dx.doi.org/10.1093/toxsci/kfp</u> | 0251   |
| 5 <u>Reynolds, CW; Foon, KA. (1984)</u> . T gamma-lymphoproliferative disease and related disorders in |        |
| 6 humans and experimental animals: A review of the clinical, cellular, and functional                  |        |
| 7 characteristics [Review]. Blood 64: 1146-1158.   |        |
| http://dx.doi.org/10.1182/blood.V64.6.1146.1146  |        |
| Roebuck, BD; Baumgartner, KJ; Macmillan, DL. (1993). Caloric restriction and intervention in           |        |
| pancreatic carcinogenesis in the rat. Cancer Res 53: 46-52.  |        |
| Roebuck, BD; Yager, JD; Longnecker, DS; Wilpone, SA. (1981). Promotion by Unsaturated Fat of           | Ĩ      |
| Azaserine-Induced Pancreatic Carcinogenesis in the Rat. Cancer Res 41: 3961-3966.                      |        |
| Ruddick, JA; Villeneuve, DC; Chu, I; Nestmann, E; Miles, D. (1981). An assessment of the               |        |
| teratogenicity in the rat and mutagenicity in Salmonella of mono-2-ethylhexyl phthalate. Bu            | ıll    |
| Environ Contam Toxicol 27: 181-186. http://dx.doi.org/10.1007/BF01611005                               |        |
| Rushbrook, CJ; Jorgenson, TA; Hodgson, JR. (1982). Dominant lethal study of di-(2-                     |        |
| ethylhexyl)phthalate and its major metabolites in ICR/SIM mice [Abstract]. Environ Mutag               | en 4:  |
| 387.   |        |
| Saillenfait, AM; Gallissot, F; Sabate, JP. (2009). Differential developmental toxicities of di-n-hexy  | 1      |
| phthalate and dicyclohexyl phthalate administered orally to rats. J Appl Toxicol 29: 510-52            | 1.     |
| http://dx.doi.org/10.1002/jat.1436   |        |
| Sanchez, JH; Abernethy, DJ; Boreiko, CJ. (1987). Lack of di-(2-ethylhexyl) phthalate activity in th    | e      |
| C3H/10T 1/2 cell transformation system. Toxicol In Vitro 1: 49-53.                                     |        |
| http://dx.doi.org/10.1016/0887-2333(87)90038-5   |        |
| Sanner, T; Rivedal, E. (1985). Tests with the Syrian hamster embryo (SHE) cell transformation ass      | ay. In |
| J Ashby; FJ de Serres; M Draper; M Ishidate, Jr.; BH Margolin; BE Matter; MD Shelby (Ec                | ds.),  |
| (pp. 665-671). Amsterdam, The Netherlands: Elsevier.   |        |
| Sato, T; Nagase, H; Sato, K; Niikawa, M; Kito, H. (1994). Enhancement of the mutagenicity of am        | ino    |
| acid pyrolysates by phthalate esters. Environ Mol Mutagen 24: 325-331.                                 |        |
| http://dx.doi.org/10.1002/em.2850240410  |        |
| Schmezer, P; Pool, BL; Klein, RG; Komitowski, D; Schmahl, D. (1988). Various short-term assays         | s and  |
| two long-term studies with the plasticizer di(2-ethylhexyl)phthalate in the Syrian golden has          | mster. |
| Carcinogenesis 9: 37-43. http://dx.doi.org/10.1093/carcin/9.1.37                                       |        |
| Seed, JL. (1982). Mutagenic activity of phthalate esters in bacterial liquid suspension assays. Envir  | on     |
| Health Perspect 45: 111-114. <u>http://dx.doi.org/10.2307/3429392</u>                                  |        |
| Selenskas, S; Teta, MJ; Vitale, JN. (1995). Pancreatic cancer among workers processing synthetic       |        |
| resins. Am J Ind Med 28: 385-398. http://dx.doi.org/10.1002/ajim.4700280308                            |        |
| Seo, KW; Kim, KB; Kim, YJ; Choi, JY; Lee, KT; Choi, KS. (2004). Comparison of oxidative stres          | s and  |
| changes of xenobiotic metabolizing enzymes induced by phthalates in rats. Food Chem Tox                | cicol  |
| 42: 107-114. http://dx.doi.org/10.1016/j.fct.2003.08.010   |        |
| Shahin, MM; Von Borstel, RC. (1977). Mutagenic and lethal effects of alpha-benzene hexachloride        | э,     |
| dibutyl phthalate and trichloroethylene in Saccharomyces cerevisiae. Mutat Res 48: 173-18              | 0.     |
| http://dx.doi.org/10.1016/0027-5107(77)90157-9   |        |
| Shin, M; Ohnishi, M; Iguchi, S; Sano, K; Umezawa, C. (1999). Peroxisome-Proliferator Regulates         | Key    |
| Enzymes of the Tryptophan-NAD+ Pathway. Toxicol Appl Pharmacol 158: 71-80.                             | 5      |
| http://dx.doi.org/10.1006/taap.1999.8683   |        |
| Short, RD; Robinson, EC; Lington, AW; Chin, AE. (1987). Metabolic and peroxisome proliferation         | n      |
| studies with di(2-ethylhexyl)phthalate in rats and monkeys. Toxicol Ind Health 3: 185-195.             |        |
| http://dx.doi.org/10.1177/074823378700300213   |        |
|  |        |
|  |        |

|      | May 2025  |
|------|---|
| 4250 | Simmon, VF; Kauhanen, K; Tardiff, RG. (1977). Mutagenic activity of chemicals identified in drinking  |
| 4251 | water. In D Scott; B Bridges; F Sobel (Eds.), Developments in Toxicology and Environmental  |
| 4252 | Science, 2 (pp. 249-258). New York, NY: Elsevier/North Holland Press.   |
| 4253 | Singh, AR; Lawrence, WH; Autian, J. (1974). Mutagenic and antifertility sensitivities of mice to di-2-  |
| 4254 | ethylhexyl phthalate (DEHP) and dimethoxyethyl phthalate (DMEP). Toxicol Appl Pharmacol   |
| 4255 | 29: 35-46. http://dx.doi.org/10.1016/0041-008X(74)90159-8   |
| 4256 | Smith-Oliver, T; Butterworth, BE. (1987). Correlation of the carcinogenic potential of di(2-  |
| 4257 | ethylhexyl)phthalate (DEHP) with induced hyperplasia rather than with genotoxic activity. Mutat   |
| 4258 | Res 188: 21-28. http://dx.doi.org/10.1016/0165-1218(87)90110-8  |
| 4259 | Smith, JH; Isenberg, JS; Pugh, G, Jr.; Kamendulis, LM; Ackley, D; Lington, AW; Klaunig, JE. (2000).   |
| 4260 | Comparative in vivo hepatic effects of Di-isononyl phthalate (DINP) and related C7-C11 dialkyl  |
| 4261 | phthalates on gap junctional intercellular communication (GJIC), peroxisomal beta-oxidation   |
| 4262 | (PBOX), and DNA synthesis in rat and mouse liver. Toxicol Sci 54: 312-321.  |
| 4263 | http://dx.doi.org/10.1093/toxsci/54.2.312   |
| 4264 | Soames, AR; Cliffe, S; Pate, I; Foster, JR. (1999). Quantitative analysis of the lobular distribution of S-   |
| 4265 | phase in rat liver following dietary administration of di(2-ethylhexyl)phthalate. Toxicol Pathol  |
| 4266 | 27: 436-440. http://dx.doi.org/10.1177/019262339902700407   |
| 4267 | Sobol, Z; Homiski, ML; Dickinson, DA; Spellman, RA; Li, D; Scott, A; Cheung, JR; Coffing, SL;   |
| 4268 | Munzner, JB; Sanok, KE; Gunther, WC; Dobo, KL; Schuler, M. (2012). Development and  |
| 4269 | validation of an in vitro micronucleus assay platform in TK6 cells. Mutat Res 746: 29-34.   |
| 4270 | http://dx.doi.org/10.1016/j.mrgentox.2012.02.005  |
| 4271 | Stenchever, MA; Allen, MA; Jerominski, L; Petersen, RV. (19/6). Effects of bis(2-ethylhexyl) phthalate  |
| 4272 | on chromosomes of human leukocytes and human fetal lung cells. J Pharm Sci 65: 1648-1651.   |
| 4273 | http://dx.doi.org/10.1002/jps.2600651121  |
| 4274 | Svoboda, DJ; Azarnoff, DL. (1979). Tumors in male rats fed ethyl chlorophenoxylsobutyrate, a  |
| 4275 | nypolipidemic drug. Cancer Res 39: 3419-3428.   |
| 4270 | <u>Iakagi, A; Sai, K; Umemura, I; Hasegawa, K; Kurokawa, Y</u> . (1990). Significant increase of 8-   |
| 4277 | proliferators di(2 athulhovul) with blate and di(2 athulhovul) adjuste. Jun I Cancer Des 81: 212  |
| 4270 | promerators $d(2-ethylnexyf)$ philarate and $d(2-ethylnexyf)$ adipate. Jph J Cancer Kes 81. 213-<br>215. http://dx.doi.org/10.1111/j.1240.7006.1000.tb02551.x |
| 4279 | Tamura H: Iida T: Watanabe T: Suga T (1991) Lack of induction of henatic DNA damage on long-  |
| 4280 | term administration of perovisome proliferators in male E-344 rats. Toxicology 69: 55-62  |
| 4281 | http://dx doi org/10.1016/0300-483X(91)90153-R  |
| 4283 | Tenkanen I : Mänttäri M: Kovanen PT: Virkkunen H: Manninen V (2006) Gemfibrozil in the  |
| 4284 | treatment of dyslipidemia: an 18-year mortality follow-up of the Helsinki Heart Study Arch  |
| 4285 | Intern Med 166: 743-748, http://dx.doi.org/10.1001/archinte.166.7.743   |
| 4286 | Tennant, RW: Margolin, BH: Shelby, MD: Zeiger, E: Haseman, JK: Snalding, J: Caspary, W: Resnick.  |
| 4287 | M: Stasiewicz, S: Anderson, B: Minor, R. (1987). Prediction of chemical carcinogenicity in  |
| 4288 | rodents from in vitro genetic toxicity assays. Science 236: 933-941.  |
| 4289 | http://dx.doi.org/10.1126/science.3554512   |
| 4290 | Thiess, A; Frentzel-Beyme, R; Wieland, R. (1978). [Mortality study in workers exposed to di-2-  |
| 4291 | ethylhexyl phthalate (DOP)]. In H Loskant (Ed.), (pp. 155–164). Stuttgart, Germany: Gentner.  |
| 4292 | Thiess, AM; Fleig, I. (1978). [Chromosome studies in workers exposed to di-2-ethylhexyl phthalate].   |
| 4293 | Zentralbl Arbeitsmed Arbeitsschutz Prophyl 28: 351-355.   |
| 4294 | Thomas, J; Haseman, JK; Goodman, JI; Ward, JM; Loughran, TP, Jr.; Spencer, PJ. (2007). A review of  |
| 4295 | large granular lymphocytic leukemia in Fischer 344 rats as an initial step toward evaluating the  |
| 4296 | Implication of the endpoint to human cancer risk assessment [Review]. Toxicol Sci 99: 3-19.   |
| 4297 | http://dx.doi.org/10.1093/toxsci/kfm098   |
|      |   |

| 1114 2020   |
|---|
| Thottassery, J; Winberg, L; Youssef, J; Cunningham, M; Badr, M. (1992). Regulation of                   |
| perfluorooctanoic acidinduced peroxisomal enzyme activities and hepatocellular growth by                |
| adrenal hormones. Hepatology 15: 316-322. <u>http://dx.doi.org/10.1002/hep.1840150223</u>               |
| Tomaszewski, KE; Heindel, SW; Jenkins, WL; Melnick, RL. (1990). Induction of peroxisomal acyl           |
| CoA oxidase activity and lipid peroxidation in primary rat hepatocyte cultures. Toxicology 65:          |
| 49-60. <u>http://dx.doi.org/10.1016/0300-483X(90)90078-U</u>  |
| Tomita, I; Nakamura, Y; Aoki, N; Inui, N. (1982). Mutagenic/carcinogenic potential of DEHP and          |
| MEHP. Environ Health Perspect 45: 119-125. <u>http://dx.doi.org/10.2307/3429394</u>                     |
| Tomonari, Y; Kurata, Y; David, RM; Gans, G; Kawasuso, T; Katoh, M. (2006). Effect of di(2-              |
| ethylhexyl) phthalate (DEHP) on genital organs from juvenile common marmosets: I.                       |
| Morphological and biochemical investigation in 65-week toxicity study. J Toxicol Environ                |
| Health A 69: 1651-1672. <u>http://dx.doi.org/10.1080/15287390600630054</u>                              |
| <u>Ioyosawa, K; Okimolo, K; Kobayashi, I; Kijima, K; Kikawa, E; Kohchi, M; Koujitani, I; Tanaka, K;</u> |
| Malsuoka, N. (2001). Di(2-ethylnexyl)phthalate induces nepatocellular adenoma in transgenic             |
| Dethol 20: 458 466 http://dx doi org/10.1020/01026220152400044  |
| Tresende L: Liu P: Pao W (2021) Phtheletes and attributable mortality: A population based               |
| longitudinal cohort study and cost analysis. Environ Pollut 118021                                      |
| https://beropet.epa.gov/beropet/index.cfm/reference/download/reference_id/0/05370                       |
| Tsutsui T: Watanabe E: Barrett IC (1993) Ability of perovisome proliferators to induce cell             |
| transformation chromosome aberrations and peroxisome proliferation in cultured Syrian hamster           |
| embryo cells. Carcinogenesis 14: 611-618. http://dx.doi.org/10.1093/carcin/14.4.611                     |
| Tucker, MJ: Orton, TC. (1995). Comparative toxicology of hypolipidaemic fibrates. Bristol. PA: Taylor   |
| and Francis.  |
| Turner, JH; Petricciani, JC; Crouch, ML; Wenger, S. (1974). An evaluation of the effects of             |
| diethylhexyl phthalate (DEHP) on mitotically capable cells in blood packs. Transfusion 14: 560-         |
| 566. http://dx.doi.org/10.1111/j.1537-2995.1974.tb04577.x   |
| U.S. CPSC. (2001). Report to the U.S. Consumer Product Safety Commission by the Chronic Hazard          |
| Advisory Panel on diisononyl phthalate (DINP).  |
| U.S. CPSC. (2010a). Toxicity review of benzyl-n-butyl phthalate. Bethesda, MD: U.S. Consumer            |
| Product Safety Commission, Directorate for Hazard Identification and Reduction.                         |
| https://www.cpsc.gov/s3fs-public/ToxicityReviewOfBBP.pdf  |
| U.S. CPSC. (2010b). Toxicity review of di-n-butyl phthalate. Bethesda, MD: U.S. Consumer Product        |
| Safety Commission, Directorate for Hazard Identification and Reduction.                                 |
| https://web.archive.org/web/20190320060443/https://www.cpsc.gov/s3fs-                                   |
| public/ToxicityReviewOfDBP.pdf  |
| <u>U.S. CPSC. (2010c)</u> . Toxicity review of Di(2-ethylhexyl) Phthalate (DEHP). Bethesda, MD.         |
| http://www.cpsc.gov//PageFiles/126533/toxicityDEHP.pdf  |
| U.S. CPSC. (2010d). Toxicity Review of Di(isodecyl) Phthalate. Washington, DC: Consumer Product         |
| Safety Commission (CPSC). <u>http://www.cpsc.gov/PageFiles/126534/toxicityDIDP.pdf</u>                  |
| U.S. CPSC. (2010e). Toxicity review of dicyclohexyl phthalate (DCHP). Bethesda, MD: U.S. Consumer       |
| Product Safety Commission, Directorate for Hazard Identification and Reduction.                         |
| https://web.archive.org/web/20190320060432/https://www.cpsc.gov/s3fs-                                   |
| public/ToxicityReviewOtDCHP.pdt   |
| U.S. CPSC. (2010f). Toxicity review of Ditsononyl Phthalate (DINP). Bethesda, MD.                       |
| nttp://www.cpsc.gov/PageFiles/126539/toxicityDINP.pdf   |

| 4344 | U.S. CPSC. (2011). Toxicity review of diisobutyl phthalate (DiBP, CASRN 84-69-5). Bethesda, MD:   |
|------|---|
| 4345 | U.S. Consumer Product Safety Commission. https://www.cpsc.gov/s3fs-   |
| 4346 | public/ToxicityReviewOfDiBP.pdf   |
| 4347 | U.S. CPSC. (2014). Chronic Hazard Advisory Panel on Phthalates and Phthalate Alternatives (with   |
| 4348 | appendices). Bethesda, MD: U.S. Consumer Product Safety Commission, Directorate for Health  |
| 4349 | Sciences. https://www.cpsc.gov/s3fs-public/CHAP-REPORT-With-Appendices.pdf  |
| 4350 | U.S. EPA. (1986). Guidelines for carcinogen risk assessment [EPA Report] (pp. 33993-34003).   |
| 4351 | (EPA/630/R-00/004). Washington, DC: U.S. Environmental Protection Agency, Risk  |
| 4352 | Assessment Forum. https://cfpub.epa.gov/ncea/raf/car2sab/guidelines_1986.pdf  |
| 4353 | U.S. EPA. (1987). Integrated Risk Information System (IRIS), chemical assessment summary, dibutyl   |
| 4354 | phthalate; CASRN 84-74-2. Washington, DC: U.S. Environmental Protection Agency, National  |
| 4355 | Center for Environmental Assessment.  |
| 4356 | https://cfpub.epa.gov/ncea/iris/iris_documents/documents/subst/0038_summary.pdf   |
| 4357 | U.S. EPA. (1988a). Integrated Risk Information System (IRIS), chemical assessment summary, butyl  |
| 4358 | benzyl phthalate; CASRN 85-68-7 [EPA Report]. Washington, DC: U.S. Environmental  |
| 4359 | Protection Agency, National Center for Environmental Assessment.  |
| 4360 | U.S. EPA. (1988b). Integrated Risk Information System (IRIS), chemical assessment summary, di(2-  |
| 4361 | ethylhexyl)phthalate (DEHP); CASRN 117-81-7. Washington, DC: U.S. Environmental   |
| 4362 | Protection Agency, National Center for Environmental Assessment.  |
| 4363 | https://heronet.epa.gov/heronet/index.cfm/reference/download/reference_id/5113322   |
| 4364 | U.S. EPA. (1991). Alpha-2u-globulin: Association with chemically induced renal toxicity and neoplasia   |
| 4365 | in the male rat [EPA Report]. (EPA625391019F. PB92143668). Washington, DC: U.S.   |
| 4366 | Environmental Protection Agency, National Center for Environmental Assessment.  |
| 4367 | https://ntrl.ntis.gov/NTRL/dashboard/searchResults.xhtml?searchQuery=PB92143668   |
| 4368 | U.S. EPA. (1999). Guidelines for carcinogen risk assessment [review draft] [EPA Report]. (NCEA-F-   |
| 4369 | 0644). Washington, DC: U.S. Environmental Protection Agency, Office of the Science Advisor.   |
| 4370 | http://www.epa.gov/raf/publications/pdfs/CANCER_GLS.PDF   |
| 4371 | U.S. EPA. (2002). Provisional Peer Reviewed Toxicity Values for butyl benzyl phthalate (CASRN 85-   |
| 4372 | 68-/): Derivation of a carcinogenicity assessment [EPA Report]. Cincinnati, OH.   |
| 43/3 | U.S. EPA. (2005). Guidelines for carcinogen risk assessment [EPA Report]. (EPA630P03001F).  |
| 43/4 | Washington, DC. <u>https://www.epa.gov/sites/production/files/2013-</u>   |
| 43/3 | <u>09/documents/cancer_guidennes_innai_5-25-05.pdi</u>  |
| 4370 | <u>U.S. EPA. (2011)</u> . Recommended use of body weight 5/4 as the default method in derivation of the oral reference does (EDA 100P 110001). Washington, DC |
| 4377 | https://www.epa.gov/sites/production/files/2013_00/documents/recommended_use_of_bw34.pdf  |
| 4370 | US EPA (2019a) Proposed designation of butyl benzyl phthalate (CASRN 85-68-7) as a high priority  |
| 4380 | <u>substance for risk evaluation</u>  |
| 4381 | US EPA (2019b) Proposed designation of Di-Ethylbexyl Phthalate (DEHP) (1 2-Benzene-   |
| 4382 | dicarboxylic acid 1 2-bis (2-ethylbexyl) ester) (CASRN 117-81-7) as a high-priority substance   |
| 4383 | for risk evaluation. Washington, DC: Office of Pollution Prevention and Toxics.   |
| 4384 | https://www.epa.gov/sites/production/files/2019-08/documents/di-ethylhexyl_phthalate_117-81-  |
| 4385 | 7 proposeddesignation 082219.pdf  |
| 4386 | U.S. EPA. (2019c). Proposed Designation of Di-isobutyl Phthalate (DIBP) (CASRN 84-69-5) as High-  |
| 4387 | Priority Substance for Risk Evaluation. https://www.epa.gov/sites/production/files/2019-  |
| 4388 | 08/documents/diisobutylphthalate_84-69-5_high-priority_proposeddesignation_082319_0.pdf   |
| 4389 | U.S. EPA. (2019d). Proposed designation of Dibutyl Phthalate (CASRN 84-74-2) as a high-priority   |
| 4390 | substance for risk evaluation. U.S. Environmental Protection Agency, Office of Chemical Safety  |
|      |   |

|              | 114 2020  |
|--------------|---|
| 4391         | and Pollution Prevention. https://www.epa.gov/sites/production/files/2019-  |
| 4392         | 08/documents/dibutylphthalate_84-74-2_high-priority_proposeddesignation_082319.pdf  |
| 4393         | U.S. EPA. (2019e). Proposed designation of dicyclohexyl phthalate (CASRN 84-61-7) as a high-priority  |
| 4394         | substance for risk evaluation (pp. 1-21). U.S. Environmental Protection Agency, Office of   |
| 4395         | Chemical Safety and Pollution Prevention. https://www.regulations.gov/document?D=EPA-HQ-  |
| 4396         | <u>OPPT-2018-0504-0009</u>  |
| 4397         | U.S. EPA. (2020a). Final scope of the risk evaluation for butyl benzyl phthalate (1,2-  |
| 4398         | benzenedicarboxylic acid, 1-butyl 2-(phenylmethyl) ester); CASRN 85-68-7 [EPA Report].  |
| 4399         | (EPA-740-R-20-015). Washington, DC: Office of Chemical Safety and Pollution Prevention.   |
| 4400         | https://www.epa.gov/sites/default/files/2020-09/documents/casrn_85-68-  |
| 4401         | 7_butyl_benzyl_phthalate_finalscope.pdf   |
| 4402         | U.S. EPA. (2020b). Final scope of the risk evaluation for di-ethylhexyl phthalate (1,2-   |
| 4403         | benzenedicarboxylic acid, 1,2-bis(2-ethylhexyl) ester); CASRN 117-81-7 [EPA Report]. (EPA-  |
| 4404         | 740-R-20-017). Washington, DC: Office of Chemical Safety and Pollution Prevention.  |
| 4405         | https://www.epa.gov/sites/default/files/2020-09/documents/casrn_117-81-7_di-  |
| 4406         | ethylhexyl_phthalate_final_scope.pdf  |
| 4407         | U.S. EPA. (2020c). Final scope of the risk evaluation for di-isobutyl phthalate (1,2-benzenedicarboxylic  |
| 4408         | acid, 1,2-bis(2-methylpropyl) ester); CASRN 84-69-5 [EPA Report]. (EPA-740-R-20-018).   |
| 4409         | Washington, DC: Office of Chemical Safety and Pollution Prevention.   |
| 4410         | https://www.epa.gov/sites/default/files/2020-09/documents/casrn_84-69-5_di-   |
| 4411         | isobutyl_phthalate_final_scope.pdf  |
| 4412         | <u>U.S. EPA. (2020d)</u> . Final scope of the risk evaluation for dibutyl phthalate (1,2-benzenedicarboxylic  |
| 4413         | acid, 1,2-dibutyl ester); CASRN 84-74-2 [EPA Report]. (EPA-740-R-20-016). Washington, DC:   |
| 4414         | Office of Chemical Safety and Pollution Prevention.   |
| 4415         | https://www.epa.gov/sites/default/files/2020-09/documents/casrn_84-74-  |
| 4416         | 2_dibutyl_phthalate_final_scope_0.pdf   |
| 4417         | <u>U.S. EPA. (2020e)</u> . Final scope of the risk evaluation for dicyclohexyl phthalate (1,2-  |
| 4418         | benzenedicarboxylic acid, 1,2-dicyclohexyl ester); CASRN 84-61-7 [EPA Report]. (EPA-740-R-  |
| 4419         | 20-019). Washington, DC: Office of Chemical Safety and Pollution Prevention.  |
| 4420         | https://www.epa.gov/sites/default/files/2020-09/documents/casrn_84-61-  |
| 4421         | <u>7_dicyclohexyl_phthalate_final_scope.pdf</u>   |
| 4422         | U.S. EPA. (2021a). Final scope of the risk evaluation for di-isodecyl phthalate (DIDP) (1,2-  |
| 4423         | benzenedicarboxylic acid, 1,2-diisodecyl ester and 1,2-benzenedicarboxylic acid, di-C9-11-  |
| 4424         | branched alkyl esters, C10-rich); CASRN 26/61-40-0 and 68515-49-1 [EPA Report]. (EPA-740-   |
| 4425         | R-21-001). Washington, DC: Office of Chemical Safety and Pollution Prevention.  |
| 4426         | https://www.epa.gov/system/files/documents/2021-08/casrn-26/61-40-0-d1-isodecy1-phthalate-  |
| 4427         | <u>Inal-scope.par</u>   |
| 4428         | <u>U.S. EPA. (2021b)</u> . Final scope of the risk evaluation for di-isononyl phthalate (DINP) (1,2-benzene-  |
| 4429         | alcarboxylic acid, 1,2-diisononyl ester, and 1,2-benzenedicarboxylic acid, di-U8-10-branched  |
| 4450         | alkyl estels, C9-ficil); CASKINS 26555-12-0 alid 06515-46-0 [EPA Report]. (EPA-740-K-21-  |
| 4431         | bttps://www.eps.gov/gustem/files/documents/2021_08/gogm_28552_12_0_di icononul_phthelate  |
| 4432         | https://www.epa.gov/system/mes/documents/2021-08/casm-28555-12-0-di-isononyi-phinalate-   |
| 4433         | <u>IIIIai-scope.pui</u><br>U.S. EDA (2022) Droft Droposed Approach for Cumulative Dick Assessment of High Drighty   |
| 4434<br>1125 | <u>U.S. LFA. (2023)</u> . Dian Floposed Approach for Cullulative KISK Assessment of High-Phonely<br>Physical and a Manufacturer Percuested Detailets under the Toxic Substances Control Act |
| 4433<br>1136 | (EDA 740 D 23 002) Washington DC: U.S. Environmental Protection A genery Office of  |
| 4430<br>1127 | (LI A-740-F-23-002). Washington, DC. U.S. Environmental Flotection Agency, Office of<br>Chemical Safety and Pollution Prevention, https://www.regulations.gov/document/EDA_HO               |
| 4437<br>1120 | ODDT 2022 0018 0000   |
| 4430         | <u>VIII-2022-0710-0007</u>  |

|      | May 2025   |
|------|--|
| 4439 | U.S. EPA. (2024a). Draft chemistry, fate, and transport assessment for Dibutyl Phthalate (DBP).        |
| 4440 | Washington, DC: Office of Pollution Prevention and Toxics.   |
| 4441 | U.S. EPA. (2024b). Draft Meta-Analysis and Benchmark Dose Modeling of Fetal Testicular                 |
| 4442 | Testosterone for Di(2-ethylhexyl) Phthalate (DEHP), Dibutyl Phthalate (DBP), Butyl Benzyl              |
| 4443 | Phthalate (BBP), Diisobutyl Phthalate (DIBP), and Dicyclohexyl Phthalate (DCHP).                       |
| 4444 | Washington, DC: Office of Pollution Prevention and Toxics.   |
| 4445 | U.S. EPA. (2024c). Draft Non-cancer Human Health Hazard Assessment for Butyl benzyl phthalate          |
| 4446 | (BBP). Washington, DC: Office of Pollution Prevention and Toxics.                                      |
| 4447 | U.S. EPA. (2024d). Draft Non-cancer Human Health Hazard Assessment for Dibutyl Phthalate (DBP).        |
| 4448 | Washington, DC: Office of Pollution Prevention and Toxics.   |
| 4449 | U.S. EPA. (2024e). Draft Non-Cancer Human Health Hazard Assessment for Dicyclohexyl Phthalate          |
| 4450 | (DCHP). Washington, DC: Office of Pollution Prevention and Toxics.                                     |
| 4451 | U.S. EPA. (2024f). Draft Non-cancer Human Health Hazard Assessment for Diethylhexyl Phthalate          |
| 4452 | (DEHP). Washington, DC: Office of Pollution Prevention and Toxics.                                     |
| 4453 | U.S. EPA. (2024g). Draft Non-cancer Human Health Hazard Assessment for Diisobutyl phthalate            |
| 4454 | (DIBP). Washington, DC: Office of Pollution Prevention and Toxics.                                     |
| 4455 | U.S. EPA. (2024h). Draft Physical and Chemical Property Assessment and Fate and Transport              |
| 4456 | Assessment for Di-ethylhexyl Phthalate (DEHP). Washington, DC: Office of Pollution                     |
| 4457 | Prevention and Toxics.   |
| 4458 | U.S. EPA. (2024i). Draft Physical Chemistry and Fate and Transport Assessment for Butyl Benzyl         |
| 4459 | Phthalate (BBP). Washington, DC: Office of Pollution Prevention and Toxics.                            |
| 4460 | U.S. EPA. (2024i). Draft physical chemistry and fate and transport assessment for dicyclohexyl         |
| 4461 | phthalate (DCHP), Washington, DC: Office of Pollution Prevention and Toxics.                           |
| 4462 | U.S. EPA. (2024k). Draft physical chemistry and fate and transport assessment for Diisobutyl phthalate |
| 4463 | (DIBP). Washington, DC: Office of Pollution Prevention and Toxics.                                     |
| 4464 | U.S. EPA. (20241). Draft Risk Evaluation for Dicyclohexyl Phthalate (DCHP). Washington, DC: Office     |
| 4465 | of Pollution Prevention and Toxics.  |
| 4466 | U.S. EPA, (2024m), Draft Systematic Review Protocol for Dicyclohexyl Phthalate (DCHP).                 |
| 4467 | Washington, DC: Office of Pollution Prevention and Toxics.   |
| 4468 | U.S. EPA, (2024n), Human Health Hazard Assessment for Diisodecyl Phthalate (DIDP), Washington,         |
| 4469 | DC: Office of Pollution Prevention and Toxics.   |
| 4470 | U.S. EPA. (2024o). Physical Chemistry Assessment for Diisodecyl Phthalate (DIDP). Washington, DC:      |
| 4471 | Office of Pollution Prevention and Toxics.   |
| 4472 | U.S. EPA. (2024p), Risk Evaluation for Dijsodecyl Phthalate (DIDP), Washington, DC: Office of          |
| 4473 | Pollution Prevention and Toxics.   |
| 4474 | U.S. EPA. (2024a). Science Advisory Committee on Chemicals Meeting Minutes and Final Report No.        |
| 4475 | 2024-2. Docket ID: EPA-HO-OPPT-2024-0073: For the Draft Risk Evaluation for Di-isodecyl                |
| 4476 | Phthalate (DIDP) and Draft Hazard Assessments for Di-isononyl Phthalate (DINP). Washington             |
| 4477 | DC. U.S. Environmental Protection Agency. Science Advisory Committee on Chemicals                      |
| 4478 | US EPA (2024r) Systematic Review Protocol for Diisodecyl Phthalate (DIDP) Washington DC.               |
| 4479 | Office of Pollution Prevention and Toxics  |
| 4480 | US EPA (2025a) Cancer Human Health Hazard Assessment for Diisononyl Phthalate (DINP)                   |
| 4481 | Washington DC: Office of Pollution Prevention and Toxics   |
| 4487 | US FPA (2025b) Draft Environmental Media and General Population and Environmental Exposure             |
| 4483 | for Diethylbexyl Phthalate (DEHP) Washington DC. Office of Pollution Prevention and Toxics             |
| 4484 | US FPA (2025c) Draft Risk Evaluation for Butyl Benzyl Phthalate (BRP) Washington, DC: Office of        |
| 1/85 | Pollution Prevention and Toxics  |
| 4405 | i onution i revenution and Toxics.   |

#### May 2025

4486 U.S. EPA. (2025d). Draft Risk Evaluation for Dibutyl Phthalate (DBP). Washington, DC: Office of 4487 Pollution Prevention and Toxics. 4488 U.S. EPA. (2025e). Draft Risk Evaluation for Diethylhexyl Phthalate (DEHP). Washington, DC: Office 4489 of Pollution Prevention and Toxics. 4490 U.S. EPA. (2025f). Draft Risk Evaluation for Diisobutyl Phthalate (DIBP). Washington, DC: Office of 4491 Pollution Prevention and Toxics. U.S. EPA. (2025g). Draft Systematic Protocol for Butyl Benzyl Phthalate (BBP). Washington, DC: 4492 4493 Office of Pollution Prevention and Toxics. 4494 U.S. EPA. (2025h). Draft Systematic Review Protocol for Dibutyl Phthalate (DBP). Washington, DC: 4495 Office of Pollution Prevention and Toxics. 4496 U.S. EPA. (2025i). Draft Systematic Review Protocol for Diethylhexyl Phthalate (DEHP). Washington, 4497 DC: Office of Pollution Prevention and Toxics. 4498 U.S. EPA. (2025j). Draft Systematic Review Protocol for Diisobutyl Phthalate (DIBP). Washington, 4499 DC: Office of Pollution Prevention and Toxics. U.S. EPA. (2025k). Non-Cancer Human Health Hazard Assessment for Diisononyl Phthalate (DINP) 4500 4501 Washington, DC: Office of Pollution Prevention and Toxics. 4502 U.S. EPA. (2025). Physical Chemistry Assessment for Diisononyl Phthalate (DINP). Washington, DC: 4503 Office of Pollution Prevention and Toxics. 4504 U.S. EPA. (2025m). Risk Evaluation for Diisononyl Phthalate (DINP). Washington, DC: Office of 4505 Pollution Prevention and Toxics. U.S. EPA. (2025n). Systematic Review Protocol for Diisononyl Phthalate (DINP) Washington, DC: 4506 4507 Office of Pollution Prevention and Toxics. 4508 University of Rochester. (1953). One month feeding tests of di-isobutyl phthalate with cover letter 4509 [TSCA Submission]. (OTS0205995. 878212229. TSCATS/017429). Confidential. 4510 https://ntrl.ntis.gov/NTRL/dashboard/searchResults/titleDetail/OTS0205995.xhtml 4511 University of Rochester, (1954). Preliminary acute toxicity tests and short term feeding tests of rats and 4512 dogs given di-isobutyl phthalate and di-butyl phthalate [TSCA Submission]. (OTS0205995. 4513 878210833. TSCATS/017427). Confidential. 4514 https://ntrl.ntis.gov/NTRL/dashboard/searchResults.xhtml?searchQuery=OTS0205995 4515 Vogel, EW; Nivard, MJ. (1993). Performance of 181 chemicals in a drosophila assay predominantly monitoring interchromosomal mitotic recombination. Mutagenesis 8: 57-81. 4516 4517 http://dx.doi.org/10.1093/mutage/8.1.57 4518 von Däniken, A; Lutz, W; Jäckh, R; Schlatter, C. (1984). Investigation of the potential for binding of Di(2-ethylhexyl) phthalate (DEHP) and Di(2-ethylhexyl) adipate (DEHA) to liver DNA in vivo. 4519 4520 Toxicol Appl Pharmacol 73: 373-387. http://dx.doi.org/10.1016/0041-008X(84)90089-9 4521 Voss, C; Zerban, H; Bannasch, P; Berger, MR. (2005). Lifelong exposure to di-(2-ethylhexyl)-phthalate 4522 induces tumors in liver and testes of Sprague-Dawley rats. Toxicology 206: 359-371. 4523 http://dx.doi.org/10.1016/j.tox.2004.07.016 4524 Wang, IJ; Lin, CC; Lin, YJ; Hsieh, WS; Chen, PC. (2014). Early life phthalate exposure and atopic 4525 disorders in children: A prospective birth cohort study. Environ Int 62: 48-54. 4526 http://dx.doi.org/10.1016/j.envint.2013.09.002 4527 Wang, X; Sheng, N; Cui, R; Zhang, H; Wang, J; Dai, J. (2017). Gestational and lactational exposure to di-isobutyl phthalate via diet in maternal mice decreases testosterone levels in male offspring. 4528 Chemosphere 172: 260-267. http://dx.doi.org/10.1016/j.chemosphere.2017.01.011 4529 4530 Ward, JM; Hagiwara, A; Anderson, LM; Lindsey, K; Diwan, BA. (1988). The chronic hepatic or renal toxicity of di(2-ethylhexyl) phthalate, acetaminophen, sodium barbital, and phenobarbital in 4531 4532 male B6C3F1 mice: autoradiographic, immunohistochemical, and biochemical evidence for

|      | May 2025   |
|------|--|
| 4533 | levels of DNA synthesis not associated with carcinogenesis or tumor promotion. Toxicol Appl                      |
| 4534 | Pharmacol 96: 494-506. http://dx.doi.org/10.1016/0041-008X(88)90009-9  |
| 4535 | Westberg, KBT; Hardell, LO; Malmqvist, N; Ohlson, CG; Axelson, O. (2005). On the use of different                |
| 4536 | measures of exposure - Experiences from a case-control study on testicular cancer and PVC                        |
| 4537 | exposure. J Occup Environ Hyg 2: 351-356. <u>http://dx.doi.org/10.1080/15459620590969046</u>                     |
| 4538 | <u>WHO. (1978)</u> . A co-operative trial in the primary prevention of ischaemic heart disease using clofibrate. |
| 4539 | Report from the Committee of Principal Investigators. British Heart Journal 40: 1069-1118.                       |
| 4540 | WHO. (1987). PRINCIPLES FOR THE SAFETY ASSESSMENT OF FOOD ADDITIVES AND  |
| 4541 | CONTAMINANTS IN FOOD (pp. 1-165). (BIOSIS/88/07735). WHO.  |
| 4542 | Yamasaki, K; Okuda, H; Takeuchi, T; Minobe, Y. (2009). Effects of in utero through lactational                   |
| 4543 | exposure to dicyclohexyl phthalate and p,p'-DDE in Sprague-Dawley rats. Toxicol Lett 189: 14-                    |
| 4544 | 20. <u>http://dx.doi.org/10.1016/j.toxlet.2009.04.023</u>  |
| 4545 | Yang, G; Zhou, X; Wang, J; Zhang, W; Zheng, H; Lu, W; Yuan, J. (2012). MEHP-induced oxidative                    |
| 4546 | DNA damage and apoptosis in HepG2 cells correlates with p53-mediated mitochondria-                               |
| 4547 | dependent signaling pathway. Food Chem Toxicol 50: 2424-2431.  |
| 4548 | http://dx.doi.org/10.1016/j.fct.2012.04.023  |
| 4549 | Yoon, JS; Mason, JM; Valencia, R; Woodruff, RC; Zimmering, S. (1985). Chemical mutagenesis testing               |
| 4550 | in Drosophila. IV. Results of 45 coded compounds tested for the National Toxicology Program.                     |
| 4551 | Environ Mutagen 7: 349-367. http://dx.doi.org/10.1002/em.2860070310  |
| 4552 | Yoshikawa, K; Tanaka, A; Yamaha, T; Kurata, H. (1983). Mutagenicity study of nine monoalkyl                      |
| 4553 | phthalates and a dialkyl phthalate using Salmonella typhimurium and Escherichia coli. Food                       |
| 4554 | Chem Toxicol 21: 221-223. http://dx.doi.org/10.1016/0278-6915(83)90239-9   |
| 4555 | Yost, EE; Euling, SY; Weaver, JA; Beverly, BEJ; Keshava, N; Mudipalli, A; Arzuaga, X; Blessinger, T;             |
| 4556 | Dishaw, L; Hotchkiss, A; Makris, SL. (2019). Hazards of diisobutyl phthalate (DIBP) exposure:                    |
| 4557 | A systematic review of animal toxicology studies [Review]. Environ Int 125: 579-594.                             |
| 4558 | http://dx.doi.org/10.1016/j.envint.2018.09.038   |
| 4559 | Zeiger, E; Haworth, S; Mortelmans, K; Speck, W. (1985). Mutagenicity testing of di(2-                            |
| 4560 | ethylhexyl)phthalate and related chemicals in Salmonella. Environ Mol Mutagen 7: 213-232.                        |
| 4561 | http://dx.doi.org/10.1002/em.2860070209  |
| 4562 | Zeiger, E; Haworth, S; Speck, W; Mortelmans, K. (1982). Phthalate ester testing in the National                  |
| 4563 | Toxicology Program's environmental mutagenesis test development program. Environ Health                          |
| 4564 | Perspect 45: 99-101. http://dx.doi.org/10.1289/ehp.824599  |
| 4565 |  |

### 4566 APPENDICES

4567

#### 4568

## Appendix A SUMMARY OF DEHP GENOTOXICITY STUDIES

4569 4570 4571

### Table\_Apx A-1. Genotoxicity of DEHP In Vitro (Studies Considered by ATSDR (2022))<sup>a</sup>

|   | Endpoint                 | Result             |                       |   |
|---|--------------------------|--------------------|-----------------------|---|
| Species (test system)                                       |                          | With<br>Activation | Without<br>Activation | Reference   |
|   | Prokaryotic organism     | ns                 |                       |   |
| Salmonella typhimurium (TA98, TA100, TA1535, TA1538)        | Gene mutation            | _                  | _                     | ( <u>Agarwal et al., 1985</u> )                   |
| typhimurium (NS)  | Gene mutation            | _                  | _                     | (Astill et al., 1986)                             |
| <i>S. typhimurium</i> (TA98, TA100, TA1535, TA1537, TA1538) | Gene mutation            | _                  | -                     | (Kirby et al., 1983)                              |
| S. typhimurium (TA100)                                      | Gene mutation            | _                  | +                     | (Kozumbo et al., 1982)                            |
| S. typhimurium (TA98)                                       | Gene mutation            | _                  | _                     | ( <u>Sato et al., 1994</u> )                      |
| S. typhimurium (TA102)                                      | Gene mutation            | _                  | _                     | ( <u>Schmezer et al., 1988</u> )                  |
| <i>S. typhimurium</i> (TA98, TA100, TA1535, TA1537, TA1538) | Gene mutation            | _                  | —                     | ( <u>Simmon et al., 1977</u> )                    |
| S. typhimurium (TA100)                                      | Gene mutation            | _                  | —                     | ( <u>Seed, 1982</u> )                             |
| S. typhimurium (TA100)                                      | Gene mutation            | +                  | NS                    | ( <u>Tomita et al., 1982</u> )                    |
| S. typhimurium (TA98, TA100)                                | Gene mutation            | _                  | _                     | ( <u>Yoshikawa et al.,</u><br><u>1983</u> )       |
| S. typhimurium (TA98, TA1537)                               | Gene mutation            | _                  | NS                    | (Kanode et al., 2017)                             |
| <i>S. typhimurium</i> (TA98, TA100, TA1535, TA1537)         | Gene mutation            | _                  | _                     | ( <u>Lee et al., 2019</u> )                       |
| <i>S. typhimurium</i> (TA98, TA100, TA1535, TA1537)         | Gene mutation            | -                  | _                     | ( <u>Zeiger et al., 1985</u> )                    |
| Escherichia coli PQ37                                       | Gene mutation            | _                  | _                     | (Sato et al., 1994)                               |
| E. coli WP2UVRA+  | Gene mutation            | _                  | _                     | ( <u>Yoshikawa et al.,</u><br><u>1983</u> )       |
| E. coli WP2UVRA   | Gene mutation            | -                  | -                     | ( <u>Yoshikawa et al.,</u><br><u>1983</u> )       |
| E. coli WP2UVRA   | Gene mutation            | _                  | _                     | (Lee et al., 2019)                                |
| S. typhimurium (TA1535/psk 1002)                            | DNA damage               | +                  | _                     | ( <u>Okai and Higashi-</u><br><u>Okai, 2000</u> ) |
| Bacillus subtilis (rec assay)                               | DNA damage               | +                  | _                     | ( <u>Tomita et al., 1982</u> )                    |
| S. typhimurium (TA100)                                      | Azaguanine<br>resistance | _                  | —                     | ( <u>Seed, 1982</u> )                             |
| Eukaryotic organisms  |                          |                    |                       |   |

|   | Endpoint            | Result             |                       |  |
|---|---------------------|--------------------|-----------------------|--|
| Species (test system)                                       |                     | With<br>Activation | Without<br>Activation | Reference                                    |
| Saccharomyces cerevisae (XV185-14C, D7, RM52, D6, D5, D6-1) | Gene mutation       | _                  | _                     | ( <u>Parry et al., 1985</u> )                |
| Saccharomyces cerevisiae (JD1, D7-144, D7)                  | Gene conversion     | —                  | _                     | (Parry et al., 1985)                         |
| S. cerevisiae (D61M, D6)                                    | Mitotic aneuploidy  | +                  | +                     | ( <u>Parry et al., 1985</u> )                |
| S. cerevisiae (D61M, D6)                                    | Mitotic segregation | -                  | _                     | ( <u>Parry et al., 1985</u> )                |
| Schizosaccharomyces pombe (P1)                              | Gene mutation       | _                  | _                     | ( <u>Parry et al., 1985</u> )                |
| Aspergillus niger (P1)                                      | Mitotic segregation | _                  | NS                    | ( <u>Parry et al., 1985</u> )                |
|   | Mammalian cells     |                    |                       |  |
| Mouse lymphoma cells  | Mutagenicity        | —                  | -                     | ( <u>Astill et al., 1986</u> )               |
| Mouse lymphoma cells  | Mutagenicity        | —                  | -                     | (Kirby et al., 1983)                         |
| Mouse lymphoma cells  | Mutagenicity        | $\pm^b$            | Ι                     | ( <u>Oberly et al., 1985</u> )               |
| Mouse lymphoma cells  | Mutagenicity        | _                  | Ι                     | ( <u>Tennant et al., 1987</u> )              |
| Human leukocytes  | DNA damage          | _                  | +                     | ( <u>Anderson et al., 1999</u> )             |
| Human lymphocytes   | DNA damage          | _                  | +                     | ( <u>Anderson et al., 1999</u> )             |
| Human HeLa cells  | DNA damage          | NS                 | +                     | (Park and Choi, 2007)                        |
| Human HepG2 cells   | DNA damage          | NS                 | +                     | ( <u>Choi et al., 2010</u> )                 |
| Human LNCaP prostate adenocarcinoma cells                   | DNA damage          | NS                 | +                     | ( <u>Erkekoglu et al.,</u><br><u>2010b</u> ) |
| Human HepaRG cells  | DNA damage          | _                  | NA                    | ( <u>Le Hégarat et al.,</u><br>2014)         |
| Human thyroid carcinoma                                     | DNA damage          | NS                 | +                     | ( <u>Kim et al., 2019</u> )                  |
| Mouse MA-10 Leydig tumor cells                              | DNA damage          | NS                 | +                     | ( <u>Erkekoglu et al.,</u><br><u>2010a</u> ) |
| Mouse lung cells  | DNA damage          | NS                 | +                     | ( <u>Wang et al., 2014</u> )                 |
| Rat hepatocytes   | DNA damage          | _                  | NA                    | ( <u>Schmezer et al., 1988</u> )             |
| Hamster hepatocytes   | DNA damage          | _                  | NA                    | (Schmezer et al., 1988)                      |
| CHO cells   | DNA damage          | _                  | _                     | (Douglas et al., 1986)                       |
| Human hepatocytes   | DNA repair          | _                  | NA                    | (Butterworth et al.,<br><u>1984</u> )        |
| Mouse hepatocytes   | DNA repair          | _                  | NA                    | (Smith-Oliver and<br>Butterworth, 1987)      |
| Rat hepatocytes   | DNA repair          | _                  | NA                    | (Astill et al., 1986)                        |
| Rat hepatocytes   | DNA repair          | _                  | NA                    | (Butterworth, 1984)                          |
| Rat hepatocytes   | DNA repair          | _                  | NA                    | ( <u>Hodgson et al., 1982</u> )              |
| Rat hepatocytes   | DNA repair          | —                  | NA                    | ( <u>Kornbrust et al.,</u><br><u>1984</u> )  |

|                                 | Endpoint                    | Result             |                       |  |
|---------------------------------|-----------------------------|--------------------|-----------------------|--|
| Species (test system)           |                             | With<br>Activation | Without<br>Activation | Reference  |
| Rat hepatocytes                 | DNA repair                  | _                  | NA                    | (Probst and Hill, 1985)                                |
| Chinese hamster V79 fibroblasts | DNA repair                  | _                  | NA                    | ( <u>Kornbrust et al.,</u><br><u>1984</u> )            |
| Human HepaRG cells              | Micronuclei                 | _                  | NA                    | ( <u>Le Hégarat et al.,</u><br>2014)                   |
| Human TK6 lymphoblastoid cells  | Micronuclei                 | NS                 | Ι                     | ( <u>Sobol et al., 2012</u> )                          |
| Rat RL4 liver cells             | Sister chromatid exchange   | —                  | NA                    | ( <u>Priston and Dean,</u><br><u>1985</u> )            |
| CHO cells                       | Sister chromatid exchange   | NS                 | _                     | ( <u>Abe and Sasaki,</u><br><u>1977</u> )              |
| CHO cells                       | Sister chromatid exchange   | _                  | _                     | ( <u>Douglas et al., 1986</u> )                        |
| CHO cells                       | Sister chromatid exchange   | NS                 | _                     | ( <u>Phillips et al., 1982</u> )                       |
| CHO cells                       | Sister chromatid exchange   | NS                 | +                     | ( <u>Tennant et al., 1987</u> )                        |
| Human hepatocytes               | Chromosomal aberrations     | _                  | NA                    | ( <u>Turner et al., 1974</u> )                         |
| Human leucocytes                | Chromosomal aberrations     | _                  | NA                    | ( <u>Stenchever et al.,</u><br><u>1976</u> )           |
| Rat RL4 liver cells             | Chromosomal aberrations     | _                  | NA                    | ( <u>Priston and Dean,</u><br><u>1985</u> )            |
| CHO cells                       | Chromosomal aberrations     | NS                 | Ι                     | ( <u>Phillips et al., 1982</u> )                       |
| CHO cells                       | Chromosomal aberrations     | NS                 | Ι                     | ( <u>Tennant et al., 1987</u> )                        |
| Chinese hamster lung (CHL/OU)   | Chromosomal aberrations     | _                  | _                     | ( <u>Lee et al., 2019</u> )                            |
| SHE cells                       | Chromosomal aberrations     | _                  | _                     | ( <u>Tsutsui et al., 1993</u> )                        |
| CH SV40-transformed liver cells | Selective DNA amplification | _                  | NA                    | (Schmezer et al., 1988)                                |
| Mouse JB6 epidermal cells       | Cell transformation         | +                  | NA                    | (Diwan et al., 1985)                                   |
| Mouse C3H/10T1/2 fibroblasts    | Cell transformation         | NS                 | _                     | ( <u>Sanchez et al., 1987</u> )                        |
| Mouse BALB 3T3 cells            | Cell transformation         | _                  | _                     | (Astill et al., 1986)                                  |
| SHE cells                       | Cell transformation         | NS                 | +                     | ( <u>Mauthe et al., 2001;</u><br>Leboeuf et al., 1996) |
| SHE cells                       | Cell transformation         | NS                 | +                     | ( <u>Mikalsen et al., 1990</u> )                       |
| SHE cells                       | Cell transformation         | NS                 | +                     | ( <u>Pant et al., 2010</u> )                           |

|                             | Endpoint            | Result             |                       |   |
|-----------------------------|---------------------|--------------------|-----------------------|---|
| Species (test system)       |                     | With<br>Activation | Without<br>Activation | Reference                                     |
| SHE cells                   | Cell transformation | NS                 | +                     | ( <u>Sanner and Rivedal,</u><br><u>1985</u> ) |
| SHE cells                   | Cell transformation | +                  | ±                     | ( <u>Tsutsui et al., 1993</u> )               |
| Rat hepatocytes             | DNA binding         | _                  | NA                    | ( <u>Gupta et al., 1985</u> )                 |
| Human fetal pulmonary cells | Aneuploidy          |                    | NA                    | ( <u>Stenchever et al.,</u><br><u>1976</u> )  |
| Rat RL4 liver cells         | Polyploidy          | _                  | NA                    | (Priston and Dean,<br>1985)                   |

<sup>*a*</sup> Adapted from Table 2-18 of ATSDR (2022).

<sup>b</sup> Mutagenic effect coincident with cytotoxicity.

Abbreviations: - = negative result; + = positive result;  $\pm$  = equivocal result; CHO = Chinese hamster ovary; DNA = deoxyribonucleic acid; NA = not applicable to mammalian cell cultures with endogenous metabolic activity; NS = not specified; SHE = Syrian hamster embryo

4572

4573

#### 4574 Table\_Apx A-2. Genotoxicity of MEHP *In Vitro* (Studies Considered by ATSDR (2022))<sup>*a*</sup>

|  | Endpoint               | Result             |                       |   |
|--|------------------------|--------------------|-----------------------|---|
| Species (test system)  |                        | With<br>Activation | Without<br>Activation | Reference                                   |
|  | Prokaryotic organism   | ns                 |                       |   |
| <i>S. typhimurium</i> (TA98, TA100, TA1535, TA1538)                  | Gene mutation          | -                  | -                     | (Agarwal et al., 1985)                      |
| S. typhimurium (NS)  | Gene mutation          | -                  | —                     | (Astill et al., 1986)                       |
| <i>S. typhimurium</i> (TA97, TA98, TA100, TA102)                     | Gene mutation          | _                  | —                     | ( <u>Dirven et al., 1991</u> )              |
| <i>S. typhimurium</i> (TA98, TA100, TA1535, TA1537, TA1538)          | Gene mutation          | -                  | -                     | ( <u>Kirby et al., 1983</u> )               |
| <i>S. typhimurium</i> (TA98, TA100, TA1535, TA1537, TA1538)          | Gene mutation          | _                  | _                     | ( <u>Ruddick et al., 1981</u> )             |
| S. typhimurium (TA100, TA102)  | Gene mutation          | _                  | _                     | ( <u>Schmezer et al., 1988</u> )            |
| S. typhimurium (TA100)   | Gene mutation          | —                  | ±                     | ( <u>Tomita et al., 1982</u> )              |
| S. typhimurium (TA98, TA100)   | Gene mutation          | _                  | _                     | ( <u>Yoshikawa et al.,</u><br><u>1983</u> ) |
| <i>S. typhimurium</i> (TA98, TA100, TA1535, TA1537)                  | Gene mutation          | _                  | —                     | (Zeiger et al., 1985)                       |
| Escherichia coli (WP2 B/r)   | Gene mutation          | NS                 | $\pm^b$               | ( <u>Tomita et al., 1982</u> )              |
| <i>E. coli</i> (WP2 <i>try</i> – [ <i>UvrA</i> + and <i>UvrA</i> –]) | Gene mutation          | -                  | —                     | ( <u>Yoshikawa et al.,</u><br><u>1983</u> ) |
| Bacillus subtilis (H17, M45)   | DNA damage (Rec assay) | NS                 | +                     | ( <u>Tomita et al., 1982</u> )              |
| Mammalian cells  |                        |                    |                       |   |
|                                       |                           | Res                | sult                  |   |  |
|---------------------------------------|---------------------------|--------------------|-----------------------|---|--|
| Species (test system)                 | Endpoint                  | With<br>Activation | Without<br>Activation | Reference   |  |
| Mouse lymphoma cells L5178Y (tk+/tk-) | Mutagenicity              | —                  | -                     | (Kirby et al., 1983)                                    |  |
| CHO cells                             | Mutagenicity              | NS                 | Ι                     | ( <u>Phillips et al., 1982</u> )                        |  |
| CHO cells (AS52)                      | Mutagenicity              | NS                 | +                     | ( <u>Chang et al., 2017</u> )                           |  |
| Human leukocytes                      | DNA damage                | NS                 | +                     | ( <u>Anderson et al., 1999</u> )                        |  |
| Human LNCaP prostatic cancer cells    | DNA damage                | NS                 | +                     | ( <u>Erkekoglu et al.,</u><br><u>2010b</u> )            |  |
| Mouse MA-10 Leydig tumor cells        | DNA damage                | NS                 | +                     | ( <u>Erkekoglu et al.,</u><br><u>2010a</u> )            |  |
| Human peripheral lymphocytes          | DNA damage                | NS                 | +                     | ( <u>Kleinsasser et al.,</u><br>2004)                   |  |
| Human nasal mucosa cells              | DNA damage                | NS                 | +                     | ( <u>Kleinsasser et al.,</u><br>2004)                   |  |
| CHO cells (AS52)                      | DNA damage                | NS                 | +                     | ( <u>Chang et al., 2017</u> )                           |  |
| Human HepG2 cells                     | Oxidative DNA<br>damage   | NS                 | +                     | ( <u>Yang et al., 2012</u> )                            |  |
| Human primary hepatocytes             | DNA repair                | -                  | NA                    | ( <u>Butterworth et al.,</u><br><u>1984</u> )           |  |
| Rat primary hepatocytes               | DNA repair                | —                  | NA                    | (Cattley et al., 1986)                                  |  |
| Mouse primary hepatocytes             | DNA repair                | _                  | NA                    | ( <u>Smith-Oliver and</u><br><u>Butterworth, 1987</u> ) |  |
| Hamster SV40 transformed cells        | DNA amplification         | NS                 | _                     | ( <u>Schmezer et al., 1988</u> )                        |  |
| Chinese hamster V79 fibroblasts       | Sister chromatid exchange | NS                 | +                     | ( <u>Tomita et al., 1982</u> )                          |  |
| Rat RL4 liver cells                   | Chromosomal aberrations   | NS                 | +                     | ( <u>Phillips et al., 1986</u> )                        |  |
| CHO cells                             | Chromosomal aberrations   | +                  | +                     | ( <u>Phillips et al., 1986</u> )                        |  |
| CHO cells                             | Chromosomal aberrations   | NS                 | +                     | ( <u>Phillips et al., 1982</u> )                        |  |
| SHE cells                             | Chromosomal aberrations   | +                  | -                     | ( <u>Tsutsui et al., 1993</u> )                         |  |
| CHO transformed cells                 | Gene mutation             | NS                 | +                     | ( <u>Chang et al., 2017</u> )                           |  |
| Mouse BALB 3T3 cells                  | Cell transformation       | -                  | _                     | ( <u>Astill et al., 1986</u> )                          |  |
| Mouse C3H/10T1/2 fibroblasts          | Cell transformation       | NS                 | _                     | ( <u>Sanchez et al., 1987</u> )                         |  |
| SHE cells                             | Cell transformation       | NS                 | +                     | ( <u>Mikalsen et al., 1990</u> )                        |  |
| SHE cells                             | Cell transformation       | +                  | —                     | ( <u>Tsutsui et al., 1993</u> )                         |  |

<sup>*a*</sup> Adapted from Table 2-19 of ATSDR (2022). <sup>*b*</sup> Mutagenic effect coincident with cytotoxicity.

Abbreviations: - = negative result; + = positive result;  $\pm$  = equivocal result; DNA = deoxyribonucleic acid; NA = not applicable to mammalian cell cultures with endogenous metabolic activity; NS = not specified

4575

#### 4576

#### 4577 Table\_Apx A-3. Genotoxicity of DEHP *In Vivo* (Studies Considered by ATSDR (2022))<sup>*a*</sup>

| Species (exposure route) Endpoint     |  | Result   | Reference                                       |
|---------------------------------------|--|----------|---|
|                                       | Mammals                                      | <u>.</u> | <u>.</u>  |
| Mouse (subcutaneous)                  | Dominant lethal test                         | +        | ( <u>Autian, 1982</u> )                         |
| Mouse (gavage)                        | Dominant lethal test                         | _        | ( <u>Rushbrook et al.,</u><br><u>1982</u> )     |
| Mouse (intraperitoneal)               | Dominant lethal test                         | +        | ( <u>Singh et al., 1974</u> )                   |
| Rat (gpt delta transgenic) (diet)     | Gene mutation in liver                       | -        | ( <u>Kanki et al., 2005</u> )                   |
| Mouse (lacZ transgenic) (NS)          | Gene mutation in liver                       | +        | (Boerrigter, 2004)                              |
| Mouse (lacZ transgenic) (NS)          | Gene mutation in kidney or spleen            | _        | ( <u>Boerrigter, 2004</u> )                     |
| Hamster embryo (gavage; via placenta) | 8AG/6TG-resistant mutation                   | +        | ( <u>Tomita et al., 1982</u> )                  |
| Mouse (NS)                            | Micronuclei in bone marrow                   | _        | ( <u>Astill et al., 1986</u> )                  |
| Mouse (intraperitoneal)               | Micronuclei in bone marrow                   | -        | (Douglas et al., 1986)                          |
| Mouse (Oral)                          | Micronuclei in bone marrow                   | -        | (Lee et al., 2019)                              |
| Human (unknown)                       | DNA damage in sperm and granulosa cells      | +        | ( <u>Al-Saleh et al., 2019</u> )                |
| Human (unknown)                       | DNA damage in peripheral blood cells         | _        | ( <u>Franken et al., 2017</u> )                 |
| Rat (gavage, diet)                    | DNA damage in liver                          | _        | (Butterworth et al.,<br><u>1984</u> )           |
| Rat (diet)                            | DNA damage in liver                          | _        | ( <u>Tamura et al., 1991</u> )                  |
| Rat (diet)                            | DNA damage in liver                          | -        | ( <u>Pogribny et al., 2008</u> )                |
| Rat (gavage)                          | DNA damage in sperm                          | +        | ( <u>Hsu et al., 2016</u> )                     |
| Rat (gavage)                          | DNA damage in blood<br>lymphocytes and sperm | +        | ( <u>Karabulut and Barlas,</u><br><u>2018</u> ) |
| Rat (gavage)                          | DNA damage in thyroid                        | +        | ( <u>Kim et al., 2019</u> )                     |
| Mouse (pipette)                       | Oxidative DNA damage in brain                | +        | ( <u>Barakat et al., 2018</u> )                 |
| Mouse (gavage)                        | Oxidative DNA damage in oocytes              | +        | ( <u>Lu et al., 2019</u> )                      |
| Rat (diet)                            | DNA base modification in liver               | _        | ( <u>Cattley and Glover,</u><br><u>1993</u> )   |
| Rat (diet)                            | DNA base modification in liver               | +        | ( <u>Takagi et al., 1990</u> )                  |
| Rat (gavage, diet)                    | DNA repair in liver                          | -        | (Butterworth et al.,<br><u>1984</u> )           |
| Rat (diet)                            | DNA repair in liver                          | -        | (Cattley et al., 1988)                          |
| Rat (gavage, diet)                    | DNA repair in liver                          | -        | (Kornbrust et al., 1984)                        |
| Rat (gavage)                          | DNA repair in liver                          | +        | (Hayashi et al., 1998)                          |

| Species (exposure route)   | Endpoint                               | Result | Reference   |  |  |  |  |
|--|--|--------|---|--|--|--|--|
| Mouse (gavage, diet)   | DNA repair in liver                    | _      | (Smith-Oliver and<br>Butterworth, 1987)                   |  |  |  |  |
| Rat (diet)   | DNA binding in liver                   | +      | ( <u>Albro et al., 1982</u> )                             |  |  |  |  |
| Rat (gavage)   | DNA binding in liver                   | _      | ( <u>Gupta et al., 1985</u> )                             |  |  |  |  |
| Rat (gavage, diet)   | DNA binding in liver                   |        | ( <u>Lutz, 1986; von</u><br><u>Däniken et al., 1984</u> ) |  |  |  |  |
| Human (occupational)   | Chromosomal aberrations in leucocytes  | _      | (Thiess and Fleig,<br><u>1978</u> )                       |  |  |  |  |
| Rat (gavage)   | Chromosomal aberrations in bone marrow | _      | (Putman et al., 1983)                                     |  |  |  |  |
| Hamster embryo (gavage; via placenta)  | Chromosomal aberrations                | +      | ( <u>Tomita et al., 1982</u> )                            |  |  |  |  |
| Hamster embryo (gavage; via placenta)  | Cell transformation                    | +      | ( <u>Tomita et al., 1982</u> )                            |  |  |  |  |
| Rat embryo (intraperitoneal; via placenta)   | Mitotic recombination                  | +      | (Fahrig and Steinkamp-<br>Zucht, 1996)                    |  |  |  |  |
| Rat (diet)   | Tetraploid nuclei in liver             | +      | (Ahmed et al., 1989)                                      |  |  |  |  |
|  | Host-mediated assay                    |        |   |  |  |  |  |
| Salmonella typhimurium (TA100); (rat host-meditated)   | Gene mutation                          | _      | ( <u>Kozumbo et al., 1982</u> )                           |  |  |  |  |
|  | Eukaryotic Organisms                   |        |   |  |  |  |  |
| Drosophila melanogaster (feeding)  | Mitotic recombination                  | _      | ( <u>Vogel and Nivard,</u><br><u>1993</u> )               |  |  |  |  |
| D. melanogaster (injection)  | Sex linked recessive lethal            | _      | ( <u>Yoon et al., 1985</u> )                              |  |  |  |  |
| <sup><i>a</i></sup> Adapted from Table 2-20 of ATSDR (2022).<br>Abbreviations: $-$ = negative result; $+$ = positive result; DNA = deoxyribonucleic acid; <i>gpt</i> = guanine phosphoribosyltransferase |  |        |   |  |  |  |  |

#### 4578 4579

4580

#### Table\_Apx A-4. Genotoxicity of MEHP In Vivo (Studies Considered by ATSDR (2022))<sup>a</sup>

| Species (exposure route)  | Endpoint                               | Result | Reference                                      |
|---|--|--------|--|
| Rat (gavage)  | DNA damage in liver                    | _      | ( <u>Elliott and Elcombe,</u><br><u>1987</u> ) |
| Rat (gavage)  | Chromosomal aberrations in bone marrow | _      | ( <u>Putman et al., 1983</u> )                 |
| Hamster embryo (gavage; via placenta)   | Chromosomal aberrations                | +      | ( <u>Tomita et al., 1982</u> )                 |
| Hamster embryo (gavage; via placenta)   | Cell transformation                    | +      | (Tomita et al., 1982)                          |
| Hamster embryo (gavage; via placenta  | 8AG/6TG-resistant mutation             | +      | ( <u>Tomita et al., 1982</u> )                 |
| <sup><i>a</i></sup> Adapted from Table 2-21 of ATSDR (2022).<br>Abbreviations: – = negative result; + = positive resu | lt                                     |        |  |

#### May 2025 Table\_Apx A-5. Summary of NTP Genotoxicity Testing of DEHP (As Reported in NTP (2021b) 4582

| Species (Test System)  | Result  |
|--|---|
| In vitro Studies   |   |
| Bacterial gene mutations: <i>Salmonella typhimurium</i> strains TA100, TA1535, TA1537, TA97, TA98 treated with 100 to 1,000 µg DEHP per plate with and without exogenous metabolic activation systems ( <i>i.e.</i> , induced hamster, rat, or mouse liver S9) | Negative with and without S9 in 6 independent assays  |
| Mouse lymphoma gene mutation assay with L5178Y $tk^{+/-}$ cells with 0.125 to 3.0 $\mu$ L/mL DEHP with and without induced rat liver S9  | Negative with and without S9 in 1 assay   |
| <i>In vitro</i> CHO cell chromosomal aberration test with and without induced rat liver S9   | Negative with and without S9 in 3 independent studies at concentrations up to $5,000 \ \mu g/mL$  |
| <i>In vitro</i> CHO cell sister chromatid exchange test with and without induced rat liver S9  | Positive in 4, equivocal in 3, and negative in 2<br>out of 9 studies without rat liver S9   |
|  | Positive or equivocal results were only observed<br>at concentrations of DEHP that induced severe<br>cell cycle delay that necessitated longer<br>incubation times. Cytotoxicity and longer<br>incubation times may have contributed to<br>increased SCE levels, rather than direct<br>interactions of DEHP with chromosomal DNA. |
|  | Negative in 9 out of 9 studies with rat liver S9  |
| In vivo Studies  |   |
| <i>In vivo</i> chromosome aberration test with female B6C3F1 mice fed diets containing 3,000 to 12,000 ppm DEHP for 14 days  | No increase in chromosomal aberrations in bone marrow cells   |
| In vivo micronucleus test in mice  | Equivocal overall result in B6C3F1 females<br>exposed to 3,000 to 12,000 ppm DEHP in feed<br>for 14 days  |
|  | Equivocal in male TgAC (FVB/N) mice and<br>positive in female mice exposed to 1,500 to<br>6,000 ppm DEHP in feed for 26 weeks   |
|  | Negative in male and female TgAC (FVB/N)<br>mice exposed dermally to 100 to 400 mg/kg-day<br>DEHP for 26 weeks  |
| Drosophila melanogaster sex-linked recessive lethal test   | Negative (adult injection)  |
|  | Negative (larval feeding)   |

#### 4585 PUBLIC RELEASE DRAFT May 2025 4585 Appendix B RODENT CARCINOGENICITY STUDY SUMMARIES

4586

4589

### 4587 **B.1 Di(2-ethylhexyl) Phthalate (DEHP)**

4588 B.1.1 Mice - Oral Exposure Studies

#### B.1.1.1 Two-year Dietary Study of B6C3F1 Mice (NTP, 1982a)

4590 NTP (1982a) reports the results of a 2-year dietary study of male and female B6C3F1mice. Male and 4591 female mice (50 per sex per dose) were administered diets containing 0, 3,000, and 6,000 ppm DEHP 4592 (equivalent to approximately 673 and 1,325 mg/kg-day for males and 799 and 1,821 mg/kg-day for 4593 females) for 103 weeks. Terminal body weight was reduced 7 and 10 percent in low- and high-dose 4594 males, respectively, and 21 and 33 percent in low- and high-dose females, respectively. Average daily 4595 feed consumption per rat was 100 and 96 percent of controls for low-dose males and females, 4596 respectively, and 96 and 100 percent of controls for high-dose males and females, respectively. No 4597 compound-related clinical signs were reported. No significant effects on survival were observed for 4598 males, however, survival was significantly reduced for low-dose females (survival of control, low- and high-dose: 34/50, 38/50, 35/50 for males; 39/50, 25/50, 33/50 for females). Dose-related, statistically 4599 4600 significant increases in hepatocellular carcinoma were observed in high-dose male mice, while 4601 combined hepatocellular carcinoma and adenoma were significantly increased in low- and high-dose 4602 male mice compared to controls (Table Apx B-1). Similarly, statistically significant increases in 4603 hepatocellular carcinoma and combined hepatocellular carcinoma and adenoma were observed in low-4604 and high-dose female mice (Table\_Apx B-1). No other tumor types were significantly increased in male or female mice at any dose. 4605

4606

4607 Under the conditions of the study, NTP concluded that DEHP was carcinogenic for B6C3F1 mice,4608 causing increased incidence of male and female mice with hepatocellular carcinomas.

4609 4610

### Table\_Apx B-1. Incidence of Liver Tumors in Male and Female B6C3F1 Mice Fed Diets Containing DEHP for Two Years (NTP, 1982a)<sup>a</sup>

| Tissue: Tumor Type   | Control     | 3,000 ppm    | 6,000 ppm    |  |  |  |  |  |
|--|-------------|--------------|--------------|--|--|--|--|--|
| Male Mice  |             |              |              |  |  |  |  |  |
| Liver: Hepatocellular carcinoma  | 9/50 (18%)  | 14/48 (29%)  | 19/50 (38%)* |  |  |  |  |  |
| Liver: Hepatocellular adenoma  | 6/50 (12%)  | 11/48 (23%)  | 10/50 (20%)  |  |  |  |  |  |
| Liver: Hepatocellular carcinoma or adenoma   | 14/50 (28%) | 25/48 (52%)* | 29/50 (58%)* |  |  |  |  |  |
| Female   | Mice        |              |              |  |  |  |  |  |
| Liver: Hepatocellular carcinoma  | 0/50        | 7/50 (14%)*  | 17/50 (34%)* |  |  |  |  |  |
| Liver: Hepatocellular adenoma  | 1/50 (2%)   | 5/50 (10%)   | 1/50 (2%)    |  |  |  |  |  |
| Liver: Hepatocellular carcinoma or adenoma   | 1/50 (2%)   | 12/50 (24%)* | 18/50 (36%)* |  |  |  |  |  |
| <sup><i>a</i></sup> Asterisk indicates statistically significant pairwise comparison to the control by Fisher exact test ( $P < 0.05$ ) when the Cochran-Armitage test was statistically significant ( $P<0.05$ ). Data from Tables 15 and 16 of (NTP, 1982a). |             |              |              |  |  |  |  |  |

May 2025

#### 4614 4615

#### B.1.1.2 Two-year Dietary Study of B6C3F1 Mice (<u>David et al., 2000a</u>; <u>David et al.,</u> 1999)

David et al. (2000a; 1999) reports the results of a 2-year dietary study of male and female B6C3F1mice. 4616 Briefly, male and female mice (65–70 per sex per dose) were administered diets containing 0, 100, 500, 4617 4618 1,500, and 6,000 ppm DEHP for up to 104 weeks (equivalent to 19, 99, 292, 1,266 mg/kg-day for males; 4619 24, 117, 354, 1,458 mg/kg-day for females). An additional recovery group was included in which male 4620 and female mice (55/sex) were fed diets containing 6,000 ppm DEHP for 78 weeks, and then control diet for an additional 26 weeks. Survival was significantly reduced for high-dose males. Adjusted 4621 4622 survival rates at study termination were 75, 80, 71, 71, and 31 percent for males and 63, 66, 73, 72, and 4623 61 percent for females across dose groups. The most common cause of death was hepatocellular 4624 neoplasia, which was most frequently observed in mice fed diets containing 1,500 and 6,000 ppm 4625 DEHP. Mean body weight gain was significantly lower in high-dose males compared to controls (mean body weight change for control and high-dose males:  $10.5 \pm 2.7$  vs.  $5.8 \pm 2.5$  grams), but was not 4626 4627 significantly affected for females in any dose group. Incidence of combined hepatocellular adenomas 4628 and carcinomas were statistically significantly increased in a dose-related manner in male mice at 500 4629 ppm DEHP and above and in female mice at 1,500 ppm DEHP and above (Table Apx B-2). No other 4630 tumor types were significantly increased in male or female mice at any dose.

4631 4632

### Table\_Apx B-2. Incidence of Liver Tumors in Male and Female B6C3F1 Mice Fed Diets Containing DEHP for Two-years (David et al., 2000a; David et al., 1999)<sup>a</sup>

| Tissue: Tumor Type                         | 0 ppm         | 100<br>ppm     | 500<br>ppm      | 1500<br>ppm     | 6000<br>ppm     | Recovery        | Historical |
|--|---------------|----------------|-----------------|-----------------|-----------------|-----------------|------------|
|  | -             | Male           | Mice            | -               | -               |                 |            |
| Liver: Hepatocellular carcinoma            | 4/70<br>(6%)  | 5/60<br>(8%)   | 9/65<br>(14%)   | 14/65<br>(22%)  | 22/70<br>(31%)  | 12/55<br>(22%)  |            |
| Liver: Hepatocellular adenoma              | 4/70<br>(6%)  | 10/60<br>(17%) | 13/65<br>(20%)  | 14/65<br>(22%)  | 19/70<br>(27%)  | 3/55<br>(5%)    |            |
| Liver: Hepatocellular carcinoma or adenoma | 8/70<br>(11%) | 14/60<br>(23%) | 21/65*<br>(32%) | 27/65*<br>(42%) | 37/70*<br>(53%) | 14/55*<br>(26%) | 41/149     |
|  |               | Female         | e Mice          |                 |                 |                 |            |
| Liver: Hepatocellular carcinoma            | 3/70<br>(4%)  | 2/60<br>(3%)   | 3/65<br>(5%)    | 10/65<br>(15%)  | 6/70<br>(23%)   | 23/55<br>(42%)  |            |
| Liver: Hepatocellular adenoma              | 0/70          | 2/60<br>(3%)   | 4/65<br>(6%)    | 9/65<br>(14%)   | 34/70<br>(49%)  | 13/55<br>(24%)  |            |
| Liver: Hepatocellular carcinoma or adenoma | 3/70<br>(4%)  | 4/60<br>(6%)   | 7/65<br>(11%)   | 19/65*<br>(29%) | 44/70*<br>(63%) | 30/55*<br>(55%) | 11/151     |

<sup>*a*</sup> Asterisk indicates statistically significant pairwise comparison to the control by Fisher exact test ( $P \le 0.05$ ) as determined by original study authors. Data from Table 6 of (David et al., 1999).

4635

#### 4636 B.1.2 Rats - Oral Exposure Studies

4637

### B.1.2.1 Two-year Dietary Study of F344 Rats (NTP, 1982a)

NTP (<u>1982a</u>) reports the results of a 2-year dietary study of male and female F344 Rats. Male and
female rats (50 per sex per dose) were administered diets containing 0, 6,000, and 12,000 ppm DEHP

4640 (equivalent to approximately 322, 674 mg/kg-day for males; 394, 774 mg/kg-day for females) for 103 weeks. Terminal body weight was reduced 11 and 15 percent in low- and high-dose males, respectively, 4641 4642 and 5 and 20 percent in low- and high-dose females, respectively. Average daily feed consumption per 4643 rat was 86 and 85 percent of controls for low-dose males and females, respectively, and 86 and 75 4644 percent of controls for high-dose males and females, respectively. No compound-related clinical signs 4645 were reported. No significant effects on survival were observed (survival of control, low- and high-dose: 4646 30/50, 28/50, 33/50 for males; 36/50, 34/50, 38/50 for females). No significant increases in MNCL or 4647 pancreatic acinar cell adenomas were observed in either sex. Compared to controls, the incidence of 4648 testicular interstitial cell tumors was significantly decreased in high-dose male rats; however, the spontaneous background rate of this tumor type was high (96%) in control males (Table\_Apx B-3). 4649 4650 Dose-related, statistically significant increases in combined neoplastic nodules and hepatocellular carcinomas were observed in high-dose male rats (incidence: 12/49 compared to 3/50 for controls). 4651 Similarly, statistically significant increases in hepatocellular carcinoma and neoplastic nodules were 4652 4653 observed in high-dose females, while the incidence of combined hepatocellular carcinomas and neoplastic nodules was significantly increased in low and high-dose females (combined incidence: 0/50, 4654 4655 6/49, 13/50) (Table\_Apx B-3).

4656

4657 Under the conditions of the study, NTP concluded that DEHP was carcinogenic for F344 rats, causing
4658 increased incidence of female rats with hepatocellular carcinomas, and inducing an increased incidence
4659 of male rats with either hepatocellular carcinomas or neoplastic nodules.

4660 4661

### Table\_Apx B-3. Incidence of Tumors in Male and Female F344 Rats Fed Diets Containing DEHP for Two Years (NTP, 1982a)<sup>a</sup>

| Tissue: Tumor Type  | Control     | 6000 ppm    | 12,000 ppm   |  |  |  |  |  |
|---|-------------|-------------|--------------|--|--|--|--|--|
| Male Rats   |             |             |              |  |  |  |  |  |
| Testis: Interstitial cell tumor   | 47/49 (96%) | 42/44 (95%) | 11/48 (23%)* |  |  |  |  |  |
| Liver: Hepatocellular carcinoma   | 1/50 (2%)   | 1/49 (2%)   | 5/49 (10%)   |  |  |  |  |  |
| Liver: Neoplastic nodule  | 2/50 (4%)   | 5/49 (10%)  | 7/49 (14%)   |  |  |  |  |  |
| Liver: Hepatocellular carcinoma or neoplastic nodule  | 3/50 (6%)   | 6/49 (12%)  | 12/49 (24%)* |  |  |  |  |  |
| Female  | Rats        |             |              |  |  |  |  |  |
| Liver: Hepatocellular carcinoma   | 0/50        | 2/49 (2%)   | 8/50 (16%)*  |  |  |  |  |  |
| Liver: Neoplastic nodule  | 0/50        | 4/49 (8%)   | 5/50 (10%)*  |  |  |  |  |  |
| Liver: Hepatocellular carcinoma or neoplastic nodule  | 0/50        | 6/49 (12%)* | 13/50 (26%)* |  |  |  |  |  |
| <sup><i>a</i></sup> Asterisk indicates statistically significant pairwise comparison to the control by Fisher exact test ( $P < 0.05$ ) when the Cochran-Armitage test was statistically significant ( $P < 0.05$ ). Data from Tables 11 and 12 of (NTP 1982a). |             |             |              |  |  |  |  |  |

4664

4665

#### B.1.2.2 Two-year Dietary Study of F344 Rats (David et al., 2000b; David et al., 1999)

David et al. (2000b; 1999) report the results of a 2-year dietary study of male and female F344 Rats.
Briefly, male and female rats (55–80 per sex per dose) were administered diets containing 0, 100, 500,
2,500, and 12,500 ppm DEHP for up to 104 weeks (equivalent to 6, 29, 147, 780 mg/kg-day for males;

4669 7, 36, 182, 939 mg/kg-day for females). An additional recovery group was included in which male and 4670 female rats (55/sex) were fed diets containing 12,500 ppm DEHP for 78 weeks, and then control diet for

4671 an additional 26 weeks. Survival was not significantly affected by treatment with DEHP, although there was trend toward lower survival for high-dose rats. Adjusted survival rates at study termination were 82, 4672 78, 78, 70, and 73 percent for males and 80, 86, 80, 76, and 70 percent for females across dose groups, 4673 4674 respectively. The most frequent cause of death was reported to be due to MNCL. Mean body weights for 4675 high-dose male and female rats were significantly lower than the control for the duration of the study. From study week 1 to 105, mean body weight gain was 226 vs. 192 grams for control and high-dose 4676 4677 males, respectively, and 149 vs. 126 grams for control and high-dose females, respectively. For females, 4678 the only tumor type significantly increased compared to concurrent controls was incidence of combined 4679 hepatocellular adenomas and carcinomas in the 100 ppm, 12,500 ppm, and recovery group. However, 4680 the effect on incidence of liver tumors in female rats was only dose-related at the high-dose group 4681 (Table\_Apx B-4). In male rats, a treatment related increase in incidence of pancreatic acinar cell adenomas was observed in the high-dose group (incidence: 0/60 vs. 5/59 in control and high-dose group, 4682 respectively) (Table\_Apx B-4). Additionally, in the two highest dose groups (i.e., 2,500 and 12,500 4683 4684 ppm) incidence of MNCL and combined hepatocellular adenomas and carcinomas was statistically significantly increased compared to concurrent controls (Table\_Apx B-4). Incidence of interstitial cell 4685 tumor in the testis was significantly decreased compared to concurrent controls (Table\_Apx B-4). 4686 4687

4688

### Table\_Apx B-4. Incidence of Tumors in Male and Female F344 Rats Fed Diets Containing DEHP for Two-years (David et al., 2000b; David et al., 1999)<sup>a</sup>

| Tissue: Tumor Type                         | 0 ppm          | 100<br>ppm     | 500<br>ppm     | 2500<br>ppm     | 12,500<br>ppm   | Recovery        | Historical |  |  |
|--|----------------|----------------|----------------|-----------------|-----------------|-----------------|------------|--|--|
| Male Rats                                  |                |                |                |                 |                 |                 |            |  |  |
| Liver: Hepatocellular carcinoma            | 1/80<br>(1%)   | 0/50           | 1/55<br>(2%)   | 3/65<br>(5%)    | 24/80<br>(34%)  | 7/55<br>(13%)   |            |  |  |
| Liver: Hepatocellular adenoma              | 4/80<br>(5%)   | 5/50<br>(10%)  | 3/55<br>(6%)   | 8/65<br>(12%)   | 21/80<br>(30%)  | 12/55<br>(22%)  |            |  |  |
| Liver: Hepatocellular carcinoma or adenoma | 5/80<br>(7%)   | 5/50<br>(10%)  | 4/55<br>(7%)   | 11/65*<br>(17%) | 34/80*<br>(43%) | 18/55*<br>(33%) | 11/323     |  |  |
| Testis: Interstitial cell tumor            | 59/64<br>(92%) | 45/50<br>(90%) | 50/55<br>(91%) | 60/65<br>(92%)  | 20/64*<br>(31%) |                 |            |  |  |
| Pancreas: Acinar cell adenoma              | 0/60           | 0/17           | 0/14           | 0/18            | 5/59*<br>(8%)   |                 |            |  |  |
| MNCL                                       | 15/65<br>(23%) | 13/50<br>(26%) | 16/55<br>(27%) | 32/65*<br>(49%) | 27/65*<br>(42%) |                 |            |  |  |
|  | -              | Femal          | e Rats         | •               | •               | •               |            |  |  |
| Liver: Hepatocellular carcinoma            | 0/80           | 1/50<br>(2%)   | 0/55           | 1/65<br>(2%)    | 14/80<br>(20%)  | 4/55<br>(7%)    |            |  |  |
| Liver: Hepatocellular adenoma              | 0/80           | 3/50<br>(6%)   | 1/55<br>(2%)   | 2/65<br>(3%)    | 8/80<br>(10%)   | 6/55<br>(11%)   |            |  |  |
| Liver: Hepatocellular carcinoma or adenoma | 0/80           | 4/50*<br>(8%)  | 1/55<br>(2%)   | 3/65<br>(5%)    | 22/80*<br>(31%) | 10/55*<br>(18%) | 4/320      |  |  |
| Pancreas: Acinar cell adenoma              | 0/60           | 0/7            | 0/10           | 0/14            | 2/60<br>(3%)    |                 |            |  |  |
| MNCL                                       | 14/65<br>(22%) | 17/50<br>(34%) | 11/55<br>(20%) | 16/65<br>(25%)  | 17/65<br>(26%)  |                 |            |  |  |

#### IDI IC DELEASE DDAET

|      |  | PUDI                           | May 20                       | 35e draf<br>)25              | 1                            |                               |                              |              |
|------|--|--------------------------------|------------------------------|------------------------------|------------------------------|-------------------------------|------------------------------|--------------|
|      | Tissue: Tumor Type   | 0 ppm                          | 100<br>ppm                   | 500<br>ppm                   | 2500<br>ppm                  | 12,500<br>ppm                 | Recovery                     | Historical   |
|      | <sup><i>a</i></sup> Asterisk indicates statistically signific<br>original study authors. Data from Table | cant pairwise<br>e 5 of (David | comparison<br>l et al., 1999 | to the contr<br>) and Tables | ol by Fisher<br>6 and 7 of ( | exact test (I<br>David et al. | $P \le 0.05$ ) as d, 2000b). | etermined by |
| 4691 |  |                                |                              |                              |                              |                               |                              |              |
| 4692 | B.1.2.3 Ninety-five  | Week Diet                      | tary Study                   | y of Male                    | F344 Rat                     | s ( <u>Rao et a</u>           | <u>al., 1987</u> )           |              |
| 4693 | Male F344 rats were fed diets con  | taining 0 c                    | or 2 percer                  | t DEHP fo                    | or 95 weel                   | cs (n = 8 a)                  | nd 10 rats i                 | in control   |
| 4694 | and DEHP dose group, respective  | ly). No liv                    | er tumors                    | were obse                    | rved in an                   | y control 1                   | rats. Four o                 | f ten rats   |
| 4695 | treated with DEHP had one or mo  | re hepatoc                     | ellular car                  | cinomas, v                   | while two                    | of ten rats                   | treated with                 | th DEHP      |
| 4696 | had neoplastic nodules. Six out of   | ten rats tr                    | eated with                   | DEHP ha                      | d neoplast                   | tic nodules                   | s or hepatod                 | cellular     |
| 4697 | carcinomas (combined) ( $P < 0.003$  | 5 by X <sup>2</sup> tes        | st).                         |                              |                              |                               |                              |              |
| 4698 |  |                                |                              |                              |                              |                               |                              |              |
| 4699 | B.1.2.4 Two-year D   | ietary Stu                     | dy of Mal                    | e F344 Ra                    | ats ( <u>Rao e</u>           | t al., 1990                   | )                            |              |
| 4700 | Male F344 rats were fed diets con  | taining 0 c                    | or 2 percer                  | t DEHP fo                    | or 108 we                    | eks ( $n = 10$                | 0 and 14 ra                  | ts in        |
| 4701 | control and DEHP dose group, res   | spectively)                    | . All rats i                 | n both gro                   | ups surviv                   | ved until so                  | cheduled no                  | ecropsy.     |
| 4702 | Terminal body weight of rats fed   | diets conta                    | ining DEF                    | HP was sig                   | nificantly                   | lower that                    | n that of co                 | ontrols      |
| 4703 | (276 vs. 378 grams). Liver tumors  | s were obse                    | erved in a                   | single mal                   | e control                    | rat, where                    | a tumor (cl                  | lassified    |
| 4704 | as a hepatocellular carcinoma) of  | 15 mm in :                     | size was o                   | bserved (7                   | Table_Apx                    | k B-5). Liv                   | vers of 11 o                 | f 14 rats    |
| 4705 | (79%) treated with DEHP contain  | ed grossly                     | visible no                   | dules mea                    | suring 1 to                  | o 15 mm i                     | n size (Tab                  | le_Apx       |
| 4706 | B-5). Grossly visible lesions less   | than 3 mm                      | in size sh                   | owed featu                   | ares consis                  | stent with                    | altered area                 | as or        |
| 4707 | neoplastic nodules, while tumors   | 3 to 5 mm                      | in size sho                  | owed featu                   | res consis                   | tent with 1                   | neoplastic r                 | iodules      |
| 4708 | and/or hepatocellular carcinoma.   | All tumors                     | greater th                   | an 5 mm s                    | showed fea                   | atures cons                   | sistent with                 | well         |
| 4709 | differentiated hepatocellular carci  | noma.                          |                              |                              |                              |                               |                              |              |
| 4710 |  |                                |                              |                              |                              |                               |                              |              |

4711

#### Table\_Apx B-5. Quantification of Liver Tumors by Size in Male F344 Rats Exposed to DEHP in 4712 the Diet for 108-weeks (Rao et al., 1990)<sup>a</sup> 4713

|  | Total       | # of rats with tumors |        |       | # of nodules per liver                |                          |                          |  |
|--|-------------|-----------------------|--------|-------|---------------------------------------|--------------------------|--------------------------|--|
| Group  | No.<br>Rats | < 3 mm                | 3-5 mm | >5 mm | < 3 mm                                | 3-5 mm                   | >5 mm                    |  |
| Control  | 10          | 0                     | 0      | 1     | 0                                     | 0                        | 1                        |  |
| 2% DEHP  | 14          | 8                     | 2      | 5     | $\frac{1.14 \pm 0.32^{b}}{(0-3)^{c}}$ | $1.14 \pm 0.32$<br>(0-1) | $1.14 \pm 0.32$<br>(0-2) |  |
| <sup>a</sup> Adapted from Table 2 in (Rao et al., 1990). |             |                       |        |       |                                       |                          |                          |  |

<sup>*b*</sup> Mean  $\pm$  SEM

<sup>c</sup> Range of number of tumors per liver.

4714

4715

#### **B.1.2.5** Lifetime Dietary Study of Male Sprague-Dawley Rats (Voss et al., 2005)

Voss et al. (2005) fed male Sprague-Dawley (SD) rats diets containing 0 (n = 390), 600 (n = 180), 1,897 4716 (n = 100), and 6,000 (n = 60) mg/kg DEHP. Rats were fed 5 grams of DEHP-diet/100 grams rat/day for 4717

6 days per week and received DEHP-free food on the seventh day only after the rest of their DEHP diet 4718

had been consumed. On this basis, rats received doses of 0, 30, 95, and 300 mg/kg-day DEHP over the 4719

entire lifetime of the animals (up to 159 weeks). Treatment with DEHP did not affect median survival 4720

times compared to control animals. Weight gain was comparable across control and all treatment groups, 4721

except for a short period around study day 300, when body weight of rats in all DEHP treated groups
was lower than the control. However, body weight of DEHP treated rats recovered to that of control
levels by around study day 500. No increase in hepatocellular adenomas and carcinomas (combined)
was observed when incidence of tumors across all rats were compared (incidence: 35/390 [9.0%],
16/180 [8.9%], 5/100 [5%], 5/60 [8.3%]). However, histopathologic examination of the liver of only rats

- 4727 found in a moribund state and sacrificed demonstrated a statistically significant dose-related increase in
- 4728 the incidence of combined hepatocellular adenomas and carcinomas in high-dose rats (Table\_Apx B-6).
- 4729 In addition to liver tumors, treatment-related, statistically significant increases in benign Leydig cell
- 4730 tumors were observed in high-dose male rats (Table\_Apx B-7).
- 4731 4732

### Table\_Apx B-6. Incidence of Liver Tumors in Male Sprague-Dawley Rats Chronically Fed Diets Containing DEHP (Voss et al., 2005)<sup>a</sup>

| Tissue: Tumor Type  | Control       | 30 mg/kg    | 95 mg/kg    | 300 mg/kg    |  |  |  |
|---|---------------|-------------|-------------|--------------|--|--|--|
| Number examined microscopically   | 167           | 84          | 53          | 31           |  |  |  |
| Hepatocellular adenomas   | 13/167 (7.8%) | 3/84 (3.6%) | 4/53 (7.5%) | 6/31 (19.4%) |  |  |  |
| Hepatocellular carcinomas   | 2/167 (1.2%)  | 3/84 (3.6%) | 0/53        | 3/31 (9.7%)  |  |  |  |
| Hepatocellular adenomas and carcinomas (combined)   | 15/167 (9.0%) | 6/84 (7.1%) | 4/53 (7.5%) | 9/31* (29%)  |  |  |  |
| <sup><i>a</i></sup> Asterisk indicates statistically significant pairwise comparison to the control ( $P \le 0.05$ ) as determined by original study authors. Data from Table 4 of (Voss et al., 2005). |               |             |             |              |  |  |  |

4735

#### 4736

## Table\_Apx B-7. Incidence of Testicular Tumors in Male Sprague-Dawley Rats Chronically Fed Diets Containing DEHP (<u>Voss et al., 2005</u>)<sup>a</sup>

| Tissue: Tumor Type              | Control      | 30 mg/kg     | 95 mg/kg            | 300 mg/kg    |
|---------------------------------|--------------|--------------|---------------------|--------------|
| Number Examined microscopically | 390          | 180          | 100                 | 60           |
| Leydig cell tumors (all)        | 64/390 (16%) | 34/180 (19%) | 21/100 (21%)        | 17/60* (28%) |
| Leydig cell tumors (unilateral) | 51/390 (13%) | 30/180 (17%) | 17/100 (17%)        | 12/60 (20%)  |
| Leydig cell tumors (bilateral)  | 13/390 (3%)  | 4/180 (2%)   | 4/100 (4%)          | 5/60 (8%)    |
| Leydig cell tumors (multifocal) | 16/390 (4%)  | 14/180 (8%)  | 5/100 (5%)          | 10/60* (17%) |
|                                 | • • • • • •  | (1/D < 0.0)  | <b>c</b> ) 1 ( ' 11 |              |

<sup>*a*</sup> Asterisk indicates statistically significant pairwise comparison to the control ( $P \le 0.05$ ) as determined by original study authors. Data from Table 6 of (Voss et al., 2005).

4739

4740 4741

### B.1.2.6 Two-year Dietary Study of Sprague-Dawley Rats (Perinatal and Postweaning Exposure Study) (NTP, 2021b)

NTP (2021b) report the results of a chronic perinatal and postweaning exposure study of DEHP.
Beginning on gestational day 6, time-mated Sprague-Dawley rats (45/group) were fed diets containing
0, 300, 1,000, 3,000 or 10,000 ppm DEHP throughout gestation and lactation. Groups of 50 male and
female F1 offspring were then fed diets containing the same respective DEHP concentration for twoyears. Mean received doses of DEHP in units of mg/kg-day for each phase of the study are shown in
Table\_Apx B-8.

| Phase of Study  | 0 ppm | 300 ppm | 1000<br>ppm | 3000<br>ppm | 10,000<br>ppm |  |
|---|-------|---------|-------------|-------------|---------------|--|
| Gestational Day 6–21                                      | 0     | 21      | 68          | 206         | 626           |  |
| Lactational Day 1-14                                      | 0     | 49      | 266         | 482         | 1244          |  |
| Two-year study (F1 males)                                 | 0     | 18      | 58          | 189         | 678           |  |
| Two-year study (F1 females)                               | 0     | 18      | 62          | 196         | 772           |  |
| <sup><i>a</i></sup> Adapted from Table 4 of (NTP, 2021b). |       |         |             |             |               |  |

### Table\_Apx B-8. DEHP Intake (mg/kg-day) during the Gestational, Perinatal, and Two-year Phases of Chronic Dietary Study of DEHP with Sprague-Dawley Rats (<u>NTP, 2021b</u>)<sup>a</sup>

4752

4753 4754 Treatment with DEHP had no effect on maternal survival, maternal clinal observations, percentage of 4755 females that produced pups, gestation length, pup sex ratio. In the high dose group, dam body weight 4756 was lower (up to 10%) compared to controls throughout gestation, with decreased body weight gain over 4757 the GD 6–9, GD 15–18, and GD 18–21 intervals. Overall, mean dam body weight gain in high-dose 4758 dams was reduced 27 percent over GD 6-21 compared to controls. Similarly, high-dose dam body 4759 weight gain was reduced 10 percent throughout the lactational period (PND 1-21). Food consumption was reduced by approximately 14 and 39 percent in high-dose dams throughout gestation and lactation, 4760 4761 respectively. On PND 1, total litter size and total live litter size was significantly reduced in the 10,000 4762 ppm group, which corresponded to a decreased number of live female offspring in the high-dose group. 4763 Offspring body weight gain was suppressed throughout PND 1-21. At weaning on PND 21, male and female offspring body weight was reduced by approximately 6 percent in the 1,000 and 3,000 ppm 4764 4765 groups, while male and female offspring body weight in the 10,000 ppm group was reduced by 53 to 55 percent. However, pup survival was unaffected, and no exposure-related clinical observations were 4766 observed, so F1 offspring from the 10,000 ppm group were carried into the postweaning phase of the 4767 4768 study. At study termination, no differences in overall survival were observed across treatment groups for 4769 male and female rats. However, terminal body weight was 30 to 32 percent lower for high-dose male 4770 and female rats compared to controls. 4771

4772 *Liver.* As can be seen from Table Apx B-9, treatment with DEHP resulted in a statistically significant 4773 increase in hepatocellular adenoma (males at 10,000 ppm and females at 3,000 ppm), hepatocellular 4774 carcinoma (females at 10,000 ppm), and combined hepatocellular adenomas and carcinomas (males at 4775 10,000 ppm and females at 3,000 ppm and above). Further, there was a statistically significant positive trend in hepatocellular carcinoma for males. Hepatocellular tumors were accompanied by numerous 4776 4777 non-neoplastic lesions in the liver of male and female rats (many of which occurred at lower doses that 4778 caused tumorigenesis), including cytoplasmic alteration of hepatocytes, hepatocellular hypertrophy, 4779 increased pigment, necrosis, eosinophilic focus, basophilic focus, and bile duct hyperplasia (see Table 4780 13 of (NTP, 2021b) for incidence data of these non-neoplastic liver lesions).

- 4781
- 4782

### Table\_Apx B-9. Incidence of Liver Tumors in SD Rats Chronically Exposed to DEHP (Perinatal and Postweaning Exposure Study) (NTP, 2021b)<sup>l</sup>

| Tissue: Tumor Type  | 0 ppm | 300 ppm   | 1000 ppm | 3000 ppm  | 10,000 ppm |  |  |
|---|-------|-----------|----------|-----------|------------|--|--|
| Male Rats   |       |           |          |           |            |  |  |
| Hepatocellular Adenoma (overall rate) <sup><i>a e</i></sup> | 0/50  | 1/49 (2%) | 0/50     | 3/50 (6%) | 8/49 (16%) |  |  |

| Tissue: Tumor Type  | 0 ppm     | 300 ppm   | 1000 ppm   | 3000 ppm   | 10,000 ppm  |
|---|-----------|-----------|------------|------------|-------------|
| Hepatocellular Adenoma (rate per litter) <sup>b</sup>                             | 0/25      | 1/25 (4%) | 0/25       | 3/25 (12%) | 7/25 (28%)  |
| Hepatocellular Adenoma (adjusted rate) <sup>c</sup>                               | 0%        | 2.4%      | 0%         | 6.7%       | 22.3%       |
| Rao-Scott-adjusted Poly-3 test <sup>d</sup>                                       | p < 0.001 | p = 0.578 | (e)        | p = 0.246  | p = 0.018   |
| Hepatocellular Carcinoma (overall rate) <sup>f</sup>                              | 1/50 (2%) | 0/49      | 0/50       | 0/50       | 3/49 (6%)   |
| Hepatocellular Carcinoma (rate per litter)  | 1/25 (4%) | 0/25      | 0/25       | 0/25       | 3/25 (12%)  |
| Hepatocellular Carcinoma (adjusted rate)  | 2.6%      | 0%        | 0%         | 0%         | 8.7%        |
| Rao-Scott-adjusted Poly-3 test  | p = 0.038 | p = 0.589 | p = 0.587  | p = 0.587  | p = 0.341   |
| Hepatocellular Adenoma or Carcinoma (combined) (overall rate) <sup><i>g</i></sup> | 1/50 (2%) | 1/49 (2%) | 0/50       | 3/50 (6%)  | 11/49 (22%) |
| Hepatocellular Adenoma or Carcinoma<br>(combined) (rate per litter)               | 1/25 (4%) | 1/25 (4%) | 0/25       | 3/25 (12%) | 9/25 (36%)  |
| Hepatocellular Adenoma or Carcinoma<br>(combined) (adjusted rate)                 | 2.6%      | 2.4%      | 0%         | 6.7%       | 30.6%       |
| Rao-Scott-adjusted Poly-3 test  | p < 0.001 | p = 0.750 | p = 0.565  | p = 0.429  | p = 0.009   |
|   | Fer       | nale Rats | -          | •          |             |
| Hepatocellular Adenoma (overall rate) <sup>h</sup>                                | 1/49 (2%) | 0/50      | 5/50 (10%) | 9/50 (18%) | 5/48 (10%)  |
| Hepatocellular Adenoma (rate per litter)  | 1/25 (4%) | 0/25      | 4/35 (6%)  | 7/25 (28%) | 5/25 (20%)  |
| Hepatocellular Adenoma (adjusted rate)  | 2.4%      | 0%        | 11.8%      | 20.9%      | 13.8%       |
| Rao-Scott-adjusted Poly-3 test  | p = 0.089 | p = 0.587 | p = 0.170  | p = 0.033  | p = 0.126   |
| Hepatocellular Carcinoma (overall rate) <sup>i</sup>                              | 0/49      | 0/50      | 0/50       | 0/50       | 8/48 (17%)  |
| Hepatocellular Carcinoma (rate per litter)  | 0/25      | 0/25      | 0/25       | 0/25       | 7/25 (28%)  |
| Hepatocellular Carcinoma (adjusted rate)  | 0%        | 0%        | 0%         | 0%         | 21.8%       |
| Rao-Scott-adjusted Poly-3 test  | p < 0.001 | (e)       | (e)        | (e)        | p = 0.023   |
| Hepatocellular Adenoma or Carcinoma (combined) (overall rate) <sup><i>j</i></sup> | 1/49 (2%) | 0/50      | 5/50 (10%) | 9/50 (18%) | 13/48 (27%) |
| Hepatocellular Adenoma or Carcinoma<br>(combined) (rate per litter)               | 1/25 (4%) | 0/25      | 4/25 (16%) | 7/25 (28%) | 11/25 (44%) |
| Hepatocellular Adenoma or Carcinoma<br>(combined) (adjusted rate)                 | 2.4%      | 0%        | 11.8%      | 20.9%      | 35.4%       |
| Rao-Scott-adjusted Poly-3 test  | p < 0.001 | p = 0.568 | p = 0.158  | p = 0.028  | p = 0.002   |

<sup>*a*</sup> Number of animals with neoplasm per number of animals necropsied.

<sup>b</sup> Number of litters with neoplasm-bearing animals per number of litters examined at site.

<sup>c</sup> Poly-3 estimated neoplasm incidence after adjustment for intercurrent mortality.

<sup>*d*</sup> Beneath the control incidence is the p value associated with the trend test. Beneath the exposed group incidence are the p values corresponding to pairwise comparisons between the control group and that exposed group. The Rao-Scott test adjusts the Poly-3 test (which accounts for differential mortality in animals that do not reach study termination) for within-litter correlation.

<sup>*e*</sup> Historical control incidence for all routes of 2-year studies (mean  $\pm$  standard deviation): 2/489 (0.44%  $\pm$  0.88%); range: 0–2%.

<sup>*f*</sup>Historical control incidence: 2/489 (0.45%  $\pm$  0.89%); range: 0–2%.

<sup>g</sup> Historical control incidence:  $4/489 (0.89\% \pm 1.06\%)$ ; range: 0-2%.

<sup>*h*</sup> Historical control incidence: 15/489 (2.65%  $\pm$  2.59%); range: 0–8%.

|   | 141  | uy 2023                              |          |          |            |
|---|--|--------------------------------------|----------|----------|------------|
| Tissue: Tumor Type  | 0 ppm  | 300 ppm                              | 1000 ppm | 3000 ppm | 10,000 ppm |
| <sup><i>i</i></sup> Historical control incidence: 1/489 (0.22%<br><sup><i>j</i></sup> Historical control incidence: 16/489 (2.879<br><sup><i>k</i></sup> (e) indicates that the value of the statistic of<br><sup><i>l</i></sup> Adapted from Table 13 in (NTP, 2021b). | $\pm 0.67\%$ ); ran<br>% $\pm 2.8\%$ ); ran<br>could not be ca | ge: 0–2%.<br>ge: 0–8%.<br>llculated. |          |          |            |
|   |  |                                      |          |          |            |

#### 4785 4786

4787 *Pancreas*. As can be seen from Table\_Apx B-10, treatment with DEHP resulted in a statistically
4788 significant increase in pancreatic acinar adenoma and combined pancreatic acinar adenoma or carcinoma
4789 in males of the 3,000 and 10,000 ppm groups. Pancreatic acinar carcinoma were observed in 3/50 males
4790 at 3000 ppm and 1/49 males at 10,000 ppm compared to 0/50 control males, however, the effect was not
4791 statistically significant. NTP also report that a clear morphological continuum from focal acinar
4792 hyperplasia to adenoma and to carcinoma was observed.

4793 4794

### Table\_Apx B-10. Incidence of Pancreatic Tumors in SD Rats Chronically Exposed to DEHP (Perinatal and Postweaning Exposure Study) (NTP, 2021b)<sup>a</sup>

| Tumor Type   | 0 ppm       | 300 ppm                 | 1000 ppm   | 3000 ppm        | 10,000 ppm  |
|--|-------------|-------------------------|------------|-----------------|-------------|
|  | Ma          | le Rats                 |            |                 | -           |
| Acinus, Hyperplasia <sup>b</sup>   | 13/50       | 9/49                    | 16/50      | 25/50           | 15/50       |
| Acinar Adenoma (overall rate) bf   | 10/50 (20%) | 7/49 (14%)              | 8/50 (16%) | 36/50 (72%)     | 22/49 (45%) |
| Acinar Adenoma (rate per litter) <sup><i>c</i></sup>                         | 8/25 (32%)  | 5/25 (20%)              | 8/25 (32%) | 24/25 (96%)     | 18/25 (72%) |
| Acinar Adenoma (adjusted rate) <sup>d</sup>                                  | 26%         | 16.6%                   | 16.9%      | 77.9%           | 62.5%       |
| Rao-Scott-adjusted Poly-3 test <sup>e</sup>                                  | p < 0.001   | p = 0.209               | p = 0.210  | p < 0.001       | p < 0.001   |
| Acinar Carcinoma (overall rate) <sup>g</sup>                                 | 0/50        | 0/49                    | 0/50       | 3/50 (6%)       | 1/49 (2%)   |
| Acinar Carcinoma (rate per litter)   | 0/25        | 0/25                    | 0/25       | 3/25 (12%)      | 1/25 (4%)   |
| Acinar Carcinoma (adjusted rate)   | 0%          | 0%                      | 0%         | 6.6%            | 2.9%        |
| Rao-Scott-adjusted Poly-3 test   | p = 0.290   | (e) <sup><i>j</i></sup> | (e)        | p = 0.250       | p = 0.534   |
| Acinar Adenoma or Carcinoma<br>(combined) (overall rate) <sup><i>h</i></sup> | 10/50 (20%) | 7/49 (14%)              | 8/50 (16%) | 38/50 (76%)     | 22/49 (45%) |
| Acinar Adenoma or Carcinoma<br>(combined) (rate per litter)                  | 8/25 (32%)  | 5/25 (20%)              | 8/25 (32%) | 25/25<br>(100%) | 18/25 (72%) |
| Acinar Adenoma or Carcinoma<br>(combined) (adjusted rate)                    | 26%         | 16.6%                   | 16.9%      | 81.2%           | 62.5%       |
| Rao-Scott-adjusted Poly-3 test   | p < 0.001   | p = 0.209N              | p = 0.210N | p < 0.001       | p < 0.001   |
|  | Fem         | ale Rats                |            |                 |             |
| Acinus, Hyperplasia  | 0/49        | 0/50                    | 0/50       | 2/50            | 3/48        |
| Acinar Adenoma (overall rate) <sup><i>i</i></sup>                            | 0/49        | 0/50                    | 0/50       | 2/50            | 1/48        |
| Acinar Adenoma (rate per litter)   | 0/25        | 0/25                    | 0/25       | 2/25            | 1/25        |
| Acinar Adenoma (adjusted rate)   | 0%          | 0%                      | 0%         | 4.6%            | 2.8%        |
| Rao-Scott-adjusted Poly-3 test   | p = 0.307   | (e)                     | (e)        | p = 0.366       | p = 0.561   |
| <sup><i>a</i></sup> Adapted from Table 14 in (NTP, 2021b).                   |             |                         |            |                 |             |

|   | IVI  | ay 2025          |                   |                  |            |  |  |  |
|---|--|------------------|-------------------|------------------|------------|--|--|--|
| Титог Туре  | 0 ppm  | 300 ppm          | 1000 ppm          | 3000 ppm         | 10,000 ppm |  |  |  |
| <sup>b</sup> Number of animals with neoplasm or lesion  | on per number o  | of animals necro | opsied.           |                  |            |  |  |  |
| <sup>c</sup> Number of litters with neoplasm-bearing a  | <sup>c</sup> Number of litters with neoplasm-bearing animals per number of litters examined at site. |                  |                   |                  |            |  |  |  |
| <sup>d</sup> Poly-3 estimated neoplasm incidence after  | adjustment for   | intercurrent m   | ortality.         |                  |            |  |  |  |
| <sup>e</sup> Beneath the control incidence is the p value associated with the trend test. Beneath the exposed group incidence are the |  |                  |                   |                  |            |  |  |  |
| p values corresponding to pairwise comparisons between the control group and that exposed group. The Rao-Scott test                   |  |                  |                   |                  |            |  |  |  |
| adjusts the Poly-3 test (which accounts for differential mortality in animals that do not reach study termination) for                |  |                  |                   |                  |            |  |  |  |
| within-litter correlation.  |  |                  |                   |                  |            |  |  |  |
| <sup>f</sup> Historical control incidence for all routes o  | f 2-year studies   | s (mean ± stand  | ard deviation): 6 | 50/488 (11.58% : | ± 9.25%);  |  |  |  |
| range: 0–28%.   |  |                  |                   |                  |            |  |  |  |
| <sup>g</sup> Historical control incidence: 4/488 (0.8% =  | ± 1.42%); range  | e: 0–4%.         |                   |                  |            |  |  |  |
| <sup>h</sup> Historical control incidence: 62/488 (12.03  | $3\% \pm 9.16\%$ ; r   | ange: 0–28%.     |                   |                  |            |  |  |  |
| <sup><i>i</i></sup> Historical control incidence: 0/489.  |  |                  |                   |                  |            |  |  |  |
| $^{j}$ (e) indicates that the value of the statistic c  | ould not be cale   | culated.         |                   |                  |            |  |  |  |
|   |  |                  |                   |                  |            |  |  |  |

4797

4798 4799 *Male Reproductive Tract.* Numerous treatment-related gross lesions were observed in the male 4800 reproductive tracts, including small testis, undescended testis, small size epididymis, incomplete 4801 preputial separation, and missing gubernaculum (see Table 15 of (NTP, 2021b) for incidence of lesions). 4802 Similarly, treatment-related non-neoplastic histopathologic lesions were noted in the testis (*i.e.*, 4803 degeneration of germinal epithelium, seminiferous tubule dysgenesis) and epididymis (*i.e.*, hypospermia) (see Table 16 of (NTP, 2021b) for incidence of lesions). A significant treatment-related 4804 4805 increase in focal hyperplasia of interstitial cells was also observed in high-dose male rats (incidence of hyperplasia across respective dose groups: 4/49, 3/49, 6/50, 5/50, 30/49). However, the incidence of 4806 interstitial cell adenomas was not significantly affected by treatment with DEHP (incidence of 4807 4808 interstitial adenoma across dose groups: 3/49, 1/49, 3/50, 5/50, 5/49). 4809

- 4810 *Uterus*. A significant positive trend with increasing exposure to DEHP in uterus endometrium 4811 adenocarcinoma and combined uterus adenoma, adenocarcinoma, squamous cell carcinoma, or 4812 squamous cell papilloma was observed (Table\_Apx B-11). However, pair-wise comparisons to the 4813 control were not statistically significant. NTP characterized this as an equivocal finding.
- 4815 Under the conditions of the study, NTP concluded:
- 4816 "Under the conditions of the perinatal and postweaning feed study (Study 1), there was clear evidence of carcinogenic activity of di(2-ethylhexyl) phthalate (DEHP) in male 4817 4818 Hsd:Sprague Dawley® SD® rats based on the increased incidences of hepatocellular adenoma or carcinoma (combined) and acinar adenoma or carcinoma (combined) 4819 4820 neoplasms (predominately adenomas) of the pancreas. There was clear evidence of 4821 carcinogenic activity of DEHP in female Hsd:Sprague Dawley® SD® rats based on the 4822 increased incidence of hepatocellular adenoma or carcinoma (combined). The occurrence 4823 of pancreatic acinar adenoma or carcinoma (combined) was considered to be related to 4824 exposure. The occurrence of uterine (including cervix) adenoma, adenocarcinoma, 4825 squamous cell carcinoma, or squamous cell papilloma (combined) in female rats may have 4826 been related to exposure." 4827
- 4828

### Table\_Apx B-11. Incidence of Uterine Tumors in SD Rats Chronically Exposed to DEHP (Perinatal and Postweaning Exposure Study) (NTP, 2021b)<sup>a</sup>

| Tissue: Tumor Type   | 0 ppm      | 300 ppm   | 1000 ppm  | 3000 ppm   | 10,000 ppm |
|--|------------|-----------|-----------|------------|------------|
| Adenoma <sup>bf</sup>  | 0/50       | 1/50      | 0/50      | 0/50       | 0/48       |
| Adenocarcinoma (overall rate) <sup>bg</sup>  | 3/50 (6%)  | 0/50      | 1/50 (2%) | 3/50 (6%)  | 6/48 (13%) |
| Adenocarcinoma (rate per litter) <sup>c</sup>  | 3/25 (12%) | 0/25      | 1/25 (4%) | 3/25 (12%) | 6/25 (24%) |
| Adenocarcinoma (adjusted rate) <sup>d</sup>  | 7%         | 0%        | 2.4%      | 7%         | 16.4%      |
| Rao-Scott-adjusted Poly-3 test <sup>e</sup>  | p = 0.008  | p = 0.147 | p = 0.325 | p = 0.653  | p = 0.184  |
| Squamous cell carcinoma (includes multiple) <sup><i>h</i></sup>  | 0/50       | 1/50      | 0/50      | 0/50       | 1/48       |
| Squamous cell papilloma (includes multiple) <sup>i</sup>   | 0/50       | 0/50      | 0/50      | 1/50       | 0/48       |
| Adenoma, adenocarcinoma, squamous<br>cell carcinoma, squamous cell papilloma<br>(combined) (overall rate) <sup>j</sup> | 3/50 (6%)  | 1/50 (2%) | 1/50 (2%) | 3/50 (6%)  | 7/48 (15%) |
| Adenoma, adenocarcinoma, squamous<br>cell carcinoma, squamous cell papilloma<br>(combined) (rate per litter)           | 3/25 (12%) | 1/25 (4%) | 1/25 (4%) | 3/25 (12%) | 7/25 (28%) |
| Adenoma, adenocarcinoma, squamous<br>cell carcinoma, squamous cell papilloma<br>(combined) (adjusted rate)             | 7%         | 2.4%      | 2.4%      | 7%         | 19%        |
| Rao-Scott-adjusted Poly-3 test   | p = 0.005  | p = 0.325 | p = 0.317 | p = 0.651  | p = 0.113  |

<sup>*a*</sup> Adapted from Table 17 in (NTP, 2021b).

<sup>b</sup> Number of animals with neoplasm or lesion per number of animals necropsied.

<sup>c</sup> Number of litters with neoplasm-bearing animals per number of litters examined at site.

<sup>d</sup> Poly-3 estimated neoplasm incidence after adjustment for intercurrent mortality.

<sup>*e*</sup> Beneath the control incidence is the p value associated with the trend test. Beneath the exposed group incidence are the p values corresponding to pairwise comparisons between the control group and that exposed group. The Rao-Scott test adjusts the Poly-3 test (which accounts for differential mortality in animals that do not reach study termination) for within-litter correlation.

<sup>*f*</sup>Historical control incidence for all routes of 2-year studies (mean  $\pm$  standard deviation): 1/350 (0.29%  $\pm$  0.76%); range: 0–2%.

<sup>*g*</sup> Historical control incidence: 20/350 (5.71%  $\pm$  3.35%); range: 2–10%.

 $^{\rm h}$  Historical control incidence: 2/350 (0.57%  $\pm$  1.51%); range: 0–4%.

<sup>*i*</sup> Historical control incidence: 0/350

<sup>*j*</sup> Historical control incidence:  $23/350 (6.57\% \pm 3.41\%)$ ; range: 2-10%.

4831

#### 4832 4833

### B.1.2.7 Two-year Dietary Study of Sprague-Dawley Rats (Postweaning Exposure Study) (NTP, 2021b)

4834 Male and female SD rats (50/sex/dose) were fed diets containing 0, 300, 1,000, 3,000, or 10,000 ppm 4835 DEHP for two-years (mean received doses: 17, 54, 170, 602 mg/kg-day for males and 17, 60, 177, 646 4836 mg/kg-day for females). Survival of male and female rats to study termination in all treatment groups 4837 was commensurate with or greater than that of control rats. At study termination, high-dose male and 4838 female rat body weight was approximately 16 and 22 percent lower than respective controls. Feed 4839 consumption by male and female rats was comparable to across treatment groups, with the exception of

4840 21 percent lower feed consumption for high-dose males during study week one. No exposure-related4841 clinical findings were observed in any treatment groups.

4842

4843 Liver. As can be seen from Table\_Apx B-12, treatment with DEHP resulted in a statistically significant 4844 increase in hepatocellular adenoma (males and females at 10,000 ppm), hepatocellular carcinoma (males 4845 at 10,000 ppm), and combined hepatocellular adenomas and carcinomas (males and females at 10,000 4846 ppm). Hepatocellular tumors were accompanied by numerous non-neoplastic lesions in the liver of male and female rats (many of which occurred at lower doses that caused tumorigenesis), including 4847 4848 cytoplasmic alteration of hepatocytes, hepatocellular hypertrophy, increased pigment, necrosis, eosinophilic focus, and clear cell focus (see Table 25 of (NTP, 2021b) for incidence data of these non-4849 4850 neoplastic liver lesions).

4851 4852

| 4853 | Table_Apx B-12. Incidence of Liver Tumors in SD Rats Exposed to DEHP in the Diet for Two- |
|------|---|
| 4854 | years (NTP, 2021b) <sup><math>k</math></sup>  |

| Tissue: Tumor Type  | 0 ppm     | 300 ppm   | 1000 ppm                | 3000 ppm  | 10,000 ppm  |  |  |  |
|---|-----------|-----------|-------------------------|-----------|-------------|--|--|--|
| Male Rats   |           |           |                         |           |             |  |  |  |
| Hepatocellular Adenoma (overall rate) <sup><i>a d</i></sup>                       | 0/50      | 1/50 (2%) | 0/50                    | 1/50 (2%) | 6/50 (12%)  |  |  |  |
| Hepatocellular Adenoma (adjusted rate) <sup>b</sup>                               | 0%        | 4.5%      | 0%                      | 2.2%      | 12.9%       |  |  |  |
| Poly-3 test <sup>c</sup>  | p < 0.001 | p = 0.251 | (e) <sup><i>j</i></sup> | p = 0.514 | p = 0.022   |  |  |  |
| Hepatocellular Carcinoma (overall rate) <sup><i>e</i></sup>                       | 0/50 (0%) | 0/50 (0%) | 0/50 (0%)               | 0/50 (0%) | 6/50 (12%)  |  |  |  |
| Hepatocellular Carcinoma (adjusted rate)  | 0%        | 0%        | 0%                      | 0%        | 12.8%       |  |  |  |
| Poly-3 test   | p < 0.001 | (e)       | (e)                     | (e)       | p = 0.022   |  |  |  |
| Hepatocellular Adenoma or Carcinoma (combined) (overall rate) <sup>f</sup>        | 0/50 (0%) | 2/50 (4%) | 0/50 (0%)               | 1/50 (2%) | 12/50 (24%) |  |  |  |
| Hepatocellular Adenoma or Carcinoma<br>(combined) (adjusted rate)                 | 0%        | 4.5%      | 0%                      | 2.2%      | 25.6%       |  |  |  |
| Poly-3 test   | p < 0.001 | p = 0.251 | (e)                     | p = 0.514 | p < 0.001   |  |  |  |
| Female Rats   |           |           |                         |           |             |  |  |  |
| Hepatocellular Adenoma (overall rate) <sup>g</sup>                                | 0/50 (0%) | 0/50 (0%) | 1/50 (2%)               | 1/50 (2%) | 13/49 (27%) |  |  |  |
| Hepatocellular Adenoma (adjusted rate)  | 0%        | 0%        | 2.4%                    | 2.3%      | 31.3%       |  |  |  |
| Poly-3 test   | p < 0.001 | (e)       | p = 0.495               | p = 0.505 | p < 0.001   |  |  |  |
| Hepatocellular Carcinoma (overall rate) h   | 0/50 (0%) | 0/50 (0%) | 0/50 (0%)               | 0/50 (0%) | 2/49 (4%)   |  |  |  |
| Hepatocellular Carcinoma (adjusted rate)  | 0%        | 0%        | 0%                      | 0%        | 4.9%        |  |  |  |
| Poly-3 test   | p = 0.018 | (e)       | (e)                     | (e)       | p = 0.226   |  |  |  |
| Hepatocellular Adenoma or Carcinoma (combined) (overall rate) <sup><i>i</i></sup> | 0/50 (0%) | 0/50 (0%) | 1/50 (2%)               | 1/50 (2%) | 14/49 (29%) |  |  |  |
| Hepatocellular Adenoma or Carcinoma<br>(combined) (adjusted rate)                 | 0%        | 0%        | 2.4%                    | 2.3%      | 33.7%       |  |  |  |
| Poly-3 test   | p < 0.001 | (e)       | p = 0.495               | p = 0.505 | p < 0.001   |  |  |  |
| <sup>a</sup> Number of animals with neoplasm per number of animals necropsied.    |           |           |                         |           |             |  |  |  |

<sup>b</sup> Poly-3 estimated neoplasm incidence after adjustment for intercurrent mortality.

| Tissue: Tumor Type0 ppm300 ppm1000 ppm3000 ppm10,000 ppm <sup>c</sup> Beneath the control incidence is the p value associated with the trend test. Beneath the exposed group incidence are the<br>p values corresponding to pairwise comparisons between the control group and that exposed group. The Rao-Scott test<br>adjusts the Poly-3 test (which accounts for differential mortality in animals that do not reach study termination) for<br>within-litter correlation.0 ppm10,000 ppm   | May 2025   |   |   |                                       |                  |  |  |  |
|--|--|---|---|---------------------------------------|------------------|--|--|--|
| <sup>c</sup> Beneath the control incidence is the p value associated with the trend test. Beneath the exposed group incidence are the p values corresponding to pairwise comparisons between the control group and that exposed group. The Rao-Scott test adjusts the Poly-3 test (which accounts for differential mortality in animals that do not reach study termination) for within-litter correlation.  | Tissue: Tumor Type   | 0 ppm   | 300 ppm   | 1000 ppm                              | 3000 ppm         | 10,000 ppm   |  |  |
| <sup><i>d</i></sup> Historical control incidence for all routes of 2-year studies (mean $\pm$ standard deviation): 2/489 (0.44% $\pm$ 0.88%); range: 0–2%.<br><sup><i>e</i></sup> Historical control incidence: 2/489 (0.45% $\pm$ 0.89%); range: 0–2%.<br><sup><i>f</i></sup> Historical control incidence: 4/489 (0.89% $\pm$ 1.06%); range: 0–2%.<br><sup><i>g</i></sup> Historical control incidence: 15/489 (2.65% $\pm$ 2.59%); range: 0–8%.<br><sup><i>h</i></sup> Historical control incidence: 1/489 (0.22% $\pm$ 0.67%); range: 0–2%.<br><sup><i>i</i></sup> Historical control incidence: 16/489 (2.87% $\pm$ 2.8%); range: 0–8%.<br><sup><i>j</i></sup> (e) indicates that the value of the statistic could not be calculated.<br><sup><i>k</i></sup> Adapted from Table 25 in (NTP, 2021b). | <ul> <li><sup>c</sup> Beneath the control incidence is the p value p values corresponding to pairwise comparial adjusts the Poly-3 test (which accounts for existinal distribution).</li> <li><sup>d</sup> Historical control incidence for all routes of 0-2%.</li> <li><sup>e</sup> Historical control incidence: 2/489 (0.45%</li> <li><sup>g</sup> Historical control incidence: 4/489 (0.89%</li> <li><sup>g</sup> Historical control incidence: 15/489 (2.65%</li> <li><sup>h</sup> Historical control incidence: 1/489 (0.22%)</li> <li><sup>i</sup> Historical control incidence: 16/489 (2.87%)</li> <li><sup>j</sup> (e) indicates that the value of the statistic c</li> <li><sup>k</sup> Adapted from Table 25 in (NTP, 2021b).</li> </ul> | e associated w<br>sons between t<br>lifferential mo<br>of 2-year studie<br>$\pm 0.89\%$ ); ran<br>$\pm 1.06\%$ ); ran<br>$\% \pm 2.59\%$ ); ra<br>$\pm 0.67\%$ ); rar<br>$\% \pm 2.8\%$ ); ran<br>could not be ca | with the trend test<br>the control group<br>ortality in animals<br>es (mean $\pm$ stand<br>age: 0–2%.<br>age: 0–2%.<br>ange: 0–8%.<br>age: 0–2%.<br>age: 0–8%.<br>lculated. | and that expose<br>s that do not read | 2/489 (0.44% ± 0 | idence are the<br>ao-Scott test<br>tion) for<br>0.88%); range: |  |  |

4855

4856

4857 Pancreas. As can be seen from Table\_Apx B-13, treatment with DEHP resulted in a statistically significant increase in pancreatic acinar adenoma (males at 3,000 and 10,000 ppm), pancreatic acinar 4858 carcinoma (males at 10,000 ppm), and combined pancreatic acinar adenoma and carcinoma (males at 4859 4860 3,000 and 10,000 ppm). The increase in pancreatic tumors was accompanied by a statistically significant increase in focal hyperplasia of the acinus in males of 3,000 and 10,000 ppm groups. Pancreatic acinar 4861 adenomas were observed in 1/50 and 1/47 females at 3,000 and 10,000 ppm (not statistically 4862 4863 significant), respectively, while pancreatic acinar carcinoma was observed in one high dose female (not statistically significant). No pancreatic tumors were observed in control females. 4864

4865 4866

### Table\_Apx B-13. Incidence of Pancreatic Tumors in SD Rats Exposed to DEHP in the Diet for Two-years (NTP, 2021b)<sup>a</sup>

| Tumor Type  | 0 ppm     | 300 ppm    | 1000 ppm                | 3000 ppm    | 10,000 ppm  |  |  |  |
|---|-----------|------------|-------------------------|-------------|-------------|--|--|--|
| Male Rats   |           |            |                         |             |             |  |  |  |
| Acinus, Hyperplasia <sup>b</sup>                                      | 7/49      | 8/50       | 9/50                    | 24/50**     | 26/50**     |  |  |  |
| Acinar Adenoma (overall rate) <sup>be</sup>                           | 1/49 (2%) | 4/50 (8%)  | 5/50 (10%)              | 23/50 (46%) | 30/50 (60%) |  |  |  |
| Acinar Adenoma (adjusted rate) <sup>c</sup>                           | 2.4%      | 9%         | 10.7%                   | 49.9%       | 64%         |  |  |  |
| Poly-3 test <sup>d</sup>  | p < 0.001 | p = 0.202  | p = 0.131               | p < 0.001   | p < 0.001   |  |  |  |
| Acinar Carcinoma (overall rate) <sup>f</sup>                          | 49 (0%)   | 1/50 (2%)  | 0/50 (0%)               | 1/50 (2%)   | 5/50 (10%)  |  |  |  |
| Acinar Carcinoma (adjusted rate)                                      | 0%        | 2.3%       | 0%                      | 2.2%        | 10.6%       |  |  |  |
| Poly-3 test   | p < 0.001 | p = 0.513  | (e) <sup><i>i</i></sup> | p = 0.515   | p = 0.043   |  |  |  |
| Acinar Adenoma or Carcinoma<br>(combined) (overall rate) <sup>g</sup> | 1/49 (2%) | 5/50 (10%) | 5/50 (10%)              | 23/50 (46%) | 33/50 (66%) |  |  |  |
| Acinar Adenoma or Carcinoma<br>(combined) (adjusted rate)             | 2.4%      | 11.2%      | 10.7%                   | 49.9%       | 69.8%       |  |  |  |
| Poly-3 test   | p < 0.001 | p = 0.119  | p = 0.131               | p < 0.001   | p < 0.001   |  |  |  |
| Female Rats   |           |            |                         |             |             |  |  |  |
| Acinus, Hyperplasia   | 0/50      | 1/50       | 1/50                    | 1/50        | 5/47*       |  |  |  |
| Acinar Adenoma (overall rate) <sup><i>h</i></sup>                     | 0/50 (0%) | 0/50 (0%)  | 0/50 (0%)               | 1/50 (2%)   | 1/47 (2%)   |  |  |  |

| Tumor Type  | 0 ppm     | 300 ppm   | 1000 ppm  | 3000 ppm  | 10,000 ppm |
|---|-----------|-----------|-----------|-----------|------------|
| Acinar Carcinoma (overall rate) <sup>h</sup>                | 0/50 (0%) | 0/50 (0%) | 0/50 (0%) | 0/50 (0%) | 1/47 (2%)  |
| Acinar Adenoma or Carcinoma (combined (overall rate) $^{h}$ | 50 (0%)   | 0/50 (0%) | 0/50 (0%) | 1/50 (2%) | 2/47 (4%)  |

\*Statistically significant at  $p \leq 0.05$  by the Poly-3 test; \*\*p $\leq 0.01$ 

<sup>a</sup> Adapted from Table 26 in (NTP, 2021b).

<sup>b</sup> Number of animals with neoplasm or lesion per number of animals necropsied.

<sup>c</sup> Poly-3 estimated neoplasm incidence after adjustment for intercurrent mortality.

<sup>d</sup> Beneath the control incidence is the p value associated with the trend test. Beneath the exposed group incidence are the p values corresponding to pairwise comparisons between the control group and that exposed group. The Rao-Scott test adjusts the Poly-3 test (which accounts for differential mortality in animals that do not reach study termination) for within-litter correlation.

<sup>*e*</sup> Historical control incidence for all routes of 2-year studies (mean  $\pm$  standard deviation): 60/488 (11.58%  $\pm$  9.25%); range: 0–28%.

<sup>*f*</sup>Historical control incidence: 4/488 (0.8%  $\pm$  1.42%); range: 0–4%.

 $^{\rm g}$  Historical control incidence: 62/488 (12.03%  $\pm$  9.16%); range: 0–28%.

<sup>*h*</sup> Historical control incidence: 0/489.

 $^{i}$  (e) indicates that the value of the statistic could not be calculated.

#### 4869 4870

*Male Reproductive Tract.* Treatment-related non-neoplastic histopathologic lesions were noted in the
testis (*i.e.*, degeneration of germinal epithelium, edema, interstitial cell hyperplasia) and epididymis (*i.e.*,
hypospermia, exfoliated germ cells in the duct) (see Table 27 of (NTP, 2021b) for incidence of lesions).
A positive trend in increasing incidence of interstitial cell adenomas was observed in male rats,
however, pairwise comparisons to the control were not statistically significant (Table\_Apx B-14).

#### 4877

## 4878 Table\_Apx B-14. Incidence of Testicular Tumors in SD Rats Exposed to DEHP in the Diet for 4879 Two-years (<u>NTP, 2021b</u>)<sup>a</sup>

| Tissue: Tumor Type  | 0 ppm      | 300 ppm   | 1000 ppm  | 3000 ppm   | 10,000 ppm  |
|---|------------|-----------|-----------|------------|-------------|
| Interstitial cell, hyperplasia, focal (includes bilateral) <sup>b</sup> | 1/50       | 1/50      | 0/50      | 4/50       | 4/50        |
| Interstitial cell, adenoma (overall rate) <sup>be</sup>                 | 7/50 (14%) | 3/50 (6%) | 3/50 (6%) | 6/50 (12%) | 15/50 (30%) |
| Interstitial cell, adenoma (adjusted rate) <sup>c</sup>                 | 16.7%      | 6.8%      | 6.5%      | 13.4%      | 32.2%       |
| Poly-3 test <sup>d</sup>  | p < 0.001  | p = 0.135 | p = 0.119 | p = 0.451  | p = 0.073   |

<sup>*a*</sup> Adapted from Table 27 in (NTP, 2021b).

<sup>b</sup> Number of animals with neoplasm or lesion per number of animals necropsied.

<sup>c</sup> Poly-3 estimated neoplasm incidence after adjustment for intercurrent mortality.

<sup>d</sup> Beneath the control incidence is the p value associated with the trend test. Beneath the exposed group incidence are the p values corresponding to pairwise comparisons between the control group and that exposed group. The Rao-Scott test adjusts the Poly-3 test (which accounts for differential mortality in animals that do not reach study termination) for within-litter correlation.

<sup>*e*</sup> Historical control incidence for all routes of 2-year studies (mean  $\pm$  standard deviation): 19/487 (4.06%  $\pm$  4.36%); range: 0–14%.

#### 4880

4881

4882 *Uterus*. As can be seen from Table\_Apx B-15, treatment with DEHP causes a significant increase in 4883 incidence of uterine endometrial adenocarcinomas and combined uterine adenoma, adenocarcinoma,

4884 squamous cell carcinoma, and squamous cell papilloma in high-dose female rats. A significant positive

trend in incidence of uterine squamous cell papilloma was also observed, however, pairwise
comparisons to the control were not significant. Additionally, chronic uterine inflammation was
observed in the 300, 1,000, and 10,000 ppm groups compared to controls.

#### 4889 Under the conditions of the study, NTP concluded:

4890 "Under the conditions of the postweaning-only feed study (Study 2), there was clear evidence 4891 of carcinogenic activity of DEHP in male Hsd:Sprague Dawley® SD® rats based on the 4892 increased incidences of hepatocellular adenoma or carcinoma (combined) and acinar 4893 adenoma or carcinoma (combined) neoplasms (predominately adenomas) of the pancreas. 4894 The occurrence of testicular interstitial cell adenoma in male rats may have been related to 4895 exposure. There was clear evidence of carcinogenic activity of DEHP in female Hsd:Sprague 4896  $Dawlev \otimes SD \otimes rats$  based on the increased incidences of hepatocellular adenoma or 4897 carcinoma (combined) and uterine (including cervix) adenoma, adenocarcinoma, squamous 4898 cell carcinoma, or squamous cell papilloma (combined). The occurrence of pancreatic 4899 acinar adenoma or carcinoma (combined) in female rats was considered to be related to 4900 exposure."

4901 4902

### Table\_Apx B-15. Incidence of Uterine Tumors in SD Rats Chronically Exposed to DEHP in the Diet for Two-years (NTP, 2021b)<sup>a</sup>

| Tissue: Tumor Type   | 0 ppm     | 300 ppm   | 1,000 ppm  | 3,000 ppm  | 10,000 ppm  |
|--|-----------|-----------|------------|------------|-------------|
| Inflammation, Chronic <sup>b</sup>   | 2/50      | 9/50*     | 6/50*      | 8/50       | 8/49*       |
| Adenoma <sup>b</sup> e   | 0/50      | 1/50      | 0/50       | 0/50       | 0/49        |
| Adenocarcinoma (overall rate) <sup>b</sup>   | 2/50 (4%) | 2/50 (4%) | 1/50 (2%)  | 4/50 (8%)  | 10/50 (20%) |
| Adenocarcinoma (adjusted rate) <sup>cf</sup>   | 4.7%      | 4.9%      | 2.4%       | 9%         | 23.8%       |
| Poly-3 test <sup>d</sup>   | p < 0.001 | p = 0.678 | p = 0.508N | p = 0.352  | p = 0.011   |
| Squamous cell carcinoma (includes multiple) <sup>g</sup>   | 0/50      | 1/50      | 0/50       | 2/50       | 1/49        |
| Squamous cell papilloma (includes multiple) <sup><i>h</i></sup>  | 0/50      | 0/50      | 0/50       | 0/50       | 2/49        |
| Adenoma, adenocarcinoma, squamous<br>cell carcinoma, squamous cell papilloma<br>(combined) (overall rate) <sup>i</sup> | 2/50 (4%) | 4/50 (8%) | 1/50 (2%)  | 6/50 (12%) | 13/50 (26%) |
| Adenoma, adenocarcinoma, squamous<br>cell carcinoma, squamous cell papilloma<br>(combined) (adjusted rate)             | 4.7%      | 9.7%      | 2.4%       | 13.4%      | 30.7%       |
| Poly-3 test  | p < 0.001 | p = 0.315 | p = 0.508N | p = 0.145  | p < 0.001   |

\*Statistically significant at  $p \le 0.05$  by the Poly-3 test.

<sup>*a*</sup> Adapted from Table 28 in (NTP, 2021b).

<sup>b</sup> Number of animals with neoplasm or lesion per number of animals necropsied.

<sup>c</sup> Poly-3 estimated neoplasm incidence after adjustment for intercurrent mortality.

<sup>*d*</sup> Beneath the control incidence is the p value associated with the trend test. Beneath the exposed group incidence are the p values corresponding to pairwise comparisons between the control group and that exposed group. The Rao-Scott test adjusts the Poly-3 test (which accounts for differential mortality in animals that do not reach study termination) for within-litter correlation.

<sup>*e*</sup> Historical control incidence for all routes of 2-year studies (mean  $\pm$  standard deviation): 1/350 (0.29%  $\pm$  0.76%); range: 0–2%.

<sup>*f*</sup>Historical control incidence:  $20/350 (5.71\% \pm 3.35\%)$ ; range: 2-10%.

| May 2025  |   |                          |  |           |            |  |  |
|---|---|--------------------------|--|-----------|------------|--|--|
| Tissue: Tumor Type0 ppm300 ppm1   |   |                          |  | 3,000 ppm | 10,000 ppm |  |  |
| <sup>g</sup> Historical control incidence: 2/350 (0.57%<br><sup>h</sup> Historical control incidence: 0/350<br><sup>i</sup> Historical control incidence: 23/350 (6.57% | $0 \pm 1.51\%$ ); rang<br>$\% \pm 3.41\%$ ); rang | ge: 0–4%.<br>age: 2–10%. |  |           |            |  |  |

#### 4905

#### 4906 B.1.3 Hamsters – Inhalation and Intraperitoneal Studies

- 4907 **B.1.3.1** Inhalation Study (Schmezer et al., 1988) Male and female Syrian golden hamsters (80/sex for the control; 65/sex for treatment group) were 4908 4909 exposed continuously to vapor concentrations of 0 or  $15 \pm 5 \,\mu g/m^3$  DEHP from 12 weeks of age until 4910 natural death (around 23 months for males; 17 months for females). Continuous exposure was 4911 maintained 5 days per week. Twice per week exposure was stopped for animal care. Treatment with 4912 DEHP had no effect on median survival, which was 709, 703, 507, and 522 days for control males, 4913 treated males, control females, and treated females, respectively. No significant increase in any tumor 4914 types were observed.
- 4915

4916

#### B.1.3.2 Intraperitoneal Injection Study (Schmezer et al., 1988)

4917 Six week old male and female Syrian golden hamsters (25/sex/group) were administered 0 or 3 grams 4918 DEHP per kilogram body weight via intraperitoneal injection. Animals were split into five treatment 4919 groups, including 1) untreated control group; 2) 1 injection of DEHP per week; 3) 1 injection of DEHP 4920 every two weeks; 4) 1 injection of DEHP every four weeks; and 5) 1 injection of DEHP every four 4921 weeks in combination of 1 injection of 1.67 mg/kg N-nitrosodimethylamine (NDMA) per week. 4922 Treatment continued for life or until animals were found in a moribund state and sacrificed. Treatment 4923 with DEHP (groups 3, 4, and 5) alone had no effect on median survival times compared to untreated 4924 controls, although treatment with DEHP and NDMA in combination significantly reduced male and female median survival times. No significant difference in tumor incidence was observed between 4925 untreated controls and DEHP-treated animals. 4926

4927

#### 4928 B.1.4 Transgenic Mice – Oral Exposure Studies

## 4929B.1.4.1Twenty-six Week Dietary Study of Wild-type and Transgenic RasH2 Mice4930(Toyosawa et al., 2001)

4931 Groups of male and female transgenic RasH2 mice (15/sex/group) were fed diets containing 0, 1,500, 4932 3,000, or 6,000 ppm for 26-weeks, while groups of male and female wild-type mice (15/sex/dose) were fed diets of 0 and 6,000 ppm DEHP for 26-weeks. No dose-related effects on survival were observed for 4933 4934 either sex or strain. Food consumptions was comparable across treatment groups for both sexes and 4935 strains of mice. Body weight gain was decreased in high-dose rasH2 males starting around study week 4936 12, and was decreased after 19 and 21 weeks of treatment with 6,000 ppm DEHP for wild-type male and 4937 female mice, respectively. At study termination, body weight was reduced approximately 10 percent in 4938 these treatment groups. Neoplastic findings attributable to DEHP exposure were limited to the liver of 4939 high-dose rasH2 male mice, and included a statistically significant increase in incidence of 4940 hepatocellular adenomas (Table Apx B-16). No hepatocellular adenomas were observed in wild-type or 4941 female rasH2 mice, and no hepatocellular carcinomas were observed in any treatment group. 4942

| 4944 | Table_Apx B-16. Summary of Neoplastic Lesions of the Liver Observed in RasH2 and Wild-type |
|------|--|
| 4945 | Mice Fed Diets Containing DEHP for 26-weeks (Toyosawa et al., 2001) <sup>a</sup>           |

| Strain of mice  | Neoplastic Lesion      | 0 ppm | 1500<br>ppm  | 3000<br>ppm   | 6000<br>ppm    |  |  |
|---|------------------------|-------|--------------|---------------|----------------|--|--|
| Male - RasH2  | Hepatocellular Adenoma | 0/15  | 1/15<br>(7%) | 2/15<br>(13%) | 4/15*<br>(27%) |  |  |
| Female - RasH2  | Hepatocellular Adenoma | 0/15  | 0/15         | 0/15          | 0/15           |  |  |
| Male - Wild-type  | Hepatocellular Adenoma | 0/15  | NA           | NA            | 0/15           |  |  |
| Female - Wild-type  | Hepatocellular Adenoma | 0/15  | NA           | NA            | 0/15           |  |  |
| NA = Not applicable, dose not tested for this strain.<br>Asterisk (*) indicates statistically significant difference compared to control at $p < 0.05$ as calculated by original study authors. |                        |       |              |               |                |  |  |

4946

4947 4948

#### B.1.4.2 Twenty-six Week Dietary and 28-week Topical Studies of Tg.AC Mice (Eastin et al., 2001)

4949 The TG.AC transgenic mouse model carries the v-HA-ras oncogene fused to the promoter of the zeta-4950 globin gene. Male and female Tg.AC mice (15/sex/dose) were exposed to DEHP topically and via oral 4951 administration. In the topical exposure study, 0, 100, 200, and 400 mg/kg DEHP was applied to a clipped area (approximately 8 cm<sup>2</sup>) of dorsal skin of male and female Tg.AC mice. DEHP was dissolved 4952 4953 in acetone and volume doses of 3.3 mL/kg were applied to the shaved backs of mice 5 days per week for 4954 28 weeks. Treatment with DEHP did not affect survival of female mice (11/15 or 73% of mice survived 4955 to scheduled necropsy in all groups), however, survival of high-dose males was reduced (survival: 4956 13/15, 11/15, 13/15, 7/15 for males across dose groups). Treatment with DEHP did not significantly increase the incidence of tumors at the site of application for either sex at any dose. 4957

4958

In the oral exposure study, male and female Tg.AC mice (15/sex/dose) were fed diets containing 0, 1,500, 3,000, and 6,000 ppm DEHP for 26 weeks (equivalent to 252, 480, 1,000 mg/kg-day for males; 273, 545, 1,143 mg/kg-day for females). Treatment with DEHP had no significant effect on terminal body weight or survival in males or females across dose groups (males that survived until scheduled necropsy: 13/15, 11/15, 13/15, 9/15; females that survived until scheduled necropsy: 10/15, 13/15, 8/15, 12/15). Treatment with DEHP did not significantly increase the incidence of proliferative changes in either sex at any dose.

- 4966
- 4967 4968

## **B.1.4.3** Thirty-nine Week Dietary Study of *Xpa<sup>-/-</sup>* Mice, C57BL/6 Mice, and *Xpa<sup>-/-</sup>* /*P53<sup>+/-</sup>* Mice (Mortensen et al., 2002)

4969 Male and female  $Xpa^{-/-}$  mice (15/sex/dose) were fed diets containing 0, 1,500, 3,000, and 6,000 ppm 4970 DEHP (equivalent to 204, 408, 862 mg/kg-day for males; 200, 401, 827 for females) for 39 weeks. 4971 Similarly, male and female wild-type and  $Xpa^{-/-}/p53^{+/-}$  mice (15/sex/dose) were fed diets containing 0 4972 and 6,000 ppm DEHP for 39 weeks (equivalent to 879 and 872 mg/kg-day for male and female wild-4973 type mice, respectively; 896 and 796 mg/kg-day for male and female  $Xpa^{-/-}/p53^{+/-}$  mice, respectively). 4974 No significant increases in tumor responses were observed across various strains and treatment groups in 4975 response to exposure to DEHP.

May 2025

#### 4977 **B.1.4.4** 4978 (Ito et al., 2007a)

### Twenty-two Month Dietary Study of Wild-type and PPARa-null Sv/129 Mice Wild-type and *Ppara*-null male mice on a Sv/129 genetic background were fed diets containing 0, 0.01,

4979 4980 0.05 percent DEHP for 22 months. Mice were sacrificed by decapitation at approximately 23 months of 4981 age. Treatment with DEHP had no effect on survival, terminal body weight, or weight gain for either 4982 strain at any dose. In wild-type mice, hepatocellular adenomas were observed in two mice of each the 4983 0.01 and 0.05 percent DEHP groups (Table\_Apx B-17), however the effect was not statistically significant. In Ppara-null mice hepatocellular adenomas, carcinomas, and cholangiocellular carcinomas 4984 4985 were observed in the high-dose group (Table\_Apx B-17). A statistically significant trend in increased total liver tumors was observed for Ppara-null mice 4986

4987 4988

#### 4989 Table\_Apx B-17. Summary of Liver Tumors in Wild-type and Ppara-null Mice Fed Diets 4990 Containing DEHP for 22 Months (Ito et al., 2007a)<sup>a</sup>

|                             | Wild-type   |          |         | Ppara-null |        |            |  |
|-----------------------------|-------------|----------|---------|------------|--------|------------|--|
|                             | 0%          | 0.01%    | 0.05%   | 0%         | 0.01%  | 0.05%      |  |
| No. necropsied              | $24(1)^{b}$ | 23 (2)   | 20(1)   | 25 (1)     | 25 (3) | 31 (3)     |  |
| Hepatocellular adenoma      | 0           | 2        | 2       | 0          | 1      | 6          |  |
| Hepatocellular carcinoma    | 0           | 0        | 0       | 1          | 0      | 1          |  |
| Cholangiocellular carcinoma | 0           | 0        | 0       | 0          | 0      | 1          |  |
| Total liver tumors          | 0 (0%)      | 2 (8.7%) | 2 (10%) | 1 (4%)     | 1 (4%) | 8* (25.8%) |  |

<sup>*a*</sup> Adapted from Table 2 in (Ito et al., 2007a).

<sup>b</sup> Number in parentheses indicates the number of deaths prior to scheduled necropsy.

Asterisk (\*) indicates a significant trend between control and 0.05% DEHP group in *Ppara*-null mice (p<0.05) as calculated by original study authors.

#### 4991

#### **Butyl Benzyl Phthalate (BBP)** 4992 **B.2**

#### 4993 **B.2.1** Studies of Mice

4994

#### Two-year Dietary Study of B6C3F1 Mice (NTP, 1982b) **B.2.1.1**

4995 NTP (1982b) reports the results of a 2-year dietary study of male and female B6C3F1mice. Male and 4996 female mice (50/sex/dose) were administered diets containing 0, 6,000, and 12,000 ppm BBP (equivalent to approximately 900 and 1,800 mg/kg-day) for two-years. Survival across treatment groups 4997 4998 was comparable, with 88, 88, and 84 percent of control, low-, and high-dose males, respectively, and 70, 4999 70, and 72 percent of control, low-, and high-dose females, respectively, survival until scheduled necropsies at study weeks 105 to 106. No treatment-related or statistically significant increases in any 5000 5001 tumor type in any tissue were observed. Under the conditions of the study, NTP concluded that BBP was 5002 "not carcinogenic for B6C3F1 mice of either sex."

#### 5004 B.2.2 Studies of Rats

5005

#### B.2.2.1 Two-year Dietary Study of F344/N Rats (NTP, 1982b)

NTP (1982b) reports the results of a two-year dietary study of male and female F344/N rats. Male and 5006 female rats (50/sex/dose) were administered diets containing 0, 6,000, and 12,000 ppm BBP (equivalent 5007 5008 to approximately 300 and 600 mg/kg-day) for two-years. Male rats died prematurely, with internal 5009 hemorrhaging being suspected at gross necropsy (but was not confirmed microscopically). At week 28, 5010 only 30 percent of high-dose males were still alive, and all male rats were sacrificed at study weeks 29 5011 to 30, when 98, 80 and 30 percent of control, low-, and high-dose males were alive, respectively. 5012 Increased mortality was not encountered in female rats, with 62, 58, and 64 percent of control, low-, and 5013 high-dose females, respectively, surviving until scheduled necropsy at 105 to 106 weeks. The only 5014 tumor type statistically significantly increased was MNCL in high-dose females (Table Apx B-18), 5015 which was observed in 18/50 (36%) high-dose females, compared to 7/49 (14%) of controls. Incidence of MNCL in high-dose females was outside the range of historical control data for female F344/N rats 5016 5017 with "all leukemias" from the laboratory conducting the study (observed in 77/399 (19%); range 12-5018 24%). No significant increase in urinary bladder transitional cell papillomas or carcinomas, or pancreatic 5019 adenomas or carcinomas were observed at any dose. Under the conditions of the study, NTP concluded 5020 that BBP was "probably carcinogenic for female F344/N rats, causing an increased incidence of 5021 mononuclear cell leukemias." Due to the high mortality observed in male rats, carcinogenicity of BBP 5022 could not be assessed.

5023 5024

### 5025Table\_Apx B-18. Incidence of MNCL in Female F344 Rats Fed Diets Containing BBP for Two-5026years (NTP, 1982b)<sup>a</sup>

| Tissue: Tumor Type   | Control | 6000 ppm<br>(300 mg/kg-<br>day) | 12,000 ppm<br>(600 mg/kg-<br>day) |  |  |  |
|--|---------|---------------------------------|-----------------------------------|--|--|--|
| MNCL   | 7/49    | 7/49                            | 18/50*                            |  |  |  |
| <sup><i>a</i></sup> Asterisk indicates statistically significant pairwise comparison to the control by Fisher exact test ( $P < 0.05$ ) when the Cochran-Armitage test was statistically significant ( $P<0.05$ ). Data from Table A2 of (NTP, 1982b). |         |                                 |                                   |  |  |  |

#### 5027

5028

#### B.2.2.2 Two-year Dietary Study of F344/N Rats (NTP, 1997b)

Male F344/N rats (60/dose) were fed diets containing 0, 3,000, 6,000, and 12,000 ppm BBP and female 5029 5030 F344/N rats (60/dose) were fed 0, 6,000, 12,000, and 24,000 ppm BBP for two years (equivalent to 120, 5031 240, 500 mg/kg-day for males; 300, 600, 1.200 mg/kg-day for females) (NTP, 1997b). Survival rates 5032 were comparable across treatment groups for male (survival to study termination: 28/50, 20/50, 22/50, 5033 22/50) and female rats (survival: 25/50, 29/50, 29/50, 29/50). No treatment-related clinical observations 5034 were reported for either sex in any dose group. Effects on food consumption were limited to females in the 24,000 ppm BBP treatment group. Food consumption was reduced in high-dose females at the start 5035 5036 of the study, but was similar to that of controls by study week 6. Body weights were reduced in high-5037 dose male (4–10% less than controls throughout most of the study; terminal body weight on study week 5038 101 was reduced 6%) and female rats (7-27%) less than controls throughout most of the study; terminal 5039 body weight on study week 101 was reduced 27%).

5040

5041 In males, a statistically significant increase in focal hyperplasia of the pancreatic acinar cell was 5042 observed in high-dose males compared to concurrent study control group (Table\_Apx B-19). This

5043 preneoplastic lesion was accompanied by a statistically significant increase in pancreatic acinar cell 5044 adenomas and pancreatic acinar cell adenomas and carcinoma (combined) in high-dose males 5045 (Table Apx B-19). Incidence of acinar cell adenomas and adenomas and carcinoma (combined) were 5046 outside the range of historical controls from NTP two-year feed studies (see footnotes b, c, and d in 5047 Table Apx B-19). In female rats, no treatment-related increases in focal hyperplasia of the pancreatic 5048 acinar cell were observed. Pancreatic acinar cell adenomas were observed in two high-dose females; 5049 however, the effect was not statistically significant, and fell within the range of historical controls from 5050 NTP two-year feed studies (see footnote e in Table\_Apx B-19). Because pancreatic neoplasms are rare in control animals and because a pancreatic tumor response was observed in males, NTP considered the 5051 low incidence of pancreatic acinar adenomas in female rats to be potentially treatment-related. 5052 5053 5054 In high-dose female rats, mild to moderate transitional epithelium hyperplasia was observed in the 5055 urinary bladder (10/50 vs. 4/50 in controls) (Table Apx B-19). Transitional epithelium papillomas were

urinary bladder (10/50 vs. 4/50 in controls) (Table\_Apx B-19). Transitional epithelium papillomas were
observed in two high-dose females. Although the incidence of papillomas in the urinary bladder was not
statistically significant, the incidence of this neoplasm exceeded the range of NTP historical control data
from two-year feed studies (see footnote f in Table\_Apx B-19). No transitional epithelium papillomas
were observed in male rats.

5061 MNCL was not significantly increased by exposure to BBP in male or female rats (Table\_Apx B-19)

5063 Overall, NTP concluded "Under the conditions of this 2-year feed study, there was some evidence of 5064 carcinogenic activity" of butyl benzyl phthalate in male F344/N rats based on the increased incidences 5065 of pancreatic acinar cell adenoma and of acinar cell adenoma or carcinoma (combined). There was 5066 equivocal evidence of carcinogenic activity of butyl benzyl phthalate in female 344/N rats based on the 5067 marginally increased incidences of pancreatic acinar cell adenoma and of transitional epithelial 5068 papilloma of the urinary bladder."

5069

5062

5070

### Table\_Apx B-19. Summary of Neoplastic Findings in the Pancreas and Urinary Bladder in F344/N Rats Fed Diets Containing BBP for Two-years (NTP, 1997b)<sup>a</sup>

|  | 0 ppm          | 3000 ppm       | 6000 ppm       | 12,000 ppm     | 24,000 ppm |  |  |  |
|--|----------------|----------------|----------------|----------------|------------|--|--|--|
| Male Rats  |                |                |                |                |            |  |  |  |
| Number Examined microscopically                          | 50             | 49             | 50             | 50             | NA         |  |  |  |
| Pancreas, Acinus, Focal Hyperplasia                      | 4/50           | 7/49           | 9/50           | 12/50*         | NA         |  |  |  |
| Pancreas, Acinus, Adenoma <sup>b</sup>                   | 3/50 (6%)      | 2/49 (4%)      | 3/50 (6%)      | 10/50* (20%)   | NA         |  |  |  |
| Pancreas, Acinus, Carcinoma <sup>c</sup>                 | 0/50           | 0/49           | 0/50           | 1/50 (2%)      | NA         |  |  |  |
| Pancreas, Acinus, Adenoma or Carcinoma <sup>d</sup>      | 3/50 (6%)      | 2/49 (4%)      | 3/50 (6%)      | 11/50* (22%)   | NA         |  |  |  |
| Urinary Bladder, Hyperplasia, Transitional<br>Epithelium | 0/50           | 0/49           | 0/50           | 2/50           | NA         |  |  |  |
| Urinary Bladder, Papilloma, Transitional<br>Epithelium   | 0/50           | 0/49           | 0/50           | 0/50           | NA         |  |  |  |
| MNCL   | 31/50<br>(62%) | 28/50<br>(56%) | 34/50<br>(68%) | 30/50<br>(60%) | NA         |  |  |  |
|  | Female         | Rats           |                |                |            |  |  |  |
| Number Examined microscopically                          | 50             | NA             | 50             | 50             | 50         |  |  |  |

|  | 0 ppm          | 3000 ppm | 6000 ppm       | 12,000 ppm     | 24,000 ppm     |
|--|----------------|----------|----------------|----------------|----------------|
| Pancreas, Acinus, Focal Hyperplasia                              | 1/ 50          | NA       | 4/50           | 2/50           | 0/50           |
| Pancreas, Acinus, Adenoma <sup>e</sup>                           | 0/50           | NA       | 0/50           | 0/50           | 2/50<br>(4%)   |
| Urinary Bladder, Hyperplasia, Transitional<br>Epithelium         | 4/50           | NA       | 0/50           | 1/50           | 10/50*         |
| Urinary Bladder, Papilloma, Transitional Epithelium <sup>f</sup> | 1/50           | NA       | 0/50           | 0/50           | 2/50           |
| MNCL   | 21/50<br>(42%) | NA       | 20/50<br>(40%) | 21/50<br>(42%) | 19/50<br>(38%) |

NA = Not Applicable (dose not tested for this sex)

Asterisk (\*) indicates significant difference ( $P \le 0.05$ ) from the control by the logistic regression test, as calculated by NTP.

<sup>a</sup> Incidence data from Tables 9 and 10 in (NTP, 1997b).

<sup>*b*</sup> Historical incidence for 2-year NTP feed studies with untreated controls (acinus, adenoma, males):  $19/1,191 (1.6\% \pm 2.4\%)$ ; range 0-10%.

<sup>c</sup> Historical incidence (acinus, carcinoma, males): 0/1,919 (0.0%)

<sup>*d*</sup> Historical incidence (acinus, adenoma or carcinoma, males): 19/1,191 (1.6%  $\pm 2.4\%$ ); range 0-10%.

 $^e$  Historical incidence (acinus, adenoma, females): 2/1,194 (0.2%  $\pm$  0.8%); range 0–4%

<sup>*f*</sup>Historical incidence (transitional epithelium papilloma): 4/1,182 (0.3%  $\pm$  0.8%); range 0–2%

5073

#### 5074 **B.**2 5075

#### B.2.2.3 Two-year Dietary Study of F344/N Rats – Study 1 (Ad Libitum and Weight-Matched Controls Protocol) (NTP, 1997a)

5076 NTP (1997a) reports the results of three studies of BBP, including several diet restriction studies. In the 5077 first study (Ad Libitum and Weight-Matched Controls Protocol), male F344/N rats (60/dose) were fed diets containing 0 or 12,000 ppm BBP, while female F344/N rats (60/dose) fed 0 or 24,000 ppm BBP in 5078 5079 feed that was available *ad libitum* for 104 weeks (equivalent to approximately 500 mg/kg-day for males 5080 and 1,200 mg/kg-day for females). Two control groups were included, including a group in which food was available ad libitum and a group in which control diet was restricted such that mean body weight 5081 5082 matched the BBP treatment group. Survival rates were similar between male and female rats dosed with 5083 BBP and the ad libitum controls, but were less than those of the weight-matched controls (survival (ad *libitum* control, weight-matched, BBP): 28/60, 33/60, 22/60 for males; 25/60, 41/60, 29/60 for females). 5084 5085 Feed consumption for BBP treated females was less than that of the *ad libitum* controls from study week 38 through the end of the study. Feed consumption for BBP treated males was comparable to that of the 5086 5087 ad libitum controls. No treatment-related clinical findings were reported for either sex. Mean body weights for BBP treated males were reduced approximately 8 percent compared to ad libitum controls 5088 5089 throughout the study. Mean body weights for BBP treated females were 80 percent that of ad libitum 5090 controls after one year and fell to 73 percent that of *ad libitum* controls by study termination.

5091

5092 Incidence of hyperplasia of the pancreatic acinus was increased in males treated with BBP compared to 5093 ad libitum and weight-matched controls (Table\_Apx B-20). Further, incidence of pancreatic acinar cell 5094 adenomas and pancreatic acinar cell adenomas and carcinomas (combined) were increased in male rats 5095 treated with BBP compared to both control groups. NTP further reported that the incidence of adenomas 5096 in BBP treated males exceeded the overall NTP historical control incidence of this tumor type in 5097 untreated male F344/N rats fed ad libitum. In female rats treated with BBP, there was no increase in 5098 hyperplasia of the pancreatic acinus, while pancreatic acinar cell adenomas were observed in 2 out of 50 female rats treated with BBP (not statistically significant) (Table\_Apx B-20). 5099

- 5101 BBP-dosed females had higher incidence of hyperplasia of the urinary bladder transitional epithelium
- 5102 (10/50) compared to ad libitum (4/50) and weight-matched (0/50) control female rats (Table\_Apx B-20).
- 5103 However, papilloma of the transitional epithelium was not significantly increased in BBP treated
- 5104 females (2/50) compared to ad libitum (1/50) or weight-matched (0/50) controls (Table\_Apx B-20). 5105
- 5106 Incidence of MNCL was comparable between ad libitum fed controls and BBP treated F344/N rats of
  5107 both sexes (Table\_Apx B-20), while weight-matched controls of both sexes had lower incidence of
  5108 MNCL (Table\_Apx B-20). Incidence of MNCL in BBP treated rats of both sexes was reported by NTP
  5109 to be within the historical control ranges for leukemia (all types) in untreated F344/N rats.
- 5110 5111

5100

# Table\_Apx B-20. Incidence of Neoplasms and Non-neoplastic Lesions of the Pancreas, Urinary Bladder, and MNCL in F344/N Rats (Ad Libitum and Weight-Matched Controls Protocols) (NTP, 1997a)<sup>a</sup>

|   | Lesion/ Tumor Type                   | Ad<br>Libitum-<br>Fed Control | Weight-<br>Matched<br>Control | 12,000 ppm<br>(males) or 24,000<br>ppm (females) |  |  |  |
|---|--------------------------------------|-------------------------------|-------------------------------|--|--|--|--|
| Male Rats   |                                      |                               |                               |  |  |  |  |
| Number Examined   |                                      | 50                            | 50                            | 50   |  |  |  |
| Pancreas  | Acinus, Focal Hyperplasia            | 4/50                          | 2/50                          | 12/50  |  |  |  |
|   | Acinus, Adenoma                      | 3/50 (6%)                     | 0/50                          | 10/50* (20%)                                     |  |  |  |
|   | Acinus, Carcinoma                    | 0/50                          | 1/50 (2%)                     | 1/50 (2%)  |  |  |  |
|   | Adenoma or Carcinoma                 | 3/50 (6%)                     | 1/50 (2%)                     | 11/50* (22%)                                     |  |  |  |
| Urinary Bladder   | Hyperplasia, Transitional Epithelium | 0/50                          | 0/50                          | 2/50   |  |  |  |
|   | Papilloma, Transitional Epithelium   | 0/50                          | 0/50                          | 0/50   |  |  |  |
| MNCL  | MNCL <sup>b</sup>                    | 31/50 (62%)                   | 15/50 (30%)                   | 30/50* (60%)                                     |  |  |  |
|   | Female Rats                          |                               |                               |  |  |  |  |
| Number Examined   |                                      | 50                            | 49                            | 50   |  |  |  |
| Pancreas  | Acinus, Focal Hyperplasia            | 1/50 (2%)                     | 0/49                          | 0/50   |  |  |  |
|   | Acinus, Adenoma                      | 0/50                          | 0/49                          | 2/50 (4%)  |  |  |  |
| Urinary Bladder   | Hyperplasia, Transitional Epithelium | 4/50 (8%)                     | 0/50                          | 10/50 (20%)                                      |  |  |  |
|   | Papilloma, Transitional Epithelium   | 1/50 (2%)                     | 0/50                          | 2/50 (4%)  |  |  |  |
| MNCL  | MNCL <sup>b</sup>                    | 21/50 (42%)                   | 13/50 (26%)                   | 19/50* (38%)                                     |  |  |  |
| Asterisk (*) indicates significant difference (P $\leq$ 0.05) from the control by the logistic regression test, as calculated by NTP.<br><sup><i>a</i></sup> Incidence data from Tables 3, 4, B1a, and B3a of (NTP, 1997a).<br><sup><i>b</i></sup> Incidence of MNCL significantly increased compared to weight-matched, but not ad libitum fed controls. |                                      |                               |                               |  |  |  |  |

5115

## 5116B.2.2.4Two-year Dietary Study of F344/N Rats – Study 2 (2-year Restricted Feed5117Protocol) (NTP, 1997a)

5118 Male F344/N rats (60/dose) were fed diets containing 0 or 12,000 ppm BBP, while female F344/N rats 5119 (60/dose) fed diets containing 0 or 24,000 ppm BBP for 104 weeks. Control animals were diet-restricted

to limit the mean body weight of controls to approximately 85 percent of the *ad libitum* control rats in
Study 1. Survival rates were similar between BBP treated males and controls (survival to 104 weeks:
34/50 vs. 31/50) and BBP treated females and controls (survival: 35/50 vs. 39/50). No clinical findings
related to BBP treatment were observed. Mean body weights of BBP-treated males remained withing 10
percent of controls throughout the duration of the study. Mean body weights of BBP-treated females
were 23 percent less than that of controls at study termination.

5126

Evidence of carcinogenicity was limited to the urinary bladder in female rats (Table\_Apx B-21). BBPdosed females had higher incidence of hyperplasia of the urinary bladder transitional epithelium (14/50)
compared to diet-restricted control female rats (0/50). Additionally, papilloma of the transitional
epithelium was observed in two female rats treated with BBP (2/50), however, the increase was not
statistically significant compared to the concurrent control. No carcinomas of the transitional epithelium
in the urinary bladder were observed.

5133

5134 No statistically significant increase in MNCL was observed in male or female rats compared to the 5135 concurrent control (incidence: 21/50 [42%] vs. 27/50 [54%] in control and BBP-treated males, 5136 respectively; 16/50 [32%] vs. 18/50 [36%] in control and BBP-treated females, respectively).

- 5137
- 5138 5139

## B.2.2.5 Two-year Dietary Study of F344/N Rats – Study 3 (Lifetime Restricted Feed Protocol) (NTP, 1997a)

Male F344/N rats (60/dose) were fed diets containing 0 or 12,000 ppm BBP, while female F344/N rats 5140 5141 (60/dose) fed diets containing 0 or 24,000 ppm BBP until survival fell to 20 percent. Control animals 5142 were diet-restricted to limit the mean body weight of controls to approximately 85 percent of the ad 5143 libitum control rats in Study 1. Survival was reduced to 20 percent during week 129 (approximately 30 5144 months) for males and week 140 for females (approximately 32 months). No clinical findings related to 5145 BBP treatment were observed. Mean body weights of BBP-treated males remained withing 10 percent 5146 of controls throughout the duration of the study. Mean body weights of BBP-treated females were 29 5147 percent less than that of controls at study termination. 5148

Evidence of carcinogenicity was limited to the urinary bladder in female rats (Table\_Apx B-21). BBPdosed females had higher incidence of hyperplasia of the urinary bladder transitional epithelium (16/50)
compared to diet-restricted control female rats (0/50). Papilloma and carcinoma of the transitional
epithelium was observed in two and four female rats treated with BBP, respectively, while one control
female rat had a papilloma at 32 months. Although a marginal increase in papillomas and carcinomas
(combined) were observed in BBP-treated female rats (6/50) compared to control female rats (1/50), the
increase was not statistically significant.

5156

5157 No statistically significant increase in MNCL was observed in male or female rats compared to the 5158 concurrent control (incidence: 39/50 [78%] vs. 36/50 [72%] in control and BBP treated males, 5159 respectively; 29/50 [58%] vs. 39/50 [78%] in control and BBP treated females, respectively).

May 2025

## Table\_Apx B-21. Incidence of Non-neoplastic and Neoplastic Findings in F344/N Rats Treated with BBP (2-year Restricted Feed and Lifetime Restricted Feed Protocols) (NTP, 1997a)<sup>a</sup>

|  |  |                | 2-Year Restricted Feed<br>Protocol                  |                | Restricted Feed<br>Protocol                         |  |  |  |
|--|--|----------------|---|----------------|---|--|--|--|
| Lesion/ Tumor Type   |  | 0 ppm          | 12,000 ppm<br>(males) or<br>24,000 ppm<br>(females) | 0 ppm          | 12,000 ppm<br>(males) or<br>24,000 ppm<br>(females) |  |  |  |
|  | Ν  | Iale Rats      |   |                |   |  |  |  |
| Number Examined  |  | 50             | 50  | 50             | 50  |  |  |  |
| Urinary Bladder  | Hyperplasia  | 1/50           | 2/50  | 0/50           | 1/50  |  |  |  |
|  | Papilloma  | 0/50           | 1/50 (2%)   | 0/50           | 1/50 (2%)   |  |  |  |
|  | Carcinomas   | 0/50           | 0/50  | 0/50           | 1/50 (2%)   |  |  |  |
| Pancreas   | Acinus, Focal Hyperplasia  | 0/50           | 3/50  | 0/50           | 2/50  |  |  |  |
|  | Acinus, Adenoma  | 0/50           | 0/50  | 0/50           | 1/50 (2%)   |  |  |  |
| MNCL   | MNCL   | 21/50<br>(42%) | 27/50<br>(54%)                                      | 39/50<br>(78%) | 36/50<br>(72%)                                      |  |  |  |
|  | Fe   | male Rats      | -   |                |   |  |  |  |
| Number Examined  |  | 50             | 50  | 49             | 50  |  |  |  |
| Urinary Bladder  | Hyperplasia  | 0/50           | 14/50*  | 0/49           | 16/50*  |  |  |  |
|  | Papilloma  | 0/50           | 2/50 (4%)   | 1/49 (2%)      | 2/50 (4%)   |  |  |  |
|  | Carcinomas   | 0/50           | 0/50  | 0/49           | 4/50 (8%)   |  |  |  |
|  | Papilloma or Carcinoma<br>(combined)   | 0/50           | 2/50 (4%)   | 1/49 (2%)      | 6/50 (12%)  |  |  |  |
| Pancreas   | Acinus, Focal Hyperplasia  | 0/50           | 3/50  | 0/50           | 1/50  |  |  |  |
|  | Acinus, Adenoma  | 0/50           | 0/50  | 0/50           | 1/50 (2%)   |  |  |  |
| MNCL   | MNCL   | 16/50<br>(32%) | 18/50<br>(36%)                                      | 29/50<br>(58%) | 39/50<br>(78%)                                      |  |  |  |
| Asterisk (*) indicates sign<br><sup><i>a</i></sup> Incidence date from Tab | Asterisk (*) indicates significant difference (P $\leq$ 0.05) from the control by the logistic regression test, as calculated by NTP.<br><sup><i>a</i></sup> Incidence date from Table 7, A1b, A3b, B1b, and B3b of (NTP,1997a). |                |   |                |   |  |  |  |

## 5165 **Appendix C** 5166

5167

### C SCIENTIFIC UNCERTAINTIES RELATED TO MONONUCLEAR CELL LEUKEMIA (MNCL) AND LEYDIG CELL TUMORS IN F344 RATS

MNCL is a spontaneously occurring neoplasm of the hematopoietic system that reduces lifespan and is 5168 5169 one of the most common tumor types occurring at a high background rate in the F344 strain of rat 5170 (Thomas et al., 2007). Historical control data from NTP have demonstrated an increase in the 5171 spontaneous background incidence of MNCL in untreated male and female F344 rats from 7.9 and 2.1 5172 percent in males and females, respectively, in 1971 to 52.5 and 24.2 percent in males and females, 5173 respectively, from 1995 through 1998 (Thomas et al., 2007). Spontaneous incidence of MNCL in other 5174 strains of rat appear to be rare. Brix et al. (2005) report the incidence of MNCL in female Harlan SD rats 5175 to be 0.5 percent in NTP 2-year studies. Further, MNCL does not appear to occur naturally in mice 5176 (Thomas et al., 2007). Similarly, as discussed by King-Herbert et al. (2006), there is also a high 5177 background rate of spontaneous testicular Leydig cell tumors (also known as interstitial cell tumors) in 5178 control F344 and F344/N rats (ranging from 86–87%). Comparatively, the background rate of Levdig 5179 cell tumors is much lower in Wistar and SD strains of rats, ranging from 0.3 to 3.4 percent (King-Herbert and Thaver, 2006). The F344/N strain of rat was used in NTP two-year chronic and 5180 carcinogenicity bioassays for nearly 30 years (King-Herbert et al., 2010; King-Herbert and Thayer, 5181 5182 2006). However, in the early 2000s NTP stopped using the F344/N strain of rat in part because of high 5183 background incidence of MNCL and testicular Leydig cell tumors, which decrease the ability of the 5184 F344 strain to detect exposure-related increases in MNCL and testicular Leydig cell tumors (King-5185 Herbert and Thayer, 2006). 5186

5187 Another source of uncertainty is lack of MOA information for induction of MNCL in F344 rats. The 5188 MOA for induction of MNCL in F344 rats is unknown. Lack of MOA information makes it difficult to 5189 determine human relevancy. There is additional uncertainty related to the human correlate to MNCL in 5190 F344 rats. Some researchers have suggested that based on the biological and functional features in the 5191 F344 rat, MNCL is analogous to LGL in humans (Caldwell et al., 1999; Caldwell, 1999; Reynolds and 5192 Foon, 1984). There are two major human LGL leukemias, including CD3+ LGL leukemia and CD3-5193 LGL leukemia with natural killer cell activity (reviewed in (Maronpot et al., 2016; Thomas et al., 5194 2007)). Thomas et al. (2007) contend that MNCL in F344 rats shares some characteristics in common 5195 with ANKCL in humans, and that ANKCL may be a human correlate. However, Maronpot et al. (2016) 5196 point out that ANKCL is extremely rare with less than 98 cases reported worldwide, and its etiology is 5197 related to infection with Epstein-Barr virus, not chemical exposure. This is in contrast to MNCL in F344 5198 rats, which is a more common form of leukemia and is not associated with a viral etiology. However, 5199 under EPA's Guidelines for Carcinogen Risk Assessment (U.S. EPA, 2005), site concordance is not 5200 always assumed between animals and humans.

5201 5202 Given the limitations and uncertainties regarding MNCL in F344 rats discussed above, during the July 5203 2024 peer review meeting of the DIDP and DINP human health hazard assessments, the SACC recommended that "the observation of an increased incidence of MNCL in a chronic bioassay 5204 5205 employing the Fisher 344 rat should not be considered a factor in the determination of the cancer 5206 classification..." and "Most Committee members agreed that given the material presented in a 5207 retrospective review, MNCL and Levdig Cell Tumors, among other tumor responses in F344 rat carcinogenicity studies lack relevance in predicting human carcinogenicity (Maronpot et al., 2016)" 5208 5209 (U.S. EPA, 2024q). Consistent with the recommendations of the SACC, EPA is not further considering 5210 MNCL as a factor in the determination of the cancer classifications for phthalates.

May 2025

# 5211Appendix DSUMMARY OF STUDIES OF DEHP EVALUATING5212PPARα ACTIVATION

5213 EPA reviewed the health effects section of ATSDR (2022), including Table 2-2, for studies that report

5214 evaluation of biomarkers of PPAR $\alpha$  activation (KE 1 in PPAR $\alpha$  MOA). Identified studies were

independently reviewed by EPA to determine effect levels (*i.e.*, NOAEL and LOAEL values) for
PPARα activation in each study.

5218 Overall, EPA identified 27 studies that evaluated various biomarkers of PPARα activation in the liver,
5219 including 18 studies of rats, 3 studies of mice, 3 studies of monkeys, 2 studies of hamsters, and 1 study
5220 of guinea pigs (Table\_Apx D-1). As can be seen from Table\_Apx D-1, the lowest identified NOAELs
5221 were 7.5 mg/kg-day for mice (Isenberg et al., 2000) and for 11 mg/kg-day for rats (Barber et al., 1987;
5222 BIBRA, 1985).

5223 5224

5217

### 5225 Table\_Apx D-1. Summary of NOAEL and LOAEL Values for PPARα Activation from *In Vivo* 5226 Animal Toxicology Studies of DEHP<sup>α</sup>

| Brief Study Details<br>(Reference)   | NOAEL/<br>LOAEL (mg/kg-<br>day)                | PPARα Biomarker at<br>LOAEL  | Comments   |
|--|--|--|--|
| Male B6C3F1 mice (5/dose)<br>exposed to 0, 500, or 6,000<br>ppm DEHP via diet for 2- or 4-<br>weeks (equivalent to 0, 7.5,<br>900 mg/kg-day) (Isenberg et<br>al., 2000) <sup><i>b</i></sup>  | 7.5 / 900                                      | ↑ PPAR-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>PBOX at 2- and 6-<br>weeks)                         | <ul> <li>↓ hepatic GJIC at 2-<br/>weeks at 900<br/>mg/kg-day (GJIC<br/>evaluated via <i>in situ</i><br/>dye transfer assay)</li> </ul>   |
| Male and female F344 rats<br>(5/sex/dose) exposed to 0, 11,<br>105, 667, 1,224, or 2,101<br>mg/kg-day DEHP [males] or 0,<br>12, 109, 643, 1,197, or 1892<br>mg/kg-day [females] for 21<br>days via feed ( <u>Barber et al.</u> ,<br><u>1987; BIBRA, 1985</u> ) | 11 / 105                                       | ↑ peroxisome<br>proliferation (electron<br>microscopy<br>quantification of<br>periportal peroxisome<br>sore) | <ul> <li>Coincided with ↓<br/>serum lipids and ↑<br/>liver weight at ≥105<br/>mg/kg-day</li> <li>38-44% ↓ body<br/>weight and 48-60%<br/>↓ food consumption<br/>at ≥1,892 mg/kg-<br/>day</li> </ul>                      |
| Male and female B6C3F1 mice<br>(60–70/sex/dose) exposed to 0,<br>19.2, 98.5, 292.2, 1,266 mg/kg-<br>day [males] or 0, 23.8, 116.8,<br>354.2, 1,458 mg/kg-day DEHP<br>[females] for 104 weeks via<br>feed (David et al., 2000a;<br>David et al., 1999)          | 19.2/ 98.5 (males)<br>23.8/ 116.8<br>(females) | ↑ PPAR-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>palmitoyl CoA oxidation<br>activity)                | <ul> <li>Coincided with ↑<br/>liver weight and<br/>cytoplasmic<br/>eosinophilia at<br/>1,266 mg/kg-day<br/>and hepatocellular<br/>neoplasms (≥98.5<br/>mg/kg-day [males];<br/>≥354.2 mg/kg-day<br/>[females])</li> </ul> |

|  | 11 <b>u</b> j 20                       | 20  |   |
|--|--|---|---|
| Brief Study Details<br>(Reference)   | NOAEL/<br>LOAEL (mg/kg-<br>day)        | PPARα Biomarker at<br>LOAEL   | Comments  |
|  |  |   | <ul> <li>hepatocellular<br/>neoplasia was the<br/>most common<br/>cause of death<br/>(≥500 mg/kg-day)</li> </ul>  |
| Male SD rats (5/group)<br>exposed to 0, 25, 100, 250, or<br>1,000 mg/kg-day DEHP for 2<br>weeks via gavage ( <u>Lake et al.</u> ,<br><u>1984</u> )   | 25 / 100                               | ↑ PPAR-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>hepatic palmitoyl-CoA<br>oxidation and carnitine<br>acetyltransferase)   | <ul> <li>Coincided with ↑<br/>relative liver weight<br/>at ≥100 mg/kg-day<br/>and ↑ liver<br/>peroxisomes<br/>(qualitative<br/>histopathological<br/>assessment via 3,3'-<br/>diaminobenzidine<br/>staining) at ≥250<br/>mg/kg-day.</li> </ul>  |
| Male and female F344 rats<br>(50–80/sex/dose) exposed to 0,<br>5.8, 29, 147, 789 mg/kg-day<br>[males] or 0, 7.3, 36, 182, 939<br>mg/kg-day DEHP [females] for<br>104 weeks via feed (David et<br>al., 2000b; David et al., 1999) | 29 / 147 (males)<br>36 / 182 (females) | ↑ PPAR-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>palmitoyl CoA<br>oxidation)  | <ul> <li>Coincided with ↑<br/>absolute liver<br/>weight and<br/>hepatocellular<br/>tumors (≥147<br/>mg/kg-day [males];<br/>939 mg/kg-day<br/>[females])</li> <li>12% reduction in<br/>survival due to<br/>MNCL</li> <li>15% ↓ body weight<br/>gain; no changes in<br/>food consumption</li> </ul> |
| Male and female Wistar albino<br>strain rats (4–6/sex/dose)<br>exposed to 0, 50, 200, or 1,000<br>mg/kg-day DEHP for 9 months<br>via diet ( <u>Mitchell et al., 1985</u> )   | ND / 50                                | <ul> <li>↑ hepatic peroxisome<br/>proliferation         <ul> <li>(ultrastructural changes</li> <li>visualized by electron</li> <li>microscopy; males and</li> <li>females)</li> </ul> </li> </ul> | <ul> <li>↓ body weight</li> <li>≥ 200 mg/kg-day</li> <li>males (9–15%) and</li> <li>1,000 mg/kg-day</li> <li>females (12%)</li> </ul>   |
| Male F344 rats (5/group)<br>exposed to 0, 1000, 6000,<br>12,000, or 20,000 ppm DEHP<br>via diet for 1-, 2-, 4-, or 6-<br>weeks (equivalent to 0, 50,   | 50 / 300                               | ↑ PPAR-dependent<br>enzyme activities at 1-<br>and 2- weeks ( <i>e.g.</i> ,<br>PBOX)  | <ul> <li>↓ hepatic GJIC at<br/>≥300 mg/kg-day</li> <li>Dose-dependent ↑<br/>PBOX</li> <li>GJIC significant<br/>only at the high</li> </ul>  |

| May 2025  |                                 |  |  |
|---|---------------------------------|--|--|
| Brief Study Details<br>(Reference)  | NOAEL/<br>LOAEL (mg/kg-<br>day) | PPARα Biomarker at<br>LOAEL  | Comments   |
| 300, 600, 1000 mg/kg-day)<br>( <u>Isenberg et al., 2000</u> ) <sup><i>b</i></sup>   |                                 |  | <ul> <li>dose (6,000 ppm) at</li> <li>4-week timepoint</li> <li>GJIC evaluated via <i>in situ</i> dye transfer assay</li> <li>Coincided with ↑ liver weights (≥300 mg/kg-day, all timepoints)</li> </ul>             |
| Female F344 rats $(18-20/\text{group})$ were exposed to 0, 0.03, 0.1, or 1.2% DEHP for up to 2 years via diet (equivalent to 0, 15, 50, 600 mg/kg-day) (Cattley et al., 1987) <sup>b</sup>  | 50 / 600                        | ↑ PPAR-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>Carnitine<br>acetyltransferase and<br>cyanide insensitive<br>palmitoyl CoA oxidase) | <ul> <li>Coincided with ↑<br/>incidence of hepatic<br/>neoplasms in high<br/>dose (6/20 animals<br/>compared to 0/18 in<br/>control)</li> <li>Sample size for<br/>enzyme activities<br/>was 11–16 /group.</li> </ul> |
| Male and female F344 rats<br>(5/sex/dose) exposed to 0, 75,<br>470, or 950 mg/kg-day DEHP<br>[males] or 0, 79, 490, or 930<br>mg/kg-day [females] for 3<br>weeks via feed followed by a<br>2-week recovery ( <u>Astill et al.</u> ,<br><u>1986</u> ) <sup>c</sup> | 75 / 470 (males)                | ↑ PPAR-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>hepatic carnitine<br>acetyltransferase)   | <ul> <li>Coincided with ↓<br/>serum lipids, ↑ liver<br/>weight at<br/>≥75 mg/kg-day</li> <li>Enzyme activity<br/>returns to control<br/>levels after<br/>recovery period</li> </ul>                                  |
|   | 79 / 490 (females)              | ↑ PPAR-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>hepatic carnitine<br>acetyltransferase)   | <ul> <li>Coincided with ↓<br/>serum lipids,↑ liver<br/>weight at<br/>≥490 mg/kg-day</li> <li>Enzyme activity<br/>returns to control<br/>levels after<br/>recovery period</li> </ul>                                  |
| Male and female marmoset<br>monkeys (5–6/group) exposed<br>to 0, 100, 500, or 2500 mg/kg-<br>day DEHP via gavage (oral)<br>for 65 weeks from 3 months of<br>age to sexual maturity (18<br>months) (Tomonari et al.,<br>2006)                                      | 100 / 500                       | ↑ PPAR-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>lauric acid ω-1-hydrolase<br>activity (females)                                     | - No significant<br>effects observed for<br>hepatic palmitoyl<br>CoA beta oxidation,<br>carnitine acetyl<br>transferase, and<br>catalase; large<br>variability across  |

| Brief Study Details<br>(Reference)  | NOAEL/<br>LOAEL (mg/kg-<br>day) | PPARα Biomarker at<br>LOAEL   | Comments  |
|---|---------------------------------|---|---|
|   |                                 |   | individual values in dose groups and controls.  |
| Male SD rats (6/group)<br>exposed to 0 or 500 mg/kg-day<br>MEHP for 2 weeks via gavage<br>(Lake et al., 1984)   | ND / 500                        | ↑ PPAR-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>hepatic palmitoyl-CoA<br>oxidation, carnitine<br>acetyltransferase)  | <ul> <li>Coincided with ↑<br/>relative liver weight</li> </ul>  |
| Male Syrian hamsters<br>(6/group) exposed to 0 or 500<br>mg/kg-day MEHP for 2 weeks<br>via gavage ( <u>Lake et al., 1984</u> )  | ND / 500                        | ↑ PPAR-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>hepatic palmitoyl-CoA<br>oxidation, carnitine<br>acetyltransferase)  | <ul> <li>Coincided with ↑<br/>relative liver weight</li> </ul>  |
| Male F344 rats (4/group)<br>exposed to 0, 11, 105, 667,<br>1223, or 2100 mg/kg-day<br>DEHP for 21 days via diet<br>(Short et al., 1987)                                   | 105 / 667                       | ↑ PPAR-dependent<br>enzyme activities (e.g.,<br>cyanide-insensitive<br>palmitoyl-CoA oxidation<br>and lauric acid<br>hydroxylation) and<br>↑ peroxisome score ( <i>i.e.</i> ,<br>"moderate increase –<br>clear increase in<br>peroxisome numbers and<br>size range"; visualized<br>via electron microscopy) | <ul> <li>Coincided with ↑<br/>liver weight at<br/>≥667 mg/kg-day</li> <li>↑ PPAR-dependent<br/>enzyme activities<br/>(≥105 mg/kg-day)</li> </ul>  |
| Male F344 rats (5–10/group)<br>exposed to 0, 23.8, 51.7, 115,<br>559, 1,093, or 2,496 mg/kg-<br>day DEHP for 28 days via feed<br>(BIBRA, 1990)                            | 109 / 643                       | ↑ peroxisome<br>proliferation (electron<br>microscopy<br>quantification of<br>periportal peroxisome<br>sore); ↑ PPAR-<br>dependent enzyme<br>activities ( <i>e.g.</i> ,<br>palmitoyl-CoA oxidase)   | <ul> <li>Coincided with ↓<br/>serum lipids and ↑<br/>liver weight at ≥646<br/>mg/kg-day</li> <li>38-44% ↓ body<br/>weight and 48-60%<br/>↓ food consumption<br/>at ≥1,892 mg/kg-<br/>day</li> </ul> |
| Male F344 rats exposed to 0,<br>0.25, 0.5, 1, and 2% DEHP via<br>diet for 30 days (equivalent to<br>0, 125, 250, 500, or 1,000<br>mg/kg-day) (Reddy et al.,<br>1986) $bc$ | 125 / 250                       | <ul> <li>↑ indicators of<br/>peroxisomal proliferation<br/>(peroxisome number and<br/>density via electron<br/>microscopy) and ↑</li> <li>PPAR-dependent<br/>enzyme activities (<i>e.g.</i>,<br/>PBOX, catalase)</li> </ul>   | <ul> <li>Coincided with ↑<br/>liver weight (≥10%<br/>at all doses tested;<br/>no statistical<br/>analysis was<br/>performed)</li> </ul>   |

| Brief Study Details<br>(Reference)   | NOAEL/<br>LOAEL (mg/kg-<br>day) | PPARα Biomarker at<br>LOAEL  | Comments   |
|--|---------------------------------|--|--|
| Male Syrian Hamsters<br>(5/group) exposed to 0, 25,<br>100, 250, or 1,000 mg/kg-day<br>DEHP for 2 weeks via gavage<br>(Lake et al., 1984)  | 250 / 1,000                     | ↑ liver peroxisomes<br>(qualitative<br>histopathological<br>assessment via 3,3'-<br>diaminobenzidine<br>staining)              | <ul> <li>Coincided with ↑<br/>relative liver weight<br/>at 1,000 mg/kg-day<br/>and ↑ hepatic<br/>palmitoyl-CoA<br/>oxidation and<br/>carnitine<br/>acetyltransferase<br/>(~200% ↑ in<br/>enzyme activity at<br/>1,000 mg/kg-day).</li> </ul>       |
| Male and female SD rats<br>(10/sex/dose) exposed to 0,<br>0.4, 3.7, 37.6, 375.2 mg/kg-day<br>[males] or 0, 0.4, 4.2, 42.2,<br>419.3 mg/kg-day [females]<br>DEHP for 13 weeks via feed<br>(Poon et al., 1997) | ND / 375.2                      | ↑ liver peroxisomes<br>(percent cell area;<br>visualized via 3,3'-<br>diaminobenzidine<br>staining)                            | <ul> <li>Coincided with ↑<br/>absolute and<br/>relative liver weight<br/>and mild<br/>hypertrophy (high<br/>dose only, both<br/>sexes)</li> <li>Peroxisome<br/>staining was only<br/>evaluated in control<br/>and high-dose<br/>animals</li> </ul> |
| Male CD-1 mice (6/group)<br>were administered 0, 1.25, or<br>2.5 mmol/kg DEHP for 2<br>weeks (equivalent to 0, 488, or<br>976 mg/kg-day) ( <u>Ito et al.</u> ,<br><u>2007b</u> ) <sup><i>b</i></sup>         | ND / 488                        | ↑ mRNA of PPARa-<br>target gene ( <i>PT</i> )  | <ul> <li>Coincided with ↑<br/>liver weights ≥488<br/>mg/kg-day; ↑<br/>mRNA at high dose<br/>(MCAD); no<br/>change in PPARa</li> </ul>  |
| Male SD rats (3/group) were<br>administered 0, 1.25, or 2.5<br>mmol/kg DEHP for 2 weeks<br>(equivalent to 0, 488, or 976<br>mg/kg-day) ( <u>Ito et al., 2007b</u> ) <sup>b</sup>                             | ND / 488                        | ↑ mRNA and protein of<br>PPARa-target gene ( <i>PT</i> )   | <ul> <li>Coincided with ↑<br/>liver weights ≥488<br/>mg/kg-day; ↑<br/>mRNA at high dose<br/>(MCAD); no<br/>change in PPARa</li> </ul>  |
| Male F344 rats (3–10/group)<br>exposed to 0 or 1.2% DEHP<br>via diet for 1 year (equivalent<br>to 0 or 600 mg/kg-day) via diet<br>(Marsman et al., 1988) <sup>b</sup>  | ND / 600                        | ↑ peroxisomal volume<br>and density (electron<br>microscopy); ↑ PPARa-<br>dependent enzyme<br>activities ( <i>e.g.</i> , PBOX) | <ul> <li>Coincided with ↑<br/>absolute liver<br/>weights; ↓body<br/>weight gain in<br/>DEHP group; no<br/>macroscopic lesions</li> </ul>   |

May 2025

| Brief Study Details<br>(Reference)  | NOAEL/<br>LOAEL (mg/kg-<br>day) | PPARα Biomarker at<br>LOAEL  | Comments  |
|---|---------------------------------|--|---|
|   |                                 |  | of the liver were<br>observed.<br>- Sample size for<br>peroxisomal<br>volume density<br>(electron<br>microscopy) was<br>3/group; sample<br>size for enzyme<br>activity assays was<br>5–10/group)  |
| Male cynomolgus monkeys<br>(2/group) were exposed to 0,<br>100, or 500 mg/kg-day DEHP<br>via gavage for 21 days.<br>Monkeys were then<br>administered radiolabeled<br>DEHP (100 mg/kg-day) on day<br>23, 24, and 25, and were<br>sacrificed on day 25 ( <u>Short et</u><br><u>al., 1987</u> ) | 500 / ND                        | NA   | <ul> <li>Low sample size</li> <li>No changes in liver<br/>weight, no changes<br/>in PPAR-dependent<br/>enzyme activities<br/>(<i>e.g.</i>, cyanide-<br/>insensitive<br/>palmitoyl-CoA<br/>oxidation and lauric<br/>acid hydroxylation)</li> </ul> |
| Male F344 rats (4–7/group)<br>exposed to 0, 500, or 4,000<br>mg/kg-day DEHP for 1 week<br>via feed ( <u>Reddy et al., 1976</u> )  | 500 / 4,000                     | ↑ PPAR-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>hepatic catalase and<br>carnitine acetyl<br>transferase activity)                                 | <ul> <li>Coincided with ↑<br/>relative liver weight<br/>at 4,000 mg/kg-day</li> </ul>   |
| Male adult cynomolgus<br>monkeys (4/group) exposed to<br>0 or 500 mg/kg-day DEHP via<br>intragastric intubation (oral)<br>for 14-days ( <u>Pugh et al., 2000</u> )  | ND / 500                        | <ul> <li>↑ indicators of<br/>peroxisomal proliferation<br/>(liver histopathology;<br/>diffuse hepatocellular<br/>vacuolation in one<br/>animal)</li> </ul> | <ul> <li>No significant<br/>effects observed for<br/>hepatic GJIC or<br/>PBOX</li> </ul>  |
| Male F344 rats (8–10/group)<br>exposed to 0 or 2% DEHP for<br>95 weeks via diet (equivalent<br>to 0 or 600 mg/kg-day) ( <u>Rao et</u><br><u>al., 1987</u> ) <sup><i>b</i></sup>   | ND / 600                        | ↑ peroxisomes; ↑<br>PPARa-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>PBOX, catalase)  | <ul> <li>Coincided with ↑<br/>hepatocellular<br/>carcinomas</li> </ul>  |
| Male F344 rats (5/group)<br>exposed to 0 or 950 mg/kg-day<br>DEHP for 4 days via gavage<br>( <u>Hasmall et al., 2000</u> )  | ND / 950                        | ↑ PPAR-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>PBOX)   | <ul> <li>Coincided with<br/>significant ↑ liver<br/>weight (24%); no<br/>significant change<br/>in body weight</li> </ul>   |

| Widy 2023  |                                 |   |  |
|--|---------------------------------|---|--|
| Brief Study Details<br>(Reference)   | NOAEL/<br>LOAEL (mg/kg-<br>day) | PPARα Biomarker at<br>LOAEL   | Comments   |
| Dunkin Hartley guinea<br>pigs (5/group) exposed to 0 or<br>950 mg/kg-day DEHP for 4<br>days via gavage ( <u>Hasmall et</u><br><u>al., 2000</u> )                           | 950 / ND                        | NA  | - No significant<br>effects observed for<br>hepatic PBOX of<br>liver weights; no<br>significant change<br>in body weight |
| Male SD rats (3/group) were<br>exposed to 0 or 2% DEHP for<br>2 weeks via diet (equivalent to<br>1,000 mg/kg-day DEHP) ( <u>Shin</u><br><u>et al., 1999</u> ) <sup>b</sup> | 1000 / ND                       | ↑ PPARa-dependent<br>enzyme activities ( <i>e.g.</i> ,<br>PBOX, catalase) | <ul> <li>Coincided with ↑<br/>liver weights ↑<br/>NAD+</li> </ul>  |
| Abhanistiana, DEUD - Di 2 athulhanul athalata, CUC - Con Junction Intercollular Communication, I OAEL -  |                                 |   |  |

*Abbreviations:* DEHP = Di-2-ethylhexyl phthalate; GJIC = Gap Junction Intercellular Communication; LOAEL = Lowest observable adverse effect level; MEHP = Mono-2-ethylhexyl phthalate; NAD+ = Nicotinamide adenine dinucleotide; ND = No data; NOAEL = No observable adverse effect level; PBOX = Peroxisomal beta oxidation; PPARa= Peroxisome proliferator-activated receptor alpha; PT = keto-acyl-CoA thiolase; MCAD = medium-chain acyl-CoA dehydrogenase

<sup>a</sup> Studies identified from (<u>ATSDR, 2022</u>) unless otherwise stated.

<sup>*b*</sup> Study did not report received doses in mg/kg-day and food consumption were not reported. To estimate the mean received doses of DEHP in mg/kg-day, when given as % DEHP in diet, the following equation was applied: % DEHP in diet \* (food factor) \* 10,000 = mean dose in mg/kg-day, where food factor = 0.15 for mice, 0.05 for rats, 0.10 for young rats, 0.04 for guinea pigs, 0.05 for monkeys. To estimate the mean received doses of DEHP in mg/kg-day, when given as ppm DEHP in diet, the following equation was applied: DEHP in diet (ppm) \* (food factor) = mean dose in mg/kg-day (<u>WHO, 1987</u>). <sup>*c*</sup> Studies identified from (IARC, 2013).